UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

FORM 10-Q

(Mark One)

X

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended January 2, 2016

to

OR

• TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from

Commission File Number: 001-11593

The Scotts Miracle-Gro Company

(Exact name of registrant as specified in its charter)

OHIO

(State or other jurisdiction of incorporation or organization)

14111 SCOTTSLAWN ROAD, MARYSVILLE, OHIO

(Address of principal executive offices)

31-1414921 (I.R.S. Employer Identification No.)

> 43041 (Zip Code)

(937) 644-0011

(Registrant's telephone number, including area code)

(Former name, former address and former fiscal year, if changed since last report)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes \boxtimes No o

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes 🗵 No o

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act.

 Large accelerated filer
 Image: Comparison of the symplectic symp

Class

Common Shares, \$0.01 stated value, no par value

Outstanding at February 5, 2016

61,411,683 Common Shares

THE SCOTTS MIRACLE-GRO COMPANY INDEX

PAGE NO.

PART I. FINANCIAL INFORMATION:

Item 1.	Financial Statements	
	Condensed Consolidated Statements of Operations — Three months ended January 2, 2016 and December 27, 2014	<u>3</u>
	Condensed Consolidated Statements of Comprehensive Income (Loss) — Three months ended January 2, 2016 and	
	<u>December 27, 2014</u>	<u>4</u>
	Condensed Consolidated Statements of Cash Flows — Three months ended January 2, 2016 and December 27, 2014	<u>5</u>
	Condensed Consolidated Balance Sheets — January 2, 2016, December 27, 2014 and September 30, 2015	<u>6</u>
	Notes to Condensed Consolidated Financial Statements	<u>7</u>
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	<u>35</u>
Item 3.	Quantitative and Qualitative Disclosures about Market Risk	44
Item 4.	Controls and Procedures	<u>44</u>
PART II. C	OTHER INFORMATION:	
Item 1.	Legal Proceedings	<u>45</u>
Item 1A.	Risk Factors	<u>45</u>
Item 2.	Unregistered Sales of Equity Securities and Use of Proceeds	<u>46</u>
Item 6.	<u>Exhibits</u>	<u>47</u>
<u>Signatures</u>		<u>48</u>
Index to E:	<u>xhibits</u>	<u>49</u>

THE SCOTTS MIRACLE-GRO COMPANY Condensed Consolidated Statements of Operations (In millions, except per common share data) (Unaudited)

	THREE MONTHS ENDED				
	JANUARY 2, 2016		EMBER 27, 2014		
Net sales	\$ 245.7	\$	216.2		
Cost of sales	198.1		186.9		
Cost of sales—impairment, restructuring and other	5.0		—		
Gross profit	 42.6		29.3		
Operating expenses:					
Selling, general and administrative	139.5		126.9		
Impairment, restructuring and other	4.3		9.6		
Other income, net	(1.0)		(1.2)		
Loss from operations	 (100.2)		(106.0)		
Costs related to refinancing	8.8		—		
Interest expense	16.3		9.7		
Loss before income taxes	 (125.3)		(115.7)		
Income tax benefit	(44.5)		(41.7)		
Net loss	\$ (80.8)	\$	(74.0)		
Net income attributable to noncontrolling interest	 (0.5)		(0.6)		
Net loss attributable to controlling interest	\$ (81.3)	\$	(74.6)		
Basic loss per common share	\$ (1.32)	\$	(1.23)		
Common shares used in basic loss per share calculation	61.5		60.8		
Diluted loss per common share	\$ (1.32)	\$	(1.23)		
Common shares and potential common shares used in diluted loss per share calculation	61.5		60.8		
Dividends declared per common share	\$ 0.47	\$	0.45		

See Notes to Condensed Consolidated Financial Statements.

THE SCOTTS MIRACLE-GRO COMPANY Condensed Consolidated Statements of Comprehensive Income (Loss) (In millions) (Unaudited)

		THREE MON	THS ENDED		
	J	JANUARY 2, 2016		ECEMBER 27, 2014	
Net loss	\$	(80.8)	\$	(74.0)	
Other comprehensive (loss) income:					
Net foreign currency translation adjustment		(2.8)		(3.0)	
Net unrealized gains (losses) on derivative instruments, net of tax of \$0.8 and (\$0.7), respectively		1.3		(1.1)	
Reclassification of net unrealized losses on derivatives to net income, net of tax of \$0.5 and \$0.6, respectively		0.8		1.0	
Net unrealized losses in pension and other post-retirement benefits, net of tax of \$0.0 and \$0.0, respectively		—		_	
Reclassification of net pension and post-retirement benefit loss to net income, net of tax of \$0.3 and \$0.5, respectively		0.5		0.8	
Total other comprehensive loss		(0.2)		(2.3)	
Comprehensive loss	\$	(81.0)	\$	(76.3)	

See Notes to Condensed Consolidated Financial Statements.

THE SCOTTS MIRACLE-GRO COMPANY Condensed Consolidated Statements of Cash Flows (In millions) (Unaudited)

	THREE MC				
	JANUARY 2, 2016	DEC	EMBER 27, 2014		
PERATING ACTIVITIES					
Net loss	\$ (80.8)	\$	(74.0		
Adjustments to reconcile net loss to net cash used in operating activities:					
Impairment, restructuring and other	_		3.0		
Costs related to refinancing	2.2		-		
Share-based compensation expense	2.2		2.1		
Depreciation	13.5		12.5		
Amortization	4.6		3.2		
Gain on sale of assets	(0.1)		-		
Changes in assets and liabilities, net of acquired businesses:					
Accounts receivable	134.4		148.9		
Inventories	(354.1)		(301.2		
Prepaid and other assets	(6.5)		(6.8		
Accounts payable	42.6		33.0		
Other current liabilities	(85.6)		(93.8		
Restructuring reserves	(10.2)		2.		
Other non-current items	1.4		5.8		
Other, net	(2.2)		(2.		
Net cash used in operating activities	(338.6)		(267.0		
VESTING ACTIVITIES					
Proceeds from sale of long-lived assets	0.1		_		
Investments in property, plant and equipment	(16.1)		(14.		
Investment in unconsolidated affiliates	(0.8)		_		
Investments in acquired businesses, net of cash acquired			(11.		
Net cash used in investing activities	(16.8)		(25.0		
NANCING ACTIVITIES					
Borrowings under revolving and bank lines of credit and term loans					
Repayments under revolving and bank lines of credit and term loans	924.3		539.0		
Proceeds from issuance of 6.000% Senior Notes	(751.2)		(167.1		
Repayment of 6.625% Senior Notes	400.0		_		
Financing and issuance fees	(200.0)		-		
Dividends paid	(10.5)		-		
Purchase of Common Shares	(28.9)		(27.4		
Payments on seller notes	—		(14.		
Excess tax benefits from share-based payment arrangements	(0.8)		-		
Cash received from the exercise of stock options	0.1		0.		
Net cash provided by financing activities	1.2		6.2		
fect of exchange rate changes on cash	334.2		337.0		
et (decrease) increase in cash and cash equivalents	(1.2)		(3.0		
	(22.4)		40.8		
ish and cash equivalents at beginning of period	71.4		89.3		
ish and cash equivalents at end of period	\$ 49.0	\$	130.1		
JPPLEMENTAL CASH FLOW INFORMATION					
Interest paid	\$ (13.2)	\$	(11.2		
Call premium on 6.000% Senior Notes	5 (13.2) (6.6)	ψ	(11.2		
Income taxes paid			(0.1		
See Notes to Condensed Consolidated Fin	(2.1)		(8.6		

THE SCOTTS MIRACLE-GRO COMPANY Condensed Consolidated Balance Sheets (In millions, except stated value per share) (Unaudited)

	JANUARY 2, 2016		DECEMBER 27, 2014		SEI	PTEMBER 30, 2015
ASSETS						
Current assets:						
Cash and cash equivalents	\$	49.0	\$	130.1	\$	71.4
Accounts receivable, less allowances of \$8.0, \$6.8 and \$8.7, respectively		206.4		185.4		191.3
Accounts receivable pledged		_				152.9
Inventories		759.2		682.8		407.6
Prepaid and other current assets		133.4		127.6		125.4
Total current assets		1,148.0		1,125.9		948.6
Property, plant and equipment, net of accumulated depreciation of \$624.5 \$606.1 and \$612.9 respectively	,	449.2		434.4		453.7
Goodwill		433.0		364.3		432.4
Intangible assets, net		658.0		308.9		663.5
Other assets		39.2		31.7		29.0
Total assets	\$	2,727.4	\$	2,265.2	\$	2,527.2
	Ψ	2,727.4	Ψ	2,200.2	Ψ	2,027.2
LIABILITIES AND SHAREHOLDER	S'EQ	UTTY				
Current liabilities:	¢	20.0	¢	26.6	¢	124.0
Current portion of debt	\$	28.8	\$	36.6	\$	134.8
Accounts payable		233.5		220.0		197.9
Other current liabilities		183.3		165.3		280.4
Total current liabilities		445.6		421.9		613.1
Long-term debt		1,503.6		1,133.3		1,028.5
Other liabilities		251.5		249.1		252.5
Total liabilities		2,200.7		1,804.3		1,894.1
Contingencies (Note 10)						
Shareholders' equity:						
Common shares and capital in excess of \$.01 stated value per share; 61.5, 60.7 and 61.4 shares issued and outstanding, respectively		402.3		399.0		400.4
Retained earnings		573.7		534.6		684.2
Treasury shares, at cost; 6.6, 7.5 and 6.7 shares, respectively		(355.2)		(398.2)		(357.1)
Accumulated other comprehensive loss		(107.0)		(88.6)		(106.8)
Total shareholders' equity - controlling interest		513.8		446.8		620.7
Noncontrolling interest		12.9		14.1		12.4
Total equity		526.7		460.9		633.1
Total liabilities and equity	\$	2,727.4	\$	2,265.2	\$	2,527.2
			-		-	

See Notes to Condensed Consolidated Financial Statements.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

NOTE 1. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

Nature of Operations

The Scotts Miracle-Gro Company ("Scotts Miracle-Gro" or "Parent") and its subsidiaries (collectively, together with Scotts Miracle-Gro, the "Company") are engaged in the manufacturing, marketing and sale of consumer branded products for lawn and garden care. The Company's primary customers include home centers, mass merchandisers, warehouse clubs, large hardware chains, independent hardware stores, nurseries, garden centers, food and drug stores, and indoor gardening and hydroponic stores. The Company's products are sold primarily in North America and the European Union. The Company also operates the Scotts LawnService[®] business, which provides residential and commercial lawn care, tree and shrub care and pest control services in the United States.

Due to the nature of the consumer lawn and garden business, the majority of the Company's sales to customers occur in the Company's second and third fiscal quarters. On a combined basis, net sales for the second and third quarters of the last three fiscal years represented in excess of 75% of annual net sales.

The Company follows a 13-week quarterly accounting cycle pursuant to which the first three fiscal quarters end on a Saturday and the fiscal year always ends on September 30. This fiscal calendar convention requires the Company to cycle forward the first three fiscal quarter ends every six years. Fiscal 2016 is the most recent year impacted by this process and, as a result, the first quarter of fiscal 2016 will have six additional days and the fourth quarter of fiscal 2016 will have five less days compared to the respective quarters of fiscal 2015. The Company's first quarter of fiscal 2016 ended on January 2, 2016 while the first quarter of fiscal 2015 ended on December 27, 2014.

Organization and Basis of Presentation

The Company's unaudited condensed consolidated financial statements for the three months ended January 2, 2016 and December 27, 2014 are presented in accordance with accounting principles generally accepted in the United States of America ("GAAP"). The condensed consolidated financial statements include the accounts of Scotts Miracle-Gro and its subsidiaries. All intercompany transactions and accounts have been eliminated in consolidation. The Company's consolidation criteria are based on majority ownership (as evidenced by a majority voting interest in the entity) and an objective evaluation and determination of effective management control. AeroGrow International, Inc. ("AeroGrow"), in which the Company has a controlling interest, is consolidated, with the equity owned by other shareholders shown as noncontrolling interest in the Condensed Consolidated Balance Sheets, and the other shareholders' portion of net earnings and other comprehensive income shown as net income (loss) or comprehensive income attributable to noncontrolling interest in the Condensed Consolidated Statements of Operations and Condensed Consolidated Statements of Comprehensive Income (Loss), respectively. In the opinion of management, interim results reflect all normal and recurring adjustments and are not necessarily indicative of results for a full year.

Certain information and footnote disclosures normally included in financial statements prepared in accordance with GAAP have been omitted or condensed pursuant to the rules and regulations of the Securities and Exchange Commission ("SEC"). Accordingly, this report should be read in conjunction with Scotts Miracle-Gro's Annual Report on Form 10-K for the fiscal year ended September 30, 2015 (the "2015 Annual Report"), which includes a complete set of footnote disclosures, including the Company's significant accounting policies.

The Company's Condensed Consolidated Balance Sheet at September 30, 2015 has been derived from the Company's audited Consolidated Balance Sheet at that date, but does not include all of the information and footnotes required by GAAP for complete financial statements.

Use of Estimates

The preparation of financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the amounts reported in the condensed consolidated financial statements and accompanying notes and related disclosures. Although these estimates are based on management's best knowledge of current events and actions the Company may undertake in the future, actual results ultimately may differ from the estimates.

Long-lived Assets

The Company had noncash investing activities of \$2.7 million and \$3.0 million during the three months ended January 2, 2016 and December 27, 2014, respectively, representing unpaid liabilities incurred during each period to acquire property, plant and equipment.

RECENT ACCOUNTING PRONOUNCEMENTS

Revenue Recognition from Contracts with Customers

In May 2014, the Financial Accounting Standards Board ("FASB") issued Accounting Standards Update No. 2014-09, *Revenue from Contracts with Customers*. This guidance requires companies to recognize revenue in a manner that depicts the transfer of promised goods or services to customers in amounts that reflect the consideration to which a company expects to be entitled in exchange for those goods or services. The new standard also will result in enhanced disclosures about the nature, amount, timing and uncertainty of revenue and cash flows arising from contracts with customers. The provisions are effective for the Company's financial statements no later than the fiscal year beginning October 1, 2018. The standard allows for either a full retrospective or a modified retrospective transition method. The Company is currently evaluating the impact of this standard on its consolidated results of operations, financial position and cash flows.

Discontinued Operations Reporting

In April 2014, the FASB issued an accounting standard update that amends the accounting guidance related to discontinued operations. This amendment defines discontinued operations as a component or group of components that is disposed of or is classified as held for sale and represents a strategic shift that has or will have a major effect on an entity's operations and financial results. This amendment also introduces new disclosures for disposals that do not meet the criteria of discontinued operations. The provisions are effective for fiscal years beginning after December 15, 2014 and apply to new disposals and new classifications of disposal groups as held for sale after the effective date. The adoption of the amended guidance impacts presentation and disclosure of future divestitures and did not have a significant impact on the Company's consolidated financial position, results of operations or cash flows as of January 2, 2016.

Going Concern

In April 2014, the FASB issued a new accounting standard that requires management to assess if there is substantial doubt about an entity's ability to continue as a going concern for each annual and interim period. If conditions or events give rise to substantial doubt, disclosures are required. The new accounting standard will be effective as of December 31, 2016 and is not expected to have an impact on the Company's financial statement disclosures.

Inventory

In July 2015, the FASB issued an accounting standard update that requires inventory to be measured "at the lower of cost and net realizable value," thereby simplifying the current guidance that requires inventory to be measured at the lower of cost or market (market in this context is defined as one of three different measures, one of which is net realizable value). The provisions are effective prospectively for fiscal years beginning after December 15, 2016 and are not expected to have a significant impact on the Company's consolidated financial position, results of operations or cash flows.

Debt Issuance Costs

In April 2015, the FASB issued an accounting standard update that requires debt issuance costs related to a recognized debt liability to be presented in the balance sheet as a direct deduction from the corresponding debt liability rather than as an asset. The provisions are effective for fiscal years beginning after December 15, 2015 and require retrospective application. The adoption of the amended guidance impacts presentation and disclosure of debt issuance costs and is not expected to have a significant impact on the Company's consolidated financial position, results of operations or cash flows.

Cloud Computing Arrangements

In April 2015, the FASB issued an accounting standard update that clarifies how customers in cloud computing arrangements should determine whether the arrangement includes a software license, and requires acquired software licenses to be accounted for as licenses of intangible assets. The provisions are effective for fiscal years beginning after December 15, 2015 and are not expected to have a significant impact on the Company's consolidated financial position, results of operations or cash flows.

Business Combinations

In September 2015, the FASB issued an accounting standard update to simplify the accounting for measurement-period adjustments by requiring an acquirer to recognize adjustments to provisional amounts that are identified during the measurement period in the reporting period in which the adjustment amounts are determined, and requiring disclosure of the portion of the amount recorded in current-period earnings by line item that would have been recorded in previous reporting periods if the adjustment to the provisional amounts had been recognized as of the acquisition date. The provisions are effective prospectively for the Company's financial statements no later than the fiscal year beginning October 1, 2016 and are not expected to have a significant impact on the Company's consolidated financial position, results of operations or cash flows.

Income Taxes

In November 2015, the FASB issued an accounting standard update to simplify the presentation of deferred income taxes by requiring that deferred income tax liabilities and assets be classified as noncurrent in a classified statement of financial position. The provisions are effective for the Company's financial statements no later than the fiscal year beginning October 1, 2017. The standard allows for either a retrospective or prospective transition method and is not expected to have a significant impact on the Company's consolidated financial position, results of operations or cash flows.

NOTE 2. ACQUISITIONS

Fiscal 2015

On October 16, 2014, Scotts LawnService[®] acquired the assets of Action Pest Control Inc. ("Action Pest"), a residential and commercial pest control provider in the Midwest, for \$21.7 million. Action Pest provides residential and commercial pest control services to homeowners and businesses throughout Indiana, Kentucky, and Illinois. This transaction provides Scotts LawnService[®] an entry into the pest control market. Included in the purchase price of \$21.7 million is non-cash investing activity of \$4.0 million representing the deferral of a portion of the purchase price into subsequent fiscal periods. The valuation of acquired assets included finite-lived identifiable intangible assets of \$6.1 million and tax deductible goodwill of \$14.1 million. Identifiable intangible assets included tradename, customer relationships and non-compete agreements with useful lives ranging between 1 to 12 years. The estimated fair values of the identifiable intangible assets were determined using an income-based approach, which includes market participant expectations of cash flows that an asset will generate over the remaining useful life discounted to present value using an appropriate discount rate. Net sales for Action Pest included in the Scotts LawnService[®] segment for the three months ended January 2, 2016 and December 27, 2014 were \$3.1 million and \$1.9 million, respectively. During fiscal 2015, Scotts LawnService[®] also acquired several other businesses that individually and in the aggregate were not significant for an aggregate purchase price of \$3.5 million, which included \$2.6 million in tax deductible goodwill.

On March 30, 2015, the Company acquired the assets of General Hydroponics, Inc. ("General Hydroponics") and Bio-Organic Solutions, Inc. ("Vermicrop") for \$120.0 million and \$15.0 million, respectively. This transaction provides the Company's Global Consumer segment with an additional entry in the indoor and urban gardening market, which is a part of the Global Consumer segment's long-term growth strategy. General Hydroponics and Vermicrop are leading producers of liquid plant food products, growing media, and accessories for the hydroponics markets. The General Hydroponics purchase price includes non-cash investing activity of \$1.0 million representing the deferral of a portion of the purchase price into fiscal 2016. The Vermicrop purchase price includes \$5.0 million of contingent consideration, which will be paid during fiscal 2016. Additionally, the Vermicrop purchase price was paid in common shares of Scotts Miracle-Gro ("Common Shares") based on the average share price at the time of payment. The valuation of acquired assets was determined during the third quarter of fiscal 2015 and included (i) \$14.2 million of inventory and accounts receivable, (ii) \$5.7 million in fixed assets, (iii) \$65.0 million of finite-lived identifiable intangible assets, and (iv) \$53.9 million of tax-deductible goodwill. Identifiable intangible assets included tradenames, customer relationships and non-compete arrangements with useful lives ranging between 5 to 26 years. The estimated fair values of the identifiable intangible assets were determined using an income-based approach, which includes market participant expectations of cash flows that an asset will generate over the remaining useful life discounted to present value using an appropriate discount rate. Net sales for General Hydroponics and Vermicrop included within the Global Consumer segment for the three months ended January 2, 2016 were \$13.5 million.

During fiscal 2015, the Company completed four acquisitions of growing media operations within the Global Consumer segment for an aggregate purchase price of \$40.2 million. These acquisitions expand the Company's growing media operations and distribution capabilities within its Global Consumer segment. The valuation of acquired assets for the transactions included (i) \$10.1 million in finite-lived identifiable intangible assets, (ii) \$11.4 million in fixed assets, (iii) \$10.7 million in tax deductible goodwill, and (iv) \$9.8 million of inventory and accounts receivable. Identifiable intangible assets include tradenames and customer relationships with useful lives ranging between 7 to 20 years. The estimated fair values of the identifiable intangible assets were determined using an income-based approach, which includes market participant expectations of cash flows that an asset will generate over the remaining useful life discounted to present value using an appropriate discount rate.

The condensed consolidated financial statements include the results of operations for these business combinations from the date of each acquisition.

NOTE 3. IMPAIRMENT, RESTRUCTURING AND OTHER

Activity described herein is classified within the "Impairment, restructuring and other" lines in the Condensed Consolidated Statements of Operations.

The following table details impairment, restructuring and other for each of the periods presented:

	THREE MONTHS ENDED						
	JANUARY 2, 2016		DECEMBER 27, 2014				
	(In m	illions)					
Restructuring and other charges	\$ 9.3	\$		9.6			
Goodwill and intangible asset impairments	—			—			
Total impairment, restructuring and other	\$ 9.3	\$		9.6			

The following table summarizes the activity related to liabilities associated with the restructuring and other charges during the three months ended January 2, 2016 (in millions):

Amounts reserved for restructuring and other charges at September 30, 2015	\$ 28.1
Restructuring and other charges	9.3
Payments and other	(19.5)
Amounts reserved for restructuring and other charges at January 2, 2016	\$ 17.9

Included in the restructuring reserves as of January 2, 2016 is \$2.8 million that is classified as long-term. Payments against the long-term reserves will be incurred as the employees covered by the restructuring plan retire or through the passage of time. The remaining amounts reserved will continue to be paid out over the course of the next twelve months.

Fiscal 2016

In the first quarter of fiscal 2016, the Company announced a series of initiatives called Project Focus designed to maximize the value of its non-core assets and concentrate its focus on emerging categories of the lawn and garden industry in its core U.S. business. On December 10, 2015, as part of this project, the Company entered into a definitive agreement to contribute its Scotts LawnService[®] business into a joint venture with TruGreen Holdings Corporation. This agreement is subject to several conditions to close including receiving regulatory approval and the joint venture obtaining financing. The transaction is expected to close in the second quarter of fiscal 2016. During the three months ended January 2, 2016, the Company recognized \$2.8 million in deal related costs associated with this transaction. In addition, during the three months ended January 2, 2016, the Company recognized \$0.9 million in deal costs related to other potential transactions associated with this project.

During the third quarter of fiscal 2015, the Company's Global Consumer segment began experiencing an increase in certain consumer complaints related to the newly reformulated Bonus S[®] fertilizer product sold in the southeastern United States indicating customers were experiencing damage to their lawns after application. During the first quarter of fiscal 2016, the Company recognized \$5.4 million in costs related to resolving these consumer complaints and the recognition of costs the Company expects to incur for current and expected consumer claims. Costs incurred to date since the inception of this matter are \$67.8 million. The Company is working through the claims process with its insurers with respect to these costs, and has received reimbursement payments of \$44.9 million through the first quarter of fiscal 2016, including \$25.0 million received during the first quarter of fiscal 2016. The Company recorded an offsetting insurance reimbursement recovery in the amount of \$4.9 million in the fourth quarter of fiscal 2015 and the remaining \$40.0 million of reimbursements have been recognized as an accrued liability on the Condensed Consolidated Balance Sheets pending the resolution of the insurer's review of claim documentation. During the first quarter of fiscal 2016, the Company paid \$13.2 million to the third party administrator of the consumer claims to pay for lawn repairs.

Fiscal 2015

During the three months ended December 27, 2014, the Company recognized \$9.6 million in restructuring costs related to termination benefits provided to U.S. and international personnel as part of the Company's restructuring of its U.S. administrative and overhead functions and the continuation of the international profitability improvement initiative. These restructuring charges included \$3.6 million of costs related to the acceleration of equity compensation expense.

NOTE 4. INVENTORIES

Inventories consisted of the following for each of the periods presented:

	 JANUARY 2, 2016		DECEMBER 27, 2014	SEPTEMBER 30, 2015
			(In millions)	
Finished goods	\$ 531.5	\$	476.4	\$ 230.2
Work-in-process	60.8		60.5	48.3
Raw materials	166.9		145.9	129.1
Total inventories	\$ 759.2	\$	682.8	\$ 407.6

Adjustments to reflect inventories at net realizable values were \$23.0 million at January 2, 2016, \$19.1 million at December 27, 2014 and \$17.8 million at September 30, 2015.

NOTE 5. MARKETING AGREEMENT

The Scotts Company LLC and Monsanto are parties to an Amended and Restated Exclusive Agency and Marketing Agreement (the "Marketing Agreement"), pursuant to which the Company has served, since its 1998 fiscal year, as Monsanto's exclusive agent for the marketing and distribution of consumer Roundup[®] herbicide products (with additional rights to new products containing glyphosate or other similar non-selective herbicides) in the consumer lawn and garden market. Under the terms of the Marketing Agreement, the Company is entitled to receive an annual commission from Monsanto as consideration for the performance of the Company's duties as agent. The annual gross commission under the Marketing Agreement is calculated as a percentage of the actual earnings before interest and income taxes of the consumer Roundup[®] business in the markets covered by the Marketing Agreement subject to the achievement of annual earnings thresholds. The Marketing Agreement also requires the Company to make annual payments of \$20.0 million to Monsanto as a contribution against the overall expenses of the consumer Roundup[®] business. From 1998 until May 15, 2015, the Marketing Agreement covered the United States and other specified countries, including Australia, Austria, Belgium, Canada, France, Germany, the Netherlands and the United Kingdom. On May 15, 2015, the territories were expanded to cover additional countries as outlined below.

In consideration for the rights granted to the Company under the Marketing Agreement in 1998, the Company paid a marketing fee of \$32 million to Monsanto. The Company deferred this amount on the basis that the payment will provide a future benefit through commissions that will be earned under the Marketing Agreement. The economic useful life over which the marketing fee is being amortized is twenty years, with a remaining unamortized amount of \$2.2 million and remaining amortization period of less than three years as of January 2, 2016.

On May 15, 2015, the Company and Monsanto entered into an Amendment to the Marketing Agreement (the "Marketing Agreement Amendment"), a Lawn and Garden Brand Extension Agreement (the "Brand Extension Agreement") and a Commercialization and Technology Agreement (the "Commercialization and Technology Agreement"). In consideration for these agreements, the Company paid \$300.0 million to Monsanto on August 14, 2015 using borrowings under its credit facility.

Among other things, the Marketing Agreement Amendment amends the Marketing Agreement in the following significant respects:

- Expands the territories in which the Company may serve as Monsanto's exclusive agent in the consumer lawn and garden market to include all countries other than Japan and countries subject to a comprehensive U.S. trade embargo or certain other embargoes and trade restrictions.
- Eliminates the initial and renewal terms that the original Marketing Agreement applied to European Union ("EU") countries. As amended, the term of the Marketing Agreement will now continue indefinitely for all included markets, including EU countries within the included markets, unless and until otherwise terminated in accordance with the Marketing Agreement.
- Revises the procedures of the Marketing Agreement relating to a potential sale of the consumer Roundup[®] business to (1) require Monsanto to negotiate exclusively with the Company with respect to any potential Roundup[®] sale for 60 days after the Company receives notice from Monsanto regarding a potential Roundup[®] sale and (2) provide the Company with a right of first offer and a right of last look in connection with a potential Roundup[®] sale to a third party. In addition, if the Company makes a bid in connection with a Roundup[®] sale, the then-applicable termination fee would serve as a credit against the purchase price and the Monsanto board of directors would not be permitted to discount the value of the Company's bid compared to a competing bid as a result of the termination fee discount.

- Requires the Company to (1) provide notice to Monsanto of certain proposals and processes that may result in a sale of the Company and (2) conduct non-exclusive negotiations with Monsanto with respect to such a sale.
- Increases the minimum termination fee payable under the Marketing Agreement to the greater of (1) \$200.0 million or (2) four times (A) the average of the program earnings before interest or income taxes for the three trailing program years prior to the year of termination, minus (B) the 2015 program earnings before interest or income taxes.
- Amends Monsanto's termination rights and provides additional rights to the Company in the event of a termination, as follows:
 - delays the effectiveness of a notice of termination given by Monsanto as a result of a change of control with respect to Monsanto or a sale of the consumer Roundup[®] business to a third party from (1) the end of the later of 12 months or the next program year to (2) the end of the fifth full program year after Monsanto gives such notice;
 - eliminates Monsanto's termination rights for a regional performance default, a change of significant ownership of the Company or an uncured or incurable egregious injury (as each are defined in the Marketing Agreement); and
 - eliminates Monsanto's termination rights in connection with a change in control of the Company or Scotts Miracle-Gro as long as the Company has determined, in its reasonable commercial opinion, that the acquirer can and will fully perform the duties and obligations of the Company under the Marketing Agreement.
- Expands the Company's termination rights to include termination for a brand decline event (as defined in the Marketing Agreement Amendment) occurring before program year 2023.
- Expands the Company's assignment rights to allow the Company to transfer its rights, interests and obligations under the Marketing Agreement with respect to (1) the North America territories and (2) one or more other included markets for up to three other assignments.
- Amends the commission structure by (1) eliminating the commission threshold for program years 2016, 2017 and 2018, (2) setting the commission threshold for the subsequent program years at \$40 million and (3) establishing the commission payable by Monsanto to the Company for each program year at an amount equal to 50% of the program earnings before interest and income taxes for such program year.

The Brand Extension Agreement provides the Company a worldwide, exclusive license to use the Roundup[®] brand on additional products offered by the Company outside of the non-selective weed category within the residential lawn and garden market. The application of the Roundup[®] brand to these additional products is subject to a product review and approval process developed between the Company and Monsanto. Monsanto will maintain oversight of its brand, the handling of brand registrations covering these new products and new territories, as well as primary responsibility for brand enforcement. The Brand Extension Agreement has an initial term of twenty years, which will automatically renew for additional successive twenty year terms, at the Company's sole option, for no additional monetary consideration.

The Commercialization and Technology Agreement provides for the Company and Monsanto to further develop and commercialize new products and technology developed at Monsanto and intended for introduction into the residential lawn and garden market. Under the Commercialization and Technology Agreement, the Company receives an exclusive first look at new Monsanto technology and products and an annual review of Monsanto's developing products and technologies. The Commercialization and Technology Agreement has a term of thirty years (subject to early termination upon a termination event under the Marketing Agreement or the Brand Extension Agreement).

The Company recorded the \$300.0 million consideration paid by the Company to Monsanto in connection with the entry into the Marketing Agreement Amendment, the Brand Extension Agreement and the Commercialization and Technology Agreement as intangible assets and the related economic useful life of such assets is indefinite. The identifiable intangible assets include the Marketing Agreement Amendment and the Brand Extension Agreement with allocated fair value of \$188.3 million and \$111.7 million, respectively. The estimated fair values of the identifiable intangible assets were determined using an income-based approach, which includes market participant expectations of cash flows that an asset will generate over the remaining useful life discounted to present value using an appropriate rate of return.

Under the terms of the Marketing Agreement, the Company performs certain functions, primarily manufacturing conversion services (in North America), distribution and logistics, and selling and marketing support, on behalf of Monsanto in the conduct of the consumer Roundup[®] business. The actual costs incurred for these activities are charged to and reimbursed by Monsanto. The Company records costs incurred under the Marketing Agreement for which the Company is the primary obligor on a gross



basis, recognizing such costs in "Cost of sales" and the reimbursement of these costs in "Net sales," with no effect on gross profit dollars or net income.

The gross commission earned under the Marketing Agreement, the contribution payments to Monsanto and the amortization of the initial marketing fee paid to Monsanto in 1998 are included in the calculation of net sales in the Company's Condensed Consolidated Statements of Operations. The elements of the net commission and reimbursements earned under the Marketing Agreement and included in "Net sales" are as follows:

THREE MONTHS ENDE					
	MBER 27, 2014				
(In millions)					
\$ — \$	—				
(5.0)	(5.0)				
(0.2)	(0.2)				
 (5.2)	(5.2)				
14.0	17.3				
\$ 8.8 \$	12.1				
	2016 (In millions) (5.0) (0.2) (5.2) 14.0 (1.1)				

NOTE 6. DEBT

The components of long-term debt are as follows:

	 JANUARY 2, 2016	DECEMBER 27, 2014			SEPTEMBER 30, 2015
Credit Facilities:					
Revolving loans	\$ 806.5	\$	922.9	\$	816.3
Term loans	300.0		—		—
Senior Notes – 6.625%	—		200.0		200.0
Senior Notes – 6.000%	400.0		—		—
Master Accounts Receivable Purchase Agreement	—		—		122.3
Other	25.9		47.0		24.7
	1,532.4		1,169.9		1,163.3
Less current portions	28.8		36.6		134.8
Long-term debt	\$ 1,503.6	\$	1,133.3	\$	1,028.5

Credit Facilities

On December 20, 2013, the Company entered into the third amended and restated credit agreement, providing the Company and certain of its subsidiaries with a five-year senior secured revolving loan facility in the aggregate principal amount of up to \$1.7 billion (the "former credit facility"). On October 29, 2015, the Company entered into the fourth amended and restated credit agreement (the "new credit agreement"), providing the Company and certain of its subsidiaries with five-year senior secured loan facilities in the aggregate principal amount of \$1.9 billion, comprised of a revolving credit facility of \$1.6 billion and a term loan in the original principal amount of \$300.0 million (the "new credit facilities"). The new credit agreement also provides the Company with the right to seek additional committed credit under the agreement in an aggregate amount of up to \$500.0 million plus an unlimited additional amount, subject to certain specified financial and other conditions. Under the new credit agreement, the Company has the ability to obtain letters of credit up to \$100.0 million. The new credit agreement replaces the former credit facility, and will terminate on October 29, 2020. Borrowings on the revolving credit facility may be made in various currencies, including U.S. dollars, euro, British pounds, Australian dollars and Canadian dollars. The terms of the new credit agreement include customary representations and warranties, affirmative and negative covenants, financial covenants, and events of default. The proceeds of borrowings on the new credit facilities may be used: (i) to finance working capital requirements and other general corporate purposes of the Company and its subsidiaries; and (ii) to refinance the amounts outstanding under the former credit facility.

Under the terms of the new credit agreement, loans bear interest, at the Company's election, at a rate per annum equal to either the ABR or Adjusted LIBO Rate (both as defined in the new credit agreement) plus the applicable margin. The new credit facilities are guaranteed by substantially all of the Company's domestic subsidiaries, and are secured by (i) a perfected first priority security interest in all of the accounts receivable, inventory and equipment of the Company and those of the Company's domestic subsidiaries that are guarantors and (ii) the pledge of all of the capital stock of the Company's domestic subsidiaries that are guarantors.

At January 2, 2016, the Company had letters of credit outstanding in the aggregate principal amount of \$27.7 million, and \$765.9 million of availability under the new credit agreement, subject to the Company's continued compliance with covenants discussed below. The weighted average interest rates on average borrowings under the new credit agreement and the former credit facility were 3.9% and 3.7% for the three months ended January 2, 2016 and December 27, 2014, respectively.

The new credit agreement contains, among other obligations, an affirmative covenant regarding the Company's leverage ratio on the last day of each quarter calculated as average total indebtedness, divided by the Company's earnings before interest, taxes, depreciation and amortization ("EBITDA"), as adjusted pursuant to the terms of the new credit facilities ("Adjusted EBITDA"). The maximum leverage ratio was 4.50 as of January 2, 2016. The Company's leverage ratio was 2.77 at January 2, 2016. The new credit agreement also includes an affirmative covenant regarding its interest coverage ratio. The interest coverage ratio is calculated as Adjusted EBITDA divided by interest expense, as described in the new credit agreement, and excludes costs related to refinancings. The minimum interest coverage ratio was 3.00 for the twelve months ended January 2, 2016. The Company's interest coverage ratio was 8.52 for the twelve months ended January 2, 2016. The new credit agreement allows the Company to make unlimited restricted payments (as defined in the new credit agreement), including increased or one-time dividend payments and Common Share repurchases, as long as the leverage ratio resulting from the making of such restricted payments is 4.00 or less. Otherwise the Company may only make restricted payments in an aggregate amount for each fiscal year not to exceed the amount set forth in the new credit agreement for such fiscal year (\$175.0 million for 2016 and 2017 and \$200.0 million for 2018 and in each fiscal year thereafter).

Senior Notes - 6.625%

On December 15, 2015, Scotts Miracle-Gro redeemed all \$200.0 million aggregate principal amount of its outstanding 6.625% senior notes due 2020 (the "6.625% Senior Notes") paying a redemption price of \$213.2 million, comprised of \$6.6 million of accrued and unpaid interest, \$6.6 million of call premium and \$200.0 million for outstanding principal amount. The \$6.6 million call premium charge was recognized within the "Costs related to refinancing" line on the Condensed Consolidated Statement of Operations in the first quarter of fiscal 2016. Additionally, the Company had \$2.2 million in unamortized bond discount and issuance costs associated with the 6.625% Senior Notes that were written-off and recognized in the "Costs related to refinancing" line on the Condensed Consolidated Statement of Operations in the first quarter of fiscal 2016.

Senior Notes - 6.000%

On October 13, 2015, Scotts Miracle-Gro issued \$400.0 million aggregate principal amount of 6.000% senior notes due 2023 (the "6.000% Senior Notes"). The net proceeds of the offering were used to repay outstanding borrowings under the former credit facility. The 6.000% Senior Notes represent general unsecured senior obligations and rank equal in right of payment with the Company's existing and future unsecured senior debt. The 6.000% Senior Notes have interest payment dates of April 15 and October 15 of each year, commencing April 15, 2016. The 6.000% Senior Notes may be redeemed, in whole or in part, on or after October 15, 2018 at applicable redemption premiums. The 6.000% Senior Notes contain customary covenants and events of default and mature on October 15, 2023. Substantially all of Scotts Miracle-Gro's domestic subsidiaries serve as guarantors of the 6.000% Senior Notes.

Master Accounts Receivable Purchase Agreement

The Company maintains a Master Accounts Receivable Purchase Agreement ("MARP Agreement"), which provides for the discretionary sale by the Company, and the discretionary purchase (outside of the commitment period specified in the MARP Agreement) by the participating banks, on a revolving basis, of accounts receivable generated by sales to three specified account debtors in an aggregate amount not to exceed \$400.0 million. The MARP Agreement is subject to renewal by mutual agreement at least annually. The Company accounts for the sale of receivables under the MARP Agreement as short-term debt and continues to carry the receivables on its Condensed Consolidated Balance Sheet, primarily as a result of the Company's right to repurchase receivables sold. There were no borrowings or receivables pledged as collateral under the MARP Agreement as of January 2, 2016 and December 27, 2014. As of January 2, 2016, there was \$42.6 million of availability under the MARP Agreement.

Interest Rate Swap Agreements

The Company has outstanding interest rate swap agreements with major financial institutions that effectively convert a portion of the Company's variable-rate debt to a fixed rate. The swap agreements had a total U.S. dollar equivalent notional amount

of \$1,300.0 million at January 2, 2016, December 27, 2014 and September 30, 2015. Interest payments made between the effective date and expiration date are hedged by the swap agreements, except as noted below. The notional amount, effective date, expiration date and rate of each of these swap agreements are shown in the table below.

Notional Amount (in millions)		Effective Date (a)	Expiration Date	Fixed Rate
\$ 50		2/14/2012	2/14/2016	3.78%
150	(b)	2/7/2012	5/7/2016	2.42%
150	(c)	11/16/2009	5/16/2016	3.26%
50	(b)	2/16/2010	5/16/2016	3.05%
100	(b)	2/21/2012	5/23/2016	2.40%
150	(c)	12/20/2011	6/20/2016	2.61%
50	(d)	12/6/2012	9/6/2017	2.96%
200		2/7/2014	11/7/2017	1.28%
150	(b)	2/7/2017	5/7/2019	2.12%
50	(b)	2/7/2017	5/7/2019	2.25%
200	(c)	12/20/2016	6/20/2019	2.12%

(a) The effective date refers to the date on which interest payments were, or will be, first hedged by the applicable swap agreement.

(b) Interest payments made during the three-month period of each year that begins with the month and day of the effective date are hedged by the swap agreement.

(c) Interest payments made during the six-month period of each year that begins with the month and day of the effective date are hedged by the swap agreement.

(d) Interest payments made during the nine-month period of each year that begins with the month and day of the effective date are hedged by the swap agreement.

Estimated Fair Values

A description of the methods and assumptions used to estimate the fair values of the Company's debt instruments is as follows:

Credit Facilities

The interest rate currently available to the Company fluctuates with the applicable LIBO rate, prime rate or Federal Funds Effective Rate and thus the carrying value is a reasonable estimate of fair value. The fair value measurement for the new credit facilities was classified in Level 2 of the fair value hierarchy.

6.000% Senior Notes

The fair value of the 6.000% Senior Notes can be determined based on the trading of the 6.000% Senior Notes in the open market. The difference between the carrying value and the fair value of the 6.000% Senior Notes represents the premium or discount on that date. Based on the trading value on or around January 2, 2016, the fair value of the 6.000% Senior Notes was approximately \$418.0 million. The fair value measurement for the 6.000% Senior Notes was classified in Level 1 of the fair value hierarchy.

Accounts Receivable Pledged

The interest rate on the short-term debt associated with accounts receivable pledged under the MARP Agreement fluctuates with the applicable LIBOR rate and thus the carrying value is a reasonable estimate of fair value. The fair value measurement for the MARP Agreement was classified in Level 2 of the fair value hierarchy.

Weighted Average Interest Rate

The weighted average interest rates on the Company's debt were 5.0% and 4.3% for the three months ended January 2, 2016 and December 27, 2014, respectively. The increase in the weighted average interest rate is due to the higher proportion of long-term fixed-rate borrowing in the Company's debt mix in fiscal 2016.

NOTE 7. RETIREMENT AND RETIREE MEDICAL PLANS

The following summarizes the components of net periodic benefit cost for the retirement and retiree medical plans sponsored by the Company:

	THREE MONTHS ENDED											
			JAN	U ARY 2, 2016								
		U.S. Pension				U.S. Medical		U.S. Pension	International Pension			U.S. Medical
						(In mi	llions	5)				
Service cost	\$	_	\$	0.3	\$	0.1	\$	_	\$	0.3	\$	0.1
Interest cost		1.1		1.7		0.3		1.0		1.9		0.3
Expected return on plan assets		(1.2)		(2.0)		_		(1.3)		(2.3)		
Net amortization		0.4		0.4		(0.3)		0.8		0.5		_
Net periodic benefit cost	\$	0.3	\$	0.4	\$	0.1	\$	0.5	\$	0.4	\$	0.4

NOTE 8. SHAREHOLDERS' EQUITY

During the three months ended January 2, 2016, Scotts Miracle-Gro did not repurchase any Common Shares pursuant to the \$500.0 million share repurchase program approved by the Scotts Miracle-Gro Board of Directors in August 2014. The program allows for repurchases of Common Shares over a five-year period starting November 1, 2014 through September 30, 2019. From the inception of this share repurchase program in the fourth quarter of fiscal 2014 through January 2, 2016, Scotts Miracle-Gro has repurchased approximately 0.2 million Common Shares for \$14.8 million.

As of January 2, 2016, the equity attributable to noncontrolling interest was \$12.9 million compared to \$12.4 million as of September 30, 2015. The \$0.5 million change is due to the net earnings from AeroGrow.

Share-Based Awards

The following is a summary of the share-based awards granted during the periods indicated:

	THRE	THREE MONTHS ENDED				
	JANUARY 2, 2016		DECEMBER 27, 2014			
Employees						
Restricted stock units	1	503	_			
Board of Directors						
Deferred stock units	1	304	1,355			
Total share-based awards	2	807	1,355			
Aggregate fair value at grant dates (in millions)	\$	0.2	\$ 0.1			

Total share-based compensation was as follows for each of the periods presented:

	 THREE MO	NTHS ENI)ED
	JANUARY 2, 2016	DEC	EMBER 27, 2014
	(In n		
Share-based compensation	\$ 2.2	\$	2.1
Tax benefit recognized	0.8		0.8

Subsequent to January 2, 2016, Scotts Miracle-Gro awarded performance share units, restricted stock units, deferred stock units and stock options covering 0.6 million Common Shares to employees and members of the Board of Directors with an estimated fair value of \$14.8 million on the date of the grant.

NOTE 9. INCOME TAXES

The effective tax rate related to continuing operations for the three months ended January 2, 2016 was 35.5%, compared to 36.0% for the three months ended December 27, 2014. The effective tax rate used for interim reporting purposes is based on management's best estimate of factors impacting the effective tax rate for the full fiscal year. There can be no assurance that the effective tax rate estimated for interim financial reporting purposes will approximate the effective tax rate determined at fiscal year end.

Scotts Miracle-Gro or one of its subsidiaries files income tax returns in the U.S. federal jurisdiction and various state, local and foreign jurisdictions. With few exceptions, which are discussed further below, the Company is no longer subject to examination by these tax authorities for fiscal years prior to 2012. The Company is currently under examination by the Internal Revenue Service and certain foreign and U.S. state and local tax authorities. The U.S. federal examination is limited to fiscal years 2011 and 2012. Regarding the foreign jurisdictions, a German audit commenced in the third quarter of 2015 covering fiscal years 2009 through 2012. In the first quarter of fiscal 2016, a Chinese (Wuhan) audit covering fiscal years 2012 through 2014 closed with no impact on the Company's consolidated financial position, results of operations or cash flows. In regard to the multiple U.S. state and local audits, the tax periods under examination are limited to fiscal years 2008 through 2013. In addition to the aforementioned audits, certain other tax deficiency notices and refund claims for previous years remain unresolved.

The Company currently anticipates that few of its open and active audits will be resolved within the next twelve months. The Company is unable to make a reasonably reliable estimate as to when or if cash settlements with taxing authorities may occur. Although audit outcomes and the timing of audit payments are subject to significant uncertainty, the Company does not anticipate that the resolution of these tax matters or any events related thereto will result in a material change to its consolidated financial position, results of operations or cash flows.

NOTE 10. CONTINGENCIES

Management regularly evaluates the Company's contingencies, including various lawsuits and claims which arise in the normal course of business, product and general liabilities, workers' compensation, property losses and other liabilities for which the Company is self-insured or retains a high exposure limit. Self-insurance reserves are established based on actuarial loss estimates for specific individual claims plus actuarially estimated amounts for incurred but not reported claims and adverse development factors applied to existing claims. Legal costs incurred in connection with the resolution of claims, lawsuits and other contingencies generally are expensed as incurred. In the opinion of management, the assessment of contingencies is reasonable and related reserves, in the aggregate, are adequate; however, there can be no assurance that final resolution of these matters will not have a material effect on the Company's financial condition, results of operations or cash flows.

Regulatory Matters

At January 2, 2016, \$5.6 million was accrued in the "Other liabilities" line in the Condensed Consolidated Balance Sheets for environmental actions, the majority of which are for site remediation. The amounts accrued are believed to be adequate to cover such known environmental exposures based on current facts and estimates of likely outcomes. Although it is reasonably possible that the costs to resolve such known environmental exposures will exceed the amounts accrued, any variation from accrued amounts is not expected to be material.

<u>Other</u>

The Company has been named as a defendant in a number of cases alleging injuries that the lawsuits claim resulted from exposure to asbestoscontaining products, apparently based on the Company's historic use of vermiculite in certain of its products. In many of these cases, the complaints are not specific about the plaintiffs' contacts with the Company or its products. The cases vary, but complaints in these cases generally seek unspecified monetary damages (actual, compensatory, consequential and punitive) from multiple defendants. The Company believes that the claims against it are without merit and is vigorously defending against them. It is not currently possible to reasonably estimate a probable loss, if any, associated with these cases and, accordingly, no reserves have been recorded in the Company's condensed consolidated financial statements. The Company is reviewing agreements and policies that may provide insurance coverage or indemnity as to these claims and is pursuing coverage under some of these agreements and policies, although there can be no assurance of the results of these efforts. There can be no assurance that these cases, whether as a result of adverse outcomes or as a result of significant defense costs, will not have a material effect on the Company's financial condition, results of operations or cash flows.

In connection with the sale of wild bird food products that were the subject of a voluntary recall in 2008, the Company has been named as a defendant in four putative class actions filed on and after June 27, 2012, which have now been consolidated in the United States District Court for the Southern District of California as *In re Morning Song Bird Food Litigation*, Lead Case No. 3:12-cv-01592-JAH-RBB. The plaintiffs allege various statutory and common law claims associated with the Company's sale of wild bird food products and a plea agreement entered into in previously pending government proceedings associated with such sales. The plaintiffs allege, among other things, a purported class action on behalf of all persons and entities in the United States who purchased certain bird food products. The plaintiffs assert hundreds of millions of dollars in monetary damages (actual, compensatory, consequential, and restitution), punitive and treble damages; injunctive and declaratory relief; pre-judgment and post-judgment interest; and costs and attorneys' fees. The Company disputes the plaintiffs' assertions and intends to vigorously defend the consolidated action. At this point in the proceedings, it is not currently possible to reasonably estimate a probable loss, if any, associated with the action and, accordingly, no reserves have been recorded in the Company's condensed consolidated financial statements with respect to the action. There can be no assurance that this action, whether as a result of an adverse outcome or as a result of significant defense costs, will not have a material adverse effect on the Company's financial condition, results of operations or cash flows.

The Company is involved in other lawsuits and claims which arise in the normal course of business. These claims individually and in the aggregate are not expected to result in a material effect on the Company's financial condition, results of operations or cash flows.

NOTE 11. DERIVATIVE INSTRUMENTS AND HEDGING ACTIVITIES

The Company is exposed to market risks, such as changes in interest rates, currency exchange rates and commodity prices. To manage a portion of the volatility related to these exposures, the Company enters into various financial transactions. The utilization of these financial transactions is governed by policies covering acceptable counterparty exposure, instrument types and other hedging practices. The Company does not hold or issue derivative financial instruments for speculative trading purposes.

Exchange Rate Risk Management

The Company uses currency forward contracts to manage the exchange rate risk associated with intercompany loans with foreign subsidiaries that are denominated in local currencies. At January 2, 2016, the notional amount of outstanding currency forward contracts was \$58.7 million, with a negative fair value of \$0.8 million. At December 27, 2014, the notional amount of outstanding currency forward contracts was \$144.5 million, with a fair value of \$1.2 million. At September 30, 2015, the notional amount of outstanding currency forward contracts was \$52.3 million, with a negative fair value of \$0.7 million. The fair value of currency forward contracts is determined using forward rates in commonly quoted intervals for the full term of the contracts. The outstanding contracts will mature over the next fiscal year.

Interest Rate Risk Management

The Company enters into interest rate swap agreements as a means to hedge its variable interest rate risk on debt instruments. Net amounts to be received or paid under the swap agreements are reflected as adjustments to interest expense. Since the interest rate swap agreements have been designated as hedging instruments, unrealized gains or losses resulting from adjusting these swaps to fair value are recorded as elements of accumulated other comprehensive income (loss) ("AOCI") within the Condensed Consolidated Balance Sheets except for any ineffective portion of the change in fair value, which is immediately recorded in interest expense. The fair value of the swap agreements is determined based on the present value of the estimated future net cash flows using implied rates in the applicable yield curve as of the valuation date.

The Company has outstanding interest rate swap agreements with major financial institutions that effectively convert a portion of the Company's variable-rate debt to a fixed rate. The swap agreements had a total U.S. dollar equivalent notional amount

of \$1,300.0 million at January 2, 2016, December 27, 2014 and September 30, 2015. Refer to "NOTE 6. DEBT" for the terms of the swap agreements outstanding at January 2, 2016. Included in the AOCI balance at January 2, 2016 was a loss of \$3.9 million related to interest rate swap agreements that is expected to be reclassified to earnings during the next twelve months, consistent with the timing of the underlying hedged transactions.

Commodity Price Risk Management

The Company enters into hedging arrangements designed to fix the price of a portion of its projected future urea requirements. The contracts are designated as hedges of the Company's exposure to future cash flow fluctuations associated with the cost of urea. The objective of the hedges is to mitigate the earnings and cash flow volatility attributable to the risk of changing prices. Since the contracts have been designated as hedging instruments, unrealized gains or losses resulting from adjusting these contracts to fair value are recorded as elements of AOCI within the Condensed Consolidated Balance Sheets. Realized gains or losses remain as a component of AOCI until the related inventory is sold. Upon sale of the underlying inventory, the gain or loss is reclassified to cost of sales. Included in the AOCI balance at January 2, 2016 was a loss of \$0.9 million related to urea derivatives that is expected to be reclassified to earnings during the next twelve months, consistent with the timing of the underlying hedged transactions.

The Company also uses derivatives to partially mitigate the effect of fluctuating diesel and gasoline costs on operating results. These financial instruments are carried at fair value within the Condensed Consolidated Balance Sheets. Changes in the fair value of derivative contracts that qualify for hedge accounting are recorded in AOCI except for any ineffective portion of the change in fair value, which is immediately recorded in earnings. The effective portion of the change in fair value remains as a component of AOCI until the related fuel is consumed, at which time the accumulated gain or loss on the derivative contract is reclassified to cost of sales. Changes in the fair value of derivatives that do not qualify for hedge accounting are recorded as an element of cost of sales. At January 2, 2016, there were no amounts included within AOCI.

The Company had the following outstanding commodity contracts that were entered into to hedge forecasted purchases:

COMMODITY	JANUARY 2, 2016	DECEMBER 27, 2014	SEPTEMBER 30, 2015
Urea	30,000 tons	58,500 tons	52,500 tons
Diesel	6,258,000 gallons	6,930,000 gallons	5,754,000 gallons
Gasoline	504,000 gallons	462,000 gallons	504,000 gallons
Heating Oil	2,520,000 gallons	7,728,000 gallons	2,772,000 gallons

Fair Values of Derivative Instruments

The fair values of the Company's derivative instruments were as follows:

		ASSETS / (LIABILITIES)					
DERIVATIVES DESIGNATED AS HEDGING		_	JANUARY 2, 2016	Ι	DECEMBER 27, 2014	s	EPTEMBER 30, 2015
INSTRUMENTS	BALANCE SHEET LOCATION				FAIR VALUE		
					(In millions)		
Interest rate swap agreements	Other assets	\$	0.1	\$	2.2	\$	—
	Other current liabilities		(7.4)		(10.2)		(8.8)
	Other liabilities		(2.5)		(5.2)		(4.6)
Commodity hedging instruments	Prepaid and other current assets				0.4		
	Other current liabilities		(1.4)		—		(1.3)
Total derivatives designated as hedging inst	ruments	\$	(11.2)	\$	(12.8)	\$	(14.7)
DERIVATIVES NOT DESIGNATED AS HEDGING INSTRUMENTS	BALANCE SHEET LOCATION						
Currency forward contracts	Prepaid and other current assets	\$	—	\$	1.2	\$	—
	Other current liabilities		(0.8)				(0.7)

Commodity hedging instruments	Other current liabilities	(6.2)	(9.1)	(3.9)
Total derivatives not designated as l	nedging instruments	(7.0)	(7.9)	 (4.6)
Total derivatives		\$ (18.2)	\$ (20.7)	\$ (19.3)

The effect of derivative instruments on AOCI and the Condensed Consolidated Statements of Operations was as follows:

	AMOUNT OF GAIN / (LOSS) RECOGNIZE								
		THREE MO	NTHS E	NDED					
DERIVATIVES IN CASH FLOW HEDGING RELATIONSHIPS		JANUARY 2, 2016		DECEMBER 27, 2014					
		(In millions)							
Interest rate swap agreements	\$	1.7	\$		(1.7)				
Commodity hedging instruments		(0.4)			0.6				
Total	\$	1.3	\$		(1.1)				

			AMOUNT OF GAIN / (LOSS)						
	RECLASSIFIED FROM	THREE MONTHS ENDED							
DERIVATIVES IN CASH FLOW HEDGING RELATIONSHIPS	AOCI INTO STATEMENT OF OPERATIONS	AOCI INTO STATEMENT			DECEMBER 27, 2014				
			(In mi	illions)					
Interest rate swap agreements	Interest expense	\$	(1.0)	\$		(1.0)			
Commodity hedging instruments	Cost of sales		0.2			—			
Total		\$	(0.8)	\$		(1.0)			

		AMOUNT OF GAIN / (LOSS)					
	RECOGNIZED IN	THREE MONTHS ENDED					
DERIVATIVES NOT DESIGNATED AS HEDGING INSTRUMENTS	STATEMENT OF OPERATIONS		JANUARY 2, 2016		DECEMBER 27, 2014		
			(In m	illions)			
Currency forward contracts	Other income, net	\$	(0.8)	\$		3.1	
Commodity hedging instruments	Cost of sales		(3.5)			(8.3)	
Total		\$	(4.3)	\$		(5.2)	

NOTE 12. FAIR VALUE MEASUREMENTS

Fair value is defined as the exchange price that would be received for an asset or paid to transfer a liability (an exit price) in the principal or the most advantageous market for the asset or liability in an orderly transaction between market participants at the measurement date. A three-level fair value hierarchy prioritizes the inputs used to measure fair value. The hierarchy requires entities to maximize the use of observable inputs and minimize the use of unobservable inputs. The three levels of inputs used to measure fair value are as follows:

Level 1 — Quoted prices in active markets for identical assets or liabilities.

Level 2 — Observable inputs other than quoted prices included in Level 1, such as quoted prices for similar assets and liabilities in active markets; quoted prices for similar assets and liabilities in markets that are not active; or other inputs that are observable or can be corroborated by observable market data.

Level 3 — Unobservable inputs that are supported by little or no market activity and that are significant to the fair value of the assets or liabilities. This includes pricing models, discounted cash flow methodologies and similar techniques that use significant unobservable inputs.

The following describes the valuation methodologies used for financial assets and liabilities measured at fair value on a recurring basis, as well as the general classification within the valuation hierarchy.

<u>Derivatives</u>

Derivatives consist of currency, interest rate and commodity derivative instruments. Currency forward contracts are valued using observable forward rates in commonly quoted intervals for the full term of the contracts. Interest rate swap agreements are valued based on the present value of the estimated future net cash flows using implied rates in the applicable yield curve as of the valuation date. Commodity contracts are measured using observable commodity exchange prices in active markets.

These derivative instruments are classified within Level 2 of the valuation hierarchy and are included within other assets and other liabilities in the Company's Condensed Consolidated Balance Sheets, except for derivative instruments expected to be settled within the next 12 months, which are included within prepaid and other current assets and other current liabilities.

Cash Equivalents

Cash equivalents consist of highly liquid financial instruments with original maturities of three months or less. The carrying value of these cash equivalents approximates fair value due to their short-term maturities.

<u>Other</u>

Other financial assets consist of investment securities in non-qualified retirement plan assets. These securities are valued using observable market prices in active markets.

The following table presents the Company's financial assets and liabilities measured at fair value on a recurring basis at January 2, 2016:

	Mark Identica	tes in Active ets for al Assets rel 1)	S O	ignificant Other bservable Inputs (Level 2)	I	Unobservable Inputs (Level 3)	Total
				(In mi	llions)		
Assets							
Cash equivalents	\$	20.2	\$	—	\$	—	\$ 20.2
Derivatives							
Interest rate swap agreements				0.1		—	0.1
Other		10.4		—		—	10.4
Total	\$	30.6	\$	0.1	\$	_	\$ 30.7
Liabilities							
Derivatives							
Interest rate swap agreements	\$		\$	(9.9)	\$	_	\$ (9.9)
Currency forward contracts				(0.8)		—	(0.8)
Commodity hedging instruments		_		(7.6)			(7.6)
Total	\$		\$	(18.3)	\$		\$ (18.3)

The following table presents the Company's financial assets and liabilities measured at fair value on a recurring basis at December 27, 2014:

	Que	oted Prices in Active Markets for Identical Assets (Level 1)	Ċ	Significant Other Dbservable Inputs (Level 2)		Unobservable Inputs (Level 3)	Total
				. ,	llions)		
Assets							
Cash equivalents	\$	88.8	\$		\$	—	\$ 88.8
Derivatives							
Interest rate swap agreements		—		2.2		—	2.2
Currency forward contracts		—		1.2		—	1.2
Commodity hedging instruments		—		0.4		—	0.4
Other		10.1		—		—	10.1
Total	\$	98.9	\$	3.8	\$		\$ 102.7
Liabilities							
Derivatives							
Interest rate swap agreements	\$	—	\$	(15.4)	\$	—	\$ (15.4)
Commodity hedging instruments				(9.1)			(9.1)
Total	\$		\$	(24.5)	\$	_	\$ (24.5)



The following table presents the Company's financial assets and liabilities measured at fair value on a recurring basis at September 30, 2015:

	- M Ide	Prices in Active arkets for ntical Assets (Level 1)	Significant Other Dbservable Inputs (Level 2)	U	nobservable Inputs (Level 3)	Total
			(In mi	llions)		
Assets						
Cash equivalents	\$	28.6	\$ _	\$	—	\$ 28.6
Other		8.9	 			 8.9
Total	\$	37.5	\$ 	\$	_	\$ 37.5
Liabilities						
Derivatives						
Interest rate swap agreements	\$	—	\$ (13.4)	\$		\$ (13.4)
Currency forward contracts		—	(0.7)			(0.7)
Commodity hedging instruments		—	(5.2)		—	(5.2)
Total	\$		\$ (19.3)	\$		\$ (19.3)

NOTE 13. SEGMENT INFORMATION

The Company divides its business into two segments: Global Consumer and Scotts LawnService[®]. This division of reportable segments is consistent with how the segments report to and are managed by the chief operating decision maker of the Company.

Segment performance is evaluated on several factors, including income (loss) before amortization, impairment, restructuring and other charges, which is not a GAAP measure. Senior management uses this measure of operating profit (loss) to evaluate segment performance because the Company believes this measure is most indicative of performance trends and the overall earnings potential of each segment.

Corporate & Other consists of revenues and expenses associated with the Company's supply agreements with Israel Chemicals, Ltd. ("ICL"), as well as corporate, general and administrative expenses and certain other income/expense items not allocated to the business segments. Corporate & Other assets primarily include deferred financing and debt issuance costs and corporate intangible assets, as well as deferred tax assets.

The following tables present summarized financial information concerning the Company's reportable segments for the periods indicated:

	THREE MONTHS ENDED			
	 JANUARY 2, 2016	DECEMBER 27, 2014		
	(In millions)			
Net sales:				
Global Consumer	\$ 189.0	\$ 163.6		
Scotts LawnService [®]	51.2	46.7		
Segment total	 240.2	210.3		
Corporate & Other	5.5	5.9		
Consolidated	\$ 245.7	\$ 216.2		
Income (loss) before income taxes:				
Global Consumer	\$ (63.1)	\$ (74.2)		
Scotts LawnService [®]	0.5	1.5		
Segment total	 (62.6)	(72.7)		
Corporate & Other	(23.7)	(20.2)		
Intangible asset amortization	(4.4)	(3.5)		
Impairment, restructuring and other	(9.5)	(9.6)		
Costs related to refinancing	(8.8)	_		
Interest expense	(16.3)	(9.7)		
Consolidated	\$ (125.3)	\$ (115.7)		

	J	ANUARY 2, 2016	D	ECEMBER 27, 2014		SEPTEMBER 30, 2015
Total assets:						
Global Consumer	\$	2,337.7	\$	1,833.7	\$	2,124.7
Scotts LawnService [®]		197.6		191.8		222.5
Corporate & Other		192.1		239.7		180.0
Consolidated	\$	2,727.4		\$ 2,265.2		2,527.2

NOTE 14. OTHER

Scotts LawnService® and TruGreen Holding Corporation Joint Venture

On December 10, 2015, the Company entered into a definitive agreement to contribute Scotts LawnService[®] into a joint venture with TruGreen Holding Corporation ("TruGreen Holdings"). Under the terms of the Contribution and Distribution Agreement, by and among the Company and TruGreen Holdings (the "Contribution Agreement"), the Company has agreed to contribute the Scotts lawn service business (the "SLS Business") to a newly formed subsidiary (the "Joint Venture") of TruGreen Holdings, in exchange for a minority equity interest of approximately 30.6% in the Joint Venture, subject to potential adjustment at closing under certain specified circumstances. A subsidiary of TruGreen Holdings has obtained debt financing commitments, pursuant to which the Joint Venture would, at closing, obtain debt financing (the "Debt Financing") and, subject to the terms of the Contribution Agreement, the Joint Venture would make a pro rata distribution of cash to the Company and TruGreen Holdings (collectively, the "Transactions"). In addition, at the closing of the Contribution, the Company and TruGreen Holdings, will either directly or through their subsidiaries, enter into a limited liability company agreement (the "LLC Agreement") governing the management of the Joint Venture.

The parties' obligations to consummate the Transactions, which is currently expected to close by the end of the second quarter of the Company's 2016 fiscal year, are conditioned upon, among other things, (i) the expiration or termination of all applicable waiting periods and clearances pursuant to certain U.S. antitrust approvals, (ii) the funding of the Debt Financing, (iii) the receipt by the Joint Venture of a solvency opinion as to the distribution of cash to the Company and TruGreen Holdings, and (iv) the delivery of certain ancillary agreements, including the LLC Agreement.

The Contribution Agreement also includes customary representations, warranties and covenants. Among other things, the Company has agreed (i) to certain customary restrictions on the conduct of the business of the SLS Business prior to the closing date of the Contribution Agreement, (ii) to cooperate with TruGreen Holdings' efforts to secure certain financing, and (iii) not to compete with the business of the Joint Venture within the scope of the existing SLS Business and certain adjacent specified activities for a period beginning on the closing date of the Contribution Agreement and ending two years from the date on which the Company no longer has the right to designate any directors to the board of the Joint Venture. The Joint Venture has also agreed not to compete against the business of the Company (except to the extent TruGreen Holdings may be doing so already) for the same period.

The Contribution Agreement provides for certain termination rights, including the right to terminate the Contribution Agreement in the event of breach or failure to perform certain covenants. The Contribution Agreement further provides that the parties may terminate the Contribution Agreement if the Transaction has not been consummated by May 30, 2016. In connection with the termination of the Contribution Agreement under specified circumstances, TruGreen Holdings may be required to pay the Company a fee of \$15 million.

Investment in Bonnie Plants, Inc.

On February 1, 2016, the Company entered into definitive agreements with Bonnie Plants, Inc. ("Bonnie") and its sole shareholder, Alabama Farmers Cooperative ("AFC"), to participate in Bonnie's business of planting, growing, developing, manufacturing, distributing, marketing and selling live plants, plant food, fertilizer and potting soil (the "Bonnie Business"). The Company's participation includes a Term Loan Agreement from the Company to AFC, with Bonnie as guarantor, in the amount of \$72.0 million (the "Term Loan") as well as a Marketing, R&D and Ancillary Services Agreement among the Company, Bonnie and AFC (the "Services Agreement") pursuant to which the Company will provide marketing, research and development and certain ancillary services to the Bonnie Business for a commission fee earned from the profits of the Bonnie Business and reimbursement of certain costs. The agreements also include multiple options that provide mechanisms for either (i) the Company to increase its economic interest in the Bonnie Business or (ii) AFC and Bonnie to repurchase the Company's economic interest in the Bonnie Business. While the board of the AFC has approved the deal, closing is dependent upon approval from the members of AFC as well as the refinancing of AFC's existing indebtedness.

NOTE 15. FINANCIAL INFORMATION FOR SUBSIDIARY GUARANTORS AND NON-GUARANTORS

The 6.000% Senior Notes were issued on October 13, 2015 and are guaranteed by certain of the Company's domestic subsidiaries and, therefore, the Company reports condensed consolidating financial information in accordance with SEC Regulation S-X Rule 3-10, *Financial Statements of Guarantors and Issuers of Guaranteed Securities Registered or Being Registered.* On January 15, 2014 and December 15, 2015, Scotts Miracle-Gro redeemed, respectively, all of its outstanding \$200.0 million aggregate principal amount of 7.25% Senior Notes and \$200.0 million aggregate principal amount of 6.625% Senior Notes, each of which were previously guaranteed by certain of its domestic subsidiaries. The guarantees are "full and unconditional," as those terms are used in Regulation S-X Rule 3-10, except that a subsidiary's guarantee will be released in certain customary circumstances, such as (1) upon any sale or other disposition of all or substantially all of the assets of the subsidiary (including by way of merger or consolidation) to any person other than Scotts Miracle-Gro or any "restricted subsidiary" under the indenture governing the 6.00% Senior Notes; (2) if the subsidiary merges with and into Scotts Miracle-Gro, with Scotts Miracle-Gro surviving such merger;

(3) if the subsidiary is designated an "unrestricted subsidiary" in accordance with the indenture governing the 6.000% Senior Notes or otherwise ceases to be a "restricted subsidiary" (including by way of liquidation or dissolution) in a transaction permitted by such indenture; (4) upon legal or covenant defeasance; (5) at the election of Scotts Miracle-Gro following the subsidiary's release as a guarantor under the new credit agreement, except a release by or as a result of the repayment of the new credit agreement; or (6) if the subsidiary ceases to be a "restricted subsidiary" and the subsidiary is not otherwise required to provide a guarantee of the 6.000% Senior Notes pursuant to the indenture governing the 6.000% Senior Notes. HGCI, Inc. and GenSource, Inc. were added as guarantors effective in the three month period ending January 2, 2016 and have been classified as Guarantors for all periods presented. HGCI, Inc. and GenSource, Inc. did not have any activity for the first quarter of 2015.

The following 100% directly or indirectly owned subsidiaries fully and unconditionally guarantee at January 2, 2016 the 6.000% Senior Notes on a joint and several basis: EG Systems, Inc.; Gutwein & Co., Inc.; Hyponex Corporation; Miracle-Gro Lawn Products, Inc.; OMS Investments, Inc.; Rod McLellan Company; Sanford Scientific, Inc.; Scotts Temecula Operations, LLC; Scotts Manufacturing Company; Scotts Products Co.; Scotts Professional Products Co.; Scotts-Sierra Investments LLC; SMG Growing Media, Inc.; Swiss Farms Products, Inc.; SMGM LLC; SLS Franchise Systems LLC; The Scotts Company LLC; The Hawthorne Gardening Company; Hawthorne Hydroponics LLC; HGCI, Inc.; and GenSource, Inc. (collectively, the "Guarantors").

The following information presents Condensed Consolidating Statements of Operations for the three months ended January 2, 2016 and December 27, 2014, Condensed Consolidating Statements of Comprehensive Income (Loss) for the three months ended January 2, 2016 and December 27, 2014, Condensed Consolidating Statements of Cash Flows for the three months ended January 2, 2016 and December 27, 2014, and Condensed Consolidating Balance Sheets as of January 2, 2016, December 27, 2014 and September 30, 2015. The condensed consolidating financial information presents, in separate columns, financial information for: Scotts Miracle-Gro on a Parent-only basis, carrying its investment in subsidiaries under the equity method; Guarantors on a combined basis; carrying their investments in subsidiaries which do not guarantee the debt (collectively, the "Non-Guarantors") under the equity method; Non-Guarantors on a combined basis; and eliminating entries. The eliminating entries primarily reflect intercompany transactions, such as interest expense, accounts receivable and payable, short and long-term debt, and the elimination of equity investments, return on investments and income in subsidiaries. Because the Parent is obligated to pay the unpaid principal amount and interest on all amounts borrowed by the Guarantors and Non-Guarantors under the previous senior secured five-year revolving loan facility), the borrowings and related interest expense for the loans outstanding of the Guarantors and Non-Guarantors are also presented in the accompanying Parent-only financial information, and are then eliminated. Included in the Parent Condensed Consolidating Statement of Cash Flows for the three months ended January 2, 2016 and December 27, 2014 are \$242.2 million and \$48.9 million, respectively, of dividends paid by the Guarantors and Non-Guarantors to the Parent representing return on investments and as such are classified within cash flows from operating activities.

THE SCOTTS MIRACLE-GRO COMPANY Condensed Consolidating Statement of Operations for the three months ended January 2, 2016 (In millions) (Unaudited)

	Parent	Subsidiary Guarantors	Non- Guarantors	Eliminations/ Consolidations	c	Consolidated
Net sales	\$ _	\$ 181.5	\$ 64.2	\$ _	\$	245.7
Cost of sales	_	150.1	48.0	_		198.1
Cost of sales—impairment, restructuring and other		5.0		—		5.0
Gross profit	_	26.4	16.2	 _		42.6
Operating expenses:						
Selling, general and administrative	—	110.4	28.7	0.4		139.5
Impairment, restructuring and other	—	4.2	0.1	—		4.3
Other income, net	—	(1.7)	0.7	—		(1.0)
Loss from operations	_	(86.5)	(13.3)	(0.4)		(100.2)
Equity income in subsidiaries	68.0	2.6		(70.6)		
Other non-operating income	(4.6)		(6.0)	10.6		—
Costs related to refinancing	8.8			—		8.8
Interest expense	 16.0	 10.0	 0.9	 (10.6)		16.3
Loss before income taxes	 (88.2)	 (99.1)	 (8.2)	 70.2		(125.3)
Income tax benefit	(7.2)	(34.3)	(3.0)	—		(44.5)
Net loss	\$ (81.0)	\$ (64.8)	\$ (5.2)	\$ 70.2	\$	(80.8)
Net income attributable to noncontrolling interest	_	_	—	0.5		0.5
Net loss attributable to controlling interest	\$ (81.0)	\$ (64.8)	\$ (5.2)	\$ 69.7	\$	(81.3)

THE SCOTTS MIRACLE-GRO COMPANY Condensed Consolidating Statement of Comprehensive Income (Loss) for the three months ended January 2, 2016 (In millions) (Unaudited)

	Parent	Subsidiary Guarantors	Non- Guarantors	 iminations/ nsolidations	Consolidated
Net loss	\$ (81.0)	\$ (64.8)	\$ (5.2)	\$ 70.2	\$ (80.8)
Other comprehensive income (loss), net of tax:					
Net foreign currency translation adjustment	(2.8)		(2.8)	2.8	(2.8)
Net change in derivatives	2.1	(0.6)	_	0.6	2.1
Net change in pension and other post-retirement benefits	0.5	0.3	0.2	(0.5)	0.5
Total other comprehensive income (loss)	(0.2)	 (0.3)	(2.6)	2.9	(0.2)
Comprehensive loss	\$ (81.2)	\$ (65.1)	\$ (7.8)	\$ 73.1	\$ (81.0)

THE SCOTTS MIRACLE-GRO COMPANY Condensed Consolidating Statement of Cash Flows for the three months ended January 2, 2016 (In millions) (Unaudited)

	Parent	Subsidiary Guarantors	G	Non- uarantors	inations/ olidations	C	onsolidated
NET CASH PROVIDED BY (USED IN) OPERATING	 						
ACTIVITIES ^(a)	\$ 233.1	\$ (271.9)	\$	(57.6)	\$ (242.2)	\$	(338.6)
INVESTING ACTIVITIES							
Proceeds from sale of long-lived assets	—	0.1		_			0.1
Investments in property, plant and equipment	—	(13.4)		(2.7)			(16.1)
Investing cash flows from (to) affiliates	(395.0)	—		—	395.0		—
Investment in unconsolidated affiliates	—	(0.8)		—			(0.8)
Net cash used in investing activities	(395.0)	(14.1)		(2.7)	395.0		(16.8)
FINANCING ACTIVITIES							
Borrowings under revolving and bank lines of credit and term loans	_	873.1		51.2	_		924.3
Repayments under revolving and bank lines of credit and term loans	_	(741.5)		(9.7)	_		(751.2)
Proceeds from issuance of 6.000% Senior Notes	400.0			_			400.0
Repayment of 6.625% Senior Notes	(200.0)	_		—			(200.0)
Financing and issuance fees	(10.5)	_					(10.5)
Dividends paid	(28.9)	(242.2)		—	242.2		(28.9)
Payments on seller notes	—	(0.8)		—			(0.8)
Excess tax benefits from share-based payment arrangements	0.1			_			0.1
Cash received from the exercise of stock options	1.2	—		_			1.2
Financing cash flows from (to) affiliates	—	393.5		1.5	(395.0)		_
Net cash provided by (used in) financing activities	 161.9	 282.1		43.0	 (152.8)		334.2
Effect of exchange rate changes on cash	 _	 		(1.2)			(1.2)
Net increase (decrease) in cash and cash equivalents	 _	 (3.9)		(18.5)	 _		(22.4)
Cash and cash equivalents at beginning of period	_	8.2		63.2	_		71.4
Cash and cash equivalents at end of period	\$ _	\$ 4.3	\$	44.7	\$ 	\$	49.0

(a) Cash received by the Parent from the Guarantors in the form of dividends in the amount of \$242.2 million represent return on investments and are included in cash flows from operating activities.

THE SCOTTS MIRACLE-GRO COMPANY Condensed Consolidating Balance Sheet As of January 2, 2016 (In millions) (Unaudited)

Parent			Subsidiary Guarantors	Non- Guarantors		Eliminations/ Consolidations			Consolidated
	ASSET	s							
\$	—	\$	4.3	\$	44.7	\$	—	\$	49.0
	—		118.9		87.5		—		206.4
			628.2		131.0		—		759.2
			95.7		37.7		—		133.4
			847.1		300.9				1,148.0
	_		396.0		53.2		—		449.2
	_		409.6		11.8		11.6		433.0
	_		613.5		33.2		11.3		658.0
	22.3		17.4		14.8		(15.3)		39.2
	396.7		_		_		(396.7)		_
	1,619.4						(1,619.4)		—
\$	2,038.4	\$	2,283.6	\$	413.9	\$	(2,008.5)	\$	2,727.4
	\$	ASSET: \$	ASSETS \$ \$ []]]	Parent Guarantors ASSETS - \$ - 4.3 - 118.9 - - 628.2 - - 95.7 - - 847.1 - - 847.1 - - 847.1 - - 613.5 - 22.3 17.4 - 396.7 - - 1,619.4 - -	Parent Guarantors ASSETS - \$ - \$ 4.3 \$ - 118.9 - 628.2 - 95.7 - 847.1 - 396.0 - 613.5 22.3 17.4 396.7 - 1,619.4 -	Parent Guarantors Guarantors ASSETS Guarantors Guarantors \$ — \$ 4.3 \$ 44.7 \$ 44.7 — \$ 4.3 \$ 44.7 87.5 — 628.2 131.0 87.5 — \$ 628.2 131.0 95.7 — 847.1 300.9 37.7 — 613.5 33.2 11.8 — 613.5 33.2 33.2 1.619.4 — — —	Parent Guarantors Guarantors Cuarantors Cuarantors ASSETS	Parent Guarantors Guarantors Consolidations ASSETS	Parent Guarantors Guarantors Consolidations ASSETS

LIABILI	TIES A	AND SHARE	ноі	DERS' EQU	ITY			
Current liabilities:								
Current portion of debt	\$	15.0	\$	18.3	\$	10.5	\$ (15.0)	\$ 28.8
Accounts payable		—		179.1		54.4		233.5
Other current liabilities		15.6		105.8		61.9		183.3
Total current liabilities		30.6		303.2		126.8	 (15.0)	445.6
Long-term debt		1,491.5		966.1		137.5	(1,091.5)	1,503.6
Other liabilities		2.4		228.7		30.6	(10.2)	251.5
Equity investment in subsidiaries		—		183.1			(183.1)	_
Intercompany liabilities		—		449.3		45.4	(494.7)	—
Total liabilities		1,524.5		2,130.4		340.3	(1,794.5)	2,200.7
Total shareholders' equity - controlling interest		513.9		153.2		73.6	(226.9)	513.8
Noncontrolling interest							12.9	12.9
Total equity		513.9		153.2		73.6	(214.0)	526.7
Total liabilities and equity	\$	2,038.4	\$	2,283.6	\$	413.9	\$ (2,008.5)	\$ 2,727.4

THE SCOTTS MIRACLE-GRO COMPANY Condensed Consolidating Statement of Operations for the three months ended December 27, 2014 (In millions) (Unaudited)

	Parent	Subsidiary Guarantors		Non- Guarantors	Eliminations/ Consolidations	C	Consolidated
Net sales	\$ _	\$ 146.5	\$	69.7	\$ 	\$	216.2
Cost of sales	—	130.5		56.4			186.9
Gross profit	 	 16.0	_	13.3			29.3
Operating expenses:							
Selling, general and administrative	—	94.5		31.9	0.5		126.9
Impairment, restructuring and other	—	8.9		0.7			9.6
Other income, net	—	(1.2)		_	—		(1.2)
Loss from operations	_	(86.2)		(19.3)	 (0.5)		(106.0)
Equity income in subsidiaries	69.2	3.4			(72.6)		
Other non-operating income	(4.5)			(5.5)	10.0		_
Interest expense	11.2	8.3		0.2	(10.0)		9.7
Loss before income taxes	(75.9)	(97.9)		(14.0)	 72.1		(115.7)
Income tax benefit	(2.4)	(33.9)		(5.4)			(41.7)
Net loss	\$ (73.5)	\$ (64.0)	\$	(8.6)	\$ 72.1	\$	(74.0)
Net income attributable to noncontrolling interest	 _	_		_	 (0.6)		(0.6)
Net loss attributable to controlling interest	\$ (73.5)	\$ (64.0)	\$	(8.6)	\$ 71.5	\$	(74.6)

THE SCOTTS MIRACLE-GRO COMPANY Condensed Consolidating Statement of Comprehensive Income (Loss) for the three months ended December 27, 2014 (In millions) (Unaudited)

	Parent	Subsidiary Guarantors	Non- Guarantors	iminations/ nsolidations	(Consolidated
Net loss	\$ (73.5)	\$ (64.0)	\$ (8.6)	\$ 72.1	\$	(74.0)
Other comprehensive income (loss), net of tax:						
Net foreign currency translation adjustment	(3.0)		(3.0)	3.0		(3.0)
Net change in derivatives	(0.1)	0.6	—	(0.6)		(0.1)
Net change in pension and other post-retirement benefits	0.8	0.5	0.3	(0.8)		0.8
Total other comprehensive income (loss)	(2.3)	 1.1	 (2.7)	1.6	_	(2.3)
Comprehensive loss	\$ (75.8)	\$ (62.9)	\$ (11.3)	\$ 73.7	\$	(76.3)

THE SCOTTS MIRACLE-GRO COMPANY Condensed Consolidating Statement of Cash Flows for the three months ended December 27, 2014 (In millions)

(Unaudited)

	Parent		Subsidiary Guarantors	Non- Guarantors	Eliminations/ Consolidations	(Consolidated
NET CASH PROVIDED BY (USED IN) OPERATING ACTIVITIES ^(a)	\$ 24.9	\$	(198.6)	\$ (40.2)	\$ (53.1)	\$	(267.0)
		-					
INVESTING ACTIVITIES							
Investments in property, plant and equipment	—		(12.4)	(2.1)			(14.5)
Investments in acquired businesses, net of cash acquired	—		(11.1)	—	—		(11.1)
Net cash used in investing activities	 _		(23.5)	(2.1)			(25.6)
FINANCING ACTIVITIES							
Borrowings under revolving and bank lines of credit and term loans	_		480.7	58.9	_		539.6
Repayments under revolving and bank lines of credit and term loans			(156.5)	(10.6)			(167.1)
Dividends paid	(27.4)		(48.9)	(3.7)	52.6		(27.4)
Purchase of Common Shares	(14.8)		_		_		(14.8)
Excess tax benefits from share-based payment arrangements	_		0.5	_	_		0.5
Cash received from the exercise of stock options	6.2		_	_	_		6.2
Intercompany financing	11.1		(12.9)	1.3	0.5		
Net cash provided by (used in) financing activities	 (24.9)		262.9	45.9	53.1		337.0
Effect of exchange rate changes on cash	 _		_	(3.6)			(3.6)
Net increase (decrease) in cash and cash equivalents	 _		40.8				40.8
Cash and cash equivalents at beginning of period	—		23.1	66.2	_		89.3
Cash and cash equivalents at end of period	\$ 	\$	63.9	\$ 66.2	\$	\$	130.1

(a) Cash received by the Parent from its subsidiaries in the form of dividends in the amount of \$48.9 million represent return on investments and are included in cash flows from operating activities. Cash received by the Guarantors from the Non-Guarantors in the form of dividends in the amount of \$3.7 million represent return on investments and are included in the cash flows from operating activities.

THE SCOTTS MIRACLE-GRO COMPANY Condensed Consolidating Balance Sheet As of December 27, 2014 (In millions) (Unaudited)

	Parent	Subsidiary Guarantors			Non- Guarantors	Eliminations/ Consolidations			Consolidated
	AS	SET	ГS						
Current assets:									
Cash and cash equivalents	\$ —	\$	63.9	\$	66.2	\$		\$	130.1
Accounts receivable, net	—		97.7		87.7				185.4
Inventories			547.2		135.6				682.8
Prepaid and other current assets			84.8		42.8				127.6
Total current assets			793.6		332.3	-			1,125.9
Property, plant and equipment, net			370.4		64.0				434.4
Goodwill	—		346.3		6.4		11.6		364.3
Intangible assets, net	—		247.4		48.5		13.0		308.9
Other assets	21.1		16.0		27.5		(32.9)		31.7
Equity investment in subsidiaries	300.7		—		—		(300.7)		—
Intercompany assets	1,266.6						(1,266.6)		
Total assets	\$ 1,588.4	\$	1,773.7	\$	478.7	\$	(1,575.6)	\$	2,265.2

LIABILITIES AND SHAREHOLDERS' EQUITY

			•			
Current liabilities:						
Current portion of debt	\$ 	\$ 16.0	\$	20.6	\$ 	\$ 36.6
Accounts payable	_	162.7		57.3		220.0
Other current liabilities	13.6	77.3		74.4		165.3
Total current liabilities	 13.6	 256.0		152.3		 421.9
Long-term debt	1,122.9	888.4		44.8	(922.8)	1,133.3
Other liabilities	5.1	224.9		47.0	(27.9)	249.1
Equity investment in subsidiaries		99.0		—	(99.0)	_
Intercompany liabilities	_	257.5		87.1	(344.6)	—
Total liabilities	 1,141.6	 1,725.8		331.2	(1,394.3)	1,804.3
Total shareholders' equity - controlling interest	446.8	47.9		147.5	(195.4)	446.8
Noncontrolling interest		_		—	14.1	14.1
Total equity	446.8	 47.9		147.5	(181.3)	460.9
Total liabilities and equity	\$ 1,588.4	\$ 1,773.7	\$	478.7	\$ (1,575.6)	\$ 2,265.2

THE SCOTTS MIRACLE-GRO COMPANY Condensed Consolidating Balance Sheet As of September 30, 2015 (In millions) (Unaudited)

	Parent		Subsidiary Guarantors	Non- Guarantors	 Eliminations/ onsolidations	Consolidated
	ASSET	s		 	 	
Current assets:						
Cash and cash equivalents	\$ —	\$	8.2	\$ 63.2	\$ —	\$ 71.4
Accounts receivable, net	—		96.9	94.4	—	191.3
Accounts receivable pledged	—		152.9	—	—	152.9
Inventories	—		318.7	88.9	—	407.6
Prepaid and other current assets	—		90.7	34.7	—	125.4
Total current assets	 _		667.4	 281.2	 	948.6
Property, plant and equipment, net	_		397.6	56.1	_	453.7
Goodwill			408.8	12.0	11.6	432.4
Intangible assets, net	_		617.0	34.8	11.7	663.5
Other assets	16.3		15.0	15.0	(17.3)	29.0
Equity investment in subsidiaries	461.3		—		(461.3)	—
Intercompany assets	1,179.4			_	(1,179.4)	_
Total assets	\$ 1,657.0	\$	2,105.8	\$ 399.1	\$ (1,634.7)	\$ 2,527.2

L	IABILITIES	5 AND SHARE	HOL	DERS' EQU	ITY			
Current liabilities:								
Current portion of debt	\$	—	\$	125.1	\$	9.7	\$ —	\$ 134.8
Accounts payable		—		141.5		56.4	—	197.9
Other current liabilities		15.5		191.9		73.0	 _	 280.4
Total current liabilities		15.5		458.5		139.1	—	 613.1
Long-term debt		1,016.3		728.4		100.1	(816.3)	1,028.5
Other liabilities		4.5		228.0		32.3	(12.3)	252.5
Equity investment in subsidiaries		—		156.2		—	(156.2)	—
Intercompany liabilities		—		296.5		47.6	(344.1)	—
Total liabilities		1,036.3		1,867.6		319.1	 (1,328.9)	 1,894.1
Total shareholders' equity - controlling interest		620.7		238.2		80.0	(318.2)	620.7
Noncontrolling interest		—		—		—	12.4	12.4
Total equity		620.7		238.2		80.0	(305.8)	633.1
Total liabilities and equity	\$	1,657.0	\$	2,105.8	\$	399.1	\$ (1,634.7)	\$ 2,527.2

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

The purpose of this discussion is to provide an understanding of the financial condition and results of operations of The Scotts Miracle-Gro Company ("Scotts Miracle-Gro") and its subsidiaries (collectively, together with Scotts Miracle-Gro, the "Company," "we" or "us") by focusing on changes in certain key measures from year-to-year. Management's Discussion and Analysis is divided into the following sections:

- Executive summary
- Results of operations
- Segment results
- Liquidity and capital resources
- Regulatory matters
- Critical accounting policies and estimates

This discussion and analysis should be read in conjunction with "Management's Discussion and Analysis of Financial Condition and Results of Operations" included in Scotts Miracle-Gro's Annual Report on Form 10-K for the fiscal year ended September 30, 2015 (the "2015 Annual Report").

EXECUTIVE SUMMARY

We are dedicated to delivering strong, long-term financial results and outstanding shareholder returns by providing products of superior quality and value to enhance consumers' lawn and garden environments. We are a leading manufacturer and marketer of consumer branded products for lawn and garden care in North America and Europe. We are Monsanto's exclusive agent for the marketing and distribution of consumer Roundup[®] non-selective herbicide products within the United States and other contractually specified countries. We have a presence in similar consumer branded products in Australia, the Far East and Latin America. In addition, with our recent acquisition of General Hydroponics and Vermicrop, and our control of AeroGrow, we are a leading producer of liquid plant food products, growing media, advanced indoor garden systems and accessories for hydroponic gardening. We also operate Scotts LawnService[®], the second largest lawn care service business in the United States. Our operations are divided into two reportable segments: Global Consumer and Scotts LawnService[®].

In the first quarter of fiscal 2016, we announced a series of initiatives called Project Focus designed to maximize the value of our non-core assets and concentrate focus on emerging categories of the lawn and garden industry in our core U.S. business. On December 10, 2015, as part of this project, we entered into a definitive agreement to contribute our Scotts LawnService[®] business into a joint venture with TruGreen Holdings Corporation ("TruGreen Holdings"), in exchange for a minority equity interest of approximately 30.6% in the joint venture, subject to potential adjustment at closing under certain specified circumstances. A subsidiary of TruGreen Holdings has obtained debt financing commitments, pursuant to which the joint venture would, at closing, obtain debt financing and, subject to the terms of the definitive agreement, the joint venture would make a pro rata distribution of cash to us and TruGreen Holdings. This agreement is subject to several conditions to close including receiving regulatory approval and the joint venture obtaining financing. The transaction is expected to close in the second quarter of fiscal 2016.

As a leading consumer branded lawn and garden company, our product development and marketing efforts are largely focused on providing innovative and differentiated products and on continually increasing brand and product awareness to inspire consumers and to create retail demand. We have successfully applied this model for a number of years by focusing on research and development and investing around 5% of our annual net sales in advertising to support and promote our products and brands. We continually explore new and innovative ways to communicate with consumers. We believe that we receive a significant return on these expenditures and anticipate a similar commitment to research and development, advertising and marketing investments in the future, with the continuing objective of driving category growth and profitably increasing market share.

Due to the seasonal nature of the lawn and garden business, significant portions of our products ship to our retail customers during our second and third fiscal quarters, as noted in the chart below. Our annual net sales are further concentrated in the second and third fiscal quarters by retailers who rely on our ability to deliver products closer to when consumers buy our products, thereby reducing retailers' pre-season inventories.

		Percent of Net Sales from Continuing Operations by Quarter			
	2015	2014	2013		
First Quarter	7.2%	6.7%	7.0%		
Second Quarter	36.5%	38.0%	36.4%		
Third Quarter	40.3%	39.3%	41.0%		
Fourth Quarter	16.0%	16.0%	15.6%		

... . .

We follow a 13-week quarterly accounting cycle pursuant to which our first three fiscal quarters end on a Saturday and our fiscal year always ends on September 30. This fiscal calendar convention requires us to cycle forward our first three fiscal quarter ends every six years. Fiscal 2016 is the most recent year impacted by this process and, as a result, our first quarter of fiscal 2016 will have six additional days and our fourth quarter of fiscal 2016 will have five less days compared to the respective quarters of fiscal 2015. Accordingly, our first quarter of fiscal 2016 ended on January 2, 2016 while our first quarter of fiscal 2015 ended on December 27, 2014. In addition, our second quarter of fiscal 2016 will end six days later than our second quarter of fiscal 2015 and these six days are in our peak selling season, which will result in a shift of net sales from our third quarter to our second quarter of fiscal 2016 of approximately \$75-\$100 million.

On August 11, 2014, we announced that the Scotts Miracle-Gro Board of Directors approved a new share repurchase authorization effective November 1, 2014, which will expire on September 30, 2019, to repurchase up to \$500 million of our Common Shares. This replaced the previous share repurchase authorization which expired on September 30, 2014. On August 3, 2015, we announced that the Scotts Miracle-Gro Board of Directors approved an increase in our quarterly cash dividend from \$0.45 to \$0.47 per Common Share. The decision to increase the amount of cash we intend to return to our shareholders reflects our continued confidence in the business and our desire to maintain a consistent capital structure.

RESULTS OF OPERATIONS

The following table sets forth the components of income and expense as a percentage of net sales:

	THREE MONTHS ENDED			
	JANUARY 2, 2016	DECEMBER 27, 2014		
Net sales	100.0 %	100.0 %		
Cost of sales	80.7	86.4		
Cost of sales—impairment, restructuring and other	2.0	—		
Gross profit	17.3	13.6		
Operating expenses:				
Selling, general and administrative	56.7	58.7		
Impairment, restructuring and other	1.8	4.4		
Other income, net	(0.4)	(0.6)		
Loss from operations	(40.8)	(48.9)		
Costs related to refinancing	3.6	—		
Interest expense	6.6	4.6		
Loss before income taxes	(51.0)	(53.5)		
Income tax benefit	(18.1)	(19.3)		
Net loss	(32.9)%	(34.2)%		

Net Sales

Net sales for the three months ended January 2, 2016 were \$245.7 million, an increase of 13.6% from net sales of \$216.2 million for the three months ended December 27, 2014. The change in net sales was attributable to:

	THREE MONTHS ENDED
	JANUARY 2, 2016
Volume	8.8
Acquisitions	8.1
Foreign exchange rates	(3.2)
Pricing	(0.1)
Change in net sales	13.6 %

The increase in net sales for the three months ended January 2, 2016 was primarily driven by:

- increased sales volume in our Global Consumer segment, driven by increased sales within the United States of controls and growing media products, as well as the impact of six additional days in the first quarter of fiscal 2016 as compared to the first quarter of fiscal 2015, which represents approximately \$15 million of net sales; and
- the addition of net sales from acquisitions, primarily from General Hydroponics and Vermicrop within our Global Consumer segment;
- partially offset by the unfavorable impact of foreign exchange rates as a result of the strengthening of the U.S. dollar relative to other currencies including Canadian dollar, euro and British pound.

Cost of Sales

The following table shows the major components of cost of sales:

		THREE MONTHS ENDED				
	JAN	JANUARY 2, 2016 DECEMBER 27, 1				
		(In mil	lions)			
Materials	\$	\$ 99.2 \$				
Distribution and warehousing		46.3		48.7		
Manufacturing labor and overhead		38.6		29.0		
Roundup [®] reimbursements		14.0		17.3		
		198.1		186.9		
Impairment, restructuring and other		5.0		—		
	\$	203.1	\$	186.9		

Factors contributing to the change in cost of sales are outlined in the following table:

	THREE M	IONTHS ENDED
	JANU	J ARY 2, 2016
	(Iı	n millions)
Volume and product mix	\$	21.3
Foreign exchange rates		(5.3)
Roundup [®] reimbursements		(3.3)
Material costs		(1.5)
		11.2
Impairment, restructuring and other		5.0
Change in cost of sales	\$	16.2

The increase in cost of sales for the three months ended January 2, 2016 was primarily driven by:

 costs related to sales from acquisitions of \$10.8 million, primarily from General Hydroponics and Vermicrop within our Global Consumer segment; and

- costs related to increased sales volume in our Global Consumer segment, as well as the impact of six additional days in the first quarter of fiscal 2016 as compared to the first quarter of fiscal 2015;
- partially offset by the favorable impact of foreign exchange rates as a result of a strengthening of the U.S. dollar relative to other currencies including Canadian dollar, euro and British pound;
- a decrease in net sales attributable to reimbursements under our Marketing Agreement for consumer Roundup®;
- lower distribution costs due to savings from fuel purchases, as well as the recognition of lower negative mark-to-market adjustments associated with our fuel hedges of \$3.5 million for the three months ended January 2, 2016 as compared to \$8.3 million for the three months ended December 27, 2014; and
- restructuring and other charges of \$5.0 million related to addressing the consumer complaints regarding our newly reformulated Bonus S[®] product.

Gross Profit

As a percentage of net sales, our gross profit rate was 17.3% and 13.6% for the three months ended January 2, 2016 and December 27, 2014, respectively. Factors contributing to the change in gross profit rate are outlined in the following table:

	THREE MONTHS ENDED
	JANUARY 2, 2016
Product mix and volume:	
Global Consumer mix and volume	2.7 %
Acquisitions	1.3
Scotts LawnService®	1.0
Roundup [®] commissions and reimbursements	0.3
Material costs	0.6
Pricing	(0.1)
	5.8 %
Impairment, restructuring and other	(2.1)
Change in gross profit rate	3.7 %

The increase in the gross profit rate for the three months ended January 2, 2016 was primarily driven by:

- favorable product mix and increased net sales volume driving improved leverage of fixed costs such as warehousing;
- lower distribution costs due to savings from fuel purchases, as well as the recognition of lower negative mark-to-market adjustments associated with our fuel hedges of \$3.5 million for the three months ended January 2, 2016 as compared to \$8.3 million for the three months ended December 27, 2014;
- the net impact of recent acquisitions increasing the gross profit rate, primarily related to General Hydroponics and Vermicrop within our Global Consumer segment;
- increase in sales within our Scotts LawnService[®] segment which has a higher gross profit rate than our Global Consumer segment; and
- decrease in costs related to reimbursements under our Marketing Agreement for consumer Roundup®.

Selling, General and Administrative Expenses

The following table sets forth the components of selling, general and administrative ("SG&A") expenses:

	 THREE MONTHS ENDED					
	 JANUARY 2, 2016	DECEMBER 27, 2014				
	(In millions)					
vertising	\$ 11.5	\$	9.5			
earch and development	11.7		10.2			
zation of intangibles	3.9		3.0			
e-based compensation	2.2		2.1			
selling, general and administrative	110.2		102.1			
	\$ 139.5	\$	126.9			

Advertising expense increased \$2.0 million driven by the timing and mix of Global Consumer media spending. The increase in other SG&A expenses of \$8.1 million is due to the impact of recent acquisitions, primarily General Hydroponics and Vermicrop, of \$3.1 million, and increased selling and variable incentive compensation of \$9.0 million, partially offset by the favorable impact of foreign exchange rates of \$2.3 million as the U.S. dollar has strengthened relative to other currencies including Canadian dollar, euro and British pound.

Impairment, Restructuring and Other (included in Operating Expenses)

In the first quarter of fiscal 2016, we announced a series of initiatives called Project Focus designed to maximize the value of our non-core assets and concentrate focus on emerging categories of the lawn and garden industry in our core U.S. business. On December 10, 2015, as part of this project, we entered into a definitive agreement to contribute our Scotts LawnService[®] business into a joint venture with TruGreen Holdings Corporation. This agreement is subject to several conditions to close including receiving regulatory approval and the joint venture obtaining financing. The transaction is expected to close in the second quarter of fiscal 2016. During the three months ended January 2, 2016, we recognized \$2.8 million in deal related costs associated with this transaction. In addition, during the three months ended January 2, 2016, we recognized an additional \$0.9 million in deal costs related to other potential transactions associated with this project.

During the third quarter of fiscal 2015, we began experiencing an increase in certain consumer complaints related to our newly reformulated Bonus S[®] fertilizer product sold in the southeastern United States indicating customers were experiencing damage to their lawns after application. During the first quarter of fiscal 2016, we recognized \$0.4 million in costs related to resolving these consumer complaints and the recognition of costs we expect to incur for current and expected consumer claims. Costs incurred to date since the inception of this matter are \$67.8 million. We are working through the claims process with our insurers with respect to these costs, and have received reimbursement payments of \$44.9 million through the first quarter of fiscal 2016, including \$25.0 million received during the first quarter of fiscal 2016. We recorded an offsetting insurance reimbursement recovery in the amount of \$4.9 million in the fourth quarter of fiscal 2015 and the remaining \$40.0 million of reimbursements have been recognized as an accrued liability on the Condensed Consolidated Balance Sheets pending the resolution of the insurer's review of claim documentation. During the first quarter of fiscal 2016, we paid \$13.2 million to the third party administrator of the consumer claims to pay for lawn repairs.

During the three months ended December 27, 2014, we recognized \$9.6 million in restructuring costs related to termination benefits provided to U.S. and international personnel as part of our restructuring of the U.S. administrative and overhead functions and the continuation of the international profitability improvement initiative.

Other Income, net

Other income is comprised of activities outside our normal business operations, such as royalty income from the licensing of certain of our brand names, franchise fee income from our Scotts LawnService[®] business, foreign exchange gains/losses, equity income/loss on unconsolidated affiliates and gains/losses from the sale of non-inventory assets. Other income was \$1.0 million for the three months ended January 2, 2016 compared to \$1.2 million for the three months ended December 27, 2014.

Costs Related to Refinancing

Costs related to refinancing were \$8.8 million for the three months ended January 2, 2016. The costs incurred were associated with the redemption of our 6.625% Senior Notes on December 15, 2015, and are comprised of \$6.6 million of call premium and \$2.2 million of unamortized bond discount and issuance costs that were written-off.

Interest Expense

Interest expense was \$16.3 million for the three months ended January 2, 2016 compared to \$9.7 million for the three months ended December 27, 2014. The increase in interest expense of \$6.6 million was driven by an increase in average borrowings of \$418.5 million, excluding the impact of foreign exchange rates, and an increase in our weighted average interest rate of 70 basis points primarily due to the higher proportion of long-term fixed-rate borrowing in our debt mix during fiscal 2016. The increase in average borrowings was driven by acquisition activity that occurred during the third quarter of fiscal 2015 related to General Hydroponics, Vermicrop and the amendment of our Marketing Agreement with Monsanto.

Income Tax Expense

The effective tax rate for the three months ended January 2, 2016 was 35.5% compared to 36.0% for the three months ended December 27, 2014. The effective tax rate used for interim purposes was based on our best estimate of factors impacting the effective tax rate for the full fiscal year. Factors affecting the estimated effective tax rate include assumptions as to income by jurisdiction (domestic and foreign), the availability and utilization of tax credits and the existence of elements of income and expense that may not be taxable or deductible. The estimated effective tax rate is subject to revision in later interim periods and at fiscal year end as facts and circumstances change during the course of the fiscal year. There can be no assurances that the effective tax rate estimated for interim financial reporting purposes will approximate the effective tax rate determined at fiscal year end.

Loss from Operations

We reported a net loss attributable to controlling interest of \$81.3 million, or \$1.32 per diluted share, for the three months ended January 2, 2016 compared to a net loss of \$74.6 million, or \$1.23 per diluted share, for the three months ended December 27, 2014. We anticipated a net loss in our first fiscal quarter due to the seasonal nature of our business, in which sales are heavily weighted to the spring and summer selling periods. For the three months ended January 2, 2016 and December 27, 2014, the pre-tax impact of impairment, restructuring and other charges was \$9.5 million and \$9.6 million, respectively. Additionally, we incurred \$8.8 million of pre-tax costs during the first quarter of fiscal 2016 related to refinancing. Excluding these items, adjusted net loss attributable to controlling interest was \$69.5 million for the three months ended January 2, 2016 compared to \$68.5 million for the three months ended December 27, 2014, primarily driven by an increase in interest expense and SG&A, partially offset by growth in net sales and gross profit. Diluted average common shares used in the diluted loss per common share calculation were 61.5 million for the three months ended January 2, 2016 compared to 60.8 million for the three months ended December 27, 2014. The increase in dilutive average common shares was primarily the result of the exercise and issuance of share-based compensation awards.

SEGMENT RESULTS

Our continuing operations are divided into two reportable segments: Global Consumer and Scotts LawnService[®]. This division of reportable segments is consistent with how the segments report to and are managed by the chief operating decision maker of the Company. Corporate & Other consists of revenues and expenses associated with our supply agreements with ICL, as well as corporate, general and administrative expenses and certain other income/expense items not allocated to the business segments.

Segment performance is evaluated based on several factors, including income (loss) before amortization, impairment, restructuring and other charges, which is not a measure recognized under GAAP. Senior management uses this measure of operating profit (loss) to evaluate segment performance because we believe this measure is most indicative of performance trends and the overall earnings potential of each segment.

The following table sets forth net sales by segment:

		THREE MONTHS ENDED					
	JANU	JANUARY 2, 2016 DECEMBER 27, 201					
		(In millions)					
Global Consumer	\$	189.0	\$	163.6			
Scotts LawnService [®]		51.2		46.7			
Segment total		240.2		210.3			
Corporate & Other		5.5		5.9			
Consolidated	\$	245.7	\$	216.2			

The following table sets forth segment income (loss) before income taxes:

	THREE MONTHS ENDED					
	JANUARY 2, 2016			DECEMBER 27, 2014		
		(In m	illions)			
Global Consumer	\$	(63.1)	\$	(74.2)		
Scotts LawnService®		0.5		1.5		
Segment total		(62.6)		(72.7)		
Corporate & Other		(23.7)		(20.2)		
Intangible asset amortization		(4.4)		(3.5)		
Impairment, restructuring and other		(9.5)		(9.6)		
Costs related to refinancing		(8.8)		—		
Interest expense		(16.3)		(9.7)		
Consolidated	\$	(125.3)	\$	(115.7)		

Global Consumer

Global Consumer segment net sales were \$189.0 million in the first quarter of fiscal 2016, an increase of 15.5%, from the first quarter of fiscal 2015 net sales of \$163.6 million. For the three months ended January 2, 2016, the favorable impacts of volume of 9.7%, which includes the impact of six additional days in the first quarter of fiscal 2016 as compared to the first quarter of fiscal 2015, and acquisitions of 10.0%, were partially offset by unfavorable changes in foreign exchange rates of 4.2%.

Net sales in the United States increased \$31.7 million, or 29.3%, for the first quarter of fiscal 2016 driven by increased volume of controls and growing media products, the impact of six additional days in the first quarter of fiscal 2016 as compared to the first quarter of fiscal 2015, as well as the impact of recent acquisitions. Excluding the impact of changes in foreign exchange rates, net sales internationally increased by \$0.4 million, or 0.8%, for the first quarter of fiscal 2016 driven by the impact of recent acquisitions.

Global Consumer segment operating loss decreased by \$11.1 million, or 15.0%, for the first quarter of fiscal 2016 as compared to the first quarter of fiscal 2015. The decrease was primarily driven by higher sales volume of controls and growing media products in the United States, including the impact of acquisitions and six additional days in the first quarter of fiscal 2016, lower distribution costs due to savings from fuel purchases and hedging, lower material costs for our growing media products and favorable product mix, partially offset by increased SG&A as a result of recent acquisitions and increased compensation expense.

Scotts LawnService®

Scotts LawnService[®] net sales increased by \$4.5 million, or 9.6%, in the first quarter of fiscal 2016 as compared to the first quarter of fiscal 2015. The increase in net sales was driven by increased customer count and the acquisition of Action Pest. The segment operating income for Scotts LawnService[®] decreased by \$1.0 million, or 66.7%, in the first quarter of fiscal 2016 as compared to the first quarter of fiscal 2015 driven by planned increases in selling costs and higher incentive compensation.

Corporate & Other

The net operating loss for Corporate & Other was \$23.7 million in the first quarter of fiscal 2016 as compared to \$20.2 million in the first quarter of fiscal 2015 driven by increased incentive compensation.

LIQUIDITY AND CAPITAL RESOURCES

Operating Activities

Cash used in operating activities totaled \$338.6 million and \$267.0 million for the three months ended January 2, 2016 and December 27, 2014, respectively. Cash used in operating activities increased \$71.6 million, which included the impact of impairment, restructuring and other and an increase in cash used for working capital. The increase in cash used for working capital was primarily due to increased inventory production in the fourth quarter of fiscal 2015 and the first quarter of fiscal 2016 to support fiscal 2016 early season build for growing media products in order to more efficiently handle sales volume during the peak season.

Investing Activities

Cash used in investing activities totaled \$16.8 million and \$25.6 million for the three months ended January 2, 2016 and December 27, 2014, respectively. Cash used for investments in property, plant and equipment during the first three months of fiscal 2016 and fiscal 2015 was \$16.1 million and \$14.5 million, respectively. During the three months ended December 27, 2014, our Global Consumer segment completed the acquisition of a growing media business for \$7.1 million and our Scotts LawnService[®] segment completed the acquisition of Action Pest for \$21.7 million. These acquisitions included cash payments of \$11.1 million during the first quarter of fiscal 2015.

Financing Activities

Financing activities provided cash of \$334.2 million and \$337.0 million for the three months ended January 2, 2016 and December 27, 2014, respectively. The decrease in cash provided by financing activities of \$2.8 million was the result of the repayment of \$200.0 million aggregate principal amount of 6.625% Senior Notes, a decrease in net borrowings under our credit facilities of \$199.4 million, payment of financing and issuance fees of \$10.5 million related to our new credit agreement and the issuance of the 6.000% Senior Notes, and a decrease in cash received from the exercise of stock options of \$5.0 million, partially offset by the issuance of \$400.0 million aggregate principal amount of 6.000% Senior Notes and a decrease in share repurchases of our Common Shares of \$14.8 million.

Cash and Cash Equivalents

Our cash and cash equivalents were held in cash depository accounts with major financial institutions around the world or invested in high quality, short-term liquid investments having original maturities of three months or less. The cash and cash equivalents balances of \$49.0 million and \$130.1 million as of as of January 2, 2016 and December 27, 2014, respectively, included \$39.6 million and \$62.6 million, respectively, held by controlled foreign corporations. Our current plans do not demonstrate a need to, nor do we have plans to, repatriate the retained earnings from these foreign corporations as the earnings are indefinitely reinvested. However, in the future, if we determine it is necessary to repatriate these funds, or we sell or liquidate any of these foreign corporations, we may be required to pay associated taxes on the repatriation, sale or liquidation.

Borrowing Agreements

Our primary sources of liquidity are cash generated by operations and borrowings under our credit facilities, which are guaranteed by substantially all of Scotts Miracle-Gro's domestic subsidiaries. On December 20, 2013, we entered into the third amended and restated credit agreement, providing us with a five-year senior secured revolving loan facility in the aggregate principal amount of up to \$1.7 billion (the "former credit facility"). On October 29, 2015, we entered into the fourth amended and restated credit agreement (the "new credit agreement"), providing us with five-year senior secured loan facilities in the aggregate principal amount of \$1.9 billion, comprised of a revolving credit facility of \$1.6 billion and a term loan in the original principal amount of \$300.0 million (the "new credit facilities"). The new credit agreement also provides us with the right to seek additional committed credit under the agreement in an aggregate amount of up to \$500.0 million plus an unlimited additional amount, subject to certain specified financial and other conditions. Under the new credit agreement, we have the ability to obtain letters of credit up to \$100.0 million. Borrowings on the revolving credit facility may be made in various currencies, including U.S. dollars, euro, British pounds, Australian dollars and Canadian dollars.

At January 2, 2016, we had letters of credit outstanding in the aggregate principal amount of \$27.7 million, and \$765.9 million of availability under the new credit agreement, subject to our continued compliance with covenants discussed below. The weighted average interest rates on average borrowings under the new credit agreement and the former credit facility were 3.9% and 3.7% for the three months ended January 2, 2016 and December 27, 2014, respectively.

On September 25, 2015, we entered into an amended and restated Master Accounts Receivable Purchase Agreement ("MARP Agreement"), which provides for the discretionary sale by us, and the discretionary (outside of the commitment period specified in the MARP Agreement) purchase by the participating banks, on a revolving basis, of accounts receivable generated by sales to three specified account debtors in an aggregate amount not to exceed \$400.0 million.

During the commitment period, which will begin no earlier than February 26, 2016 and end no later than June 17, 2016, the banks will be required to purchase such accounts receivable in an aggregate committed amount not to exceed \$160.0 million. The commencement and continuation of the commitment period will be subject to, among other things, the absence of any termination event under the MARP Agreement or any default or event of default under the new credit agreement. Under the MARP Agreement, the banks have the opportunity to purchase those accounts receivable offered by us at a discount (from the agreed base value thereof) effectively equal to the seven-day LIBOR plus 0.75%. The MARP Agreement has an initial stated termination date of August 26, 2016, which may be extended by mutual agreement. There were no borrowings or receivables pledged as collateral under the MARP Agreement as of January 2, 2016 and December 27, 2014. As of January 2, 2016, there was \$42.6 million of availability under the MARP Agreement.

On October 13, 2015, we issued \$400.0 million aggregate principal amount of 6.000% senior notes due 2023 (the "6.000% Senior Notes"). The net proceeds of the offering were used to repay outstanding borrowings under our former credit facility. The 6.000% Senior Notes represent general unsecured senior obligations and rank equal in right of payment with our existing and future unsecured senior debt. The 6.000% Senior Notes have interest payment dates of April 15 and October 15 of each year, commencing April 15, 2016. The 6.000% Senior Notes may be redeemed, in whole or in part, on or after October 15, 2018 at applicable redemption premiums. The 6.000% Senior Notes contain customary covenants and events of default and mature on October 15, 2023. Substantially all of our domestic subsidiaries serve as guarantors of the 6.000% Senior Notes.

On December 15, 2015, we used a portion of our available credit facility borrowings to redeem all \$200.0 million aggregate principal amount of our outstanding 6.625% senior notes due 2020 (the "6.625% Senior Notes"), paying a redemption price of \$213.2 million, comprised of \$6.6 million of accrued and unpaid interest, \$6.6 million of call premium and \$200.0 million for outstanding principal amount.

We were in compliance with all debt covenants as of January 2, 2016. Our new credit agreement contains, among other obligations, an affirmative covenant regarding our leverage ratio on the last day of each quarter, calculated as our net indebtedness divided by adjusted earnings before interest, taxes, depreciation and amortization. The maximum leverage ratio was 4.50 as of January 2, 2016. Our leverage ratio was 2.77 at January 2, 2016. Our new credit agreement also includes an affirmative covenant regarding our interest coverage. The minimum interest coverage ratio was 3.00 for the twelve months ended January 2, 2016. Our interest coverage ratio was 8.52 for the twelve months ended January 2, 2016. The new credit agreement allows us to make unlimited restricted payments (as defined in the new credit agreement), including increased or one-time dividend payments and Common Share repurchases, as long as the leverage ratio resulting from the making of such restricted payments is 4.00 or less. Otherwise we may only make restricted payments in an aggregate amount for each fiscal year not to exceed the amount set forth in the new credit agreement for such fiscal year (\$175.0 million for 2016 and 2017 and \$200.0 million for 2018 and in each fiscal year thereafter).

We continue to monitor our compliance with the leverage ratio, interest coverage ratio and other covenants contained in the new credit agreement and, based upon our current operating assumptions, we expect to remain in compliance with the permissible leverage ratio and interest coverage ratio throughout fiscal 2016. However, an unanticipated shortfall in earnings, an increase in net indebtedness or other factors could materially affect our ability to remain in compliance with the financial or other covenants of our new credit agreement, potentially causing us to have to seek an amendment or waiver from our lending group which could result in repricing of our credit facilities. While we believe we have good relationships with our lending group, we can provide no assurance that such a request would result in a modified or replacement credit agreement on reasonable terms, if at all.

Judicial and Administrative Proceedings

We are party to various pending judicial and administrative proceedings arising in the ordinary course of business, including, among others, proceedings based on accidents or product liability claims and alleged violations of environmental laws. We have reviewed these pending judicial and administrative proceedings, including the probable outcomes, reasonably anticipated costs and expenses, and the availability and limits of our insurance coverage, and have established what we believe to be appropriate reserves. We do not believe that any liabilities that may result from these pending judicial and administrative proceedings are reasonably likely to have a material effect on our financial condition, results of operations or cash flows; however, there can be no assurance that future quarterly or annual operating results will not be materially affected by these proceedings, whether as a result of adverse outcomes or as a result of significant defense costs.

Contractual Obligations

There have been no material changes, with the exception of changes to our borrowing agreements, outside of the ordinary course of business, in our outstanding contractual obligations since the end of fiscal 2015 and through January 2, 2016. The following table summarizes our future cash outflows for contractual obligations under our borrowing agreements as of January 2, 2016:

		Payments Due by Period							
Contractual Cash Obligations	Total	Less	Than 1 Year	1-3 Yea	ars		3-5 Years		ore Than 5 Years
		(In millions)							
Debt obligations	\$ 1,532.4	\$	28.8	\$	41.3	\$	1,062.3	\$	400.0
Interest expense on debt obligations	363.7		68.2	1	17.4		98.4		79.7

We have long-term debt obligations and interest payments due primarily under the 6.000% Senior Notes and our new credit facilities. Amounts in the table represent scheduled future maturities of long-term debt principal for the periods indicated.

The interest payments for our new credit facilities are based on outstanding borrowings as of January 2, 2016. Actual interest expense will likely be higher due to the seasonality of our business and associated higher average borrowings.

REGULATORY MATTERS

We are subject to local, state, federal and foreign environmental protection laws and regulations with respect to our business operations and believe we are operating in substantial compliance with, or taking actions aimed at ensuring compliance with, such laws and regulations. We are involved in several legal actions with various governmental agencies related to environmental matters. While it is difficult to quantify the potential financial impact of actions involving these environmental matters, particularly remediation costs at waste disposal sites and future capital expenditures for environmental control equipment, in the opinion of management, the ultimate liability arising from such environmental matters, taking into account established reserves, should not have a material effect on our financial condition, results of operations or cash flows. However, there can be no assurance that the resolution of these matters will not materially affect our future quarterly or annual results of operations, financial condition or cash flows. Additional information on environmental matters affecting us is provided in the 2015 Annual Report, under "ITEM 1. BUSINESS — Regulatory Considerations" and "ITEM 3. LEGAL PROCEEDINGS."

CRITICAL ACCOUNTING POLICIES AND ESTIMATES

The preceding discussion and analysis of our consolidated results of operations and financial condition should be read in conjunction with our condensed consolidated financial statements included elsewhere in this Quarterly Report on Form 10-Q. The 2015 Annual Report includes additional information about us, our operations, our financial condition, our critical accounting policies and accounting estimates, and should be read in conjunction with this Quarterly Report on Form 10-Q.

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

Market risks have not changed significantly from those disclosed in the 2015 Annual Report.

ITEM 4. CONTROLS AND PROCEDURES

The Scotts Miracle-Gro Company (the "Registrant") maintains "disclosure controls and procedures," as such term is defined under Exchange Act Rule 13a-15(e), that are designed to ensure that information required to be disclosed in the Registrant's Exchange Act reports is recorded, processed, summarized and reported within the time periods specified in the SEC's rules and forms, and that such information is accumulated and communicated to the Registrant's management, including its principal executive officer and its principal financial officer, as appropriate, to allow timely decisions regarding required disclosure. In designing and evaluating the disclosure controls and procedures, the Registrant's management recognized that any controls and procedures, no matter how well designed and operated, can provide only reasonable assurance of achieving the desired control objectives, and in reaching a reasonable level of assurance, the Registrant's management necessarily was required to apply its judgment in evaluating the cost-benefit relationship of possible controls and procedures.

With the participation of the principal executive officer and principal financial officer of the Registrant, the Registrant's management has evaluated the effectiveness of the Registrant's disclosure controls and procedures (as defined in Rule 13a-15(e) under the Securities Exchange Act of 1934, as amended (the "Exchange Act")) as of the end of the fiscal quarter covered by this Quarterly Report on Form 10-Q. Based upon that evaluation, the Registrant's principal executive officer and principal financial officer have concluded that the Registrant's disclosure controls and procedures were effective at the reasonable assurance level.

In addition, there were no changes in the Registrant's internal control over financial reporting (as defined in Rule 13a-15(f) under the Exchange Act) that occurred during the Registrant's fiscal quarter ended January 2, 2016 that have materially affected, or are reasonably likely to materially affect, the Registrant's internal control over financial reporting.

PART II—OTHER INFORMATION

ITEM 1. LEGAL PROCEEDINGS

There have been no material changes to the pending legal proceedings disclosed in the 2015 Annual Report.

ITEM 1A. RISK FACTORS

The Company's risk factors as of January 2, 2016 have not changed materially from those described in "ITEM IA. RISK FACTORS" in the 2015 Annual Report.

Cautionary Note Regarding Forward-Looking Statements

This Quarterly Report on Form 10-Q, including the exhibits hereto and the information incorporated by reference herein, contains "forward-looking statements" within the meaning of Section 27A of the Securities Act of 1933, as amended, and Section 21E of the Securities Exchange Act of 1934, as amended, which are subject to risks and uncertainties. Other than statements of historical fact, information regarding activities, events and developments that we expect or anticipate will or may occur in the future, including, but not limited to, information relating to our future growth and profitability targets and strategies designed to increase total shareholder value, are forward-looking statements based on management's estimates, assumptions and projections. Forward-looking statements also include, but are not limited to, statements regarding our future economic and financial condition and results of operations, the plans and objectives of management and our assumptions regarding our performance and such plans and objectives, as well as the amount and timing of repurchases of Common Shares. These forward-looking statements generally can be identified through the use of words such as "guidance," "outlook," "projected," "believe," "target," "predict," "estimate," "forecast," "strategy," "may," "goal," "expect," "anticipate," "intend," "plan," "foresee," "likely," "will," "should" and other similar words and variations.

Forward-looking statements contained in this Quarterly Report on Form 10-Q are predictions only and actual results could differ materially from management's expectations due to a variety of factors, including those described in "ITEM 1A. RISK FACTORS" in the 2015 Annual Report. All forward-looking statements attributable to us or persons working on our behalf are expressly qualified in their entirety by such risk factors.

The forward-looking statements that we make in this Quarterly Report on Form 10-Q are based on management's current views and assumptions regarding future events and speak only as of their dates. We disclaim any obligation to update developments of these risk factors or to announce publicly any revisions to any of the forward-looking statements that we make, or to make corrections to reflect future events or developments, except as required by the federal securities laws.

ITEM 2. UNREGISTERED SALES OF EQUITY SECURITIES AND USE OF PROCEEDS

The payment of future dividends, if any, on the Common Shares will be determined by the Board of Directors in light of conditions then existing, including the Company's earnings, financial condition and capital requirements, restrictions in financing agreements, business conditions and other factors. The terms of the new credit agreement allow the Company to make unlimited restricted payments (as defined in the new credit agreement), including increased or one-time dividend payments and Common Share repurchases, so long as the leverage ratio resulting from the making of such restricted payments is 4.00 or less. Otherwise the Company may only make restricted payments in an aggregate amount for each fiscal year not to exceed the amount set forth in the new credit agreement for such fiscal year (\$175.0 million for 2016 and 2017 and \$200.0 million for 2018 and in each fiscal year thereafter). Our leverage ratio was 2.77 at January 2, 2016.

(a) Issuer Purchases of Equity Securities

The following table shows the purchases of Common Shares made by or on behalf of Scotts Miracle-Gro or any "affiliated purchaser" (as defined in Rule 10b-18(a)(3) under the Securities Exchange Act of 1934, as amended) of Scotts Miracle-Gro for each fiscal month in the three months ended January 2, 2016:

Period	Total Number of Common Shares Purchased(1)	Average Price Paid per Common Share(2)				Approximate Dollar /alue of Common Shares That May Yet be Purchased Under the Plans or Programs(3)
October 1 through October 31, 2015	1,553	\$	65.77	_	\$	485,186,044
November 1 through November 28, 2015	1,455	\$	70.22	—	\$	485,186,044
November 29 through January 2, 2016	2,290	\$	69.46	—	\$	485,186,044
Total	5,298	\$	68.60			

- All of the Common Shares purchased during the quarter were purchased in open market transactions. The total number of Common Shares purchased (1)during the quarter includes 5,298 Common Shares purchased by the trustee of the rabbi trust established by the Company as permitted pursuant to the terms of The Scotts Company LLC Executive Retirement Plan (the "ERP"). The ERP is an unfunded, non-qualified deferred compensation plan which, among other things, provides eligible employees the opportunity to defer compensation above specified statutory limits applicable to The Scotts Company LLC Retirement Savings Plan and with respect to any Executive Management Incentive Pay (as defined in the ERP), Performance Award (as defined in the ERP) or other bonus awarded to such eligible employees. Pursuant to the terms of the ERP, each eligible employee has the right to elect an investment fund, including a fund consisting of Common Shares (the "Scotts Miracle-Gro Common Stock Fund"), against which amounts allocated to such employee's account under the ERP, including employer contributions, will be benchmarked (all ERP accounts are bookkeeping accounts only and do not represent a claim against specific assets of the Company). Amounts allocated to employee accounts under the ERP represent deferred compensation obligations of the Company. The Company established the rabbi trust in order to assist the Company in discharging such deferred compensation obligations. When an eligible employee elects to benchmark some or all of the amounts allocated to such employee's account against the Scotts Miracle-Gro Common Stock Fund, the trustee of the rabbi trust purchases the number of Common Shares equivalent to the amount so benchmarked. All Common Shares purchased by the trustee are purchased on the open market and are held in the rabbi trust until such time as they are distributed pursuant to the terms of the ERP. All assets of the rabbi trust, including any Common Shares purchased by the trustee, remain, at all times, assets of the Company, subject to the claims of its creditors. The terms of the ERP do not provide for a specified limit on the number of Common Shares that may be purchased by the trustee of the rabbi trust.
- (2) The average price paid per Common Share is calculated on a settlement basis and includes commissions.
- (3) On August 11, 2014, Scotts Miracle-Gro announced that its Board of Directors authorized the repurchase of up to \$500 million of Common Shares over a five-year period (starting November 1, 2014 through September 30, 2019). The dollar amounts in the "Approximate Dollar Value of Common Shares That May Yet be Purchased Under the Plans or Programs" column reflect the remaining amounts that were available for repurchase under the \$500 million authorized repurchase program.

ITEM 6. EXHIBITS

See Index to Exhibits at page 49 for a list of the exhibits included herewith.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

THE SCOTTS MIRACLE-GRO COMPANY

Date: February 11, 2016

/s/ THOMAS RANDAL COLEMAN

Printed Name: Thomas Randal Coleman Title: Executive Vice President and Chief Financial Officer

THE SCOTTS MIRACLE-GRO COMPANY QUARTERLY REPORT ON FORM 10-Q FOR THE QUARTERLY PERIOD ENDED JANUARY 2, 2016

INDEX TO EXHIBITS

EXHIBIT NO.	DESCRIPTION	LOCATION
4.1	Fifth Supplemental Indenture, dated October 26, 2015, among The Scotts Miracle-Gro Company, the Guarantors (as defined therein) and U.S. Bank National Association, as trustee	Incorporated herein by reference to the Registrant's Annual Report on Form 10-K for the fiscal year ended September 30, 2015 [Exhibit 4.1(f)]
4.2	Indenture, dated as of October 13, 2015, by and among The Scotts Miracle-Gro Company, the Guarantors (as defined therein) and U.S. Bank National Association, as trustee	Incorporated herein by reference to the Registrant's Current Report on Form 8-K filed October 14, 2015 [Exhibit 4.1]
4.3	Form of 6.000% Senior Notes due 2023 (included in Exhibit 4.1).	Incorporated herein by reference to the Registrant's Current Report on Form 8-K filed October 14, 2015 [Exhibit 4.2]
4.4	Registration Rights Agreement, dated as of October 13, 2015, by and among The Scotts Miracle-Gro Company, the guarantors named therein and Merrill Lynch, Pierce, Fenner & Smith Incorporated, as representative of the several initial purchasers named therein	Incorporated herein by reference to the Registrant's Current Report on Form 8-K filed October 14, 2015 [Exhibit 4.3]
10.1	Purchase Agreement, dated October 7, 2015, among The Scotts Miracle-Gro Company, the subsidiary guarantors named therein and Merrill Lynch, Pierce, Fenner & Smith Incorporated, as representative of the several initial purchasers named therein	Incorporated herein by reference to the Registrant's Current Report on Form 8-K filed October 14, 2015 [Exhibit 10.1]
10.2	Fourth Amended and Restated Credit Agreement, dated as of October 29, 2015, by and among The Scotts Miracle-Gro Company, as a Borrower; the Subsidiary Borrowers (as defined therein); JPMorgan Chase Bank, N.A., as Administrative Agent; Bank of America, N.A. and Wells Fargo Bank, National Association, as Co-Syndication Agents; CoBank, ACB, Mizuho Bank, LTD., Coöperatieve Centrale Raiffeisen-Boerenleenbank B.A. "Rabobank Nederland", New York Branch, TD Bank N.A. and U.S. Bank National Association, as Co-Documentation Agents; and the several other banks and other financial institutions from time to time parties thereto	Incorporated herein by reference to the Registrant's Current Report on Form 8-K filed November 3, 2015 [Exhibit 10.1]
10.3	Amendment No. 1, dated as of February 8, 2016, to Fourth Amended and Restated Credit Agreement dated October 29, 2015, by and among The Scotts Miracle-Gro Company, as a Borrower; the Subsidiary Borrowers (as defined therein); JPMorgan Chase Bank, N.A., as Administrative Agent; Bank of America, N.A. and Wells Fargo Bank, National Association, as Co-Syndication Agents; CoBank, ACB, Mizuho Bank, LTD., Coöperatieve Rabobank U.S., New York Branch (formerly known as Coöperatieve Centrale Raiffeisen-Boerenleenbank B.A. "Rabobank Nederland", New York Branch), TD Bank N.A. and U.S. Bank National Association, as Co-Documentation Agents; and the several other banks and other financial institutions from time to time parties thereto	*
10.4	Fourth Amended and Restated Guarantee and Collateral Agreement, dated as of October 29, 2015, made by The Scotts Miracle-Gro Company, each domestic Subsidiary Borrower under the Fourth Amended and Restated Credit Agreement, and certain of its and their domestic subsidiaries, in favor of JPMorgan Chase Bank, N.A., as Administrative Agent	Incorporated herein by reference to the Registrant's Current Report on Form 8-K filed November 3, 2015 [Exhibit 10.2]
10.5	Contribution and Distribution Agreement, dated as of December 10, 2015, by and among the The Scotts Miracle-Gro Company and TruGreen Holding Corporation	*

31.1	Rule 13a-14(a)/15d-14(a) Certifications (Principal Executive Officer)	*
31.2	Rule 13a-14(a)/15d-14(a) Certifications (Principal Financial Officer)	*
32	Section 1350 Certifications (Principal Executive Officer and Principal Financial Officer)	*
101.INS	XBRL Instance Document	*
101.SCH	XBRL Taxonomy Extension Schema	*
101.CAL	XBRL Taxonomy Extension Calculation Linkbase	*
101.DEF	XBRL Taxonomy Extension Definition Linkbase	*
101.LAB	XBRL Taxonomy Extension Label Linkbase	*
101.PRE	XBRL Taxonomy Extension Presentation Linkbase	*

* Filed or furnished herewith

EXECUTION COPY

AMENDMENT NO. 1

Dated as of February 8, 2016

to

FOURTH AMENDED AND RESTATED CREDIT AGREEMENT

Dated as of October 29, 2015

THIS AMENDMENT NO. 1 (this "<u>Amendment</u>") is made as of February 8, 2016 by and among The Scotts Miracle-Gro Company, an Ohio corporation (the "<u>Company</u>"), The Scotts Company LLC, an Ohio limited liability company, Scotts Australia PTY Ltd., a company incorporated in Australia, Scotts Canada Ltd., a company organized under the laws of Canada, Scotts Holdings Limited, a private limited company incorporated in England and Wales, The Scotts Company (UK) Limited, a private limited company incorporated in England and Wales, The Scotts Company (UK) Limited, a private limited company and the other Subsidiary Borrowers, the "<u>Borrowers</u>"), the financial institutions listed on the signature pages hereof and JPMorgan Chase Bank, N.A., as Administrative Agent (the "<u>Administrative Agent</u>"), under that certain Fourth Amended and Restated Credit Agreement dated as of October 29, 2015 by and among the Borrowers, the Lenders and the Administrative Agent (as amended, restated, supplemented or otherwise modified from time to time, the "<u>Credit Agreement</u>"). Capitalized terms used herein and not otherwise defined herein shall have the respective meanings given to them in the Credit Agreement.

WHEREAS, the Company has requested that the requisite Lenders and the Administrative Agent agree to a certain amendment to the Credit Agreement;

WHEREAS, the Borrowers, the Lenders party hereto and the Administrative Agent have agreed to amend the Credit Agreement on the terms and conditions set forth herein;

NOW, THEREFORE, in consideration of the premises set forth above, the terms and conditions contained herein, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Borrowers, the Lenders party hereto and the Administrative Agent hereby agree to enter into this Amendment.

1. <u>Amendment to the Credit Agreement</u>. Effective as of the date of satisfaction of the conditions precedent set forth in <u>Section 2</u> below, the parties hereto agree that the Credit Agreement is hereby amended as follows:

(a) <u>Section 5.11(b)</u> of the Credit Agreement is hereby amended by replacing the word "promptly" appearing therein with the words "promptly and in any event within thirty (30) days of such creation or acquisition (or such later date as is agreed to by the Administrative Agent in its reasonable discretion)".

2. <u>Conditions of Effectiveness</u>. The effectiveness of this Amendment is subject to the conditions precedent that the Administrative Agent shall have received (i) counterparts of this Amendment duly executed by the Borrowers, the Required Lenders and the Administrative Agent, (ii) counterparts of the Consent and Reaffirmation attached as <u>Exhibit A</u> hereto duly executed by the Loan Parties and (iii) payment and/or reimbursement of the Administrative Agent's and its affiliates' reasonable and documented out-of-pocket fees and expenses (including, to the extent invoiced, reasonable fees and expenses of counsel for the Administrative Agent) in connection with the Loan Documents.

3. <u>Representations and Warranties of the Borrowers</u>. Each Borrower hereby represents and warrants as follows:

(a) This Amendment and the Credit Agreement as modified hereby constitute legal, valid and binding obligations of such Borrower, enforceable against such Borrower in accordance with their respective terms, except as enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting the enforcement of creditors' rights generally and by general equitable principles (whether enforcement is sought by proceedings in equity or at law).

(b) As of the date hereof and after giving effect to the terms of this Amendment, (i) no Event of Default or Default has occurred and is continuing and (ii) the representations and warranties of such Borrower set forth in the Credit Agreement, as amended hereby, are true and correct in all material respects (or, in the case of any representation or warranty qualified by materiality or Material Adverse Effect, in all respects), except to the extent such representations and warranties specifically refer to an earlier date (in which case such representations and warranties shall be true and correct in all material respects as of such earlier date).

4. <u>Reference to and Effect on the Credit Agreement</u>.

(a) Upon the effectiveness hereof, each reference to the Credit Agreement in the Credit Agreement or any other Loan Document shall mean and be a reference to the Credit Agreement as amended hereby.

(b) Each Loan Document and all other documents, instruments and agreements executed and/or delivered in connection therewith shall remain in full force and effect and are hereby ratified and confirmed.

(c) The execution, delivery and effectiveness of this Amendment shall not operate as a waiver of any right, power or remedy of the Administrative Agent or the Lenders, nor constitute a waiver of any provision of the Credit Agreement, the Loan Documents or any other documents, instruments and agreements executed and/or delivered in connection therewith.

(d) This Amendment is a Loan Document under (and as defined in) the Credit Agreement.

5. <u>Governing Law</u>. This Amendment shall be construed in accordance with and governed by the laws of the State of New

York.

6. <u>Headings</u>. Section headings in this Amendment are included herein for convenience of reference only and shall not constitute a part of this Amendment for any other purpose.

7. <u>Counterparts</u>. This Amendment may be executed by one or more of the parties hereto on any number of separate counterparts, and all of said counterparts taken together shall be deemed to constitute one and the same instrument. Delivery of an executed counterpart of a signature page of this Amendment by telecopy, e-mailed.pdf or any other electronic means that reproduces an image of the actual executed signature page shall be effective as delivery of a manually executed counterpart of this Amendment.

[Signature Pages Follow]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be duly executed and delivered by their respective authorized officers as of the day and year first above written.

THE SCOTTS MIRACLE-GRO COMPANY, as the Company

By: /s/ THOMAS RANDAL COLEMAN

Name: Thomas Randal Coleman

Title: EVP and Chief Financial Officer

THE SCOTTS COMPANY LLC, as a Subsidiary Borrower

By: /s/ THOMAS RANDAL COLEMAN

Name: Thomas Randal Coleman

Title: EVP and Chief Financial Officer

EG SYSTEMS, INC., as a Subsidiary Borrower

By: /s/ THOMAS RANDAL COLEMAN

Name: Thomas Randal Coleman

Title: EVP and Chief Financial Officer

GUTWEIN & CO., INC., as a Subsidiary Borrower

By: /s/ THOMAS RANDAL COLEMAN

Name: Thomas Randal Coleman

Title: EVP and Chief Financial Officer

HYPONEX CORPORATION, as a Subsidiary Borrower

By: /s/ THOMAS RANDAL COLEMAN

Name: Thomas Randal Coleman

Title: EVP and Chief Financial Officer

SCOTTS MANUFACTURING COMPANY, as a Subsidiary Borrower

By: /s/ THOMAS RANDAL COLEMAN

Name: Thomas Randal Coleman

Title: EVP and Chief Financial Officer

SCOTTS TEMECULA OPERATIONS, LLC, as a Subsidiary Borrower

By: /s/ THOMAS RANDAL COLEMAN

Name: Thomas Randal Coleman

Title: EVP and Chief Financial Officer

SMG GROWING MEDIA, INC., as a Subsidiary Borrower

By: /s/ THOMAS RANDAL COLEMAN

Name: Thomas Randal Coleman

Title: EVP and Chief Financial Officer

SCOTTS AUSTRALIA PTY LTD., as a Subsidiary Borrower

By: /s/ AIMEE M. DELUCA

Name: <u>Aimee M. DeLuca</u>

Title: <u>Director</u>

SCOTTS CANADA LTD., as a Subsidiary Borrower

By: <u>/s/ THOMAS RANDAL COLEMAN</u>

Name: Thomas Randal Coleman

Title: EVP and Chief Financial Officer

SCOTTS HOLDINGS LIMITED, as a Subsidiary Borrower

By: /s/ AIMEE M. DELUCA

Name: <u>Aimee M. DeLuca</u>

Title: <u>Director</u>

THE SCOTTS COMPANY (UK) LIMITED, as a Subsidiary Borrower

By: /s/ AIMEE M. DELUCA

Name: <u>Aimee M. DeLuca</u>

Title: <u>Director</u>

SCOTTS TREASURY EEIG, as a Subsidiary Borrower

By: /s/ MARK J. WEAVER

Name: <u>Mark J. Weaver</u>

Title: <u>Authorized Signatory</u>

By: <u>/s/ TONY YUNG</u>

Name: <u>Tony Yung</u>

Title: <u>Executive Director</u>

By: /s/ NICHOLAS CHENG

Name: <u>Nicholas Cheng</u>

Title: <u>Director</u>

By: /s/ PETER R. MARTINETS

Name: Peter R. Martinets

Title: Managing Director

COBANK, ACB, as a Lender

By: <u>/s/ HAL NELSON</u>

Name: Hal Nelson

Title: Vice President

MIZUHO BANK, LTD., as a Lender

By: <u>/s/ DAVID LIM</u>

Name: David Lim

Title: <u>Authorized Signatory</u>

COÖPERATIEVE RABOBANK U.A., NEW YORK BRANCH (formerly known as COÖPERATIEVE CENTRALE RAIFFEISEN-BOERENLEENBANK B.A. "RABOBANK NEDERLAND", NEW YORK BRANCH), as a Lender

By: /s/ ERIN THOMAS-WALKER

Name: Erin Thomas-Walker

Title: Vice President

By: /s/ PETER DUNCAN

Name: Peter Duncan

Title: Managing Director

TD BANK, N.A., as a Lender

By: /s/ MICHELE DRAGONETTI

Name: Michele Dragonetti

Title: Senior Vice President

U.S. BANK NATIONAL ASSOCIATION, as a Lender

By: <u>/s/ JEFFREY E. HASTINGS</u>

Name: Jeffrey E. Hastings

Title: Market President, Central Ohio

THE BANK OF NOVA SCOTIA, as a Lender

By: /s/ RAFAEL TOBON

Name: Rafael Tobon

Title: Director

CITIZENS BANK OF PENNSYLVANIA, as a Lender

By: <u>/s/ CARL S. TABACJAR, JR.</u>

Name: Carl S. Tabacjar, Jr.

Title: Vice President

FIFTH THIRD BANK, as a Lender

By: /s/ MICHAEL J. SCHALTZ, JR.

Name: Michael J. Schaltz, Jr.

Title: Vice President

COMPASS BANK, as a Lender

By: /s/ SANDRA CENTA

Name: Sandra Centa

Title: Senior Vice President

SUMITOMO MITSUI BANKING CORPORATION, as a Lender

By: /s/ DAVID W. KEE

Name: David W. Kee

Title: Managing Director

THE NORTHERN TRUST COMPANY, as a Lender

By: <u>/s/ JOHN DI LEGGE</u>

Name: John Di Legge

Title: Vice President

GREENSTONE FARM CREDIT SERVICES, ACA/FLCA, as a Lender

By: /s/ SHANE PRICHARD

Name: Shane Prichard

Title: Lending Officer

By: <u>/s/ JERRY LI</u>

Name: Jerry Li

Title: <u>Authorized Signatory</u>

TRISTATE CAPITAL BANK, as a Lender

By: <u>/s/ ELLEN FRANK</u>

Name: Ellen Frank

Title: Senior Vice President

EXHIBIT A

Consent and Reaffirmation

- Save for the Administrative Agent (as defined below), each of the undersigned hereby acknowledges receipt of a copy of the foregoing (a) Amendment No. 1 to the Fourth Amended and Restated Credit Agreement (as the same may be amended, restated, supplemented or otherwise modified from time to time, the "Credit Agreement"), dated as of October 29, 2015, by and among The Scotts Miracle-Gro Company, an Ohio corporation (the "Company"), The Scotts Company LLC, an Ohio limited liability company, Scotts Australia PTY Ltd., a company incorporated in Australia, Scotts Canada Ltd., a company organized under the laws of Canada, Scotts Holdings Limited, a private limited company incorporated in England and Wales, The Scotts Company (UK) Limited, a private limited company incorporated in England and Wales, Scotts Treasury EEIG, a European economic interest grouping (each together with the Company and the other Subsidiary Borrowers, the "Borrowers"), the Lenders and JPMorgan Chase Bank, N.A., as Administrative Agent (the "Administrative Agent"), which Amendment No. 1 is dated as of February 8, 2016 and is by and among the Borrowers, the financial institutions listed on the signature pages thereof and the Administrative Agent (the "Amendment"). Capitalized terms used in this Consent and Reaffirmation and not defined herein shall have the meanings given to them in the Credit Agreement. Without in any way establishing a course of dealing by the Administrative Agent or any Lender, each of the undersigned consents to the Amendment and reaffirms the terms and conditions of the Collateral Agreement and any other Loan Document executed by it and acknowledges and agrees that the Collateral Agreement and each and every such Loan Document executed by the undersigned in connection with the Credit Agreement remains in full force and effect and is hereby reaffirmed, ratified and confirmed. All references to the Credit Agreement contained in the above-referenced documents shall be a reference to the Credit Agreement as so modified by the Amendment and as the same may from time to time hereafter be amended, modified or restated.
- (b) In addition, notwithstanding any other provision of the Credit Agreement, Scotts-Sierra Investments, LLC hereby irrevocably and unconditionally undertakes to pay to the Administrative Agent, as creditor in its own right and not as representative of the other Lenders, sums equal to and in the currency of each amount payable by Scotts-Sierra Investments, LLC to each of the Lenders under each of the Loan Documents to which it is a party as and when that amount falls due for payment under the relevant Loan Document or would have fallen due but for any discharge resulting from failure of another Lender to take appropriate steps, in insolvency proceedings affecting Scotts-Sierra Investments, LLC, to preserve its entitlement to be paid that amount. The Administrative Agent shall have its own independent right to demand payment of the amounts payable by Scotts-Sierra Investments, LLC under this paragraph (b), irrespective of any discharge of Scotts-Sierra Investments, LLC's obligation to pay those amounts to the other Lenders resulting from failure by them to take appropriate steps, in insolvency proceedings affecting Scotts-Sierra Investments, LLC, to preserve their entitlement to be paid those amounts. Any amount due and payable by Scotts-Sierra Investments, LLC to the Administrative Agent under this paragraph (b) shall be decreased to the extent that the other Lenders have received (and are able to retain) payment in full of the corresponding amount under the other provisions of the relevant Loan Documents and any amount due and payable by Scotts-Sierra Investments, LLC to the other Lenders under those provisions shall be decreased to the extent that the Administrative Agent has received (and is able to retain) payment in full of the corresponding amount under this paragraph (b). The rights of the Lenders (other than the Administrative Agent) to receive payment of amounts payable by Scotts-Sierra Investments, LLC under the Loan Documents to which it is a party are several and are separate and independent

from, and without prejudice to, the rights of the Administrative Agent to receive payment under this paragraph (b). All amounts from time to time received or recovered by the Administrative Agent under this paragraph (b) and/or in connection with the realization and enforcement of all or any part of the Security Documents shall be held by the Administrative Agent in trust to apply in accordance with the terms of the relevant Loan Documents.

(c) This Consent and Reaffirmation shall be construed in accordance with and governed by the laws of the State of New York.

Dated February 8, 2016

[Signature Page Follows]

MIRACLE-GRO LAWN PRODUCTS, INC.

By: <u>/s/ THOMAS RANDAL</u> <u>COLEMAN</u> Name: Thomas Randal Coleman Title: EVP and Chief Financial Officer

OMS INVESTMENTS, INC.

By: <u>/s/ AIMEE M.</u> <u>DELUCA</u> Name: Aimee M. DeLuca Title: President and CEO

SCOTTS PRODUCTS CO.

By: <u>/s/ THOMAS RANDAL</u> <u>COLEMAN</u> Name: Thomas Randal Coleman Title: EVP and Chief Financial Officer

SCOTTS PROFESSIONAL PRODUCTS CO.

By: <u>/s/ THOMAS RANDAL</u> <u>COLEMAN</u> Name: Thomas Randal Coleman Title: EVP and Chief Financial Officer

SCOTTS-SIERRA INVESTMENTS LLC

By: <u>/s/ AIMEE M.</u> <u>DELUCA</u>

Name: Aimee M. DeLuca Title: President and CEO

SWISS FARMS PRODUCTS, INC.

By: <u>/s/ AIMEE M.</u> <u>DELUCA</u> Name: Aimee M. DeLuca Title: President and CEO

SANFORD SCIENTIFIC, INC.

By: <u>/s/ THOMAS RANDAL</u> <u>COLEMAN</u> Name: Thomas Randal Coleman Title: EVP and Chief Financial Officer

ROD MCCLELLAN COMPANY

By: <u>/s/ THOMAS RANDAL</u> <u>COLEMAN</u> Name: Thomas Randal Coleman Title: EVP and Chief Financial Officer

SMGM LLC

By: <u>/s/ THOMAS RANDAL</u> <u>COLEMAN</u> Name: Thomas Randal Coleman Title: EVP and Chief Financial Officer

SLS FRANCHISE SYSTEMS LLC

By: <u>/s/ JIM</u> <u>GIMESON</u>

Name: Jim Gimeson Title: President

GENSOURCE, INC.

By: <u>/s/ AIMEE M.</u> <u>DELUCA</u> Name: Aimee M. DeLuca Title: Secretary

HAWTHORNE HYDROPONICS LLC

By: <u>/s/ IVAN C.</u> <u>SMITH</u> Name: Ivan C. Smith Title: Secretary

HGCI, INC.

By: <u>/s/ AIMEE M.</u> <u>DELUCA</u> Name: Aimee M. DeLuca Title: Vice President

THE HAWTHORNE GARDENING COMPANY

By: <u>/s/ IVAN C.</u> <u>SMITH</u> Name: Ivan C. Smith

Title: Secretary

JPMORGAN CHASE BANK, N.A., as Administrative Agent

By:<u>/s/ TONY YUNG</u> Name: Tony Yung

Title: Executive Director

.

CONTRIBUTION AND DISTRIBUTION AGREEMENT

between

THE SCOTTS MIRACLE-GRO COMPANY

and

TRUGREEN HOLDING CORPORATION

dated as of December 10, 2015

ARTICLE I

THE TRANSACTIONS

Scotts Internal Reorganization	2
Formation of Newco	4
Contribution & Distributions	4
The Closing	4
Deliveries at Closing	5
Repayment of Certain Debt Prior to Closing	6
	Formation of Newco Contribution & Distributions The Closing Deliveries at Closing

ARTICLE II

CASH, INDEBTEDNESS AND WORKING CAPITAL ADJUSTMENTS

2.1	Closing Adjustment
-----	--------------------

6

ARTICLE III

REPRESENTATIONS AND WARRANTIES OF SABRETOOTH

3.1	Incorporation and Authority	10
3.2	Capitalization and Subsidiaries	11
3.3	No Conflict	11
3.4	Consents and Approvals	12
3.5	SLS Financial Statements	13
3.6	No Undisclosed Material Liabilities	13
3.7	Litigation	14
3.8	Employee Benefits; Labor	14
3.9	Taxes	17
3.10	Absence of Certain Changes or Events	18
3.11	Title to Assets; Sufficiency of Assets	19
3.12	Real Property	19
3.13	Intellectual Property	20
3.14	Contracts	21
3.15	Compliance with Laws; Permits	23
3.16	Anti-Corruption; Sanctions	24
3.17	Insurance	25
3.18	Environmental Matters	25
3.19	Financial Advisors	27
3.20	Transactions with Affiliates	27
3.21	Customers and Suppliers	27
3.22	Investment Intent	27

i

3.24 SLS Franchises

ARTICLE IV

REPRESENTATIONS AND WARRANTIES OF TIGER HOLDINGS

4.1	Incorporation and Authority	30
4.2	Capitalization and Subsidiaries	31
4.3	No Conflict	32
4.4	Consents and Approvals	32
4.5	TruGreen Financial Statements	33
4.6	No Undisclosed Material Liabilities	34
4.7	Litigation	34
4.8	Employee Benefits; Labor	34
4.9	Taxes	37
4.10	Absence of Certain Changes or Events	38
4.11	Title to Assets; Sufficiency of Assets	39
4.12	Real Property	39
4.13	Intellectual Property	40
4.14	Contracts	41
4.15	Compliance with Laws; Permits	43
4.16	Anti-Corruption; Sanctions	43
4.17	Insurance	44
4.18	Environmental Matters	45
4.19	Financial Advisors	46
4.20	Transactions with Affiliates	46
4.21	Customers and Suppliers	46
4.22	Financing	46
4.23	TruGreen Holdings Subsidiaries	47
4.24	TruGreen Holdings Acknowledgment	47
4.25	TruGreen Franchises	48

ARTICLE V

CERTAIN COVENANTS

5.1	Conduct of the SLS Business Prior to the Closing	50
5.2	Conduct of the TruGreen Entities Prior to the Closing	53
5.3	Efforts; Third-Party Approvals	55
5.4	Access to Information; Other Contacts	58
5.5	Books and Records; Access; Assistance	59
5.6	Confidentiality	60
5.7	Employees	61
5.8	Intercompany Arrangements	65
5.9	Insurance	66

5.10	Financing; Solvency Opinion	67
5.11	Debt Financing Cooperation	69
5.12	Intellectual Property	72
5.13	SLS Internal IT Systems and Data Separation; Replication of Shared Contracts	73
5.14	Certain Payments	74
5.15	Further Assurances	74
5.16	Transfers of Assets or Liabilities Following the Closing	74
5.17	Audited SLS Financial Statements	74
5.18	Franchises	75
5.19	Transition Services Agreement	75

ARTICLE VI

TAX MATTERS

6.4		=0
6.1	Tax Indemnification	76
6.2	Procedures Relating to Certain Tax Indemnification	78
6.3	Tax Returns	79
6.4	Transaction Taxes	79
6.5	Cooperation	80
6.6	Tax Sharing Agreements	80
6.7	Straddle Period	80
6.8	Computational Matters	81
6.9	Exclusivity	81

ARTICLE VII

CONDITIONS TO THE CLOSING

7.1	Conditions to Obligations of Scotts and the Non-SLS Subsidiaries	81
7.2	Conditions to Obligations of TruGreen Holdings	82
7.3	Frustration of Closing Conditions	83

ARTICLE VIII

CERTAIN RESTRICTIONS

8.1	Non-Competition	84
8.2	Non-Solicitation	85
8.3	Specific Performance	86
8.4	Severability	87

ARTICLE IX

INDEMNIFICATION

Indemnification by TruGreen Holdings	87
Indemnification by Scotts	88
Limitations on Indemnification	89
Sole Remedy	90
Computation of Indemnity Payments	91
Procedures for Indemnification	91
	Indemnification by Scotts Limitations on Indemnification Sole Remedy Computation of Indemnity Payments

ARTICLE X

TERMINATION AND WAIVER

10.1	Termination	93
10.2	Effect of Termination	94
10.3	Enforcement	96

ARTICLE XI

GENERAL PROVISIONS

11.1	Expenses	96
11.2	Notices	97
11.3	Severability	98
11.4	Entire Agreement	98
11.5	Assignment	99
11.6	No Third-Party Beneficiaries	99
11.7	Amendment; Waiver	99
11.8	Governing Law; Jurisdiction	99
11.9	Public Announcements	100
11.10	Waiver of Jury Trial	100
11.11	Disclosure Generally	101
11.12	No Presumption Against Drafting Party	101
11.13	Time Periods	101
11.14	Execution of Agreement	102
11.15	Scotts and TruGreen Holdings	102
11.16	No Recourse.	102

ARTICLE XII

CERTAIN DEFINITIONS

12.1	Certain Defined Terms	103
12.2	Interpretation	121

iv

EXHIBIT A – FORM OF LLC AGREEMENT

EXHIBIT B – FORM OF MERGER AGREEMENT

EXHIBIT C – FORM OF FINANCIAL STATEMENTS CERTIFICATE

EXHIBIT D – CALCULATION PRINCIPLES

EXHIBIT E – FORM OF CLARK CONSULTING AGREEMENT

EXHIBIT F – FORM OF CLARK INDEMNIFICATION AGREEMENT

EXHIBIT G – FORM OF SABRETOOTH CONSULTING AGREEMENT

EXHIBIT H – FORM OF SABRETOOTH INDEMNIFICATION AGREEMENT

v

CONTRIBUTION AND DISTRIBUTION AGREEMENT

This CONTRIBUTION AND DISTRIBUTION AGREEMENT, dated as of December 10, 2015 (this "<u>Agreement</u>"), is entered into by and between The Scotts Miracle-Gro Company, an Ohio corporation ("<u>Scotts</u>"), and TruGreen Holding Corporation, a Delaware corporation ("<u>TruGreen Holdings</u>" and, together with Scotts, the "<u>Parties</u>").

WHEREAS, Scotts conducts the SLS Business through its Subsidiary, EG Systems, Inc. (prior to or following the Conversion (defined below), as the context may require, "Scotts Systems");

WHEREAS, TruGreen Holdings conducts business exclusively through TruGreen Limited Partnership, a Delaware limited partnership ("<u>TruGreen LP</u>"), and the other TruGreen Entities;

WHEREAS, in connection with the transactions contemplated by this Agreement and prior to the Closing, TruGreen Holdings intends to (<u>1</u>) cause the formation of a newly-formed Delaware limited liability company ("<u>Newco</u>"), all of the equity interests of which outstanding prior to the Closing (the "<u>TruGreen Newco Interests</u>") will be directly or indirectly held by TruGreen Holdings and (<u>2</u>) cause 100% of the equity interests in each of the TruGreen Entities to be directly or indirectly held by Newco prior to the Closing;

WHEREAS, in connection with the transactions contemplated by this Agreement and prior to the Closing, Scotts intends to (<u>1</u>) transfer all the stock of Scotts Systems to a newly formed Delaware corporation, (2) immediately thereafter, cause the conversion of Scotts Systems from a corporation into a limited liability company and (<u>3</u>) cause the transfer of all of the SLS Assets held by Scotts and the Non-SLS Subsidiaries to Scotts Systems;

WHEREAS, it is intended that, at the Closing, Scotts will contribute all of the equity interests in Scotts Systems after the Conversion (the "<u>SLS Interests</u>") to Newco in exchange for the Scotts Newco Interests, and TruGreen Holdings (and/or one or more of its Subsidiaries that directly hold the TruGreen Newco Interests, the "<u>TruGreen Newco Holders</u>") and Scotts will enter into a limited liability company agreement of Newco substantially in the form attached hereto as <u>Exhibit A</u> (with only those changes as may be agreed by each of Scotts and TruGreen Holdings acting in good faith, the "<u>LLC Agreement</u>");

WHEREAS, immediately following the contribution of the SLS Interests to Newco, TruGreen Holdings intends to cause Newco to contribute, directly or indirectly, the SLS Interests to TruGreen LP;

WHEREAS, substantially concurrently with the transactions described above, TruGreen Holdings intends to cause TruGreen LP to incur third-party indebtedness as more fully described below;

WHEREAS, immediately following the transactions described above, TruGreen Holdings intends to cause (<u>1</u>) TruGreen LP to distribute, directly or indirectly, all of the proceeds of the debt issuance described above, net of transaction expenses, to Newco, (<u>2</u>) Newco to in turn distribute such proceeds to Scotts and the TruGreen Newco Holders on a pro rata basis and (<u>3</u>) the TruGreen Newco Holders to in turn distribute, directly or indirectly, the portion of such proceeds received by the TruGreen Newco Holders to TruGreen Holdings;

WHEREAS, substantially concurrently with the execution of this Agreement, TruGreen Holdings intends to enter into an Agreement and Plan of Merger with CDRTG Parent, Inc. and CDRTG Merger Sub, Inc. ("<u>Merger Sub</u>") in the form attached hereto as <u>Exhibit B</u> (the "<u>Merger Agreement</u>"), pursuant to which, subject to the terms and conditions therein, Merger Sub will merge with and into TruGreen Holdings immediately following the closing of the transactions described above with TruGreen Holdings surviving as the surviving corporation (the "<u>Merger</u>");

WHEREAS, in connection with the transactions contemplated by this Agreement, Scotts desires, and TruGreen Holdings desires to cause Newco, to enter into a transition services agreement (the "<u>Transition Services Agreement</u>"), which will be negotiated following the date hereof in accordance with <u>Section 5.19</u>, to provide for certain services and other arrangements among Scotts and the Non-SLS Subsidiaries, on the one hand, and Newco and its Subsidiaries (including the SLS Entities), on the other hand, following the closing of the transactions contemplated by this Agreement as more fully described herein and therein;

WHEREAS, in connection with the transactions contemplated by this Agreement, Scotts desires, and TruGreen Holdings desires to cause Newco, to enter into an employee leasing agreement, which will be negotiated following the date hereof in accordance with <u>Section 5.7(c)</u>, to provide for certain employee leasing arrangements among Scotts and the Non-SLS Subsidiaries, on the one hand, and Newco and its Subsidiaries (including the SLS Entities), on the other hand, following the closing of the transactions contemplated by this Agreement as more fully described herein and therein; and

WHEREAS, in connection with the transactions contemplated by this Agreement, Scotts desires, and TruGreen Holdings desires to cause Newco, to enter into a trademark licensing agreement, which will be negotiated following the date hereof in accordance with <u>Section 5.12(b)</u>, to provide for the licensing of certain trademarks by Scotts and/or one or more of the Non-SLS Subsidiaries to Newco and its Subsidiaries (including the SLS Entities) following the closing of the transactions contemplated by this Agreement as more fully described herein and therein.

NOW, THEREFORE, in consideration of the mutual agreements and covenants hereinafter set forth, and for good and valuable consideration, the receipt, sufficiency and adequacy of which are hereby acknowledged, the Parties hereby agree as follows:

ARTICLE I

THE TRANSACTIONS

1.1 Scotts Internal Reorganization.

The following transactions shall be consummated at any time, and in any order, prior to the Contribution (collectively, the "Scotts Internal Reorganization"):

(a) Scotts shall (<u>i</u>) form a newly-formed Delaware corporation ("<u>SLS Holdings</u>"), (<u>ii</u>) contribute all of the stock of Scotts Systems to SLS Holdings and (<u>iii</u>) immediately thereafter cause Scotts Systems to convert to a Delaware or Indiana limited liability company (the "<u>Conversion</u>"); and

(b) Except as otherwise expressly provided in this Agreement (including in <u>Section 5.3(c)</u> and <u>Section 5.7</u>) or any Ancillary Agreement, (i) Scotts shall, and shall cause the Non-SLS Subsidiaries to, assign, transfer, convey and deliver to one or more of the SLS Entities, all of such entities' right, title and interest in and to all of the SLS Assets held by Scotts and the Non-SLS Subsidiaries, (ii) Scotts shall cause the SLS Entities to accept, assume and agree to perform, discharge and fulfill all SLS Liabilities, in each case, regardless of (A) when or where such Liabilities arose or arise and (B) whether the facts upon which they are based occurred prior to, on or subsequent to the Scotts Internal Reorganization, and (iii) Scotts shall, or shall cause one or more Non-SLS Subsidiaries to, accept, assume and agree to perform, discharge and fulfill all Retained Liabilities.

Pursuant to this Section 1.1 and unless otherwise set forth in this Agreement (including in Section 5.3(c) and Section 5.7) or any Ancillary Agreement, from and after the Scotts Internal Reorganization, (\underline{x}) the SLS Entities, taken together, shall hold, or have a right to hold, all right, title and interest in the SLS Assets (in each case, to the extent of any right, title and interest held by Scotts or any of its Subsidiaries prior to Closing, as applicable) and (\underline{y}) Scotts or one or more Non-SLS Subsidiaries shall be liable for, and shall faithfully perform, fulfill and discharge fully in due course in accordance with their respective terms, all Retained Liabilities, in each case, regardless of (\underline{i}) when or where such Liabilities arose or arise, (\underline{ii}) whether the facts upon which they are based occurred prior to, on or subsequent to the Scotts Internal Reorganization, (\underline{iii}) where or against whom such Liabilities are asserted or determined (including any Liabilities arising out of claims made by any of Scotts or the Non-SLS Subsidiaries, on the one hand, or any SLS Entity, on the other, or any of their past or present respective directors, officers, employees, agents, Subsidiaries or Affiliates against any of Scotts or the Non-SLS Subsidiaries or any SLS Entity, as the case may be) and (\underline{iv}) whether arising from or alleged to arise from negligence, recklessness, violation of Law, fraud, misrepresentation or otherwise, and (\underline{z}) the SLS Entities shall be solely liable for (and Scotts and the Non-SLS Subsidiaries shall have no liability for) and shall faithfully perform, fulfill and discharge fully in due course in accordance with their respective terms, all SLS Liabilities, in each case, regardless of (\underline{i}) when or where such SLS Liabilities arose or arise, (\underline{ii}) whether the facts upon which they are based occurred prior to, on or subsequent to the Scotts Internal Reorganization, (\underline{iii}) where or against whom such SLS

Liabilities are asserted or determined (including any SLS Liabilities arising out of claims made by any of Scotts or the Non-SLS Subsidiaries, on the one hand, or any SLS Entity, on the other, or any of their past or present respective directors, officers, employees, agents, Subsidiaries or Affiliates against any of Scotts or the Non-SLS Subsidiaries or any SLS Entity, as the case may be) and (<u>iv</u>) whether arising from or alleged to arise from negligence, recklessness, violation of Law, fraud, misrepresentation or otherwise. For the avoidance of doubt, nothing in this <u>Section 1.1</u> is intended to limit any indemnification rights and obligations set forth in <u>Article IX</u>.

1.2 Formation of Newco.

Prior to the Closing, TruGreen Holdings shall form Newco as a direct or indirect wholly owned Subsidiary and shall cause Newco to directly or indirectly own 100% of the equity interests of each of the TruGreen Entities (such transactions, the "<u>Newco</u> <u>Restructuring</u>").

1.3 <u>Contribution & Distributions</u>.

(a) At the Closing, subject to the terms and conditions set forth in this Agreement, Scotts shall cause SLS Holdings to assign and transfer to Newco, and TruGreen Holdings shall cause Newco to acquire, all of SLS Holdings' right, title and interest in and to the SLS Interests, free and clear of all Encumbrances other than (<u>i</u>) Encumbrances imposed by federal and state securities Laws and (<u>ii</u>) Encumbrances that may be created by or on behalf of Newco or any of its Subsidiaries (excluding, for the avoidance of doubt, the SLS Entities) (the "<u>Contribution</u>"). Immediately following the Contribution, TruGreen Holdings shall cause Newco to contribute, directly or indirectly, the SLS Interests to TruGreen LP.

(b) At the Closing, subject to the terms and conditions set forth in this Agreement, concurrently with, and in consideration of, the Contribution, TruGreen Holdings shall cause Newco to issue (the "<u>Issuance</u>") to Scotts limited liability company interests (of the same class as that held, directly or indirectly, by TruGreen Holdings) in Newco (the "<u>Scotts Newco</u> <u>Interests</u>") representing a percentage of the outstanding limited liability company interests in Newco as of the Closing equal to the Scotts Ownership Percentage, disregarding any interests in Newco issued on the Closing Date in connection with the "<u>Equity</u> <u>Incentive Plan</u>" (as defined in the LLC Agreement).

(c) At the Closing and immediately following the Contribution and the Issuance, TruGreen Holdings shall cause (<u>i</u>) TruGreen LP to incur the Debt Financing, (<u>ii</u>) TruGreen LP to distribute the Financing Proceeds, directly or indirectly, to Newco, (<u>iii</u>) Newco to distribute the Financing Proceeds on a pro rata basis to the TruGreen Newco Holders and Scotts, disregarding any interests in Newco issued on the Closing Date in connection with the "<u>Equity Incentive Plan</u>" (as defined in the LLC Agreement) (which, for the avoidance of doubt, shall be issued after the Distributions) and (iv) the TruGreen Newco Holders to distribute the portion of the Financing Proceeds received by the TruGreen Newco Holders in clause (iii) above, directly or indirectly, to TruGreen Holdings (such distributions, the "<u>Distributions</u>" and, together with the Contribution and Issuance, the "<u>Transactions</u>").

1.4 The Closing.

Unless this Agreement shall have been terminated pursuant to <u>Article X</u>, the closing of the Transactions (the "<u>Closing</u>") shall take place at 10:00 a.m. New York time at the offices of Debevoise & Plimpton, 919 Third Avenue, New York, New York, (<u>a</u>) on the third Business Day after each of the conditions set forth in <u>Article VII</u> has been satisfied or waived (other than those conditions that by their terms are to be satisfied by actions taken at the Closing, but subject to the satisfaction or written waiver of those conditions), or (<u>b</u>) at such other date, time or place as is agreed to in writing by the Parties. The date on which the Closing occurs is the closing date (the "<u>Closing Date</u>"). Unless the Parties agree otherwise, the Closing will be deemed to have occurred at 11:59 p.m. New York time on the day immediately prior to the Closing Date.

1.5 <u>Deliveries at Closing</u>.

(a) At or prior to the Closing, Scotts shall deliver or cause to be delivered to TruGreen Holdings the following:

(i) one or more transfer agreements (the "<u>Assignment Agreements</u>") evidencing the transfer of the SLS Assets held by Scotts and the Non-SLS Subsidiaries from such Persons to one or more SLS Entities and evidencing the assumption of the Retained Liabilities by Scotts or one or more Non-SLS Subsidiaries, each in a form reasonably satisfactory to TruGreen Holdings;

(ii) a certificate of conversion reflecting the Conversion and any other similar filings required by the State of Delaware or Indiana to effectuate the Conversion, as applicable, in a form evidencing the filing and effectiveness thereof with the Secretary of State of the State of Delaware and/or Indiana, as applicable;

(iii) an executed equity interest transfer instrument and equity interest certificates evidencing the transfer of the SLS Interests from SLS Holdings to Newco;

- (iv) the duly executed certificate to be delivered by Scotts pursuant to <u>Section 7.2(c)</u>;
- (v) the LLC Agreement duly executed by Scotts;
- (vi) the Transition Services Agreement duly executed by Scotts;
- (vii) the Employee Leasing Agreement duly executed by Scotts;

(viii) the Trademark Licensing Agreement duly executed by Scotts and/or one or more of the Non-SLS Subsidiaries;

(ix) each other Ancillary Agreement to which Scotts or any of its Subsidiaries is a party executed by a duly authorized representative of Scotts and/or its Subsidiaries, as applicable;

(x) a statement meeting the requirements of Section 1.1445-2(b) of the Treasury Regulations, to the effect that SLS Holdings is not a "foreign person" within the meaning of Section 1445 of the Code and the Treasury Regulations thereunder; and

(xi) evidence of the resignation or removal, effective as of the Closing Date, of each Person who is an officer or a director of any SLS Entity immediately prior to the Closing Date and who will be an employee or officer of Scotts or any Non-SLS Subsidiary immediately after the Closing Date.

(b) At or prior to the Closing, TruGreen Holdings shall deliver or cause to be delivered to Scotts the following:

(i) the duly executed certificate to be delivered by TruGreen Holdings pursuant to <u>Section 7.1(c)</u>;

(ii) the certificate of formation of Newco in a form evidencing the filing and effectiveness thereof with the Secretary of State of the State of Delaware, and copies of executed documentation evidencing the transfer to Newco of all of the equity interests in the other TruGreen Entities;

(iii) the LLC Agreement duly executed by the TruGreen Newco Holders;

- (iv) the certificate of incorporation of Parent;
- (v) the Transition Services Agreement duly executed by Newco;
- (vi) the Employee Leasing Agreement duly executed by Newco;

(vii) the Trademark Licensing Agreement duly executed by Newco;

(viii) evidence that the Scotts Directors (as defined in the LLC Agreement) shall be appointed to the board of directors of Newco and the board of directors or similar governing body of each TruGreen Entity and SLS Entity, in each case effective as of the close of business on the Closing Date; and

(ix) each other Ancillary Agreement to which TruGreen Holdings or any of its Subsidiaries is a party executed by a duly authorized representative of TruGreen Holdings and/or any of its Subsidiaries, as applicable.

1.6 <u>Repayment of Certain Debt Prior to Closing</u>.

Each Party shall use its reasonable best efforts to pay, satisfy and discharge Indebtedness (disregarding, for the purposes of this <u>Section 1.6</u>, clause (iv) of the definition of "Indebtedness") as of immediately prior to the Closing to the extent necessary to cause the Closing Indebtedness (disregarding, for the purposes of this <u>Section 1.6</u>, clause (iv) of the definition of "Indebtedness" in the definition of "Closing Indebtedness") of the SLS Entities (in the case of Scotts) or the TruGreen Entities (in the case of TruGreen Holdings) to be zero.

ARTICLE II

CASH, INDEBTEDNESS AND WORKING CAPITAL ADJUSTMENTS

2.1 Closing Adjustment.

(a) Not more than 15 and not less than four Business Days prior to the anticipated Closing Date, (\underline{x}) Scotts shall deliver to TruGreen Holdings with respect to the SLS Entities and (\underline{y}) TruGreen Holdings shall deliver to Scotts with respect to the TruGreen Entities, its (\underline{i}) reasonable good faith estimate of such entities' Closing Working Capital, prepared in accordance with the Calculation Principles ("<u>Estimated Closing Working Capital</u>"), (\underline{i}) reasonable good faith estimate of such entities' Closing Indebtedness ("Estimated Closing <u>Working Capital</u>"), (\underline{i}) reasonable good faith estimate of such entities' Closing Indebtedness ("<u>Estimated Closing Working Capital</u>"), (\underline{i}) is a calculation of the Closing Working Capital of the SLS Entities, and contained in <u>TruGreen Schedule 2.1(a)</u> is a calculation of the Closing Working Capital of the TruGreen Entities (each such group of entities, a "<u>Group</u>") assuming the Closing Date occurred on September 30 and based on information available as of the date of this Agreement.

(b) The "<u>Closing Adjustment</u>" shall be, (<u>x</u>) with respect to Scotts, an amount equal to the following with all amounts in respect of the SLS Entities and (<u>y</u>) with respect to TruGreen Holdings, an amount equal to the following with all amounts in respect of the TruGreen Entities:

(i) the amount of Estimated Closing Cash, less

(ii) the amount of Estimated Closing Indebtedness, plus

(iii) if the Estimated Closing Working Capital is greater than the applicable Reference Amount, an amount equal to such excess, and *less*

(iv) if the Estimated Closing Working Capital is less than the applicable Reference Amount, an amount equal to such deficit.

(c) If a Party's Closing Adjustment is less than zero, such Party shall contribute, or cause to be contributed, by wire transfer of immediately available funds, an

amount equal to the absolute value of such Closing Adjustment to Newco immediately following the Transactions. If a Party's Closing Adjustment is greater than zero, TruGreen Holdings shall cause Newco to distribute, by wire transfer of immediately available funds, the amount of such Closing Adjustment to such Party (or to such Person(s) as may be specified in writing by such party) immediately following the Transactions.

(d) Within 90 calendar days after the Closing, Newco shall prepare and deliver to Scotts and TruGreen Holdings a statement (the "<u>Post-Closing Statement</u>"), prepared in accordance with the Calculation Principles, setting forth, with respect to each Group, (i) Closing Working Capital, (ii) Closing Cash, (iii) Closing Indebtedness and (iv) a detailed calculation of each of the aforementioned. The Post-Closing Statement shall not introduce any new or different accounting methods, policies, practices, procedures, classifications, judgments or estimation methodologies other than those set forth in the Calculation Principles.

(e) Each Party shall (i) permit the other Party and its representatives to have reasonable access to the books, records and other documents of such Party and its Subsidiaries (including work papers, schedules, financial statements and memoranda) (x) pertaining to and/or (y) used or to be used in connection with, the preparation of the Post-Closing Statement and the calculation of the Closing Working Capital, Closing Cash and Closing Indebtedness, in each case, in respect of either Group, and provide the other Party with copies thereof and (ii) provide the other Party and its representatives reasonable access to its and its Subsidiaries' employees, accountants and other representatives as reasonably requested by the other Party in order to prepare or verify the accuracy of the Post-Closing Statement. Scotts shall notify TruGreen Holdings in writing of its acceptance or dispute of any amounts reflected on the Post-Closing Statement within 45 calendar days after its receipt of such statement. Any such notice of disagreement (the "Notice of Disagreement") shall specify those items or amounts as to which Scotts disagrees and describe the basis of such disagreement (and shall include Scotts's proposed changes to the calculation of Closing Working Capital, Closing Cash, and/or Closing Indebtedness, in each case, in respect of either Group, as applicable). Scotts shall be deemed to have agreed with all items and amounts included in the Post-Closing Statement delivered pursuant to <u>Section 2.1(c)</u> that it does not specifically identify in the Notice of Disagreement. Notwithstanding anything to the contrary in this Section 2.1(e), if, based on the information provided to Scotts, Scotts is unable to specify the items or amounts as to which it disagrees, describe the basis of such disagreement or propose specified changes, Scotts may indicate as such in its Notice of Disagreement and TruGreen Holdings and Newco shall use their reasonable best efforts to provide the requisite information as promptly as practicable in order for Scotts to provide such determination (and Scotts shall use its reasonable best efforts to deliver a revised Notice of Disagreement with specified items, amounts and changes as promptly as practicable), and the 45-day review period shall be extended to the extent reasonably necessary to allow Scotts to deliver a revised Notice of Disagreement.

(f) In the event of a dispute with respect to the Post-Closing Statement, the Parties shall negotiate in good faith to reconcile their differences and any resolution by them as to any disputed amounts shall be final, binding and conclusive on the parties. If the Parties are unable to reach a resolution to such effect within 30 calendar days after TruGreen Holdings's

receipt of the Notice of Disagreement, the Parties shall submit the amounts remaining in dispute for resolution to the New York, New York office of Grant Thornton LLP or such other independent accounting firm of international reputation as is mutually agreed to and appointed by the Parties (such independent accounting firm being herein referred to as the "<u>Accounting Firm</u>"). The Parties shall use their commercially reasonable efforts to cause the Accounting Firm to, as soon as practicable but in any event within 30 calendar days after such submission, determine and report to the Parties upon the remaining disputed amounts with respect to the Post-Closing Statement, and such report shall be final, binding and conclusive on the Parties and shall constitute an arbitral award upon which a judgment may be entered in any court having jurisdiction thereof. The Accounting Firm shall be authorized to resolve only those items remaining in dispute between the Parties, within the range of the difference between each Party's position with respect thereto, and such resolution shall be based solely on the written materials submitted by the Parties and the Calculation Principles and not on independent review. Each Party shall bear and pay a percentage of the fees and disbursements of the Accounting Firm that is equal to the percentage of the total disputed dollar amount determined in favor of the other Party by the Accounting Firm.

(g) No later than four Business Days after the date that Closing Working Capital, Closing Cash and Closing Indebtedness, in each case, with respect to each Group, shall be finally determined in accordance with <u>Sections 2.1(d)</u> and (<u>f)</u> (the "<u>Final Determination Date</u>"), a Party or Newco, as applicable, shall make the following payments (each such payment, a "<u>Post-Closing Adjustment Amount</u>"):

(i) If the Closing Working Capital of a Group as finally determined in accordance with <u>Sections 2.1(d)</u> and (<u>f</u>) is less than the Estimated Closing Working Capital of such Group, Scotts, in the case of the SLS Entities, or TruGreen Holdings, in the case of the TruGreen Entities, shall contribute the amount of such shortfall to Newco.

(ii) If the Closing Working Capital of a Group as finally determined in accordance with <u>Sections 2.1(d)</u> and (f) is greater than the Estimated Closing Working Capital of such Group, TruGreen Holdings shall cause Newco to distribute the amount of such excess to Scotts, in the case of the SLS Entities, or TruGreen Holdings, in the case of the TruGreen Entities.

(iii) If the Closing Cash of a Group as finally determined in accordance with <u>Sections 2.1(d)</u> and (<u>f</u>) is less than the Estimated Closing Cash of such Group, Scotts, in the case of the SLS Entities, or TruGreen Holdings, in the case of the TruGreen Entities, shall contribute the amount of such difference to Newco.

(iv) If the Closing Cash of a Group as finally determined in accordance with <u>Sections 2.1(d)</u> and <u>(f)</u> is greater than the Estimated Closing Cash of such Group, TruGreen Holdings shall cause Newco to distribute the amount of such difference to Scotts, in the case of the SLS Entities, or TruGreen Holdings, in the case of the TruGreen Entities.

(v) If the Closing Indebtedness of a Group as finally determined in accordance with <u>Sections 2.1(d)</u> and <u>(f)</u> is less than the Estimated Closing Indebtedness of such Group, TruGreen Holdings shall cause Newco to distribute the amount of such difference to Scotts, in the case of the SLS Entities, or TruGreen Holdings, in the case of the TruGreen Entities.

(vi) If the Closing Indebtedness of a Group as finally determined in accordance with <u>Sections 2.1(d)</u> and <u>(f)</u> is greater than the Estimated Closing Indebtedness of such Group, Scotts, in the case of the SLS Entities, or TruGreen Holdings, in the case of the TruGreen Entities, shall contribute the amount of such difference to Newco.

(h) Any payment to be made as a result of a Post-Closing Adjustment Amount pursuant to <u>Section 2.1(h)</u> shall be paid by wire transfer of immediately available funds on the fourth Business Day following the Final Determination Date, interest thereon for the period commencing on the fourth Business Day following the Final Determination Date through the date on which such payment is made calculated at the Prime Rate. Such interest shall be payable at the same time as the payment to which it relates and shall be calculated daily on the basis of a year of 365 days and the actual number of days elapsed. The aggregate payments to be made by a Party or Newco to the other as a result of all Post-Closing Adjustment Amounts pursuant to <u>Section 2.1(g)</u> and this <u>Section 2.1(h)</u> shall be netted against all Post-Closing Adjustment Amounts owed to such Person by the other as a result of Post-Closing Adjustment Amounts over to be made by TruGreen Holdings to Newco, or by Newco to TruGreen Holdings, pursuant to <u>Section 2.1(g)</u> and this <u>Section 2.1(h)</u> shall be made directly or indirectly, as determined by TruGreen Holdings.

(i) Any payment made pursuant to this <u>Article II</u> shall be treated as a contribution to, or a distribution from, Newco for tax purposes.

ARTICLE III

REPRESENTATIONS AND WARRANTIES OF SABRETOOTH

Except as disclosed in the corresponding section or sub-section of the disclosure schedules of Scotts attached hereto (collectively, the "<u>Scotts Schedules</u>"), Scotts hereby represents and warrants to TruGreen Holdings as follows:

3.1 Incorporation and Authority.

(a) Each of Scotts, the Non-SLS Subsidiaries conducting any SLS Business and the SLS Entities is an entity duly organized, validly existing and (where such concept is applicable) in good standing (or local equivalent) under the Laws of its jurisdiction of organization, formation or incorporation, as applicable, and has all requisite corporate power and

authority to conduct its business as currently conducted. Except as would not constitute a SLS Material Adverse Effect, each of Scotts, the Non-SLS Subsidiaries conducting any SLS Business and the SLS Entities is (i) duly qualified or licensed as a foreign corporation for the transaction of business and (ii) (where such concept is applicable) in good standing under the laws of each other jurisdiction in which the character of properties it owns or leases, or conducts any SLS Business, makes such qualification or license necessary.

(b) Scotts has made available to TruGreen Holdings complete copies of the respective articles of incorporation or bylaws (or similar organizational documents, as amended) of each of the SLS Entities as in effect as of the date hereof. None of the SLS Entities is in violation of any provision of such articles of incorporation or bylaws (or similar organizational documents).

(c) Each of Scotts, the Non-SLS Subsidiaries and the SLS Entities has all necessary corporate power and authority to enter into this Agreement and the Ancillary Agreements to which it will be a party, to carry out and perform its obligations hereunder and thereunder and to consummate the transactions contemplated on its part hereby and thereby. Scotts, the Non-SLS Subsidiaries and the SLS Entities have all necessary corporate power and authority to effect the Scotts Internal Reorganization. The Scotts Internal Reorganization, the execution and delivery by Scotts of this Agreement and by Scotts, the Non-SLS Subsidiaries and SLS Entities of the Ancillary Agreements to which they will be a party, and the consummation by Scotts, the Non-SLS Subsidiaries and the SLS Entities of the transactions contemplated on their part hereby and thereby, have been duly authorized by all necessary company action on the part of each of Scotts, the Non-SLS Subsidiaries and the SLS Entities. This Agreement has been duly executed and delivered by Scotts, and, to the extent Scotts or any Non-SLS Subsidiary is a party thereto, the Ancillary Agreements will at the Closing, be duly executed and delivered by Scotts or such Non-SLS Subsidiary, as applicable, and (assuming due authorization, execution and delivery by the other parties thereto) this Agreement constitutes a legal, valid and binding obligation of Scotts, and when executed and delivered, to the extent Scotts or any Non-SLS Subsidiary, as applicable, is a party thereto, the Ancillary Agreements will constitute, legal, valid and binding obligations of Scotts or such Non-SLS Subsidiary, enforceable against Scotts or such Non-SLS Subsidiary, as applicable, in accordance with their terms, except as enforceability may be limited by bankruptcy, insolvency, reorganization, moratorium or other similar Laws affecting creditors' rights and general principles of equity affecting the availability of specific performance and other equitable remedies (regardless of whether enforcement is sought in a proceeding at law or in equity).

3.2 Capitalization and Subsidiaries.

(a) Set forth on <u>Scotts Schedule 3.2(a)</u> is (<u>i</u>) the jurisdiction of incorporation or legal organization and (<u>ii</u>) as of the date hereof (A) the number of authorized, issued and outstanding shares or other equity securities of each of the SLS Entities (exclusive of the effects of the Transactions) and (B) there are no other authorized, issued or outstanding shares of capital stock or other equity securities of the SLS Entities (exclusive of the effects of the Transactions). All of the issued and outstanding SLS Interests and other

equity securities of each of the SLS Entities are (\underline{x}) legally and beneficially owned by Scotts (either directly or indirectly), (\underline{y}) held free and clear of any Encumbrances and (\underline{z}) have been validly issued, are fully paid and nonassessable and have not been issued in violation of any preemptive or similar rights.

(b) There are (and at the Closing, after giving effect to the Scotts Internal Reorganization and the Transactions, there will be) no outstanding options, warrants, calls, rights or any other agreements, trust, proxies or similar arrangements affecting the sale, issuance or voting of any equity securities of any SLS Entity, or any securities or other instruments convertible into, exchangeable for or evidencing the right to purchase any equity securities of any SLS Entity.

(c) No SLS Entity has any outstanding indebtedness for borrowed money, outstanding bonds, debentures, notes or other securities, the holders of which have the right to vote (or which are convertible into or exchangeable for securities having the right to vote) with the stockholders of the SLS Entities.

(d) None of the SLS Entities has (or at the Closing, after giving effect to the Scotts Internal Reorganization and the Transactions, will have) any Subsidiaries or own (or at the Closing, after giving effect to the Scotts Internal Reorganization and the Transactions, will own) any equity interests in any other Person, in each case, except for other SLS Entities.

3.3 No Conflict.

None of the Scotts Internal Reorganization, or the execution, delivery and performance by Scotts and the Non-SLS Subsidiaries of this Agreement or the Ancillary Agreements to which they will be a party, or the consummation of transactions contemplated hereby or thereby (including, for the avoidance of doubt, the Transactions and operation of the SLS Business as currently conducted after the Closing but excluding the effects of any TruGreen Franchise Agreement or other Contract to which a TruGreen Entity is a party) will:

(a) conflict with or violate the respective articles of incorporation or bylaws (or similar organizational documents) of any of Scotts, the Non-SLS Subsidiaries or the SLS Entities;

(b) assuming all Consents referred to in <u>Scotts Schedule 3.4</u> have been obtained, conflict with or violate any Law applicable to any of Scotts, the Non-SLS Subsidiaries or the SLS Entities, or to any of their respective properties, rights or assets;

(c) result in any breach of, or constitute a default (or event which with the giving of notice or lapse of time, or both, would become a default) under, or give to any Person any right of termination, amendment, rescission, acceleration or cancellation of, or any right of notice or Consent with respect to, any Contract, Permit, franchise or other instrument to which any of Scotts, the Non-SLS Subsidiaries or the SLS Entities is a party or by which any of the assets, properties or rights of any of Scotts, the Non-SLS Subsidiaries or the SLS Entities is

bound or affected (but with respect to Scotts and the Non-SLS Subsidiaries, only such assets, properties or rights relating to the SLS Business or the SLS Entities); or

(d) result in the creation or imposition of any Encumbrance, other than Permitted Encumbrances, on any SLS

Assets,

except, in the case of clauses (b), (c) and (d), where such conflicts, violations, rescissions, accelerations, breaches, defaults, terminations, cancellations or failures to obtain such Consents would not constitute a SLS Material Adverse Effect.

3.4 Consents and Approvals.

The execution and delivery by Scotts of this Agreement does not, and the Scotts Internal Reorganization and the execution and delivery by Scotts, the Non-SLS Subsidiaries and SLS Entities of the Ancillary Agreements, to the extent they are a party thereto, will not, and the performance by Scotts, the Non-SLS Subsidiaries and SLS Entities of this Agreement (including, for the avoidance of doubt, the Transactions) and the Ancillary Agreements, to the extent they are a party thereto, and the consummation of the transactions contemplated hereby and thereby will not, require any of Scotts, the Non-SLS Subsidiaries or the SLS Entities to seek or obtain any Consent from, or make any filing or notification with, any Governmental Body or other Person, except for (\underline{w}) those Consents that may be required for Scotts and its Affiliates to provide services pursuant to the Transition Services Agreement or to separate the Shared Contracts other than the Designated Shared Contracts (without giving effect to any updates to the list of Designated Shared Contracts), (\underline{x}) those Consents set forth on <u>Scotts Schedule 3.4(\underline{x})</u>, (\underline{y}) compliance with the applicable requirements of the HSR Act or any other Competition Laws of the jurisdictions set forth on <u>Scotts Schedule 3.4(\underline{y} </u>) and (\underline{z}) those Consents from, and filings with, Persons (other than Governmental Bodies) the failure of which to be made or obtained would not, constitute a SLS Material Adverse Effect.

3.5 <u>SLS Financial Statements</u>.

(a) The unaudited combined balance sheets of the SLS Business as at September 30, 2014 and September 30, 2015, and the related unaudited combined statements of operations, comprehensive loss, and cash flows of the SLS Business for the years ended September 30, 2013, September 30, 2014 and September 30, 2015, (collectively, the "<u>SLS Financial Statements</u>") are included in <u>Scotts Schedule 3.5(a)</u>. The SLS Financial Statements (i) have been prepared based on the separate books and records of the SLS Business, (<u>ii</u>) have been prepared in accordance with GAAP consistently applied throughout the periods indicated and (<u>iii</u>) present fairly in all material respects the combined financial position, the results of operations and cash flows of the SLS Business as of the respective dates thereof and for the respective periods indicated therein after taking account of the Scotts Internal Reorganization, except, in the case of clauses (ii) and (iii), (x) as otherwise noted therein and reflected in the SLS Financial Statements do not contain all footnotes required to be included in audited financials (collectively, with respect to the first sentence of this <u>Section</u> <u>3.5(a)</u>, the "<u>SLS</u>

<u>Unaudited Financial Statements Representation</u>"). Upon delivery of the Audited SLS Financial Statements in accordance with <u>Section 5.17</u> of this Agreement, the foregoing SLS Unaudited Financial Statements Representation shall be null and void and replaced in its entirety with the representation set forth in the third paragraph of the Financial Statements Certificate (other than for the definition of "SLS Financial Statements" to the extent referred to elsewhere in this Agreement).

(b) Scotts maintains, and has maintained since October 1, 2013, a system of internal control over financial reporting (as defined in Rules 13a-15(f) and 15d-15(f) of the Exchange Act) as required by Rule 13a-15 and Rule 15a-15 of the Exchange Act designed to ensure reasonable assurances regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with GAAP. Since October 1, 2013, until the date of this Agreement, Scotts has disclosed, based on the most recent evaluation of internal control prior to the date of this Agreement, to its auditors (<u>i</u>) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect in any material respect, Scotts's ability to record, process, summarize and report financial information and (<u>ii</u>) any fraud, whether or not material, that involves management or other employees who have a significant role in Scotts's internal control over financial reporting.

(c) Since October 1, 2013, no director or officer, or, to the Knowledge of Scotts, external auditor or external accountant of Scotts or its Subsidiaries (including the SLS Entities), or any employee (other than officers) or representative of Scotts or its Subsidiaries (including the SLS Entities), has received any material complaint, allegation, assertion or claim, whether written or oral, regarding the accounting or auditing practices, procedures, methodologies or methods of the SLS Business or the SLS Entities that would reasonably be expected to, in each case, impact the consolidated financial statements of Scotts or their respective internal accounting controls, including any material complaint, allegation, assertion or claim that any SLS Entity has engaged in questionable accounting or auditing practices.

3.6 No Undisclosed Material Liabilities.

No SLS Entity has, after giving effect to the completion of the Scotts Internal Reorganization and the Transactions, any Liabilities, whether known, unknown, absolute, accrued, contingent or otherwise and whether due or to become due, in each case that are required under GAAP to be reflected or reserved against in the combined financial statements of the SLS Business, except (<u>a</u>) as set forth in <u>Scotts Schedule 3.6</u>, (<u>b</u>) Liabilities disclosed or reserved against in the SLS Financial Statements as at the Balance Sheet Date, (<u>c</u>) Liabilities that were incurred after the Balance Sheet Date in the ordinary course of business, (<u>d</u>) Liabilities which, individually or in the aggregate, would not be material to the SLS Business, taken as a whole, (<u>e</u>) Liabilities that are Tax liabilities incurred in connection with the Scotts Internal Reorganization or the Transactions and (<u>f</u>) Liabilities that are specifically contemplated by, or are pursuant to this Agreement or the Ancillary Agreements and incurred on or after the date hereof. No SLS Entity is a party to, or has any commitment to become a party to, any joint venture, off-balance sheet partnership or any similar Contract (including any Contract relating to any

transaction or relationship between or among the SLS Entities, on the one hand, and Scotts or any Non-SLS Subsidiary, including any structured finance, special purpose or limited purpose Person, on the other hand, or any "off-balance sheet arrangement" (as defined in Item 303(a) of Regulation S-K of the SEC)).

3.7 Litigation.

As of the date hereof, there is no claim, suit, action, litigation, arbitration, mediation, investigation or proceeding (administrative or judicial) by or before any Governmental Body ("<u>Proceeding</u>") pending or, to the Knowledge of Scotts, threatened against Scotts or any of its Subsidiaries (including the SLS Entities), or any of their respective assets, rights or properties which bring into question the validity of this Agreement or that would reasonably be expected, individually or in the aggregate, to be materially adverse to the SLS Business or the SLS Entities, taken as a whole. There are no Governmental Orders seeking or purporting to enjoin or restrain the execution, delivery and performance by Scotts or any Non-SLS Subsidiary of this Agreement or that would reasonably be expected, individually or that would reasonably be expected, individually or in the aggregate, to be materially adverse to the SLS Business or the SLS Entities (including the SLS Entities) of the transactions contemplated hereby or that would reasonably be expected, individually or in the aggregate, to be materially adverse to the SLS Business or the SLS Entities, taken as a whole.

3.8 Employee Benefits; Labor.

(a) <u>Scotts Schedule 3.8(a)</u> lists, as of the date of this Agreement, each material Employee Plan (other than a Scotts Benefit Plan) that any SLS Entity sponsors, maintains or contributes to, or is required to maintain or contribute to for the benefit of any current or former SLS Employee (without regard to materiality, the "<u>SLS Entity Benefit Plans</u>"). No SLS Entity Benefit Plan provides benefits to any individual other than current or former employees of the SLS Entities, Persons who provide or have provided services in respect of the SLS Business and any beneficiaries and dependents of the foregoing.

(b) <u>Scotts Schedule 3.8(b)</u> lists, as of the date of this Agreement, each material Employee Plan that Scotts or any Non-SLS Subsidiary maintains or contributes to, or is required to maintain or contribute to as of the date hereof for the benefit of any current or former SLS Employee and any other current or former employee of Scotts or any Non-SLS Subsidiary (without regard to materiality, the "<u>Scotts Benefit Plans</u>," and, together with the SLS Entity Benefit Plans, the "<u>SLS Business Benefit Plans</u>").

(c) With respect to each material SLS Entity Benefit Plan, Scotts has provided TruGreen Holdings, to the extent applicable, complete and correct copies of: all written plan documents and descriptions of all unwritten plans (and all amendments thereto); all trust agreements, insurance contracts or other funding arrangements therefor; the two most recent actuarial and trust reports; the two most recent Forms 5500 and all schedules thereto; the most recent IRS determination letters; all current summary plan descriptions and summaries of material modification (or with respect to the foregoing items, the equivalent thereto required by any foreign Governmental Body); and copies of all material communications, received from or sent to any Governmental Body, relating to SLS Entity Benefit Plans in the last three (3) years.

(d) The SLS Entity Benefit Plans have been maintained and operated in material compliance with their terms and, to the extent applicable, provisions of ERISA, the Code and all other applicable Laws, domestic or foreign. None of the SLS Entity Benefit Plans are presently under audit or examination (nor has notice been received of a potential audit or examination) by the IRS, the Department of Labor or any other Governmental Body, domestic or foreign. Other than routine claims for benefits, there are no material Proceedings pending or, to the Knowledge of Scotts, threatened with respect to any SLS Entity Benefit Plan. No non-exempt "prohibited transaction," within the meaning of Section 4975 of the Code and Section 406 of ERISA, has occurred or is reasonably expected to occur with respect to any of the SLS Entity Benefit Plans. No SLS Entity Benefit Plan is maintained outside the jurisdiction of the United States, or which covers any SLS Employee residing or working outside of the United States.

(e) Each SLS Entity Benefit Plan (or any trust related thereto) intended to be qualified or otherwise receive favorable Tax treatment under applicable Law (including under Section 401(a) of the Code) has, if necessary, been determined by the applicable Governmental Body or Tax Authority to be so qualified (including by way of receipt of a favorable determination letter or opinion letter from the IRS, upon which it may rely regarding its qualified status under the applicable Law), and to the Knowledge of Scotts, there are no existing circumstances or events that would reasonably be expected to result in any revocation of, or a change to, such qualification or entitlement to favorable Tax treatment under applicable Law.

(f) With respect to current or former SLS Employees, Scotts and its Subsidiaries (including the SLS Entities) are and since October 1, 2013 have been, in compliance with all applicable Laws respecting employment and labor, including discrimination, harassment, workers' compensation, classification of workers as exempt or non-exempt or as employees or independent contractors, overtime compensation, unlawful retaliation, all titles of the Americans with Disabilities Act, meal and rest breaks, immigration, employee leave issues, workers' compensation, unemployment insurance, wages and hours, occupational safety and health, fair labor standards and the WARN Act and any similar state, local or foreign laws, except such noncompliance that would not, individually or in the aggregate, reasonably be expected to be materially adverse to the SLS Business or the SLS Entities, taken as a whole.

(g) No SLS Entity is a party to or bound by, and none of the SLS Employees are covered by, any collective bargaining agreement and there are no labor unions, employee representatives, works councils or other organizations representing, purporting to represent or attempting to represent any SLS Employees. Since October 1, 2013, there has not occurred or, to the Knowledge of Scotts, been threatened any strike, sympathy strike, slowdown, picketing, work stoppage, concerted refusal to work overtime or other similar labor activity with respect to any SLS Employees. There are no labor disputes currently subject to any grievance procedure, arbitration or litigation, which, if adversely decided, may reasonably, individually or in the aggregate, create a liability in excess of \$250,000 and there is no representation petition pending or, to the Knowledge of Scotts, threatened with respect to any SLS Employee.

(h) No SLS Entity Benefit Plan or a plan of any of its ERISA Affiliates is subject to, or has in the six (6) years immediately preceding the Closing Date been subject to, Title IV of ERISA. Neither Scotts nor any of its ERISA Affiliates, with respect to the SLS Business or SLS Employees, has in the past six (6) years contributed to or been obligated to contribute to or has any current or contingent liability or obligation in respect of: (i) a "multiemployer plan" within the meaning of Section 4001(a)(3) of ERISA, (ii) a "multiple employer plan" within the meaning of Sections 4063 or 4064 of ERISA (iii) a "multiple employer welfare arrangement" within the meaning of Section 3(40) of ERISA or (iv) a self-insured welfare benefit plan.

(i) All contributions and premiums required to have been paid by Scotts and its Subsidiaries (including the SLS Entities) or any of its ERISA Affiliates to any SLS Business Benefit Plan under the terms of any such plan or its related trust, insurance contract or other funding arrangement, or pursuant to any applicable Law (including ERISA and the Code) or collective bargaining agreement have been paid within the time prescribed by any such plan, agreement or applicable Law or properly accrued under GAAP on or before the Closing Date. No SLS Entity has a contract or commitment to create any additional plan, agreement, arrangement or policy which, upon its creation, will be an SLS Entity Benefit Plan or to modify any existing SLS Entity Benefit Plan in a manner which would increase materially the expense of maintaining the SLS Entity Benefit Plan, considered in the aggregate, except as required by applicable Law or tax qualification requirement or as necessary to conform an SLS Entity Benefit Plan to applicable Law. There have been no modifications of SLS Entity Benefit Plans or creation of new SLS Entity Benefit Plans since the date of the last balance sheet that would be prohibited by the preceding sentence. No SLS Entity has terminated any material SLS Entity Benefit Plans since the last balance sheet date.

(j) No SLS Entity Benefit Plan provides welfare benefits after termination of employment, except to the extent required by Section 4980B(f) of the Code or any similar provision of state, local or foreign Law.

(k) Neither this Agreement nor the consummation of the transactions contemplated hereby (either alone or in combination with any other event), including, for the avoidance of doubt, the Scotts Internal Reorganization and the Transactions, would reasonably be expected to (i) result in any payment becoming due or increase the amount of any compensation due to any current or former employee, consultant, officer or director of the SLS Entities, (ii) result in any forgiveness of indebtedness, trigger any funding obligation under any SLS Business Benefit Plan or impose any restrictions or limitations on Scotts's or the SLS Entities' rights to administer, amend or terminate any SLS Business Benefit Plan or (iii) result in the acceleration of time of payment or vesting of any such benefits, or increase the amount of compensation or benefits due to any employee, consultant, officer or director of the SLS Entities. No amount that would be received (whether in cash or property or the vesting of property) as a result of the consummation of the Transactions by any SLS Employee who is a "disqualified individual" (as such term is defined in Treasury Regulations Section 1.280G–1) under any SLS Business Benefit Plans or otherwise would be characterized as an "excess parachute payment" (as defined in Section 280G(b)(1) of the Code). No SLS Entity is required to gross up

or reimburse a payment to any SLS Employee for Taxes incurred under Sections 4999, 409A or 457A of the Code.

(1) The execution and delivery by Scotts of this Agreement, the Scotts Internal Reorganization, the execution and delivery by Scotts, the Non-SLS Subsidiaries and the SLS Entities of the Ancillary Agreements, to the extent they are a party thereto, the performance by Scotts, the Non-SLS Subsidiaries and the SLS Entities of this Agreement (including, for the avoidance of doubt, the Transactions) and the Ancillary Agreements, to the extent they are a party thereto, and the consummation of the transactions contemplated hereby and thereby would not reasonably be expected to require Scotts or any of its Subsidiaries (including the SLS Entities) to seek or obtain any Consent from, engage in consultation with, or issue any notice to or make any filing with (as applicable) any SLS Employees or any representatives, labor unions, works councils or similar organizations representing SLS Employees, or any Governmental Body with respect to SLS Employees (except to the extent previously made or obtained prior to the date hereof).

3.9 <u>Taxes</u>.

(a) (i) All material Tax Returns required to be filed by or with respect to the SLS Entities or the SLS Assets have been filed in a timely manner (within any applicable extension periods) and all such Tax Returns are true, correct and complete in all material aspects, (ii) all material Taxes required to be paid with respect to the SLS Entities (whether or not shown on such returns) or the SLS Assets have been timely paid in full and (iii) there exist no material liens for Taxes with respect to any of the SLS Assets, except for liens for Taxes not yet due and payable. No jurisdiction (whether within or outside the United States) in which any SLS Entity has not filed a particular type of Tax Return or paid a particular type of Tax has asserted that such SLS Entity is required to file such Tax Return or pay such type of Tax in such jurisdiction.

(b) There are no outstanding agreements or waivers extending the statutory period for assessment or collection of any Taxes of the SLS Entities or with respect to the SLS Assets and no written powers of attorney with respect to any such Taxes. The time for filing any Tax Return with respect to the SLS Entities or the SLS Assets has not been extended, other than any extension to which a SLS Entity is entitled under applicable Law without the consent of the relevant Tax Authority or Governmental Body.

(c) There are no audits, Proceedings, investigations or other actions pending or threatened in writing against the SLS Entities or the SLS Assets in respect of Taxes or Tax Returns. No Tax Authority has asserted or threatened to assert, in each case in writing, any deficiency, claim or issue with respect to Taxes or any adjustment to Taxes against the SLS Entities or with respect to the SLS Assets for any taxable period for which the period of assessment or collection remains open.

(d) None of the SLS Entities will be required to include any item of income in, or exclude any item of deduction from, taxable income for any taxable period (or portion thereof) ending after the Closing Date as a result of (i) any change in method of accounting for a

taxable period ending on or prior to the Closing Date, (<u>ii</u>) any installment sale or open transaction, (<u>iii</u>) any prepaid amount received or paid or (<u>iv</u>) any election pursuant to Section 108(i) of the Code (or any similar provision of state, local or foreign Law) made with respect to any taxable period ending on or prior to the Closing Date.

(e) None of the SLS Entities during the five-year period ending on the date hereof, has been a distributing corporation or a controlled corporation in a transaction intended to be governed by Section 355 of the Code.

(f) All material Taxes that the SLS Entities were required to withhold or collect have been duly withheld or collected and have been paid, to the extent required by applicable Law, to the proper Tax Authority. The SLS Entities have complied in all material respects with the rules and regulations relating to the withholding and remittance of Taxes.

(g) None of the SLS Entities (<u>i</u>) has received or applied for a Tax ruling or entered into a closing agreement pursuant to Section 7121 of the Code (or any predecessor provision or any similar provision of state, local or foreign Law), in either case that would be binding upon any SLS Entity after the Closing Date, (<u>ii</u>) is or has been a member of any affiliated, consolidated, combined or unitary group for purposes of filing Tax Returns or paying Taxes or (<u>iii</u>) has any liability for the Taxes of any Person (other than for any other SLS Entity) whether under Treasury Regulation Section 1.1502-6 or any similar provision of state, local or foreign Law, as a transferee or successor, pursuant to any Tax sharing or indemnity agreement or other contractual agreements, or otherwise.

(h) None of the SLS Entities has engaged in any "listed transactions" within the meaning of Treasury Regulation Section 1.6011-4(b)(2).

(i) Other than in connection with the Conversion, there are no SLS Entities for which an election to change the U.S. federal tax classification of such SLS Entities has been made within the five-year period ending on the date of this Agreement.

(j) All material transactions entered into by any of the SLS Entities with any related party have been carried out at arm's length, and each of the SLS Entities has complied in all material respects with all applicable transfer pricing disclosure, reporting, documentation and other similar requirements.

3.10 Absence of Certain Changes or Events.

(a) Since September 30, 2015 and through the date of this Agreement, there has not been any event, development or state of circumstances that constitutes a SLS Material Adverse Effect.

(b) Between September 30, 2015 and the date hereof (<u>i</u>) the SLS Business had been conducted in the ordinary course consistent with past practices other than matters in connection with the Transactions and the Scotts Internal Reorganization, and (<u>ii</u>) neither Scotts

nor any of its Subsidiaries (including the SLS Entities) has taken any action that would, if taken after the date hereof, require the consent of TruGreen Holdings under <u>Section 5.1(b)</u>.

3.11 <u>Title to Assets; Sufficiency of Assets</u>.

(a) Except as would not, individually or in the aggregate, reasonably be expected to be materially adverse to the SLS Business or to the SLS Entities, taken as a whole, Scotts and its Subsidiaries (including the SLS Entities) have (and as of the Closing the SLS Entities themselves will have, other than with respect to rights or assets (x) to be provided (or for which the benefit is to be provided) to the SLS Entities by Scotts and the Non-SLS Subsidiaries pursuant to the Ancillary Agreements, (y) that are (or for which the benefit is), between the date hereof and the Closing Date, offered in writing to the Chief Financial Officer of TruGreen Holdings to be provided by Scotts for inclusion in any such Ancillary Agreement but which TruGreen Holdings has notified Scotts in writing that it will not require (or fails to reply on a reasonably timely basis) pursuant to any such Ancillary Agreement (the "Declined Services") or (z) pursuant to or under Shared Contracts) good and valid (and, in the case of SLS Owned Real Property, good, valid and marketable fee simple) title to, or have a valid leasehold or licensed interest in, or has a legal right to use, all of the tangible and intangible assets, properties and rights (real and personal, including all Intellectual Property) that (i) are primarily used or held for use in connection with the SLS Business, (ii) are reflected on the SLS Financial Statements or (iii) were acquired by the SLS Business after the Balance Sheet Date for use primarily by, or in connection with, the SLS Business, in each case other than the Scotts Marks (collectively, the "SLS Assets"), in each case, free and clear of all Encumbrances, except for Permitted Encumbrances, other than any SLS Assets disposed of by the SLS Entities or the SLS Business (x) after the Balance Sheet Date and prior to the date hereof in the ordinary course of business or (y) after the date hereof and not in violation of this Agreement.

(b) The SLS Assets, together with the Shared Contracts to be replicated pursuant to <u>Section 5.13(b)</u>, Intellectual Property that will be licensed pursuant to <u>Section 5.12</u> (including under the Trademark License Agreement) and the services and Intellectual Property that will be provided or licensed pursuant to the Transition Services Agreement and the services and employees to be provided pursuant to the Employee Leasing Agreement, are sufficient to conduct the SLS Business in substantially the same manner as conducted, as of the Balance Sheet Date, the date hereof and immediately prior to the Closing.

3.12 <u>Real Property</u>.

(a) <u>Scotts Schedule 3.12(a)</u> sets forth an accurate and complete list of each real property owned by any SLS Entity or that otherwise primarily relates to the SLS Business and is owned by Scotts or any Non-SLS Subsidiary, in each case, as of the date hereof (together with all structures, facilities, improvements and fixtures presently or hereafter located thereon or attached thereto, the "<u>SLS Owned Real Property</u>"), including the address and owner of each parcel of such SLS Owned Real Property.

(b) <u>Scotts Schedule 3.12(b)</u> sets forth an accurate and complete list of each real property leased, subleased, licensed or occupied by any SLS Entity or that otherwise relates primarily to the SLS Business and is leased, subleased, licensed or occupied by Scotts or any Non-SLS Subsidiary (the "<u>SLS Leased Real Property</u>"), including all leases, subleases or licenses (together with any and all amendments and modifications thereto and any guarantees thereof) in effect as of the date relating thereto (collectively, the "<u>SLS Real Property Leases</u>") and identifying the tenant and address for each SLS Real Property Lease. Scotts has made available to TruGreen Holdings accurate and complete copies of each material SLS Real Property Lease. No Transferred SLS Employee is primarily located at any location other than a SLS Owned Real Property or a SLS Leased Real Property.

(c) Neither Scotts nor any of its Subsidiaries (including the SLS Entities) has leased, subleased or granted to any Person any right to possess, lease or occupy any portion of the SLS Owned Real Property or the SLS Leased Real Property.

(d) Neither Scotts nor any of its Subsidiaries (including the SLS Entities) holds, has granted or is obligated under any option, right of first offer, right of first refusal or other contractual right to purchase, acquire, sell or dispose of the SLS Owned Real Property or the SLS Leased Real Property or any portion thereof or interest therein or any other real property.

3.13 Intellectual Property.

(a) <u>Scotts Schedule 3.13(a)</u> sets forth an accurate and complete list of (<u>i</u>) all registered Intellectual Property and all applications for registration for Intellectual Property used or held for use primarily by the SLS Business or that are owned by the SLS Entities or are material to the SLS Business (the "<u>SLS Registered Intellectual Property</u>") and (<u>ii</u>) all unregistered Marks that are material to the SLS Business and primarily used or held for use by the SLS Business, in each case, owned, beneficially or of record, by (<u>x</u>) the SLS Entities or (<u>y</u>) Scotts or any Non-SLS Subsidiaries as of the date of this Agreement, and the owner thereof. The SLS Registered Intellectual Property (other than the Scotts Marks) and each of the unregistered Marks (other than the Scotts Marks), Copyrights and Trade Secrets that are owned by the SLS Entities, Scotts or any Non-SLS Subsidiaries and are used or held for use primarily by the SLS Business (collectively, the "<u>SLS Owned Intellectual Property</u>") are owned by such SLS Entities, Scotts or any Non-SLS Subsidiaries (and after giving effect to the Transactions will be owned by Newco or its Subsidiaries (including the SLS Entities), unless licensed to Newco or one of its Subsidiaries by Scotts or a Non-SLS Subsidiary) free and clear of all Encumbrances other than Permitted Encumbrances.

(b) Other than as set forth on <u>Scotts Schedule 3.13(b)</u>, there is no pending action, opposition, interference, reexamination, proceeding, or, to the Knowledge of Scotts, written threatened claim against any SLS Entity, Scotts or any Non-SLS Subsidiaries or to which any of them is party, in respect of any SLS Owned Intellectual Property or any other Intellectual Property material to the SLS Business.

(c) Other than as set forth on <u>Scotts Schedule 3.13(c)</u>: (i) to the Knowledge of Scotts, the conduct of the SLS Business does not infringe, misappropriate or violate any Intellectual Property of any Person and (ii) to the Knowledge of Scotts, no Person is infringing, misappropriating or violating the SLS Owned Intellectual Property, except in each case of (i) and (ii) as would not, constitute a SLS Material Adverse Effect.

(d) Except as would not constitute a SLS Material Adverse Effect, the SLS Entities have used (and Scotts has otherwise in relation to the SLS Business used) commercially reasonable efforts to ensure protection of the SLS Owned Intellectual Property, including the confidentiality of their Trade Secrets.

(e) Scotts has implemented reasonable backup and disaster recovery arrangements designed to provide for the continued operation of the SLS Business in the event of a disaster or business interruption. The SLS Internal IT Systems owned by the SLS Entities or that constitute SLS Assets (i) are in good repair and operating condition and are adequate and suitable in all material respects for the purposes for which they are being used or held for use, (ii) conform in all material respects with their related documentation and (iii) to the Knowledge of Scotts do not contain any virus, Trojan, or other malicious code that would reasonably be expected to interfere in any material respect with the SLS Business. The SLS Entities have not (and Scotts has not otherwise in relation to the SLS Business), in the past three (3) years, experienced any data security breach or unauthorized data release incident that has had a material and adverse effect on the SLS Business.

(f) Solely relating to the SLS Business, Scotts and each of its Subsidiaries (including the SLS Entities) is in compliance with all contractual and requirements under applicable Law pertaining to data protection or information privacy and security, including any privacy policy concerning the collection or use of data or information, except as would not constitute a SLS Material Adverse Effect.

3.14 Contracts.

<u>Scotts Schedule 3.14</u> sets forth an accurate and complete list of all Contracts in effect as of the date hereof to which any SLS Entity, Scotts or any Non-SLS Subsidiary (but with respect to Scotts and any Non-SLS Subsidiary, only such Contracts relating to the SLS Business and satisfying the requirements set forth in any of clauses (a)-(n) below to the extent as they relate to the SLS Business), is a party or shall be a party after giving effect to the Scotts Internal Reorganization or by which any property or asset of any SLS Entity, Scotts or any Non-SLS Subsidiary (but with respect to Scotts and any Non-SLS Subsidiary, only such Contracts relating to the SLS Business and satisfying the requirements set forth in any of clauses (a)-(n) below to the extent as they relate to the SLS Business) is bound or affected (other than insurance policies, purchase orders or standard sales orders entered into by Scotts, any Non-SLS Entity or any SLS Entity, as applicable, in the ordinary course of business) which by its terms:

(a) is not terminable at will within three months and requires future expenditures or other performance with respect to goods, equipment or services having an annual value in excess of \$100,000;

(b) relates to the obligation of the SLS Business to purchase products, materials, supplies, goods, equipment, other assets or services pursuant to which payments of \$250,000 or more were made during the 12-month period ending on December 31, 2014;

(c) relates to the obligation of the SLS Business to sell products or services pursuant to which payments of \$250,000 or more were received during the 12-month period ending on December 31, 2014;

(d) relates to any outstanding Indebtedness, other than SLS Intercompany Balances or SLS Intracompany Obligations, having an outstanding principal amount in excess of \$1,000,000;

(e) (i) limits the ability of any SLS Entity to compete in any line of business or with any Person in any geographic area or that would so limit the freedom of the SLS Entities or any of their Affiliates after the Closing (including the TruGreen Entities) or (ii) contains exclusivity obligations binding on any SLS Entity or, following the Closing, any of its Affiliates (including the TruGreen Entities) in each case, other than SLS Franchise Agreements;

(f) requires any capital commitment or capital expenditure (including any series of related expenditures) by the SLS Business of greater than \$500,000;

(g) relates to the acquisition or disposition of any business, assets, properties or rights under which the SLS Business has any future liability with respect to an "earn-out," contingent purchase price, deferred purchase price or similar contingent payment obligation, or any indemnification obligation that could result in payments in excess of \$100,000;

(h) constitutes a contract or agreement with any officer, employee, director, stockholder or other Affiliate of the SLS Entities (other than (\underline{x}) SLS Entity Benefit Plans and Scotts Benefit Plans and (\underline{y}) SLS Intercompany Contracts that have been terminated prior to the Closing without any additional liability or obligation to any party);

(i) constitutes an agreement that contains any indemnification obligations of the SLS Business, or credit support or guarantee relating to such indemnification obligations, other than any such indemnification, credit support or guarantee obligations incurred in the ordinary course of business or that requires credit support or indemnification obligations of less than \$100,000;

(j) is an Intellectual Property Contract that is material to the SLS Business (excluding Intellectual Property Contracts for commercially available off-the-shelf Software that is not the subject of a negotiated agreement and for which the aggregate one-time or annual

amounts paid or payable to or by the SLS Business related to such Contract are less than \$50,000);

(k) is a contract that creates (or governs the operation of) any partnership, joint venture, alliance or similar agreement or arrangement, in each case that is material to the operation of SLS Entities or the SLS Business, taken as a whole (other than this Agreement and the Ancillary Agreements);

- (l) is an SLS Franchise Agreement;
- (m) is a SLS Real Property Lease that has annual rent obligations in excess of \$500,000; or

(n) is a Shared Contract (excluding any Contract for commercially available off-the-shelf Software that is not the subject of a negotiated agreement and for which the aggregate one-time or annual amounts paid or payable to or by Scotts or any of its Subsidiaries related to such Contract are less than \$250,000)

(the Contracts described in clauses (a) through (n), the "SLS Material Contracts").

Scotts has made available to TruGreen Holdings a true and complete copy of each SLS Material Contract (including, as of the date hereof, all modifications and amendments thereto and written waivers thereunder). Except as would not constitute a SLS Material Adverse Effect, each SLS Material Contract to which Scotts or any of its Subsidiaries (including the SLS Entities), as applicable, is a party is in full force and effect in accordance with its respective terms, except after the date hereof, for any such SLS Material Contracts that expire according to their terms after the date hereof. Except, after the date hereof for any such SLS Material Contracts that expire according to their terms after the date hereof, each SLS Material Contract is valid, binding and enforceable against Scotts or such Subsidiary, as applicable, and, to the Knowledge of Scotts, each other party thereto in accordance with its terms, except as enforceability may be limited by bankruptcy, insolvency, reorganization, moratorium or other similar Laws affecting creditors' rights and general principles of equity affecting the availability of specific performance and other equitable remedies (regardless of whether enforcement is sought in a proceeding at law or in equity) and except as would not constitute a SLS Material Adverse Effect. Neither Scotts nor any of its Subsidiaries (including the Non-SLS Subsidiaries and SLS Entities) is in default, violation or breach in any material respect under (or, to the Knowledge of Scotts, is alleged to be in default or breach in any material respect under) any such SLS Material Contract to which it is a party, or has within the last 12 months provided or received written notice of any intention or threat to terminate or not to renew, any such SLS Material Contract, except as could not reasonably be expected to be material to the SLS Business, taken as a whole. To the Knowledge of Scotts, no event or circumstance has occurred and is continuing that constitutes or, with notice or the passage of time, or both, would constitute, a default, violation or breach in any respect under any such SLS Material Contract by any party thereto, or result in a termination thereof or would cause or permit the acceleration of or other changes of or to any right or obligation or the loss by Scotts or any of its Subsidiaries (including

the SLS Entities) of any benefit thereunder (other than the expiration of any SLS Material Contract according to its terms, based on the mere passage of time), in each case, except for such defaults, breaches, violations terminations, accelerations or changes as would not, individually or in the aggregate, reasonably be expected to have a material and adverse effect on the ability of the SLS Entities to continue to conduct the SLS Business as currently conducted following the Closing.

3.15 <u>Compliance with Laws; Permits</u>.

(a) Other than as set forth on <u>Scotts Schedule 3.15</u>, and with respect to Scotts and the Non-SLS Subsidiaries, solely relating to the SLS Business, each of Scotts, its Subsidiaries (including the SLS Entities) and the SLS Business is, and since October 1, 2013 has been, in compliance with, and has not received written notice of any default or violation of, any Laws applicable to Scotts and its Subsidiaries (including the SLS Entities) and/or the SLS Business, as applicable, in each case, except as would not, constitute a SLS Material Adverse Effect. Scotts and its Subsidiaries hold all Permits that are required for the conduct of the SLS Business as now being conducted, except where the failure to have any such Permits would not constitute a SLS Material Adverse Effect. Except as would not constitute a SLS Material Adverse Effect, (i) all such Permits are in full force and effect, (ii) no violations are recorded in respect of any such Permits and (iii) no Proceedings are pending or, to the Knowledge of Scotts, threatened to revoke, suspend, cancel or limit any such Permit.

(b) No representation or warranty is made under this <u>Section 3.15</u> with respect to Taxes or environmental matters, which representations and warranties are covered exclusively by <u>Sections 3.9</u> (Taxes) and <u>3.18</u> (Environmental Matters) respectively.

3.16 Anti-Corruption; Sanctions.

(a) None of Scotts, its Subsidiaries (including the SLS Entities) or their respective officers, directors, agents, distributors, employees and other Persons involved in or acting for or on behalf of the SLS Business (but with respect to Scotts and the Non-SLS Subsidiaries, solely relating to the SLS Business) (the "<u>SLS Relevant Persons</u>") has directly or indirectly violated or taken any act in furtherance of violating any provision of the U.S. Foreign Corrupt Practices Act of 1977 (as amended), the U.K. Bribery Act 2010 or any other anti-corruption or anti-bribery Laws applicable to the SLS Entities or the SLS Business (collectively, the "<u>Anti-Corruption Laws</u>").

(b) None of Scotts, its Subsidiaries (including the SLS Entities) or the SLS Relevant Persons (but with respect to Scotts and the Non-SLS Subsidiaries, solely relating to the SLS Business) has (i) directly or indirectly used any corporate funds or assets, or taken any act in the furtherance of an offer, payment, promise to pay, gift, bribe, rebate, loan, payoff, kickback, political contribution, entertainment expense, or any other transfer of value to a Government Official for the purpose of inducing such Government Official to do any act or make any decision in an official capacity (including a decision to fail to perform an official function, or use his or her or its influence with a Governmental Body in order to affect any act or decision of such

Governmental Body) for the purpose of assisting any Person to obtain or retain any business, or to facilitate efforts of any Person to transact business or for any other improper purpose (*e.g.*, to obtain a tax rate lower than allowed by Law), or (ii) established or maintained, or is maintaining, any unlawful fund of corporate monies or other properties with an intent of carrying out any of the acts set forth in the preceding clause (i).

(c) To the Knowledge of Scotts, there is no investigation, allegation, request for information, or other inquiry by any Governmental Body regarding the SLS Entities' or the SLS Business' actual or possible violation of the Anti-Corruption Laws.

(d) To the Knowledge of Scotts, none of the officers, directors, employees or agents (or their employees) involved in the SLS Business is a Government Official or consultant to any Government Official, and there is no existing family relationship between any officer, director, employee, member or agent (or their employees) of the SLS Entities and any Government Official.

(e) None of Scotts, its Subsidiaries (including the SLS Entities) or the SLS Relevant Persons (but, with respect to Scotts and the Non-SLS Subsidiaries, solely relating to the SLS Business and, with respect to SLS Relevant Persons other than Scotts, its Subsidiaries (including the SLS Entities) and their respective officers and directors, solely to the Knowledge of Scotts) has directly or indirectly (i) circumvented the internal accounting controls of Scotts or any of the SLS Entities, (ii) falsified any of the books, records or accounts of Scotts or any of the SLS Entities or (iii) made false or misleading statements to, or attempted to coerce or fraudulently influence, an accountant in connection with any audit, review or examination of the financial statements of Scotts or any of the SLS Entities.

(f) None of Scotts, its Subsidiaries (including the SLS Entities) or, to the Knowledge of Scotts, the SLS Relevant Persons (but with respect to Scotts, the SLS Relevant Persons and the Non-SLS Subsidiaries, solely relating to the SLS Business) is (i) the subject or, to the Knowledge of Scotts, target of any sanctions administered or enforced by the U.S. government (including, without limitation, the Office of Foreign Assets Control of the U.S. Department of the Treasury and the U.S. State Department), the United Nations Security Council, the European Union, Her Majesty's Treasury, or other relevant sanctions authority (collectively, "<u>Sanctions</u>"), (ii) located organized or resident in a country or territory that is the subject of Sanctions (collectively, "<u>Sanctioned Country</u>"), (iii) to the Knowledge of Scotts engaged in, directly or indirectly, any material dealings or transactions with or relating to any person or entity that at the time of the dealing or transaction is or was the subject or the target of Sanctions nor has any received notice of, or otherwise become aware of, any claim, action, suit, or proceeding or investigation, or inquiry, formal or informal, against it with respect to Sanctions.

3.17 Insurance.

<u>Scotts Schedule 3.17</u> contains an accurate and complete list of all material policies of insurance maintained as of the date of this Agreement with respect to the SLS Business (the "<u>SLS Policies</u>"). Except as would not, constitute a SLS Material Adverse Effect (<u>a</u>) none of Scotts or any of its Subsidiaries is in default with respect to its obligations under any SLS Policy and (<u>b</u>) none of Scotts or any of its Subsidiaries has received, since the time any SLS Policy was last renewed or issued (whichever is later), written notice of termination, cancellation or non-renewal of any such SLS Policy from any of its insurance brokers or carriers. As of the date hereof, there is no material claim by Scotts or any of its Subsidiaries primarily relating to the SLS Business pending under any SLS Policy as to which coverage has been denied or disputed by the underwriters of such SLS Policy or in respect of which such underwriters have reserved their rights.

3.18 Environmental Matters.

(a) Each SLS Entity complies and, while owned by Scotts, has complied with all applicable Environmental Laws, except where the failure to comply would not, constitute a SLS Material Adverse Effect.

(b) Each SLS Entity holds and is in compliance with all Permits required under applicable Environmental Laws to operate as it currently operates, except where the failure to hold such Permits would not, constitute a SLS Material Adverse Effect.

(c) Since October 1, 2013, no SLS Entity has received any written notice from any Person regarding any actual or alleged violation of Environmental Laws, or any liabilities or alleged liabilities for investigation costs, cleanup costs, response costs, corrective action costs, personal injury, property damage, natural resources damages or attorney fees under Environmental Laws, the subject of which would constitute a SLS Material Adverse Effect.

(d) There is no Proceeding by or before any Governmental Body pending or, to the Knowledge of Scotts, threatened in writing against any SLS Entity, relating to alleged liability under Environmental Laws, that if determined adversely would, constitute a SLS Material Adverse Effect.

(e) No Release of Hazardous Substances has occurred on, at, under or from or, to the Knowledge of Scotts, is affecting any property currently owned or leased by any SLS Entity and, to the Knowledge of Scotts, no Release of Hazardous Substances has occurred on, at, under or from or is affecting any property formerly owned or leased by a SLS Entity, in either case, in a manner that would, constitute a SLS Material Adverse Effect.

(f) No SLS Entity has, during the period of Scotts's ownership of such SLS Entity, and to the Knowledge of Scotts, no SLS Entity has, during any period prior to Scotts's ownership of any such SLS Entity, disposed of or arranged for the disposal of any Hazardous Substance that has resulted, or would reasonably be expected to result, in investigation costs,

cleanup costs, response costs, corrective action costs, personal injury, property damage, natural resources damages or attorney fees under or pursuant to any applicable Environmental Laws, which would constitute a SLS Material Adverse Effect.

(g) Scotts has made available to TruGreen Holdings all environmental assessments, investigations and studies in the possession, custody or control of Scotts or any of its Subsidiaries (including the SLS Entities) prepared since October 1, 2013 and prior to the date hereof (other than those subject to attorney-client privilege, for which summaries have been made available to TruGreen Holdings prior to the date hereof), which identify issues at properties owned, leased or operated by any SLS Entity or the SLS Business that would constitute a SLS Material Adverse Effect.

(h) Notwithstanding any other provisions of this <u>Article III</u>, other than as set forth in <u>Section 3.4</u> and this <u>Section 3.18</u>, Scotts makes no representations or warranties, either express or implied, with respect to matters arising under any Environmental Law or concerning any Hazardous Substance.

3.19 Financial Advisors.

Other than J.P. Morgan Securities LLC, no Person has acted, directly or indirectly, as an agent, broker, investment banker, finder or financial advisor for Scotts or any of its Subsidiaries in connection with the transactions contemplated by this Agreement, and no Person is entitled to any fee or commission or like payment from any SLS Entity in respect thereof. Scotts has made available to TruGreen Holdings, solely for informational purposes, complete and correct copies of all letter agreements between Scotts or any SLS Entity and J.P. Morgan Securities LLC pursuant to which J.P. Morgan Securities LLC could be entitled to any payment in connection with the Transactions or the Merger.

3.20 Transactions with Affiliates.

(a) <u>Scotts Schedule 3.20(a)</u> lists all written material SLS Intercompany Contracts as of the Balance Sheet Date.

(b) To the Knowledge of Scotts, no officer, director or employee of Scotts or any of its Subsidiaries, or any family member, relative or Affiliate of any such officer, director or employee, (<u>i</u>) owns, directly or indirectly, any interest in (<u>A</u>) any asset or other property used in or held for use in the SLS Business or (<u>B</u>) any Person that is a supplier, customer or competitor of the SLS Business or any SLS Entity, (<u>ii</u>) serves as an officer, director or employee of any Person that is a supplier, customer or competitor of any SLS Entity or (<u>iii</u>) is a debtor or creditor of any SLS Entity.

3.21 <u>Customers and Suppliers</u>.

<u>Scotts Schedule 3.21(a)</u> lists the names of the 10 largest customers of the SLS Business, as determined by annual revenue to the SLS Business for the 2014 fiscal year (the "<u>Major SLS</u>

<u>Customers</u>"). <u>Scotts Schedule 3.21(b)</u> lists the names of the 10 largest suppliers of the SLS Business, based on payments made by the SLS Business during the 2014 fiscal year (the "<u>Major SLS Suppliers</u>"). As of the date of this Agreement, neither Scotts nor any Non-SLS Subsidiary or SLS Entity has received written notice (or, to Scotts's Knowledge, any other notice) that any Major SLS Customer or Major SLS Supplier has terminated or threatened in writing to terminate any material agreement with Scotts or any Non-SLS Subsidiary or SLS Entity, except as would not constitute a SLS Material Adverse Effect.

3.22 Investment Intent.

Scotts is acquiring the Scotts Newco Interests for its own account for investment purposes only and not with a view to any public distribution thereof or with any intention of selling, distributing or otherwise disposing of the Scotts Newco Interests in a manner that would violate the registration requirements of the Securities Act or any similar provisions of any applicable Law. Scotts agrees that the Scotts Newco Interests may not be sold, transferred, offered for sale, pledged, hypothecated or otherwise disposed of without registration under the Securities Act and any applicable state securities Laws, except pursuant to an exemption from such registration under the Securities Act and such Laws. Scotts is able to bear the economic risk of holding the Scotts Newco Interests for an indefinite period (including total loss of its investment), and (either alone or together with its representatives) has sufficient knowledge and experience in financial and business matters so as to be capable of evaluating the merits and risk of its investment.

3.23 Scotts Acknowledgment.

(a) Except for the representations and warranties expressly set forth in this <u>Article III</u>, none of Scotts, the Non-SLS Subsidiaries or the SLS Entities or any of their Affiliates nor any other Person on behalf of any of the foregoing makes any express or implied representation or warranty with respect to Scotts, the Non-SLS Subsidiaries, the SLS Entities or the SLS Business or other businesses of any of the foregoing with respect to any other information provided, or made available, to TruGreen Holdings, its representatives or Affiliates in connection with the Transactions, the Scotts Internal Reorganization and the other transactions contemplated by this Agreement, including the accuracy or completeness thereof. Without limiting the foregoing, neither Scotts nor any other Person resulting from TruGreen Holdings', its representatives' or Affiliates' use of any information, documents, projections, forecasts, business plans or other material made available to TruGreen Holdings, its representatives or Affiliates, including any information made available in the electronic data room run by R.R. Donnelley & Sons Company and maintained by Scotts for purposes of the Transactions, the Scotts Internal Reorganization and the other transactions, functional "break-out" discussions, responses to questions submitted on behalf of TruGreen Holdings, its representatives or Affiliates or in any other form in connection with the Transaction memorandum, management presentatives or Affiliates or in any other form in connection with the Transaction memorandum, management presentatives or in any other form in connections, the Scotts Internal Reorganization and the other

transactions contemplated by this Agreement, unless and to the extent any such information is expressly included in a representation or warranty contained in this <u>Article III</u>.

(b) Scotts acknowledges that it has made its own assessment of the present condition and the future prospects of the TruGreen Entities and, following the Closing, its Subsidiaries (including the SLS Entities) and the SLS Business and is sufficiently experienced to make an informed judgment with respect thereto. Except with respect to the representations and warranties of TruGreen Holdings expressly set forth in <u>Article IV</u>, there is and has been no reliance by Scotts, its Affiliates or any of their respective representatives on any express or implied representation or warranty of TruGreen Holdings in connection with the transactions contemplated hereby.

3.24 SLS Franchises.

(a) Neither Scotts nor any of the Non-SLS Subsidiaries own or operate any franchise system. Other than the SLS Franchises, the SLS Entities have not owned or operated any franchise system.

(b) No SLS Franchisee or other Person has any enforceable right of first refusal, option or other right or arrangement to sign any SLS Franchise Agreement or acquire any SLS Franchise.

(c) All funds administered by or paid to the SLS Entities on behalf of one or more SLS Franchisees, including funds that SLS Franchisees contributed for advertising and promotion (i) have been administered and used in accordance with all Laws, all descriptions in the FDDs, and all SLS Franchise Agreements, and (ii) the SLS Entities have in all material respects properly accounted for all payments made by each SLS Franchisee with respect to any such fund or cooperative. There are no loans owed to or owing from any such funds or cooperatives. To the Knowledge of Scotts, there are no claims that any of the expenditures from any such funds or cooperatives have been improperly collected, accounted for, maintained, used or applied. Scotts Schedule 3.24(c) contains a listing of all established advertising cooperatives regarding the SLS Franchises, which Scotts Schedule also identifies whether each such cooperative is presently operating.

(d) Since October 1, 2013, no "Financial Performance Representation" (as defined in the FTC Rule) has ever been made to any SLS Franchisee or prospective franchisee by the SLS Entities, or an officer, general partner, limited partner, salesperson, or representation of the SLS Entities, except Financial Performance Representations which have been incorporated properly into the applicable FDD.

(e) The SLS Entities are, and since October 1, 2013 have been, in compliance with all Franchise Laws and have not offered or sold any Franchise in violation of any Franchise Law (including any violation for failure to file on a timely basis all required amendments and renewals of the registrations and exemptions under the Franchise Laws and any violation for failure to file franchise advertisements). With respect to the relations of the SLS Entities with

existing and former SLS Franchisees, and all terminations, non-renewals, and transfers of SLS Franchises since October 1, 2013, the SLS Entities have complied with all the proper notice of default, time to cure, and the actual termination of any SLS Franchise Agreement required by any Franchise Law.

(f) <u>Scotts Schedule 3.24(f)</u> contains a list of all independent sales agents, sales brokers and referral sources used by either SLS Entity since October 1, 2013, to assist with the sale of Franchises and a list of all written and oral agreements or arrangements with such sales agents, sales brokers and referral sources currently in effect under which either SLS Entity has authorized such Person to sell or promote the offer or sale of Franchises. Neither SLS Entity is in default under any such agreement or arrangement. None of such sales agents, sales brokers and referral sources provided information to prospective franchisees that materially differed from the information contained in the FDDs. All FDDs issued by either SLS Entity since October 1, 2013, accurately disclose the necessary information pertaining to such independent sales agents, sales brokers and referral sources.

(g) The operations manuals of the SLS Entities do not impose any obligations or set forth any requirements that are materially inconsistent with any of the SLS Franchise Agreements or the FDDs that they have previously used or currently use, to offer or sell Franchises.

(h) To the Knowledge of Scotts, no franchise association or organization is acting as a representative of any group of five or more SLS Franchisees. Any franchise council or advisory group (whether independently formed or sponsored by either SLS Entity) presently in place is advisory in nature.

(i) The Transactions will not require the consent or approval by any SLS Franchisee, council, association, or other third party.

ARTICLE IV

REPRESENTATIONS AND WARRANTIES OF TIGER HOLDINGS

Except as disclosed in the corresponding section or sub-section of the disclosure schedules of TruGreen Holdings attached hereto (collectively, the "<u>TruGreen Schedules</u>"), TruGreen Holdings hereby represents and warrants to Scotts as follows:

4.1 Incorporation and Authority.

(a) Each of TruGreen Holdings and the TruGreen Entities is an entity duly organized, validly existing and (where such concept is applicable) in good standing (or local equivalent) under the Laws of its jurisdiction of organization, formation or incorporation, as applicable, and has all requisite corporate power and authority to conduct its business as currently conducted. Except as would not constitute a TruGreen Material Adverse Effect, each of the TruGreen Entities is (i) duly qualified or licensed as a foreign corporation for the

transaction of business and (ii) (where such concept is applicable) in good standing under the laws of each other jurisdiction in which the character of properties it owns or leases, or conducts any business makes such qualification or license necessary.

(b) TruGreen Holdings has made available to Scotts complete copies of the respective articles of incorporation or bylaws (or similar organizational documents, as amended) of each of TruGreen Holdings and the TruGreen Entities as in effect as of the date hereof. None of TruGreen Holdings or the TruGreen Entities is in violation of any provision of such articles of incorporation or bylaws (or similar organizational documents).

(c) Each of TruGreen Holdings and the TruGreen Entities has all necessary corporate power and authority to enter into this Agreement and the Ancillary Agreements to which it will be a party, to carry out and perform its obligations hereunder and thereunder and to consummate the transactions contemplated on its part hereby and thereby. TruGreen Holdings and the TruGreen Entities have all necessary corporate power and authority to effect the Merger. The execution and delivery by TruGreen Holdings of this Agreement and the Merger Agreement, and by TruGreen Holdings and the TruGreen Entities of the Ancillary Agreements to which they will be a party, and the consummation by TruGreen Holdings and the TruGreen Entities of the transactions contemplated on their part hereby and thereby, have been duly authorized by all necessary company action on the part of TruGreen Holdings and the TruGreen Entities. This Agreement has been duly executed and delivered by TruGreen Holdings, and, to the extent TruGreen Holdings or any TruGreen Entity is a party thereto, the Ancillary Agreements will at the Closing be, duly executed and delivered by TruGreen Holdings or such TruGreen Entity, as applicable, and (assuming due authorization, execution and delivery by the other parties thereto) this Agreement constitutes a legal, valid and binding obligation of TruGreen Holdings, and when executed and delivered, to the extent TruGreen Holdings or any TruGreen Entity, as applicable, is a party thereto, the Ancillary Agreements will constitute, legal, valid and binding obligations of TruGreen Holdings or such TruGreen Entity, enforceable against TruGreen Holdings or such TruGreen Entity, as applicable, in accordance with their terms, except as enforceability may be limited by bankruptcy, insolvency, reorganization, moratorium or other similar Laws affecting creditors' rights and general principles of equity affecting the availability of specific performance and other equitable remedies (regardless of whether enforcement is sought in a proceeding at law or in equity).

4.2 Capitalization and Subsidiaries.

(a) Set forth on <u>TruGreen Schedule 4.2(a)</u> is (i) the jurisdiction of incorporation or legal organization and (ii) as of the date hereof (A) the number of authorized, issued and outstanding shares or other equity securities of TruGreen Holdings and each of its Subsidiaries (including the TruGreen Entities) and (B) there are no other authorized, issued or outstanding shares of capital stock or other equity securities of TruGreen Holdings or any of its Subsidiaries (including the TruGreen Entities). All of the issued and outstanding equity securities of each of the Subsidiaries of TruGreen Holdings (including the TruGreen Entities) are (\underline{x}) legally and beneficially owned by TruGreen Holdings (either directly or indirectly), (y) held free and clear of any Encumbrances and (\underline{z}) have been validly issued, are fully paid and

nonassessable and have not been issued in violation of any preemptive or similar rights. As of the date hereof and after giving effect to the Transactions and the Merger, (I) at least 51% of the issued and outstanding equity securities of TruGreen Holdings will be legally and beneficially owned by funds managed by Clayton, Dubilier & Rice free and clear of any Encumbrances and (II) all of the issued and outstanding equity securities of TruGreen Holdings have been validly issued, are fully paid and nonassessable and have not been issued in violation of any preemptive or similar rights.

(b) There are (and at the Closing, after giving effect to the Transactions, and after giving effect to the Merger, there will be, except as set forth in the LLC Agreement) no outstanding options, warrants, calls, rights or any other agreements, trust, proxies or similar arrangements affecting the sale, issuance or voting of any equity securities of TruGreen Holdings or any of its Subsidiaries (including the TruGreen Entities), or any securities or other instruments convertible into, exchangeable for or evidencing the right to purchase any equity securities of TruGreen Holdings or any of its Subsidiaries (including the TruGreen Holdings or any of its Subsidiaries). None of TruGreen Holdings or any of its Subsidiaries (including the TruGreen Holdings or other securities, the holders of which have the right to vote (or which are convertible into or exchangeable for securities having the right to vote) with the stockholders of TruGreen Holdings and/or its Subsidiaries (including the TruGreen Entities).

(c) None of the TruGreen Entities has (or at the Closing, after giving effect to the Transactions, or after giving effect to the Merger, will have) any Subsidiaries or owns (or at the Closing, after giving effect to the Transactions, or after giving effect to the Merger, will own) any equity interests in any other Person, except for other TruGreen Entities and the SLS Entities (after giving effect to the Transactions).

4.3 No Conflict.

None of the execution, delivery and performance by TruGreen Holdings and the TruGreen Entities of this Agreement, the Ancillary Agreements or the Merger Agreement (in each case to the extent a party thereto), or the consummation of transactions contemplated hereby or thereby (including, for the avoidance of doubt, the Transactions, the Merger and the operation of the business of the TruGreen Entities as currently conducted after the Closing but excluding the effects of any SLS Franchise Agreement or other Contract to which an SLS Entity is a party) will:

(a) conflict with or violate their respective articles of incorporation or bylaws (or similar organizational documents);

(b) assuming all Consents referred to in <u>TruGreen Schedule 4.4</u> have been obtained, conflict with or violate any Law applicable to TruGreen Holdings or the TruGreen Entities, or to any of their respective properties, rights or assets;

(c) result in any breach of, or constitute a default (or event which with the giving of notice or lapse of time, or both, would become a default) under, or give to any Person any right of termination, amendment, rescission, acceleration or cancellation of, or any right of notice or Consent with respect to, any Contract, Permit, franchise or other instrument to which TruGreen Holdings or any TruGreen Entity (or any other Subsidiary thereof) is a party or by which any of the assets, properties or rights of TruGreen Holdings or the TruGreen Entities (or any other Subsidiary thereof) is bound or affected; or

(d) result in the creation or imposition of any Encumbrance other than Permitted Encumbrances on any assets of the TruGreen Entities,

except, in the case of clauses (b), (c) and (d), where such conflicts, violations, rescissions, accelerations, breaches, defaults, terminations, cancellations or failures to obtain such Consents would not constitute a TruGreen Material Adverse Effect.

4.4 Consents and Approvals.

The execution and delivery by TruGreen Holdings of this Agreement and the Merger Agreement, does not, and the execution and delivery by TruGreen Holdings and the TruGreen Entities of the Ancillary Agreements, to the extent they are a party thereto, will not, and the performance by TruGreen Holdings and the TruGreen Entities of this Agreement and the Merger Agreement (including, for the avoidance of doubt, the Transactions and the Merger) and the Ancillary Agreements, to the extent they are a party thereto, and the consummation of the transactions contemplated hereby and thereby will not, require TruGreen Holdings or any TruGreen Entity to seek or obtain any Consent from, or make any filing or notification with, any Governmental Body or other Person, except for (\underline{x}) those Consents set forth on TruGreen Schedule 4.4(\underline{x}), (\underline{y}) compliance with the applicable requirements of the HSR Act or any other Competition Laws of the jurisdictions set forth on TruGreen Schedule 4.4(\underline{y}) and (\underline{z}) those Consents from, and filings with, Persons (other than Governmental Bodies) the failure of which to be made or obtained would not constitute a TruGreen Material Adverse Effect.

4.5 <u>TruGreen Financial Statements</u>.

(a) The audited combined statement of financial position of TruGreen Holdings as at December 31, 2014, and the related audited combined statements of operations and comprehensive loss, shareholders' equity and cash flows of TruGreen Holdings for the year ended December 31, 2014, together with all related notes and schedules thereto, accompanied by the reports thereon of TruGreen Holdings's independent auditors (collectively, the "<u>Audited TruGreen Financial Statements</u>") and the unaudited combined statement of financial position of TruGreen Holdings as at December 31, 2013 and September 30, 2015, and the related combined statements of operations and comprehensive loss, shareholders' equity and cash flows of TruGreen Holdings for the periods ended December 31, 2012, December 31, 2013 and September 30, 2015, together with all related notes and schedules thereto (together with the Audited TruGreen Financial Statements, the "<u>TruGreen Financial Statements</u>") are included in <u>TruGreen Schedule 4.5(a)</u>. The TruGreen Financial Statements (<u>i</u>) have, with respect to the TruGreen Financial Statements as at and for the periods ending December 31, 2014 and

September 30, 2015 only, been prepared based on the books and records of TruGreen Holdings, (<u>ii</u>) have been prepared in accordance with GAAP consistently applied (subject, in the case of interim period (September 30) unaudited statements, to normal year-end adjustments which will not, individually or in the aggregate, be material in nature or amount to the TruGreen Entities, taken as a whole) and (<u>iii</u>) present fairly in all material respects the combined financial position, the results of operations and cash flows of the TruGreen Entities as of the respective dates thereof and for the respective periods indicated therein, except, in the case of clauses (ii) and (iii), as otherwise noted therein and reflected in the TruGreen Financial Statements.

(b) TruGreen Holdings maintains, and has maintained since January 14, 2014, a system of internal controls over financial reporting designed to ensure reasonable assurances regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with GAAP. Since January 14, 2014, until the date of this Agreement, TruGreen Holdings has disclosed, based on the most recent evaluation of internal controls prior to the date of this Agreement, to its auditors (<u>i</u>) all significant deficiencies and material weaknesses in the design or operation of internal controls over financial reporting which are reasonably likely to adversely affect in any material respect TruGreen Holdings' ability to record, process, summarize and report financial information and (<u>ii</u>) any fraud, whether or not material, that involves management or other employees who have a significant role in TruGreen Holdings' internal controls over financial reporting.

(c) Since January 14, 2014, no director or officer, to the Knowledge of TruGreen Holdings, external auditor or external accountant of TruGreen Holdings or its Subsidiaries (including the TruGreen Entities) or any employee (other than officers) or representative of TruGreen Holdings or its Subsidiaries (including the TruGreen Entities) has received any material complaint, allegation, assertion or claim, whether written or oral, regarding the accounting or auditing practices, procedures, methodologies or methods of TruGreen Holdings or the TruGreen Entities that would reasonably be expected to, in each case, impact the combined financial statements of TruGreen Holdings or their respective internal accounting controls, including any material complaint, allegation, assertion or claim that any TruGreen Entity has engaged in questionable accounting or auditing practices.

4.6 No Undisclosed Material Liabilities.

No TruGreen Entity, after giving effect to the completion of the Merger and the Transactions, has any Liabilities, whether known, unknown, absolute, accrued, contingent or otherwise and whether due or to become due, in each case that are required under GAAP to be reflected or reserved against in the financial statements of TruGreen Holdings or disclosed in the notes thereto, except (<u>i</u>) as set forth in <u>TruGreen Schedule 4.6</u>, (<u>ii</u>) Liabilities disclosed or reserved against in the TruGreen Financial Statements as at the Balance Sheet Date or specifically disclosed in the notes thereto, (<u>iii</u>) Liabilities that were incurred after the Balance Sheet Date in the ordinary course of business, (<u>iv</u>) Liabilities which, individually or in the aggregate, would not be material to the TruGreen Entities, taken as a whole, and (<u>v</u>) Liabilities that are specifically contemplated by, or are pursuant to this Agreement or the Ancillary Agreements and incurred on or after the date hereof. No TruGreen Entity is a party to, or has

any commitment to become a party to, any joint venture, off-balance sheet partnership or any similar Contract (including any Contract relating to any transaction or relationship between or among the TruGreen Entities, on the one hand, and TruGreen Holdings or any of its Affiliates (other than Newco and its Subsidiaries following the Closing) including any structured finance, special purpose or limited purpose Person, on the other hand, or any "off-balance sheet arrangement" (as defined in Item 303(a) of Regulation S-K of the SEC)).

4.7 Litigation.

As of the date hereof, there is no Proceeding pending or, to the Knowledge of TruGreen Holdings, threatened against TruGreen Holdings or any of its Subsidiaries (including the TruGreen Entities), or any of their respective assets, rights or properties which bring into question the validity of this Agreement or that would reasonably be expected, individually or in the aggregate, to be materially adverse to the TruGreen Entities, taken as a whole. There are no Governmental Orders seeking or purporting to enjoin or restrain the execution, delivery and performance by TruGreen Holdings of this Agreement or the consummation by TruGreen Holdings or any of its Subsidiaries (including the TruGreen Entities) of the transactions contemplated hereby or that would reasonably be expected, individually or in the aggregate, to be materially adverse to the TruGreen Entities, taken as a whole.

4.8 Employee Benefits; Labor.

(a) <u>TruGreen Schedule 4.8(a)</u> lists, as of the date of this Agreement, each material Employee Plan that TruGreen Holdings or any of the TruGreen Entities sponsors, maintains or contributes to, or is required to maintain or contribute to for the benefit of any current or former TruGreen Employee (without regard to materiality, the "<u>TruGreen Benefit Plans</u>"). No TruGreen Benefit Plan provides benefits to any individual other than current or former employees of the TruGreen Entities, Persons who provide or have provided services in respect of the business of the TruGreen Entities and any beneficiaries and dependents of the foregoing.

(b) With respect to each material TruGreen Benefit Plan, TruGreen Holdings has provided Scotts, to the extent applicable, complete and correct copies of: all written plan documents and descriptions of all unwritten plans (and all amendments thereto); all trust agreements, insurance contracts or other funding arrangements therefor; the two most recent actuarial and trust reports; the two most recent Forms 5500 and all schedules thereto; the most recent IRS determination letters; all current summary plan descriptions and summaries of material modification (or with respect to the foregoing items, the equivalent thereto required by any foreign Governmental Body); and copies of all material communications, received from or sent to any Governmental Body, relating to TruGreen Benefit Plans in the last three (3) years.

(c) The TruGreen Benefit Plans have been maintained and operated in material compliance with their terms and, to the extent applicable, provisions of ERISA, the Code and all other applicable Laws, domestic or foreign. None of the TruGreen Benefit Plans are presently under audit or examination (nor has notice been received of a potential audit or

examination) by the IRS, the Department of Labor or any other Governmental Body, domestic or foreign. Other than routine claims for benefits, there are no material Proceedings pending or, to the Knowledge of TruGreen Holdings, threatened with respect to any TruGreen Benefit Plan. No non-exempt "prohibited transaction" within the meaning of Section 4975 of the Code and Section 406 of ERISA, has occurred or is reasonably expected to occur with respect to any of the TruGreen Benefit Plans. No TruGreen Benefit Plan is maintained outside the jurisdiction of the United States, or which covers any TruGreen Employee residing or working outside of the United States.

(d) Each TruGreen Benefit Plan (or any trust related thereto) intended to be qualified or otherwise receive favorable Tax treatment under applicable Law (including under Section 401(a) of the Code) has, if necessary, been determined by the applicable Governmental Body or Tax Authority to be so qualified (including by way of receipt of a favorable determination letter or opinion letter from the IRS upon which it may rely regarding its qualified status under the applicable Law), and to the Knowledge of TruGreen Holdings, there are no existing circumstances or events that would reasonably be expected to result in any revocation of, or a change to, such qualification or entitlement to favorable Tax treatment under applicable Law.

(e) With respect to current or former TruGreen Employees, TruGreen Holdings is and since January 14, 2014 has been, and its Subsidiaries (including the TruGreen Entities) are and since October 1, 2013 have been, in compliance with all applicable Laws respecting employment and labor, including discrimination, harassment, workers' compensation, classification of workers as exempt or non-exempt or as employees or independent contractors, overtime compensation, unlawful retaliation, all titles of the Americans with Disabilities Act, meal and rest breaks, immigration, employee leave issues, workers' compensation, unemployment insurance, wages and hours, occupational safety and health, fair labor standards and the WARN Act and any similar state, local or foreign laws, except such noncompliance that would not, individually or in the aggregate, reasonably be expected to be materially adverse to the TruGreen Entities, taken as a whole.

(f) The TruGreen Entities are not a party to or bound by, and none of the TruGreen Employees are covered by, any collective bargaining agreement and there are no labor unions, employee representatives, works councils or other organizations representing, purporting to represent or attempting to represent any TruGreen Employees. Since October 1, 2013, there has not occurred or, to the Knowledge of TruGreen Holdings, been threatened any material strike, sympathy strike, slowdown, picketing, work stoppage, concerted refusal to work overtime or other similar labor activity with respect to any TruGreen Employee. There are no labor disputes currently subject to any grievance procedure, arbitration or litigation, which, if adversely decided, may reasonably, individually or in the aggregate, create a liability in excess of \$250,000 and there is no representation petition pending or, to the Knowledge of TruGreen Holdings, threatened with respect to any TruGreen Employee.

(g) No TruGreen Benefit Plan or a plan of any of its ERISA Affiliates is subject to, or has in the six (6) years immediately preceding the Closing Date been subject to,

Title IV of ERISA. None of TruGreen Holdings, the TruGreen Entities or their ERISA Affiliates has in the past six (6) years contributed or been obligated to contribute to or has any current or contingent liability or obligation in respect of: (i) a "multiemployer plan" within the meaning of Section 4001(a)(3) of ERISA, (ii) a "multiple employer plan" within the meaning of Sections 4063 or 4064 of ERISA, (iii) a "multiple employer welfare arrangement" within the meaning of Section 3(40) of ERISA or (iv) a self-insured welfare benefit plan.

(h) All contributions and premiums required to have been paid by TruGreen Holdings and its Subsidiaries (including the TruGreen Entities) or any of its ERISA Affiliates to any TruGreen Benefit Plan under the terms of any such plan or its related trust, insurance contract or other funding arrangement, or pursuant to any applicable Law (including ERISA and the Code) or collective bargaining agreement have been paid within the time prescribed by any such plan, agreement or applicable Law or properly accrued under GAAP on or before the Closing Date. No TruGreen Entity has a contract or commitment to create any additional plan, agreement, arrangement or policy which, upon its creation, will be a TruGreen Benefit Plan or to modify any existing TruGreen Benefit Plan in a manner which would increase materially the expense of maintaining the TruGreen Benefit Plan, considered in the aggregate, except as required by applicable Law or tax qualification requirement or as necessary to conform a TruGreen Benefit Plan to applicable Law. There have been no modifications of TruGreen Benefit Plans or creation of new TruGreen Benefit Plans since the date of the last balance sheet that would be prohibited by the preceding sentence. No TruGreen Entity has terminated any material TruGreen Benefit Plans since the last balance sheet date.

(i) No TruGreen Benefit Plan provides welfare benefits after termination of employment, except to the extent required by Section 4980B(f) of the Code or any similar provision of state, local or foreign Law.

(j) Neither this Agreement, nor the Merger Agreement, nor the consummation of the transactions contemplated hereby or thereby (either alone or in combination with any other event), including, for the avoidance of doubt, the Merger and the Transactions, would reasonably be expected to (i) result in any payment becoming due or increase the amount of any compensation due to any current or former employee, consultant, officer or director of the TruGreen Entities, (ii) result in any forgiveness of indebtedness, or trigger any funding obligation under any TruGreen Benefit Plan or impose any restrictions or limitations on any TruGreen Entity's rights to administer, amend or terminate any TruGreen Business Benefit Plan or (iii) result in the acceleration of time of payment or vesting of any such benefits, or increase the amount of compensation or benefits due to any employee, consultant, officer or director of the TruGreen Entities. No amount that would be received (whether in cash or property or the vesting of property) as a result of the consummation of the Transactions or the Merger by any TruGreen Employee who is a "disqualified individual" (as such term is defined in Treasury Regulations Section 1.280G–1) under any TruGreen Benefit Plans or otherwise would be characterized as an "excess parachute payment" (as defined in Section 280G(b)(1) of the Code). No TruGreen Entity is required to gross up or reimburse a payment to any TruGreen Employee for Taxes incurred under Sections 4999, 409A or 457A of the Code.

(k) The execution and delivery by TruGreen Holdings of this Agreement and the Merger Agreement, the execution and delivery by TruGreen Holdings and the TruGreen Entities of the Ancillary Agreements, to the extent they are a party thereto, the performance by TruGreen Holdings and the TruGreen Entities of this Agreement and the Merger Agreement (including, for the avoidance of doubt, the Transactions and the Merger) and the Ancillary Agreements, to the extent they are a party thereto, and the consummation of the transactions contemplated hereby and thereby would not reasonably be expected to require TruGreen Holdings or any of its Subsidiaries (including the TruGreen Entities) to seek or obtain any Consent from, engage in consultation with, or issue any notice to or make any filing with (as applicable) any TruGreen Employees or any representatives, labor unions, works councils or similar organizations representing TruGreen Employees, or any Governmental Body with respect to TruGreen Employees (except to the extent previously made or obtained prior to the date hereof).

4.9 <u>Taxes</u>.

(a) (i) All material Tax Returns required to be filed by or with respect to the TruGreen Entities or the assets of the TruGreen Entities have been filed in a timely manner (within any applicable extension periods) and all such Tax Returns are true, correct and complete in all material aspects, (ii) all material Taxes required to be paid with respect to the TruGreen Entities or the assets of the TruGreen Entities (whether or not shown on such returns) have been timely paid in full and (iii) there exist no material liens for Taxes with respect to any assets of the TruGreen Entities, except for liens for Taxes not yet due and payable. No jurisdiction (whether within or outside the United States) in which any TruGreen Entity has not filed a particular type of Tax Return or paid a particular type of Tax has asserted that such TruGreen Entity is required to file such Tax Return or pay such type of Tax in such jurisdiction.

(b) There are no outstanding agreements or waivers extending the statutory period for assessment or collection of any Taxes of the TruGreen Entities or with respect to the assets of the TruGreen Entities and no written powers of attorney with respect to any such Taxes. The time for filing any Tax Return with respect to the TruGreen Entities or the assets of the TruGreen Entities has not been extended, other than any extension to which a TruGreen Entity is entitled under applicable Law without the consent of the relevant Tax Authority or Governmental Body.

(c) There are no audits, Proceedings, investigations or other actions pending or threatened in writing against the TruGreen Entities or the assets of the TruGreen Entities in respect of Taxes or Tax Returns. No Tax Authority has asserted or threatened to assert, in each case in writing, any deficiency, claim or issue with respect to Taxes or any adjustment to Taxes against the TruGreen Entities or with respect to the assets of the TruGreen Entities for any taxable period for which the period of assessment or collection remains open.

(d) None of the TruGreen Entities will be required to include any item of income in, or exclude any item of deduction from, taxable income for any taxable period (or portion thereof) ending after the Closing Date as a result of (i) any change in method of

accounting for a taxable period ending on or prior to the Closing Date, (<u>ii</u>) any installment sale or open transaction, (<u>iii</u>) any prepaid amount received or paid, or (<u>iv</u>) any election pursuant to Section 108(i) of the Code (or any similar provision of state, local or foreign Law) made with respect to any taxable period ending on or prior to the Closing Date.

(e) None of the TruGreen Entities during the five-year period ending on the date hereof, has been a distributing corporation or a controlled corporation in a transaction intended to be governed by Section 355 of the Code.

(f) All material Taxes that the TruGreen Entities were required to withhold or collect have been duly withheld or collected and have been paid, to the extent required by applicable Law, to the proper Tax Authority. The TruGreen Entities have complied in all material respects with the rules and regulations relating to the withholding and remittance of Taxes.

(g) None of the TruGreen Entities (<u>i</u>) has received or applied for a Tax ruling or entered into a closing agreement pursuant to Section 7121 of the Code (or any predecessor provision or any similar provision of state, local or foreign Law), in either case that would be binding upon any SLS Entity after the Closing Date, (<u>ii</u>) is or has been a member of any affiliated, consolidated, combined or unitary group for purposes of filing Tax Returns or paying Taxes or (<u>iii</u>) has any liability for the Taxes of any Person (other than for any other TruGreen Entity) whether under Treasury Regulation Section 1.1502-6 or any similar provision of state, local or foreign Law, as a transferee or successor, pursuant to any Tax sharing or indemnity agreement or other contractual agreements, or otherwise.

(h) None of the TruGreen Entities has engaged in any "listed transactions" within the meaning of Treasury Regulation Section 1.6011-4(b)(2).

(i) <u>TruGreen Schedule 4.9(i)</u> lists all of the TruGreen Entities for which an election to change the U.S. federal tax classification of such TruGreen Entities has been made within the five-year period ending on the date of this Agreement.

(j) All material transactions entered into by any of the TruGreen Entities with any related party have been carried out at arm's length, and each of the TruGreen Entities has complied in all material respects with all applicable transfer pricing disclosure, reporting, documentation and other similar requirements.

4.10 Absence of Certain Changes or Events.

(a) Since September 30, 2015 and through the date of this Agreement, there has not been any event, development or state of circumstances that constitutes a TruGreen Material Adverse Effect.

(b) Between September 30, 2015 and the date hereof (<u>i</u>) the business of the TruGreen Entities had been conducted in the ordinary course consistent with past practices other

than matters in connection with the Transactions, the Newco Restructuring or the Merger and (<u>ii</u>) neither TruGreen Holdings nor any of its Subsidiaries (including the TruGreen Entities) has taken any action that would, if taken after the date hereof, require the consent of Scotts under <u>Section 5.1(b)</u>.

4.11 <u>Title to Assets; Sufficiency of Assets</u>.

(a) Except as would not, individually or in the aggregate, reasonably be expected to be materially adverse to TruGreen Holdings or to the TruGreen Entities, taken as a whole, each TruGreen Entity has good and valid (and, in the case of TruGreen Owned Real Property, good, valid and marketable fee simple) title to, or have a valid leasehold or licensed interest in, all of its tangible and intangible assets, properties and rights (real and personal, including all Intellectual Property), in each case, free and clear of all Encumbrances, except for Permitted Encumbrances.

(b) The assets that are owned, leased or licensed by the TruGreen Entities constitute all of the assets, properties and rights required for the conduct of their business in all material respects as of the Balance Sheet Date, the date hereof and immediately following the Closing.

(c) Other than for its rights under this Agreement or set forth in <u>TruGreen Schedule 5.8(b)</u>, none of TruGreen Holdings or its Affiliates, other than the TruGreen Entities, has or at the Closing will have, any right, title or interest (other than as a direct or indirect owner of the TruGreen Entities) in any of the assets, properties or rights owned, leased or licensed by the TruGreen Entities that are material to the conduct of the business of the TruGreen Entities.

4.12 <u>Real Property</u>.

(a) <u>TruGreen Schedule 4.12(a)</u> sets forth an accurate and complete list of each real property owned by any TruGreen Entity as of the date hereof (together with all structures, facilities, improvements and fixtures presently or hereafter located thereon or attached thereto, the "<u>TruGreen Owned Real Property</u>"), including the address and owner of each parcel of such TruGreen Owned Real Property.

(b) <u>TruGreen Schedule 4.12(b)</u> sets forth an accurate and complete list of each real property leased, subleased, licensed or occupied by any TruGreen Entity (the "<u>TruGreen Leased Real Property</u>"), including all leases, subleases or licenses (together with any and all amendments and modifications thereto and any guarantees thereof) in effect as of the date relating thereto (collectively, the "<u>TruGreen Real Property Leases</u>") and identifying the tenant and address for each TruGreen Real Property Lease. TruGreen Holdings has made available to Scotts accurate and complete copies of each material TruGreen Real Property Lease.

(c) None of the TruGreen Entities has leased, subleased or granted to any Person any right to possess, lease or occupy any portion of the TruGreen Owned Real Property or the TruGreen Leased Real Property.

(d) None of the TruGreen Entities holds, has granted or is obligated under any option, right of first offer, right of first refusal or other contractual right to purchase, acquire, sell or dispose of the TruGreen Owned Real Property or the TruGreen Leased Real Property or any portion thereof or interest therein or any other real property.

4.13 Intellectual Property.

(a) <u>TruGreen Schedule 4.13(a)</u> sets forth an accurate and complete list of (i) all registered Intellectual Property and all applications for registration for Intellectual Property used or held for use by the TruGreen Entities (the "<u>TruGreen Registered</u> <u>Intellectual Property</u>") and (<u>ii</u>) all unregistered Marks that are material to the business of the TruGreen Entities and owned, beneficially or of record, by the TruGreen Entities as of the date of this Agreement, and the owner thereof. The TruGreen Registered Intellectual Property and each of the unregistered Marks, Copyrights and Trade Secrets that are owned by the TruGreen Entities and are material to their business (collectively, the "<u>TruGreen Owned Intellectual Property</u>") are owned by the TruGreen Entities free and clear of all Encumbrances other than Permitted Encumbrances.

(b) Other than as set forth on <u>TruGreen Schedule 4.13(b)</u>, there is no pending action, opposition, interference, reexamination, proceeding or, to the Knowledge of TruGreen Holdings, written threatened claim against any TruGreen Entity or to which any TruGreen Entity is party in respect of any TruGreen Owned Intellectual Property or any other Intellectual Property material to the TruGreen Entities' business.

(c) Other than as set forth on <u>TruGreen Schedule 4.13(c)</u>: (i) to the Knowledge of TruGreen Holdings, the conduct of the TruGreen Entities' business does not infringe, misappropriate or violate any Intellectual Property of any Person and (<u>ii</u>) to the Knowledge of TruGreen Holdings, no Person is infringing, misappropriating or violating the TruGreen Owned Intellectual Property, except in each case of (i) and (ii) as would not constitute a TruGreen Material Adverse Effect.

(d) Except as would not constitute a TruGreen Material Adverse Effect, the TruGreen Entities have used commercially reasonable efforts to ensure protection of the TruGreen Owned Intellectual Property, including the confidentiality of their Trade Secrets.

(e) The TruGreen Entities have implemented reasonable backup and disaster recovery arrangements designed to provide for the continued operation of their business in the event of a disaster or business interruption. The TruGreen Internal IT Systems (<u>i</u>) are in good repair and operating condition and are adequate and suitable in all material respects for the purposes for which they are being used or held for use, (<u>ii</u>) conform in all material respects with their related documentation and (<u>iii</u>) to the Knowledge of TruGreen Holdings do not contain any virus, Trojan, or other malicious code that would reasonably be expected to interfere in any material respect with their business. The TruGreen Entities have not, in the past three (3) years, experienced any data security breach or unauthorized data release incident that has had a material and adverse effect on their business.

(f) Each of the TruGreen Entities is in compliance with all contractual and requirements under applicable Law pertaining to data protection or information privacy and security, including any privacy policy concerning the collection or use of data or information, except as would not, constitute a TruGreen Material Adverse Effect.

4.14 Contracts.

<u>TruGreen Schedule 4.14</u> sets forth an accurate and complete list of all Contracts in effect as of the date hereof to which TruGreen Holdings or any of its Subsidiaries (including the TruGreen Entities) is a party or by which any property or asset of TruGreen Holdings or any of its Subsidiaries (including the TruGreen Entities) is bound or affected (other than insurance policies, purchase orders or standard sales orders entered into by TruGreen Holdings or any of its Subsidiaries (including the TruGreen Entities), as applicable, in the ordinary course of business) which by its terms:

(a) is not terminable at will within three months and requires future expenditures or other performance with respect to goods, equipment or services having an annual value in excess of \$100,000;

(b) relates to the obligation of the TruGreen Entities to purchase products, materials, supplies, goods, equipment, other assets or services pursuant to which payments of \$250,000 or more were made during the 12-month period ending on September 30, 2015;

(c) relates to the obligation of the TruGreen Entities to sell products or services pursuant to which payments of \$250,000 or more were received during the 12-month period ending on September 30, 2015;

(d) relates to any outstanding Indebtedness, other than TruGreen Intercompany Balances or TruGreen Intracompany Obligations, having an outstanding principal amount in excess of \$1,000,000;

(e) (i) limits the ability of any TruGreen Entity to compete in any line of business or with any Person in any geographic area or that would so limit the freedom of the TruGreen Entities or any of their Affiliates after the Closing (including the SLS Entities) or (ii) contains exclusivity obligations binding on any TruGreen Entity or, following the Closing, any of its Affiliates (including the SLS Entities), in each case, other than TruGreen Franchise Agreements;

(f) requires any capital commitment or capital expenditure (including any series of related expenditures) by the TruGreen Entities of greater than \$500,000;

(g) relates to the acquisition or disposition of any business, assets, properties or rights under which the TruGreen Entities have any future liability with respect to an "earn-out," contingent purchase price, deferred purchase price or similar contingent payment

obligation, or any indemnification obligation that could result in payments in excess of \$100,000;

(h) constitutes a contract or agreement with any officer, employee, director, stockholder or other Affiliate of the TruGreen Entities (other than (\underline{x}) TruGreen Benefit Plans and (\underline{y}) TruGreen Intercompany Contracts that have been terminated prior to the Closing without any additional liability or obligation to any party);

(i) constitutes a contract or agreement with ServiceMaster Global Holdings, Inc. or any of its Affiliates;

(j) constitutes an agreement that contains any indemnification obligations of any of the TruGreen Entities, or credit support or guarantee relating to such indemnification obligations, other than any such indemnification, credit support or guarantee obligations incurred in the ordinary course of business or that requires credit support or indemnification obligations of less than \$100,000;

(k) is an Intellectual Property Contract that is material to the business of the TruGreen Entities (excluding Intellectual Property Contracts for commercially available off-the-shelf Software that is not the subject of a negotiated agreement and for which the aggregate one-time or annual amounts paid or payable to or by the TruGreen Entities related to such Contract are less than \$50,000);

(l) is a contract that creates (or governs the operation of) any partnership, joint venture, alliance or similar agreement or arrangement, in each case that is material to the operation of TruGreen Entities, taken as a whole (other than this Agreement and the Ancillary Agreements);

- (m) is a TruGreen Franchise Agreement; or
- (n) is a TruGreen Real Property Lease that has annual rent obligations in excess of \$500,000

(the Contracts described in clauses (a) through (n), the "TruGreen Material Contracts").

TruGreen Holdings has made available to Scotts a true and complete copy of each TruGreen Material Contract (including, as of the date hereof, all modifications and amendments thereto and written waivers thereunder). Except as would not constitute a TruGreen Material Adverse Effect, each TruGreen Material Contract to which a TruGreen Entity is a party is in full force and effect in accordance with its respective terms, except after the date hereof, for any such TruGreen Material Contracts that expire according to their terms after the date hereof. Except, after the date hereof for any such TruGreen Material Contracts that expire according to their terms after the date hereof, each TruGreen Material Contract is valid, binding and enforceable against the TruGreen Entity party thereto and, to the Knowledge of TruGreen Holdings, each other party thereto in accordance with its terms, except as enforceability may be limited by

bankruptcy, insolvency, reorganization, moratorium or other similar Laws affecting creditors' rights and general principles of equity affecting the availability of specific performance and other equitable remedies (regardless of whether enforcement is sought in a proceeding at law or in equity) and except as would not constitute a TruGreen Material Adverse Effect. No TruGreen Entity is in default, violation or breach in any material respect under (or, to the Knowledge of TruGreen Holdings, is alleged to be in default or breach in any material respect under) any such TruGreen Material Contract to which it is a party, or has within the last 12 months provided or received written notice of any intention or threat to terminate or not to renew, any such TruGreen Material Contract, except as could not reasonably be expected to be material to the TruGreen Entities, taken as a whole. To the Knowledge of TruGreen Holdings, no event or circumstance has occurred and is continuing that constitutes or, with notice or the passage of time, or both, would constitute, a default, violation or breach in any respect under any such TruGreen Material Contract by any party thereto, or result in a termination thereof or would cause or permit the acceleration of or other changes of or to any right or obligation or the loss by TruGreen Material Contract according to its terms, based on the mere passage of time), in each case, except for such defaults, breaches, violations terminations, accelerations or changes as would not, individually or in the aggregate, reasonably be expected to have a material and adverse effect on the ability of the TruGreen Entities to continue to conduct the business of TruGreen Entities as currently conducted following the Closing.

4.15 Compliance with Laws; Permits.

(a) Other than as set forth on <u>TruGreen Schedule 4.15</u>, each TruGreen Entity is, and since October 1, 2013 has been, in compliance with, and has not received written notice of any default or violation of, any Laws applicable to the TruGreen Entities, except as would not constitute a TruGreen Material Adverse Effect. The TruGreen Entities hold all Permits that are required for the conduct of their business as now being conducted, except where the failure to have any such Permits would not constitute a TruGreen Material Adverse Effect. Except as would not constitute a TruGreen Material Adverse Effect. Except as would not constitute a TruGreen Material Adverse Effect, (i) all such Permits are in full force and effect, (ii) no violations are recorded in respect of any such Permits and (iii) no Proceedings are pending or, to the Knowledge of TruGreen Holdings, threatened to revoke, suspend, cancel or limit any such Permit.

(b) No representation or warranty is made under this <u>Section 4.15</u> with respect to Taxes or environmental matters, which representations and warranties are covered exclusively by <u>Sections 4.9</u> (Taxes) and <u>4.18</u> (Environmental Matters) respectively.

4.16 Anti-Corruption; Sanctions.

(a) None of the TruGreen Entities, or any of their respective officers, directors, agents, distributors, employees and other Persons involved in or acting for or on behalf of the TruGreen Entities (the "<u>TruGreen Relevant Persons</u>"), has directly or indirectly violated or taken any act in furtherance of violating any provision of any Anti-Corruption Law.

(b) None of the TruGreen Entities or any TruGreen Relevant Persons has (i) directly or indirectly used any corporate funds or assets or taken any act in the furtherance of an offer, payment, promise to pay, gift, bribe, rebate, loan, payoff, kickback, political contribution, entertainment expense, or any other transfer of value to a Government Official for the purpose of inducing such Government Official to do any act or make any decision in an official capacity, including a decision to fail to perform an official function, or use his or her or its influence with a Governmental Body in order to affect any act or decision of such Governmental Body for the purpose of assisting any Person to obtain or retain any business, or to facilitate efforts of any Person to transact business or for any other improper purpose (*e.g.*, to obtain a tax rate lower than allowed by Law), or (<u>ii</u>) established or maintained, or is maintaining, any unlawful fund of corporate monies or other properties with an intent of carrying out any of the acts set forth in the preceding clause (i).

(c) To the Knowledge of TruGreen Holdings, there is no investigation, allegation, request for information, or other inquiry by any Governmental Body regarding the TruGreen Entities' actual or possible violation of the Anti-Corruption Laws.

(d) To the Knowledge of TruGreen Holdings, none of the TruGreen Entities' officers, directors, employees or agents (or their employees) is a Government Official or consultant to any Government Official, and there is no existing family relationship between any officer, director, employee, member or agent (or their employees) of the TruGreen Entities and any Government Official.

(e) None of the TruGreen Entities or any TruGreen Relevant Person (with respect to TruGreen Relevant Persons other than the TruGreen Entities and their Affiliates and their respective officers and directors, solely to the Knowledge of TruGreen Holdings) has directly or indirectly (<u>i</u>) circumvented the internal accounting controls of any of the TruGreen Entities, (<u>ii</u>) falsified any of the books, records or accounts of any of the TruGreen Entities or (<u>iii</u>) made false or misleading statements to, or attempted to coerce or fraudulently influence, an accountant in connection with any audit, review or examination of the TruGreen Financial Statements.

(f) None of the TruGreen Entities or, to the Knowledge of TruGreen Holdings, the TruGreen Relevant Persons is (i) the subject or, to the Knowledge of TruGreen Holdings, target of any Sanctions, (ii) located organized or resident in a Sanctioned Country, (iii) to the Knowledge of TruGreen Holdings engaged in, directly or indirectly, any material dealings or transactions with or relating to any person or entity that at the time of the dealing or transaction is or was the subject or the target of Sanctions or in, with or relating to any Sanctioned Country or (iv) the subject or target of a governmental investigation relating to Sanctions nor has any received notice of, or otherwise become aware of, any claim, action, suit, or proceeding or investigation, or inquiry, formal or informal, against it with respect to Sanctions.

4.17 Insurance.

<u>TruGreen Schedule 4.17</u> contains an accurate and complete list of all material policies of insurance maintained as of the date of this Agreement with respect to any TruGreen Entity (the "<u>TruGreen Policies</u>"). Except as would not constitute a TruGreen Material Adverse Effect (<u>a</u>) no TruGreen Entity is in default with respect to its obligations under any TruGreen Policy and (<u>b</u>) no TruGreen Entity has received, since the time any TruGreen Policy was last renewed or issued (whichever is later), written notice of termination, cancellation or non-renewal of any such TruGreen Policy from any of its insurance brokers or carriers. As of the date hereof, there is no material claim by any of the TruGreen Entities pending under any TruGreen Policy as to which coverage has been denied or disputed by the underwriters of such TruGreen Policy or in respect of which such underwriters have reserved their rights. TruGreen Holdings acknowledges that after the Closing, the SLS Entities shall cease to be insured by, have access or availability to, be entitled to make claims on, be entitled to claim benefits from or seek coverage under any of Scotts's or its Subsidiaries and/or Affiliates' Insurance Arrangements, other than as set forth in <u>Section 5.9</u>.

4.18 Environmental Matters.

(a) Each of the TruGreen Entities complies and, while owned by TruGreen Holdings, has complied with all applicable Environmental Laws, except where the failure to comply would not constitute a TruGreen Material Adverse Effect.

(b) Each of the TruGreen Entities holds and is in compliance with all Permits required under applicable Environmental Laws to operate as it currently operates, except where the failure to hold such Permits would not constitute a TruGreen Material Adverse Effect.

(c) Since October 1, 2013, none of the TruGreen Entities has received any written notice from any Person regarding any actual or alleged violation of Environmental Laws, or any liabilities or alleged liabilities for investigation costs, cleanup costs, response costs, corrective action costs, personal injury, property damage, natural resources damages or attorney fees under Environmental Laws, the subject of which would, constitute a TruGreen Material Adverse Effect.

(d) There is no Proceeding by or before any Governmental Body pending or, to the Knowledge of TruGreen Holdings, threatened in writing against any of the TruGreen Entities, relating to alleged liability under Environmental Laws, that if determined adversely would constitute a TruGreen Material Adverse Effect.

(e) No Release of Hazardous Substances has occurred on, at, under or from or, to the Knowledge of TruGreen Holdings, is affecting any property currently owned or leased by any of the TruGreen Entities and, to the Knowledge of TruGreen Holdings, no Release of Hazardous Substances has occurred on, at, under or from or is affecting any property formerly owned or leased by any of the TruGreen Entities, in either case, in a manner that would, constitute a TruGreen Material Adverse Effect.

(f) No TruGreen Entity has, during the period of TruGreen Holdings' ownership of such TruGreen Entity, and to the Knowledge of TruGreen Holdings, no TruGreen Entity has, during any period prior to TruGreen Holdings' ownership of any such TruGreen Entity, disposed of or arranged for the disposal of any Hazardous Substance that has resulted, or would reasonably be expected to result, in investigation costs, cleanup costs, response costs, corrective action costs, personal injury, property damage, natural resources damages or attorney fees under or pursuant to any applicable Environmental Laws, which would constitute a TruGreen Material Adverse Effect.

(g) TruGreen Holdings has made available to Scotts all environmental assessments, investigations and studies in the possession, custody or control of TruGreen Holdings or any of its Subsidiaries (including the TruGreen Entities) prepared since October 1, 2013 and prior to the date hereof (other than those subject to attorney-client privilege, for which summaries have been made available to Scotts prior to the date hereof), which identify issues at properties owned, leased or operated by any of the TruGreen Entities that would, constitute a TruGreen Material Adverse Effect.

(h) Notwithstanding any other provisions of this <u>Article IV</u>, other than as set forth in <u>Section 4.4</u> and this <u>Section 4.18</u>, TruGreen Holdings makes no representations or warranties, either express or implied, with respect to matters arising under any Environmental Law or concerning any Hazardous Substance.

4.19 Financial Advisors.

Other than Merrill Lynch, Pierce, Fenner & Smith Incorporated, no Person has acted, directly or indirectly, as an agent, broker, investment banker, finder or financial advisor for TruGreen Holdings or any of its Subsidiaries in connection with the transactions contemplated by this Agreement, and no Person is entitled to any fee or commission or like payment from any of the TruGreen Entities in respect thereof. TruGreen Holdings has made available to Scotts, solely for informational purposes, complete and correct copies of all letter agreements between TruGreen Holdings or any TruGreen Entity and Merrill Lynch, Pierce, Fenner & Smith Incorporated pursuant to which Merrill Lynch, Pierce, Fenner & Smith Incorporated could be entitled to any payment in connection with the Transactions or the Merger.

4.20 Transactions with Affiliates.

(a) <u>TruGreen Schedule 4.20(a)(i)</u> lists all written material TruGreen Intercompany Contracts as of the Balance Sheet Date and <u>TruGreen Schedule 4.20(a)(ii)</u> lists any agreement or arrangement by and between TruGreen Holdings or any of its Subsidiaries, on the one hand, and Clayton, Dubilier & Rice, LLC or any its Affiliates, on the other hand.

(b) To the Knowledge of TruGreen Holdings, no officer, director or employee of any of the TruGreen Entities, or any family member, relative or Affiliate of any such officer, director or employee, (i) owns, directly or indirectly, any interest in (<u>A</u>) any asset or other

property used in or held for use in the business of the TruGreen Entities or (<u>B</u>) any Person that is a supplier, customer or competitor of any of the TruGreen Entities, (<u>ii</u>) serves as an officer, director or employee of any Person that is a supplier, customer or competitor of any of the TruGreen Entities or (<u>iii</u>) is a debtor or creditor of any of the TruGreen Entities.

4.21 Customers and Suppliers.

<u>TruGreen Schedule 4.21(a)</u> lists the names of the 10 largest customers of the TruGreen Entities, as determined by annual revenue to the TruGreen Entities for the 2014 fiscal year (the "<u>Major TruGreen Customers</u>"). <u>TruGreen Schedule 4.21(b)</u> lists the names of the 10 largest suppliers of the TruGreen Entities, based on payments made by the TruGreen Entities during the 2014 fiscal year (the "<u>Major TruGreen Suppliers</u>"). As of the date of this Agreement, no TruGreen Entity has received written notice (or, to the Knowledge of TruGreen Holdings, any other notice) that any Major TruGreen Customer or Major TruGreen Supplier has terminated or threatened in writing to terminate any material agreement with any of the TruGreen Entities, except as would not constitute a TruGreen Material Adverse Effect.

4.22 <u>Financing</u>. TruGreen Holdings has delivered to Scotts complete, true and correct copies of: (i) the executed commitment letter, dated as of the date hereof by and among JPMorgan Chase Bank, N.A., J.P. Morgan Securities LLC, Credit Suisse AG, Credit Suisse Securities (USA) LLC, ING Capital LLC, Natixis, New York Branch, Coöperatieve Centrale Raiffeisen-Boerenleenbank B.A. "Rabobank Nederland", New York Branch and Rabo Securities USA, Inc. (collectively, the "Lenders") and TruGreen LP (the "Debt Financing Commitment") and any related exhibits, schedules, annexes and supplements, pursuant to which, upon the terms and subject to the conditions set forth therein, the Lenders have agreed to lend the amounts set forth therein to TruGreen LP (the "Debt Financing"); and (ii) the related executed fee letter (the "Fee Letter"). The Debt Financing Commitment and the Fee Letter have not been amended or modified prior to the date of this Agreement (provided that the existence or exercise of the "flex" provisions contained in the Fee Letter shall not constitute an amendment or modification of the Debt Financing Commitment), and, as of the date hereof, the respective commitments contained in the Debt Financing Commitment have not been withdrawn, terminated or rescinded in any respect, and, to the Knowledge of TruGreen Holdings, no such withdrawal, rescission, amendment or modification is contemplated (except for replacements as permitted by <u>Section 5.10(a)</u>). There are no other agreements, side letters or arrangements to which TruGreen LP is a party relating to the Debt Financing Commitment. As of the date hereof, the Debt Financing Commitment is in full force and effect and constitutes the legal, valid and binding obligation of TruGreen LP and, to the Knowledge of TruGreen Holdings, the other parties thereto (subject in each case to the effect of bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium, receivership or similar Laws relating to or affecting creditors' rights generally and by general equity principles, whether considered in Proceedings in equity or at law). Other than as expressly set forth in the Debt Financing Commitment (including as may be set forth in any such Debt Financing Commitment as it may be amended after the date hereof in compliance with, and not in violation of, the provisions hereof), there are no conditions precedent related to the funding of the full net proceeds of the Debt Financing under any agreement relating to the Debt Financing to which TruGreen LP or any of its Affiliates is a party. As of the date hereof,

TruGreen LP is not in breach of any of the terms or conditions set forth in the Debt Financing Commitment and no event has occurred which, with or without notice, lapse of time or both, would constitute a breach or default (or with notice or lapse of time or both would constitute a default) by TruGreen LP under the Debt Financing Commitment, or, to the Knowledge of TruGreen Holdings, the other parties to the Debt Financing Commitment. TruGreen LP has fully paid or caused to be fully paid all commitment fees or other fees required to be paid on or prior to the date hereof pursuant to the Debt Financing Commitment. As of the date hereof, none of TruGreen Holdings or any of its Affiliates has any reason to believe that the full amount of the Debt Financing will not be available at the Closing.

4.23 TruGreen Holdings Subsidiaries.

TruGreen Holdings and its Subsidiaries other than the TruGreen Entities do not (and, as of the Closing, such entities and Newco will not) hold or own any assets, properties or rights or have any Liabilities apart from those (\underline{x}) related to their direct or indirect ownership of the equity interests of the TruGreen Entities, (\underline{y}) incidental to their formation and continuation as corporate, limited partnership or limited liability company entities and (\underline{z}) related to this Agreement and the transactions contemplated hereby or, in the case of TruGreen Holdings, the Merger Agreement and the transactions contemplated thereby.

4.24 TruGreen Holdings Acknowledgment.

(a) Except for the representations and warranties expressly set forth in this Article IV, none of TruGreen Holdings, the TruGreen Entities or any of their Affiliates nor any other Person on behalf of any of the foregoing makes any express or implied representation or warranty with respect to TruGreen Holdings, the TruGreen Entities or the businesses of any of the foregoing with respect to any other information provided, or made available, to Scotts, its representatives or Affiliates in connection with the Transactions, the Merger or the other transactions contemplated by this Agreement or the Merger Agreement, including the accuracy or completeness thereof. Without limiting the foregoing, neither TruGreen Holdings nor any other Person will have or be subject to any liability or other obligation to Scotts, its representatives or Affiliates or any other Person resulting from Scotts's, its representatives' or Affiliates' use of any information, documents, projections, forecasts, business plans or other material made available to Scotts, its representatives or Affiliates, including any information made available in the electronic data room run by R.R. Donnelley & Sons Company and maintained by TruGreen Holdings for purposes of the Transactions and the other transactions, functional "break-out" discussions, responses to questions submitted on behalf of Scotts, its representatives or Affiliates or in any other form in connection with the Transactions, the Merger or the other transactions contemplated by this Agreement, unless and to the extent any such information is expressly included in a representation or warranty contained in this Article IV.

(b) TruGreen Holdings acknowledges that it has made its own assessment of the present condition and the future prospects of the SLS Entities and the SLS Business and is

sufficiently experienced to make an informed judgment with respect thereto. Except with respect to the representations and warranties of Scotts expressly set forth in Article III, there is and has been no reliance by TruGreen Holdings, its Affiliates or any of their respective representatives on any express or implied representation or warranty of Scotts in connection with the transactions contemplated hereby or by the Merger Agreement.

4.25 <u>TruGreen Franchises</u>.

(a) Neither TruGreen Holdings nor any of its Subsidiaries, other than the TruGreen Entities, own or operate any franchise system. Other than the TruGreen Franchises, the TruGreen Entities have not owned or operated any franchise system.

(b) No TruGreen Franchisee or other Person has any enforceable right of first refusal, option or other right or arrangement to sign any TruGreen Franchise Agreement or acquire any TruGreen Franchise.

(c) All funds administered by or paid to the TruGreen Entities on behalf of one or more TruGreen Franchisees, including funds that TruGreen Franchisees contributed for advertising and promotion (i) have been administered and used in accordance with all Laws, all descriptions in the FDDs, and all TruGreen Franchise Agreements, and (ii) the TruGreen Entities have in all material respects properly accounted for all payments made by each TruGreen Franchisee with respect to any such fund or cooperative. There are no loans owed to or owing from any such funds or cooperatives. To the Knowledge of TruGreen Holdings, there are no claims that any of the expenditures from any such funds or cooperatives have been improperly collected, accounted for, maintained, used or applied. <u>TruGreen Schedule 4.25(c)</u> contains a listing of all established advertising cooperatives regarding the TruGreen Franchises, which TruGreen Schedule also identifies whether each such cooperative is presently operating.

(d) Since October 1, 2013, no "Financial Performance Representation" (as defined in the FTC Rule) has ever been made to any TruGreen Franchisee or prospective franchisee by the TruGreen Entities, or an officer, general partner, limited partner, salesperson, or representation of the TruGreen Entities, except Financial Performance Representations which have been incorporated properly into the applicable FDD.

(e) The TruGreen Entities are, and since October 1, 2013 have been, in compliance with all Franchise Laws and have not offered or sold any Franchise in violation of any Franchise Law (including any violation for failure to file on a timely basis all required amendments and renewals of the registrations and exemptions under the Franchise Laws and any violation for failure to file franchise advertisements). With respect to the relations of the TruGreen Entities with existing and former TruGreen Franchisees, and all terminations, non-renewals, and transfers of TruGreen Franchises since October 1, 2013, the TruGreen Entities have complied with all the proper notice of default, time to cure, and the actual termination of any TruGreen Franchise Agreement required by any Franchise Law.

(f) <u>TruGreen Schedule 4.25(f)</u> contains a list of all independent sales agents, sales brokers and referral sources used by any TruGreen Entity since October 1, 2013, to assist with the sale of Franchises and a list of all written and oral agreements or arrangements with such sales agents, sales brokers and referral sources currently in effect under which any TruGreen Entity has authorized such Person to sell or promote the offer or sale of Franchises. No TruGreen Entity is in default under any such agreement or arrangement. None of such sales agents, sales brokers and referral sources provided information to prospective franchisees that materially differed from the information contained in the FDDs. All FDDs issued by any TruGreen Entity since October 1, 2013, accurately disclose the necessary information pertaining to such independent sales agents, sales brokers and referral sources.

(g) The operations manuals of the TruGreen Entities do not impose any obligations or set forth any requirements that are materially inconsistent with any of the TruGreen Franchise Agreements or the FDDs that they have previously used or currently use, to offer or sell Franchises.

(h) To the Knowledge of TruGreen Holdings, no franchise association or organization is acting as a representative of any group of five or more TruGreen Franchisees. Any franchise council or advisory group (whether independently formed or sponsored by any TruGreen Entity) presently in place is advisory in nature.

(i) The Transactions will not require the consent or approval by any TruGreen Franchisee, council, association, or other third party.

ARTICLE V

CERTAIN COVENANTS

5.1 <u>Conduct of the SLS Business Prior to the Closing</u>.

(a) Except as may be required by applicable Law, as expressly permitted or contemplated by this Agreement, as set forth on <u>Scotts Schedule 5.1(b)</u> or with respect to such further matters as may be consented to by TruGreen Holdings in writing, such consent not to be unreasonably withheld, conditioned or delayed, between the date hereof and the earlier of the Closing and the date, if any, that this Agreement is terminated pursuant to <u>Section 10.1</u>, Scotts shall and shall cause the Non-SLS Subsidiaries to, in each case solely to the extent relating to the SLS Business, and shall cause the SLS Entities to, conduct their business in the ordinary course of business, consistent with past practices; <u>provided</u>, <u>however</u>, that no action or omission by Scotts, any of the Non-SLS Subsidiaries or the SLS Entities with respect to matters specifically addressed by any provision of <u>Section 5.1(b)</u> shall constitute a breach of this <u>Section 5.1(a)</u> unless such action would constitute a breach of such specific provision.

(b) Without limiting the generality of Section 5.1(a) (other than pursuant to the proviso thereto), except (w) as expressly permitted or contemplated by this Agreement (including any agreements related to, and the consummation of, the Scotts Internal Reorganization), (x) with respect to Scotts and the Non-SLS Subsidiaries to the extent not relating to the SLS Business, (y) as required by applicable Law or (z) as set forth on Scotts Schedule 5.1(b), between the date hereof and the earlier of the Closing and the date, if any, this Agreement is terminated pursuant to Section 10.1, Scotts shall not, and shall not permit its Subsidiaries (including the SLS Entities) to, do any of the following unless the prior written consent of TruGreen Holdings (which shall not be unreasonably withheld, delayed or conditioned) has been obtained:

(1) amend or otherwise modify the respective articles of incorporation or bylaws (or similar organizational documents) of any SLS Entity;

(2) incur, guaranty or assume any Indebtedness or debt obligation other than obligations under Contracts incurred in the ordinary course of business consistent with past practices;

(3) cancel, forgive, pay, discharge, satisfy, compromise or settle any material debt, claim or obligation of the SLS Business other than (i) in the ordinary course of business or as required pursuant to any Contract in effect as of the date hereof or that is permitted to be entered into after the date hereof in accordance with this <u>Section 5.1</u> or (ii) solely involving the payment of money or monetary damages by Scotts or its Subsidiaries prior to the Closing;

(4) create or otherwise incur any Encumbrance on any material SLS Asset other than Permitted Encumbrances and Encumbrances that will be released in full prior to the Closing;

(5) acquire (<u>A</u>) by merging or consolidating with, or by purchasing all or a substantial portion of the assets, properties or rights of, or by any other manner, any business of any Person or any division thereof or (<u>B</u>) any capital stock, assets, properties or rights of any other Person, in each case, other than those acquired in the ordinary course of business consistent with past practices;

(6) except (<u>A</u>) as required pursuant to the terms of any SLS Business Benefit Plan or by Law or (<u>B</u>) as contractually required, (i) increase annual base salary or hourly base wages of any SLS Employee other than in the ordinary course of business and consistent with past practices for non-executive and non-management-level employees, (<u>ii</u>) grant to any SLS Employee any material increase in compensation or benefits, (iii) grant to any SLS Employee any increase in severance or termination pay, (<u>iv</u>) enter into or amend any employment, consulting, indemnification, severance or termination agreement with any SLS Employee, (<u>v</u>) enter into or amend any collective bargaining agreement with any SLS Employee, (<u>vi</u>) establish, adopt, enter into or amend or terminate any SLS Entity Benefit Plan or, with respect to SLS Employees, any Scotts

Benefit Plan, in a manner that would increase the liability of any SLS Entity, other than broad based changes which cover Scotts employees generally (<u>vii</u>) accelerate the vesting, material payment or funding of any compensation or benefits under any SLS Entity Benefit Plan or, if it would result in liability to a SLS Entity (or, following the Closing, Newco or any of its Subsidiaries), under any Scotts Benefit Plan (<u>viii</u>) promote, hire or terminate (other than for cause) any SLS Employee who will be (after the date of promotion or hire) or is (in case of termination) an SLS Employee, in each case, other than promotion, hire or termination of non-executive and non-management employees (and who will be such employees after promotion or hire) in the ordinary course of business, (<u>ix</u>) loan any funds to any SLS Employee or (<u>x</u>) award or promise to award to any SLS Employee any equity or equity-based award;

(7) make any (<u>i</u>) change in any material accounting principles, methods or practices followed by any SLS Entity or any material change in amortization policies or rates or (<u>ii</u>) revaluation of any material SLS Asset, except in any such case (clause (<u>i</u>) or (<u>ii</u>)), (x) insofar as required solely by reason of preparing the initial audited financial statements for the SLS Business on a standalone basis or (y) as required by applicable Law or GAAP, as then in effect;

(8) (<u>A</u>) sell, lease, encumber, license or otherwise dispose of any properties, rights or SLS Assets, except operational assets (including inventory and equipment) in the ordinary course of business or (<u>B</u>) acquire title to any real property;

(9) make any loans, advances or capital contributions to, or investments in, any Person (other than pursuant to any existing obligations under any Contract) other than the advancement of trade credit to customers or expenses to employees;

(10) authorize for issuance, issue, sell, deliver or agree or commit to issue, sell or deliver (whether through the issuance or granting of options, warrants, convertible or exchangeable securities, commitments, subscriptions, rights to purchase or otherwise) any shares of capital stock or any other securities of any SLS Entity, or amend any of the terms of any such securities or make any changes (by combination, reorganization or otherwise) in the equity capital structure of any SLS Entity;

aggregate;

(11) make any commitments for capital expenditures in an amount in excess of \$500,000 in the

(12) make any material payments or grant any material discounts to customers or suppliers of the SLS Business, other than in the ordinary course of business consistent with past practices;

(13) fail to pay or satisfy when due any material account payable or other material liability of the SLS Business in the ordinary course of business

consistent with past practices (including, in each case, the timing of any such payments), other than any such liability that is being contested in good faith by the SLS Business;

(14) amend, cancel, compromise or waive any material claim or right of the SLS Business, other than in the ordinary course of business;

(15) (<u>A</u>) enter into, transfer, terminate, modify or amend (other than extensions at the end of a term in the ordinary course of business) any SLS Material Contract or other Contract that, if in effect on the date hereof, would have been a SLS Material Contract or (<u>B</u>) waive any right of material value under any SLS Material Contract or other Contract that, if in effect on the date hereof, would have been a SLS Material Contract;

(16) make or change any Tax election, change any annual Tax accounting period, adopt or change any method of Tax accounting, amend any Tax Returns or file any claims for Tax refunds, enter into any closing agreement, settle any Tax claim, audit or assessment or surrender any right to claim a Tax refund, offset or other reduction in Tax liability, in each case if any such act would have a material and adverse effect on Newco or TruGreen Holdings following the Closing Date;

(17) fail to maintain insurance consistent with past practice for the benefit of the SLS Entities, the SLS Business or the SLS Assets; or

(18) authorize or announce an intention to do any of the foregoing, or enter into any Contract or arrangement to do any of the foregoing or any other action, which would reasonably be expected to prevent or materially delay or materially impede the consummation of the Closing or the other transactions contemplated by this Agreement.

Nothing contained in this Agreement shall give TruGreen Holdings, directly or indirectly, the right to control or direct the business of Scotts and its Subsidiaries (including the SLS Business) prior to the Closing. Prior to the Closing, Scotts shall, directly or indirectly, exercise and be responsible for, consistent with the terms and conditions of this Agreement and applicable Law (including the HSR Act), complete control and supervision over its business (including the SLS Business).

5.2 <u>Conduct of the TruGreen Entities Prior to the Closing</u>.

(a) Except as may be required by applicable Law, as expressly permitted or contemplated by this Agreement, as set forth on <u>TruGreen Schedule 5.2(b)</u> or with respect to such further matters as may be consented to by Scotts in writing, such consent not to be unreasonably withheld, conditioned or delayed, between the date hereof and the earlier of the Closing, and the date, if any, that this Agreement is terminated pursuant to <u>Section 10.1</u>, TruGreen Holdings shall cause each of the TruGreen Entities to conduct its business in the ordinary course of business, consistent with past practices; <u>provided</u>, <u>however</u>, that no action or

omission by TruGreen Holdings or any of the TruGreen Entities with respect to matters specifically addressed by any provision of <u>Section 5.2(b)</u> shall constitute a breach of this <u>Section 5.2(a)</u> unless such action or omission would constitute a breach of such specific provision.

(b) Without limiting the generality of <u>Section 5.2(a)</u> (other than pursuant to the proviso thereto), except (x) as expressly permitted or contemplated by this Agreement (including any agreements related to, and the completion of, the Newco Restructuring) or as expressly required by the Merger Agreement as in effect on the date hereof, (y) as required by applicable Law or (z) as set forth on <u>TruGreen Schedule 5.2(b)</u>, between the date hereof and the earlier of the Closing, and the date that this Agreement is terminated pursuant to <u>Section 10.1</u> (if any), TruGreen Holdings shall not, and shall not permit its Subsidiaries (including the TruGreen Entities) to, do any of the following unless the prior written consent of Scotts (which shall not be unreasonably withheld, delayed or conditioned) has been obtained:

(1) amend or otherwise modify the respective articles of incorporation or bylaws (or similar organizational documents) of any TruGreen Entity;

(2) incur, guaranty or assume any Indebtedness or debt obligation other than obligations under Contracts incurred in the ordinary course of business consistent with past practices;

(3) cancel, forgive, pay, discharge, satisfy, compromise or settle any material debt, claim or obligation other than (i) in the ordinary course of business or as required pursuant to any Contract in effect as of the date hereof or that is permitted to be entered into after the date hereof in accordance with this <u>Section 5.2</u> or (ii) solely involving the payment of money or monetary damages by TruGreen Holdings or its Subsidiaries prior to the Closing;

(4) create or otherwise incur any Encumbrance on any material asset other than Permitted Encumbrances and Encumbrances that will be released in full prior to the Closing;

(5) acquire (<u>A</u>) by merging or consolidating with, or by purchasing all or a substantial portion of the assets, properties or rights of, or by any other manner, any business of any Person or any division thereof or (<u>B</u>) any capital stock, assets, properties or rights of any other Person, in each case, other than those acquired in the ordinary course of business consistent with past practices;

(6) except (<u>A</u>) as required pursuant to the terms of any TruGreen Benefit Plan or by Law or (<u>B</u>) as contractually required (i) increase in annual base salary or hourly base wages of any TruGreen Employee other than in the ordinary course of business and consistent with past practices for non-executive and non-management-level employees, (<u>ii</u>) grant to any TruGreen Employee any material increase in compensation or benefits, (iii) grant to any TruGreen Employee any increase in severance or termination pay, (<u>iv</u>) enter into or amend any employment, consulting,

indemnification, severance or termination agreement with any TruGreen Employee, (\underline{v}) enter into or amend any collective bargaining agreement with any TruGreen Employee, (\underline{vi}) establish, adopt, enter into or amend or terminate any TruGreen Benefit Plan in a manner that would increase the liability of any of the TruGreen Entities, other than broad based changes which cover TruGreen Employees generally, (\underline{vii}) accelerate the vesting, material payment or funding of any compensation or benefits under any TruGreen Benefit Plan, (\underline{vii}) promote, hire or terminate (other than for cause) any TruGreen Employee, in each case, other than promotion, hire or termination of non-executive and non-management employees (and who will be employees after promotion or hire) in the ordinary course of business, (\underline{ix}) loan any funds to any TruGreen Employee or (\underline{x}) award or promise to award to any TruGreen Employee any equity or equity-based award;

(7) make any (<u>A</u>) change in any material accounting principles, methods or practices, or any material change in amortization policies or rates, or (<u>B</u>) revaluation of any material asset, in any such case in clause (<u>A</u>) or (<u>B</u>), other than as required by applicable Law or GAAP, as then in effect;

(8) (<u>A</u>) sell, lease, encumber, license or otherwise dispose of any properties, rights or assets, except operational assets (including inventory and equipment) in the ordinary course of business or (<u>B</u>) acquire title to any real property;

(9) make any loans, advances or capital contributions to, or investments in, any Person (other than pursuant to any existing obligations under any Contract) other than the advancement of trade credit to customers or expenses to employees;

(10) authorize for issuance, issue, sell, deliver or agree or commit to issue, sell or deliver (whether through the issuance or granting of options, warrants, convertible or exchangeable securities, commitments, subscriptions, rights to purchase or otherwise) any shares of capital stock or any other securities, or amend any of the terms of any such securities or make any changes (by combination, reorganization or otherwise) in its equity capital structure;

(11) make any commitments for capital expenditures in an amount in excess of \$500,000 in the

aggregate;

(12) make any material payments or grant any material discounts to its customers or suppliers, other than in the ordinary course of business consistent with past practices;

(13) fail to pay or satisfy when due any material account payable or other material liability in the ordinary course of business consistent with past

practices (including, in each case, the timing of any such payments), other than any such liability that is being contested in good faith;

(14) amend, cancel, compromise or waive any material claim or right, other than in the ordinary

course of business;

(15) (<u>A</u>) enter into, transfer, terminate, modify or amend (other than extensions at the end of a term in the ordinary course of business) any TruGreen Material Contract or other Contract that, if in effect on the date hereof, would have been a TruGreen Material Contract, (<u>B</u>) waive any right of material value under any TruGreen Material Contract or other Contract that, if in effect on the date hereof, would have been a TruGreen Material Contract, or (C) enter into, join or become a party to, the Merger Agreement (except this clause (C) shall not apply to TruGreen Holdings);

(16) make or change any Tax election, change any annual Tax accounting period, adopt or change any method of Tax accounting, amend any Tax Returns or file any claims for Tax refunds, enter into any closing agreement, settle any Tax claim, audit or assessment or surrender any right to claim a Tax refund, offset or other reduction in Tax liability, in each case if any such act would have a material and adverse effect on Newco or Scotts following the Closing Date;

(17) fail to maintain insurance consistent with past practice; or

(18) authorize or announce an intention to do any of the foregoing, or enter into any Contract or arrangement to do any of the foregoing or any other action, which would reasonably be expected to prevent or materially delay or materially impede the consummation of the Closing or the other transactions contemplated by this Agreement.

Nothing contained in this Agreement shall give Scotts, directly or indirectly, the right to control or direct the business of TruGreen Holdings and its Subsidiaries prior to the Closing. Prior to the Closing, TruGreen Holdings shall, directly or indirectly, exercise and be responsible for, consistent with the terms and conditions of this Agreement and applicable Law (including the HSR Act), complete control and supervision over its business.

5.3 Efforts; Third-Party Approvals.

(a) On the terms and subject to the conditions of this Agreement, each party shall use its reasonable best efforts to cause the Closing to occur, including using reasonable best efforts to cause the conditions to the Closing in such party's control set forth in <u>Article VII</u> to be satisfied (provided that (i) without limitation, nothing herein shall create any obligation on the part of Scotts or any of its Affiliates with respect to the conditions set forth in 7.1(a)-(c), (g), (i)-(k) and (ii) Scotts's obligations with respect to matters covered by 7.2(h) are solely set forth in the applicable provisions of Section 5.11) and taking all actions reasonably necessary to comply promptly with all legal requirements that may be imposed on it or any of its Subsidiaries with

respect to the Closing and using its reasonable best efforts to cause the conditions to the closing of the transactions contemplated by the Merger Agreement to be satisfied; <u>provided</u> that nothing in this Agreement or any Ancillary Agreement shall be construed as requiring (i) any Person to pay or commit to pay any amount to (or incur any extraordinary or unreasonable obligation in favor of) any Person from whom any Consent or assignment may be required (other than nominal filing or application fees) or to agree to any material amendment to any Contract, (<u>ii</u>) either Party to waive compliance with any condition, covenant or agreement set forth in this Agreement or any Ancillary Agreement or (<u>iii</u>) TruGreen Holdings or any of its Affiliates to waive compliance with any condition, covenant or agreement set forth in the Merger Agreement as in effect on the date hereof. TruGreen Holdings shall not agree to any amendment to the terms of the Merger Agreement in any manner that would reasonably be expected to materially delay or prevent the Closing.

(b) Each Party shall, as promptly as practicable, and in any event within 10 Business Days of the Closing (i) make, or cause to be made, all filings and submissions (including those under the HSR Act or any other Competition Laws) required under any Law applicable to such party or any of its Affiliates, and give such reasonable undertakings as may be required in connection therewith and (ii) use reasonable best efforts to obtain, or cause to be obtained, all consents, authorizations and approvals from all Governmental Bodies necessary to be obtained by such Party or any of its Affiliates, in each case in connection with this Agreement or the Ancillary Agreements or the consummation of the transactions contemplated hereby or thereby, provided that (A) Scotts and its Subsidiaries shall not give any undertakings, make any commitments or enter into any agreements that would be binding upon Newco or any of its Subsidiaries, including, after the Closing, the SLS Entities, without the prior written consent of TruGreen Holdings, except as would not be materially adverse to Newco and its Subsidiaries, including, after the Closing, the SLS Entities, taken as a whole, (B) TruGreen Holdings and its Subsidiaries shall not give any undertakings, make any commitments or enter into any agreements that would be binding upon Newco or any of its Subsidiaries, including, after the Closing, the SLS Entities, without the prior written consent of Scotts, except as would not be materially adverse to Newco and its Subsidiaries, including, after the Closing, the SLS Entities, taken as a whole, (C) TruGreen Holdings and its Subsidiaries shall not give any undertakings, make any commitments or enter into any agreements that would be binding upon Scotts or any of its Subsidiaries (other than those binding on the SLS Entities after the Closing that are otherwise permitted pursuant to the foregoing clause (B)) without the prior written consent of Scotts, (D) neither Scotts nor any of its Affiliates shall be required to take or not take any action related to any existing business of Scotts other than the SLS Business (provided, further, that the foregoing exception shall not apply to the payment of amounts by Scotts or any Affiliate that are included in reimbursable Scotts Transaction Expenses, unless it could reasonably be expected to impair the overall benefit to be realized from the consummation of the transactions contemplated by this Agreement and the Ancillary Agreements), and (E) neither Party nor any of its Affiliates shall be required to take any action that involves divestiture of an existing business of such Party or any of its Affiliates, that involves unreasonable expense or that could reasonably be expected to impair the overall benefit expected to be realized from the consummation of the transactions contemplated by this Agreement and the Ancillary Agreements. Each Party shall keep the other apprised, to the extent permitted by applicable Laws, of the status of any communications with,

and any inquiries or request for additional information from, Governmental Bodies pursuant to the HSR Act or any other Competition Laws, and each party hereto shall use its reasonable best efforts to defend against any lawsuit, action or proceeding, judicial or administrative, challenging this Agreement or the transactions contemplated hereby.

(c) In connection with the matters contemplated by this Agreement, each of Scotts and TruGreen Holdings shall, and shall cause their respective Subsidiaries to, use commercially reasonable efforts prior to the Closing Date to give all notices to third parties and to obtain all third-party Consents and assignments in connection with the transactions contemplated hereby and by the Ancillary Agreements, the Debt Financing Commitment (in the case of Scotts) and the Merger Agreement (in the case of TruGreen Holdings); provided, however, that neither Scotts nor TruGreen Holdings nor their respective Subsidiaries shall be required to pay or commit to pay any amount to (or incur any extraordinary or unreasonable obligation in favor of) any Person from whom any such Consent or assignment may be required (other than nominal filing or application fees) or agree to any material amendment, supplement or waiver to any Contract. To the extent that Scotts or TruGreen Holdings is unable to obtain any required third-party Consents prior to the Closing despite such efforts, each of Scotts, TruGreen Holdings and each of their respective Subsidiaries shall use its respective commercially reasonable efforts to obtain (or cause to be obtained), as promptly as practicable following the Closing, all such Consents; provided, however, that no Person shall be required to pay or commit to pay any amount to (or incur any extraordinary or unreasonable obligation in favor of) any Person from whom any such Consent or assignment may be required (other than nominal filing or application fees) or agree to any material amendment, supplement or waiver to any Contract. With respect to any SLS Material Contract (other than a Shared Contract) or any TruGreen Material Contract requiring a third-party Consent that has not been obtained prior to the Closing, Scotts, the Non-SLS Subsidiaries, TruGreen Holdings, the TruGreen Entities and the SLS Entities, as applicable shall use their commercially reasonable efforts to cooperate with each other in any lawful and economically feasible arrangement (including by license) to provide the TruGreen Entities and the SLS Entities with the benefits under such SLS Material Contract or TruGreen Material Contract, as applicable; provided that Newco and its Subsidiaries shall undertake to pay or satisfy the corresponding liabilities for the enjoyment of such benefit to the extent the TruGreen Entities and the SLS Entities would have been responsible therefor if such Consent had been obtained prior to the Closing.

(d) Each party hereto shall keep the other reasonably apprised of the status of matters relating to completion of the transactions contemplated by this Agreement, the Ancillary Agreements and the Merger Agreement, including promptly furnishing the other with copies of (i) written notices or other written communications received by TruGreen Holdings, on the one hand, or Scotts, on the other hand, as the case may be, or any of its representatives, from any Governmental Body with respect to the transactions contemplated by this Agreement, any Ancillary Agreement or the Merger Agreement or from any Person alleging that the Consent of such Person is or may be required in connection with the transactions contemplated by this Agreement, any Ancillary Agreement or related to the Merger.

(e) From and after the date of this Agreement until the earlier of the Closing Date and termination of this Agreement in accordance with <u>Section 10.1</u>, Scotts shall give prompt notice to TruGreen Holdings, and TruGreen Holdings shall give prompt notice to Scotts, of (<u>i</u>) any representation or warranty made by it contained in this Agreement, or in the case of TruGreen Holdings, the Merger Agreement, becoming untrue or inaccurate and (<u>ii</u>) the failure by it to comply with or satisfy in any material respect any covenant or agreement to be complied with or satisfied by it under this Agreement, or, in the case of TruGreen Holdings, the Merger Agreement, in the case of each of clauses (i) and (ii), which could reasonably be expected to result in a failure to satisfy a condition to the other Party's obligation to effect the Closing or, in the case of TruGreen Holdings, a failure to satisfy a condition to the other Party to the Merger Agreement to consummate the transactions contemplated by the Merger Agreement.

5.4 Access to Information; Other Contacts.

(a) From and after the date of this Agreement until the earlier of the Closing Date or termination of this Agreement in accordance with Section 10.1, (i) Scotts shall allow TruGreen Holdings and its accountants, counsel and other representatives, at TruGreen Holdings's expense, to make such reasonable investigation of the business, operations and properties of the SLS Business and the SLS Entities as reasonably necessary in connection with the transactions contemplated by this Agreement and (ii) TruGreen Holdings shall allow Scotts and its accountants, counsel and other representatives, at Scotts's expense, to make such reasonable investigation of the business, operations and properties of the TruGreen Entities as reasonably necessary in connection with the transactions contemplated by this Agreement. Such investigation shall include reasonable access to the respective directors. officers, employees, agents and representatives (including legal counsel and independent accountants) of the SLS Entities or TruGreen Entities, as applicable, and Scotts and the Non-SLS Subsidiaries to the extent related to the SLS Business, and the properties and books and records of the TruGreen Entities or SLS Entities, as applicable, and Scotts and the Non-SLS Subsidiaries to the extent related to the SLS Business. The SLS Entities, Scotts and the Non-SLS Subsidiaries, shall furnish TruGreen Holdings and its representatives with such financial, operating and other data and information and copies of documents with respect to the SLS Entities, or Scotts or the Non-SLS Subsidiaries to the extent related to the SLS Business, or any of the transactions contemplated by this Agreement as TruGreen Holdings shall from time to time reasonably request. TruGreen Holdings and its Subsidiaries, shall furnish Scotts and its representatives with such financial, operating and other data and information and copies of documents with respect to the TruGreen Entities, or any of the transactions contemplated by this Agreement as Scotts shall from time to time reasonably request. All access and investigation pursuant to this Section 5.4 shall occur only upon reasonable notice during normal business hours. Notwithstanding anything to the contrary in this Agreement, neither Party nor any of its Affiliates shall be required to disclose any information to the other Party if such disclosure would reasonably be expected to (i) jeopardize any attorney-client or other legal privilege or (ii) contravene any applicable Laws, fiduciary duty or Contract entered into prior to the date hereof; provided that at the request of a Party, the other Party shall use commercially reasonable efforts to cause such disclosure to be permitted or to develop an alternative to providing such information reasonably acceptable to the requesting Party.

(b) During the period between the date hereof and the Closing, (i) TruGreen Holdings and its representatives may contact or communicate with the customers, suppliers and licensors of the SLS Business in connection with the transactions contemplated hereby only with the prior written consent of Scotts, which shall not be unreasonably withheld but may be conditioned upon a designee of Scotts being present at any meeting or conference, and (ii) Scotts and its representatives may contact or communicate with the customers, suppliers and licensors of the TruGreen Entities in connection with the transactions contemplated hereby only with the prior written consent of TruGreen Holdings, which shall not be unreasonably withheld but may be conditioned upon a designee of TruGreen Holdings being present at any meeting or conference, in the case of each of clauses (i) and (ii), subject to applicable Law. For the avoidance of doubt, nothing in this <u>Section 5.4</u> shall prohibit either Party or its Affiliates from contacting the customers, suppliers and licensors of the SLS Business or the TruGreen Entities, as applicable, in the ordinary course of the businesses of such Party or its Affiliates for the purpose of selling its or such Affiliates' products or for any other purpose unrelated to the transactions contemplated by this Agreement and the Ancillary Agreements shall not be discussed in any such contacts. Notwithstanding anything to the contrary in this <u>Section 5.4(b)</u>, all such contacts and communications shall be subject to the restrictions set forth in <u>Section 5.6</u> and in the Confidentiality Agreement.

5.5 Books and Records; Access; Assistance.

(a) Subject to <u>Section 6.5</u>, for a period of six years after the Closing Date, Scotts and TruGreen Holdings shall preserve and retain (or cause to be preserved and retained) all accounting, legal, auditing and other books and records (including any documents relating to any Proceedings arising out of or with respect to the operation or conduct of the SLS Business or the TruGreen Entities, as applicable, and those described on <u>Scotts Schedule 5.5(a)</u>) over which it has control to the extent (i) relating to the conduct of the TruGreen Entities or (ii) primarily relating to the conduct of the SLS Business, as applicable, prior to the Closing Date. Notwithstanding the foregoing, during such six-year period, a Party may dispose of any such books and records that are offered to, but not accepted by, the other Party and Newco.

(b) From and after the Closing, if, in order to properly prepare documents required to be filed with Governmental Bodies (including Tax Authorities) or its financial statements, or for any other reasonable business purpose (including in connection with any securities offering or the management, investigation, defense, prosecution and disposition of any matters and/or Proceedings described on <u>Scotts Schedule 9.2(d)</u>, including any counterclaims with respect thereto, and any Liabilities or contingent Liabilities with respect thereto (whether now or in the future existing, the "<u>Specified Matters</u>")), it is necessary that Scotts (or its Affiliates or its or their successors), on the one hand, or TruGreen Holdings (or its Affiliates or its or their successors), on the other hand, be furnished with additional information of the type described in <u>Section 5.5(a)</u> above, and such information is in the possession of the other Party or Newco (or its Subsidiaries), except as would, in such Party's or Newco's (or Newco's Subsidiaries'), as applicable, reasonable discretion (<u>i</u>) jeopardize any attorney-client or other

legal privilege (provided, that (x) the privilege with respect to any matter for which Scotts or the Non-SLS Subsidiaries have retained liability or have agreed to indemnify the TruGreen Indemnified Parties pursuant to Section 9.2(d) shall remain with Scotts to the extent relating solely to such liability as is retained or indemnified by Scotts or the Non-SLS Subsidiaries and (y) if requested by Scotts, TruGreen Holdings and its Subsidiaries (including, after the Closing, the SLS Entities) shall cooperate in good faith to enter into customary joint defense agreements proposed by Scotts to protect any such privilege, including any joint privilege) or (ii) contravene any applicable Law, fiduciary duty or Contract (provided that at the request of a Party, the other Party shall use commercially reasonable efforts, including causing Newco and its Subsidiaries to use commercially reasonable efforts, to cause the furnishing of such information to be permitted or to develop an alternative to providing such information reasonably acceptable to the requesting Party), such other Party agrees to use its commercially reasonable efforts, including causing Newco and its Subsidiaries to use commercially reasonable efforts, to furnish such information to the Party that requires such information and provide reasonable access to any employees or other relevant personnel that may be reasonably required to be consulted with in connection with any such purpose, in each case, at the cost and expense of the party being furnished such information or access to such employees or personnel, as applicable. After the Closing Date, except in the case of an adversarial Proceeding by Scotts and/or its Affiliates, on the one hand, against TruGreen Holdings and its Affiliates on the other hand, or vice versa, each Party hereto shall use its commercially reasonable efforts to make available to the other Party, upon written request, its and its Affiliates' former, current and future directors, officers, employees, other personnel and agents to the extent that any such person (giving consideration to business demands of such directors, officers, employees, other personnel and agents) may reasonably be required (including, if applicable, by serving as a witness) in connection with the management, investigation, defense, prosecution and disposition of any Specified Matter or related Proceeding in which the requesting party may from time to time be involved, regardless of whether such Specified Matter or Proceeding is a matter with respect to which indemnification may be sought hereunder. The requesting party shall bear all costs and expenses in connection therewith.

(c) Nothing in this <u>Section 5.5</u> shall be deemed to modify, diminish or otherwise limit any information rights to which a party is entitled pursuant to the LLC Agreement.

5.6 Confidentiality.

(a) Each Party acknowledges that the information being provided to it in connection with the transactions contemplated hereby is subject to the terms of the Confidentiality Agreement, the terms of which are incorporated herein by reference. Effective upon the Closing, the obligations under the Confidentiality Agreement shall terminate, except with respect to provisions regarding disclosure and use of confidential information not related to the SLS Business, the SLS Entities or the TruGreen Entities, which shall continue in accordance with the terms of the Confidentiality Agreement. If for any reason this Agreement is terminated prior to the Closing Date, the Confidentiality Agreement shall nonetheless continue in full force and effect in accordance with its terms.

(b) Except as required by applicable Law, TruGreen Holdings shall, and shall cause the TruGreen Entities and the SLS Entities after the Closing to, to the extent any of them has any nonpublic information relating to Scotts and its Subsidiaries, but excluding information relating to either the SLS Business or the SLS Entities (any such information, "Scotts Information"), (i) treat the Scotts Information strictly confidentially and (<u>ii</u>) to the extent the Scotts Information is known to be documented or exist in written, photographic or other physical form, return to Scotts or destroy such information (and any copies made thereof) upon Scotts's request.

5.7 Employees.

(a) <u>Scotts Schedule 5.7(a)</u> lists (i) each SLS Employee as of the date hereof, including, to the extent permitted by applicable Law, such SLS Employee's name, title, work location, employer, identification number and leave status and (<u>ii</u>) each Retained Scotts Employee as of the date hereof, including to the extent permitted by applicable Law, such Retained Scotts Employee's name, title, work location and employer. <u>Scotts Schedule 5.7(a)</u> may be amended by Scotts with the prior written consent of TruGreen Holdings (which shall not be unreasonably withheld) at any time prior to Closing to (x) add any other employee as SLS Employee who primarily supports the SLS Business (in which event, if applicable, such employee will be removed from being a Retained Scotts Employee) or becomes a new employee of any SLS Entity in compliance with <u>Section 5.1</u> or (y) remove any employee from being an SLS Employee) or whose employment is terminated.

(b) Except as otherwise expressly set forth herein or the Employee Leasing Agreement, from and after the Closing Date, Scotts and its Subsidiaries shall retain all Liabilities under the Scotts Benefit Plans, and, from and after the ELA Expiration Date, Newco and its Affiliates (including the SLS Entities) shall assume or retain, as applicable, all Liabilities under the SLS Entity Benefit Plans. Effective as of the ELA Expiration Date (or, with respect to Inactive Employees, the date they become Transferred SLS Employees), the Transferred SLS Employees shall cease active participation in, and receipt of any additional accruals under, the Scotts Benefit Plans. Without limiting the scope of this <u>Section 5.7</u>, subject to the terms of the Employee Leasing Agreement (including the requirement that Newco will reimburse Scotts and its Affiliates for the cost of any such liabilities incurred during the Leasing Term, as such term is defined in the Employee Leasing Agreement) Scotts and its Subsidiaries shall retain and be responsible for: (i) medical, prescription drug, dental, vision, workers' compensation, short- and long-term disability, life insurance, accidental death and dismemberment and other welfare benefit claims that are incurred by Transferred SLS Employees and their beneficiaries and dependents prior to the ELA Expiration Date (or, with respect to Inactive Employees, the date they become Transferred SLS Employees); and (ii) claims relating to COBRA coverage attributable to "qualifying events" occurring prior to the ELA Expiration Date (or, with respect to the terms of the Employees) with respect to any former SLS Employees and their beneficiaries and dependents. Subject to the terms of the Employee Leasing Agreement, Newco and its Affiliates (including the SLS Entities) shall

assume and be responsible for: (y) medical, prescription drug, dental, vision, workers' compensation, short- and long-term disability, life insurance, accidental death and dismemberment and other welfare benefit claims that are incurred by Transferred SLS Employees on or after the ELA Expiration Date (or, with respect to Inactive Employees, the date they become Transferred SLS Employees); and (z) claims relating to COBRA coverage attributable to "qualifying events" occurring on or after the ELA Expiration Date (or, with respect to Inactive Employees) with respect to any Transferred SLS Employees and their beneficiaries and dependents. For purposes of this <u>Section 5.7(b)</u>, (I) medical, dental, prescription drug and vision claims shall be considered incurred when the treatment for a given condition or medical supplies are provided, and not when the condition arose, (II) claims for workers' compensation and short- and long-term disability benefits shall be incurred upon the date of the event, occurrence or circumstance giving rise to the claim for workers' compensation, (III) claims of life insurance shall be incurred upon death and (IV) claims for accidental death and dismemberment benefits shall be incurred when the accident occurred. Notwithstanding anything to the contrary in this Agreement, the provisions of this <u>Section 5.7(b)</u> shall in all events be subject to the provisions of the Employee Leasing Agreement.

(c) Prior to the Closing, (i) the employment of each SLS Employee who is employed by an SLS Entity shall (unless such Person has ceased to be employed by Scotts or any Non-SLS Subsidiary prior thereto) be transferred to Scotts or a Non-SLS Subsidiary, and each such employee shall cease to be an employee of, an SLS Entity, without any interruption or cessation of employment, terms and conditions of employment or break in service and (ii) the employment of each Retained Scotts Employee who is employed by an SLS Entity as of the date of this Agreement shall (unless such Person has ceased to be employed by an SLS Entity prior thereto) be transferred from, and each such employee shall cease to be an employee of, an SLS Entity, without any interruption or cessation of employment, terms and conditions of employment or break in service. Retained Scotts Employees and all other individuals employed by Scotts and its Subsidiaries at the Closing shall remain employees of Scotts or its Subsidiaries immediately following the Closing. Prior to the Closing, the Parties will negotiate in good faith and enter into a sublease agreement in respect of the building identified in Item 1 of Scotts Schedule 3.12(b) and into an Employee Leasing Agreement ("Employee Leasing Agreement") having the terms described in the ELA Term Sheet attached hereto as Scotts Schedule 5.7(c). From the Closing until the expiration date (the "ELA Expiration Date") of the Employee Leasing Agreement, SLS Employees will provide services to Newco pursuant to the terms of the Employee Leasing Agreement. Newco shall, or shall cause TruGreen LP and its Affiliates to offer employment to each SLS Employee (including all full-time and part-time SLS Employees and SLS Employees who are on leave under the Family and Medical Leave Act or any similar U.S. state law, on vacation or who have taken a personal day or occasional absence day as currently defined in Scotts's policies and all SLS Employees who are on an approved leave of absence, or temporary layoff during the off-peak season of the SLS Business, whether paid or unpaid (but excluding any Inactive Employees who have not returned to active employment status (including with any reasonable accommodation) as of the ELA Expiration Date)) in accordance with the provisions of this Section 5.7 and the Employee Leasing Agreement, with such employment to be effective as of the first day following the ELA Expiration Date (any SLS

Employee who accepts Newco's or its Affiliates' offer of employment, a "<u>Transferred SLS Employee</u>"). Notwithstanding the foregoing, prior to the ELA Expiration Date, the Parties may agree in good faith to use an alternative method for SLS Employees to become employees of Newco and its Affiliates (including, if SLS Employees are then all employed by one entity, for such entity to be transferred or contributed to Newco under the terms consistent with the intent of the terms and provisions of this Agreement). Scotts shall bear and be responsible for all Liabilities with respect to Inactive Employees arising during the period of time such Inactive Employees are employed by Scotts or a Non-SLS Subsidiary following the Closing. If any such Inactive Employee returns to active employment (including with any reasonable accommodation) prior to the ELA Expiration Date, such Inactive Employee shall cease being an Inactive Employee. Any Inactive Employee who returns to active status (including with any reasonable accommodation) prior to the ELA Expiration Date and the one-year anniversary of the Closing Date, at which time any such individual shall be offered continued employment with Newco or any of its Subsidiaries (including the SLS Entities). If and when the Inactive Employee returns to active status and commences employment with Newco or any of its Subsidiaries, such Inactive Employee shall be considered a Transferred SLS Employee and the Inactive Employee shall become eligible for coverage and benefits under all employee benefit plans or programs maintained by Newco or any of its Subsidiaries under the same terms and conditions that apply to other Transferred SLS Employees.

(d) Unless the Parties agree otherwise in writing and subject to the Employee Leasing Agreement, as of the Closing Date and until December 31, 2016, Newco shall, or shall cause TruGreen LP and its Affiliates to, provide Transferred SLS Employees with (A) compensation (including salary and short-term target incentive bonus opportunities) that are no less favorable in the aggregate than the compensation provided to each Transferred SLS Employee immediately prior to the Closing Date and (B) without duplication of items described in clause (A), employee benefits (but excluding equity-based incentives, severance benefits, retention bonuses and other change in control entitlements, and retirement compensation benefits under a defined benefit plan) that are substantially comparable in the aggregate to the employee benefits (subject to the same exclusions) provided to each Transferred SLS Employee immediately prior to the Closing Date; provided, however, that a change in benefits by Scotts or any Non-SLS Entity (which was not directed or requested by Newco, any of its Affiliates or Leased Employees, pursuant to the terms of the Employee Leasing Agreement) during the period between the Closing and the ELA Expiration Date shall not constitute a breach by Newco or any of its Affiliates of this <u>Section 5.7(d)</u>. Nothing herein shall be deemed to limit the right of Newco, TruGreen LP or any of their respective Affiliates to (x) terminate the employment of any Transferred SLS Employee at any time, (y) change or modify the terms or conditions of employment of any Transferred SLS Employee or (z) change or modify any SLS Entity Benefit Plan, TruGreen Benefit Plan or other employee benefit plan or arrangement in accordance with its terms.

(e) For purposes under the employee benefit plans, programs and arrangements established or maintained by Newco and its respective Affiliates in which Transferred SLS Employees may be eligible to participate after the ELA Expiration Date (the

"New Benefit Plans") (including any severance plans), each Transferred SLS Employee shall be credited with the same amount of service as was credited by the SLS Entities as of the ELA Expiration Date under similar or comparable SLS Business Benefit Plans or Scotts Benefit Plans in which such Transferred SLS Employee participated immediately prior to the ELA Expiration Date (except (x) for purposes of benefit accrual under defined benefit plans and retiree medical arrangements or (y) to the extent such credit would result in a duplication of benefits). In addition, and without limiting the generality of the foregoing, (i) with respect to any New Benefit Plans in which the Transferred SLS Employees may be eligible to participate following the ELA Expiration Date, each Transferred SLS Employee will immediately be eligible to participate in such New Benefit Plans, without any waiting time, to the extent coverage under such New Benefit Plans replaces coverage under a similar or comparable SLS Business Benefit Plan or Scotts Benefit Plan in which such Transferred SLS Employee was eligible to participate immediately before such commencement of participation and (ii) for purposes of each New Benefit Plan providing medical, dental, pharmaceutical and/or vision benefits to any Transferred SLS Employee, Newco shall cause all preexisting condition exclusions and actively-at-work requirements of such New Benefit Plan to be waived for such Transferred SLS Employee and his or her covered dependents, to the extent any such exclusions or requirements were waived or were inapplicable under any similar or comparable SLS Business Benefit Plan or Scotts Benefit Plan in which such Transferred SLS Employee participated immediately prior to the ELA Expiration Date. Newco shall use commercially reasonable efforts to cause any eligible expenses incurred by such Transferred SLS Employee and his or her covered dependents during the portion of the plan year of the SLS Business Benefit Plan or Scotts Benefit Plan, as applicable, ending on the date such Transferred SLS Employee's participation in the corresponding New Benefit Plan begins to be taken into account under such New Benefit Plan for purposes of satisfying all deductible, coinsurance and maximum out-of-pocket requirements applicable to such Transferred SLS Employee and his or her covered dependents for the applicable plan year as if such amounts had been paid in accordance with such New Benefit Plan. Effective as of the ELA Expiration Date, Newco shall, or shall cause TruGreen LP and its Affiliates to establish, to the extent it does not currently exist, a defined contribution plan with a 401(k) feature, or cause an existing defined contribution plan with a 401(k) feature to cover the Transferred SLS Employees who were participants in, or eligible to participate in, the 401(k) plan made available by Scotts and its Affiliates to the Transferred SLS Employees (the "SLS 401(k) Plan") prior to the ELA Expiration Date. Newco shall, or shall cause TruGreen LP and its Affiliates to permit Transferred SLS Employees to rollover any distribution from the SLS 401(k) Plan to a 401(k) plan maintained by Newco or a TruGreen Entity (the "Newco 401(k) Plan"). In furtherance of the foregoing, Newco shall, or shall cause TruGreen LP and its Affiliates, to permit any Transferred SLS Employee who is a participant in the SLS 401(k) Plan and who has a loan outstanding as of the ELA Expiration Date to rollover the note evidencing the loan, provided that such rollover is made within 60 days following the ELA Expiration Date. The rollover of distributions from the SLS 401(k) Plan to the Newco 401(k) Plan shall be in cash, except that outstanding loan balances shall be transferred in the form of notes or other documentation evidencing such loans.

(f) TruGreen Holdings shall assume and be solely responsible for all liabilities in respect of claims made by any SLS Employee for notice, retention, severance,

retrenchment or other termination pay or benefits (including claims for constructive dismissal, termination indemnities, any damages arising from a breach of the SLS Employee's employment contract, and any payments required to be made under any applicable Law in respect of the termination of the SLS Employee's employment), in each case, arising out of or in connection with the termination of employment or constructive dismissal of any Transferred SLS Employee by Newco or any of its Affiliates after the ELA Expiration Date.

(g) Unless Parties agree, prior to December 31, 2016, Newco shall, or shall cause its Affiliates to, provide vacation benefits and paid holidays to Transferred SLS Employees for so long as they are employed with Newco or its Affiliates that are at least as favorable as those provided under the applicable vacation program of Newco or its Affiliates. Newco and its Affiliates shall credit each Transferred SLS Employee with all vacation and paid time off that such Transferred SLS Employee is entitled to use but has not used as of the ELA Expiration Date, and Newco and its Affiliates shall assume all Liability for the payment of such amounts.

(h) Without limiting the generality of the foregoing, and unless the Parties agree otherwise in writing, Newco shall, or shall cause TruGreen LP and its Affiliates to provide severance compensation and benefits to any terminated Transferred SLS Employees and pay out retention awards in accordance with Scotts <u>Schedule 5.7(h)</u>.

(i) The terms of this <u>Section 5.7</u> are included for the sole benefit of the respective Parties and shall not confer any rights or remedies upon any Transferred SLS Employee or former employee of the SLS Business, any participant or beneficiary in any SLS Entity Benefit Plan or any other Person or Governmental Body (whether as a third party beneficiary or otherwise) other than the Parties. Nothing contained in this Agreement shall (<u>i</u>) constitute or be deemed to constitute an amendment to any SLS Entity Benefit Plan or other compensation or benefit plan, policy, program or arrangement of any SLS Entity, any TruGreen Entity or any other Person, (<u>ii</u>) obligate Newco or any of its Subsidiaries to (<u>A</u>) maintain any particular benefit plan or arrangement or (<u>B</u>) retain the employment of any particular employee; or (<u>iii</u>) prevent Newco, TruGreen LP or any of their Subsidiaries from amending or terminating any benefit plan or arrangement.

5.8 Intercompany Arrangements.

(a) Scotts shall cause (<u>i</u>) each SLS Intercompany Balance to be eliminated prior to the Closing, (<u>ii</u>) all SLS Intracompany Obligations that exist as of the date hereof or that are created after the date hereof, other than for trade payables and receivables, to be settled in full in cash prior to the Closing and (<u>iii</u>) each Contract between a SLS Entity, on the one hand, and Scotts or any of the Non-SLS Subsidiaries, on the other hand (each, an "<u>SLS Intercompany Contract</u>"), other than this Agreement and the Ancillary Agreements, to be terminated prior to the Closing without any obligation under any SLS Intercompany Contract surviving such termination other than the Contracts set forth on <u>Scotts Schedule 5.8(a)</u>.

(b) TruGreen Holdings shall cause (<u>i</u>) each TruGreen Intercompany Balance to be eliminated prior to the Closing, (<u>ii</u>) all TruGreen Intracompany Obligations that exist as of

the date hereof or that are created after the date hereof, other than for trade payables and receivables, to be settled in full in cash prior to the Closing and (<u>iii</u>) each Contract between a TruGreen Entity, on the one hand, and TruGreen Holdings or any of its Affiliates (other than the TruGreen Entities), on the other hand (each, a "<u>TruGreen Intercompany Contract</u>"), other than this Agreement, the Ancillary Agreements and any agreements (<u>A</u>) for the provision of commercial services or the supply or purchase of commercial goods with any portfolio company of any fund managed by Clayton, Dubilier & Rice, LLC (other than TruGreen Holdings) that are on an arms-length basis or (<u>B</u>) under Clayton, Dubilier & Rice, LLC's affiliate procurement programs, to be terminated prior to the Closing without any obligation under any TruGreen Intercompany Contract surviving such termination other than the Contracts set forth on <u>TruGreen Schedule 5.8(b</u>).

(c) Neither (<u>i</u>) the satisfaction or termination of any SLS Intercompany Balance or TruGreen Intercompany Balance nor (<u>ii</u>) the termination of any SLS Intercompany Contract or TruGreen Intercompany Contract shall be deemed a breach or violation of any provision of this Agreement.

5.9 Insurance.

(a) From and after the Closing Date, the SLS Entities shall cease to be insured by, have access or availability to, be entitled to make claims on, be entitled to claim benefits from or seek coverage under any of Scotts's or its Subsidiaries and/or Affiliates' insurance policies or any of their self-insured programs (for the purposes of this <u>Section 5.9</u>, "Scotts" shall include, where appropriate to the context, its Subsidiaries and/or Affiliates) (collectively, the "<u>Insurance Arrangements</u>"), other than (i) insurance policies or self-insurance programs acquired directly by and in the name of the SLS Entities, (ii) with respect to any claim, act, omission, event, circumstance, occurrence or Loss insured under any occurrence-based insurance policy to the extent it occurred or existed before the Closing Date and (iii) with respect to any claim, act, omission, event, circumstance policy that was known to Scotts or its Subsidiaries and/or Affiliates prior to the Closing (the insurance policies described in clauses (ii) and (iii), the "<u>Available Insurance Policies</u>").

(b) It shall be a condition to any pursuit of insurance benefits or coverage permitted by this <u>Section 5.9</u>, that:

(i) the SLS Entities shall promptly notify Scotts of all such claims or efforts to seek benefits or coverage under the applicable Available Insurance Policy and Scotts shall reasonably cooperate with the SLS Entities in pursuing all such claims, <u>provided</u> that the SLS Entities shall be solely responsible for notifying the relevant insurance companies of, and complying with all conditions for, such claims;

(ii) the applicable SLS Entity shall exclusively bear (and neither Scotts nor any of its Affiliates shall have any obligation to repay or reimburse any TruGreen Entity or SLS Entity therefor) the amount of any and all deductibles, retentions or self-insurance associated with claims under the Available Insurance Policies, whether such

claims are made by any of the SLS Entities, their respective employees or third parties, and shall be liable for all uninsured, uncovered, unavailable or uncollectible amounts of such claims; and

(iii) the SLS Entities shall not, without the prior written consent of Scotts, which shall not be unreasonably withheld, conditioned or delayed, (A) settle, release, commute, buy-back or otherwise resolve disputes (other than specific insurance coverage disputes that do not implicate coverage under the policy for claims other than the disputed claim) with respect to any Available Insurance Policies, or amend, modify or waive any rights under any such Available Insurance Policies or (B) assign the Available Insurance Policies or any rights or claims thereunder.

(c) Scotts shall retain the exclusive right to control all of its Insurance Arrangements, including the Available Insurance Policies, and the benefits and amounts payable thereunder, including the right to exhaust, settle, release, commute, buyback or otherwise resolve disputes with respect to any of its insurance policies and programs and to amend, modify or waive any rights under any such insurance policies and programs, notwithstanding whether any such policies or programs apply to any claims the SLS Entities have made or could make in the future; <u>provided</u> that Scotts and the Non-SLS Subsidiaries shall not take any such action unless such action would apply generally to coverage of claims by the SLS Entities and claims by the Non-SLS Subsidiaries; <u>provided</u>, <u>further</u>, that Scotts shall not be obligated to indemnify the SLS Entities for the exhaustion of the limits of liability under the Available Insurance Policies.

(d) Nothing in this Agreement is intended to waive or abrogate in any way Scotts's own rights to insurance coverage for any Liability, whether relating to the SLS Entities or otherwise.

(e) Whenever this <u>Section 5.9</u> requires any SLS Entity to take any action after the Closing, such requirement shall be deemed to involve an undertaking on the part of TruGreen Holdings to take such action or to cause such SLS Entity to take such action.

5.10 Financing; Solvency Opinion.

(a) TruGreen Holdings shall use its reasonable best efforts to take or cause to be taken, all actions, and do, or cause to be done, all things, necessary proper or advisable to obtain and arrange for TruGreen LP to obtain the Debt Financing on the terms and conditions (including any "flex" provisions) described in the Debt Financing Commitment and the Fee Letter, including using its reasonable best efforts to (i) except as otherwise permitted by this <u>Section 5.10(a)</u>, maintain in effect the Debt Financing Commitment in accordance with the terms and subject to the conditions thereof, (<u>ii</u>) assist in the satisfaction on a timely basis of all conditions applicable to TruGreen LP in obtaining the Debt Financing at the Closing as set forth therein, (<u>iii</u>) negotiate and enter into definitive agreements with respect to the Debt Financing on terms and conditions (including any "flex" provisions) no less favorable to TruGreen LP than those contemplated by the Debt Financing Commitment and the Fee Letter and which permit

Newco to perform its indemnification obligations pursuant to, and in accordance with, terms of this Agreement and (iv) obtain the financing no less favorable to TruGreen LP than that contemplated by the Debt Financing Commitment, including fully enforcing the rights of TruGreen LP under the Debt Financing Commitment, as applicable. TruGreen Holdings shall provide copies thereof to Scotts on a current basis and otherwise keep Scotts reasonably informed on a current basis of the status of its efforts to arrange the Debt Financing and promptly provide Scotts with copies of all executed material definitive agreements with respect to the Debt Financing Commitment. Neither TruGreen Holdings nor any of its Subsidiaries shall be permitted to modify, supplement, restate, assign, substitute or replace any of the Debt Financing Commitment; provided that TruGreen Holdings and its Subsidiaries may, without the consent of Scotts, (x) amend, replace or modify the Debt Financing Commitment to add or replace lenders, lead arrangers, book runners, syndication agents or similar entities or (y) otherwise amend, replace or modify, or consent to any waiver of any provision or remedy under, the Debt Financing Commitment, so long as such amendment, replacement, modification or waiver does not (A) reduce the aggregate amount of the Debt Financing to less than \$650 million or (B) impose new or additional conditions, or otherwise expand, amend or modify any of the conditions, to the receipt of Debt Financing in a manner that would reasonably be expected to (I) delay or prevent the Closing, (II) make the funding of the Debt Financing (or satisfaction of the conditions to obtaining the Debt Financing) less likely to occur or (III) adversely impact the ability of TruGreen LP to enforce its rights against the other parties to the Debt Financing Commitment, the ability of TruGreen LP to consummate the transactions contemplated hereby or the likelihood of consummation of the transactions contemplated hereby (it being agreed and understood that any amendment, modification, supplement, replacement or waiver relating to conditionality or the availability of funding (other than a reduction that would not reduce the aggregate amount of the Debt Financing to less than \$650 million) shall be deemed to require Scotts's prior written consent hereunder). TruGreen LP shall promptly deliver to Scotts copies of any amendment, modification, waiver or replacement of the Debt Financing Commitment, Subject to the terms and conditions of this Agreement (including Section 5.11), in the event any portion of the Debt Financing becomes unavailable on the terms and conditions contemplated in the Debt Financing Commitment, TruGreen Holdings shall promptly notify Scotts and shall use its reasonable best efforts to obtain and arrange for alternative financing from alternative sources in an amount equal to the amount of the Debt Financing contemplated by the Debt Financing Commitment; provided that TruGreen Holdings and TruGreen LP shall not be required to accept any alternative financing that is less favorable to it than the Debt Financing contemplated by the Debt Financing Commitment (including any "market flex" provision set forth in any fee letters provided with the Debt Financing Commitment). TruGreen Holdings shall promptly (upon reasonable request by Scotts) deliver to Scotts true and complete copies of all drafts of any alternative financing commitments and all final agreements pursuant to which any such alternative source shall have committed to provide TruGreen Holdings with any portion of the Debt Financing, References to "Debt Financing" in this Agreement shall include the financing contemplated by the Debt Financing Commitment as permitted by this Section 5.10(a) to be amended, modified or replaced and any alternative financings in lieu of the Debt Financing Commitment (with any such amendments and modifications as are permitted by this Section 5.10(a)), and references to "Debt Financing Commitment" and "Fee Letter" shall include such documents as permitted by this Section 5.10(a) to be amended, modified or replaced, from and

after such amendment, modification or replacement and any analogous documents contemplated by the immediately preceding sentence.

(b) TruGreen Holdings shall, and shall cause TruGreen LP to, give Scotts prompt notice of (i) any material breach or any event or circumstance that, with or without notice, lapse of time or both, would reasonably be expected to give rise to any material breach by any party to the Debt Financing Commitment of which TruGreen Holdings or TruGreen LP becomes aware, (ii) the receipt of any written notice or other written communication from any Person with respect to any actual breach, default or repudiation by any party to the Debt Financing Commitment, (iii) the indication in writing by a counterparty that it will not provide, or it refuses to provide, all or any portion of the financing contemplated by the Debt Financing Commitment on the terms set forth therein, and (iv) any termination of the Debt Financing Commitment. TruGreen Holdings shall, and shall cause TruGreen LP to promptly provide any information reasonably requested by Scotts relating to any circumstances referred to in this <u>Section 5.10(b)</u>.

(c) TruGreen Holdings shall, and shall cause Newco to, (i) use its reasonable best efforts to obtain the Solvency Opinion including, but not limited to, reducing the amount of the Debt Financing (and thus reduce the amount available to fund the Distributions) in an amount necessary to obtain such opinion (<u>provided</u> that none of TruGreen Holdings, TruGreen LP or Newco shall be required to reduce the amount of the Debt Financing to such an extent that the pro rata portion of the Financing Proceeds available to be distributed to the TruGreen Newco Holders would be less than \$410,000,000 in the aggregate), (ii) to the extent permitted by applicable Law, promptly provide to Scotts any written materials and/or analysis (including any drafts of the Solvency Opinion) provided to or prepared by TruGreen Holdings or Newco or any of their respective Affiliates in connection therewith and consider in good faith any documentary or other comments from Scotts thereto and (iii) use its commercially reasonable efforts to have the Solvency Opinion reflect that Scotts may rely on the Solvency Opinion.

5.11 Debt Financing Cooperation.

(a) Prior to the Closing, Scotts shall, and shall cause its Subsidiaries (including the SLS Entities) to, and shall use its reasonable best efforts to cause its and their representatives to, provide to TruGreen Holdings and TruGreen LP such cooperation reasonably requested by TruGreen Holdings and TruGreen LP to assist TruGreen Holdings and TruGreen LP in causing the conditions in the Debt Financing Commitment to be satisfied and such cooperation as is otherwise necessary or reasonably requested by TruGreen Holdings and TruGreen LP in connection with TruGreen LP's obtaining the Debt Financing in accordance with its terms, including cooperation that consists of:

(i) participating in a customary and reasonable number of meetings, presentations, road shows, due diligence sessions, drafting sessions and sessions with ratings agencies, in each case at times and locations mutually agreed and reasonably coordinated in advance thereof;

(ii) providing customary authorization letters to the Financing Sources authorizing the distribution of information to prospective lenders or investors (including customary 10b-5 and material non-public information representations);

(iii) executing and delivering any securities purchase agreement, credit agreement, indentures, notes, guarantees, pledge and security documents, supplemental indentures, currency or interest hedging arrangements, other definitive financing documents, and certificates or documents and back-up therefor and for legal opinions as may be reasonably requested by TruGreen Holdings or TruGreen LP and otherwise reasonably facilitating the pledging of collateral; provided that none of Scotts or its Subsidiaries or any of their respective officers or employees shall be required to execute any document in connection with this <u>Section 5.11(a)</u>, which document would be effective at any time before the time that will be immediately prior to the Closing, other than the execution by the SLS Entities of any note purchase agreement being entered into at the time of "pricing" of any high yield bonds issued in lieu of all or a portion of the Debt Financing, and any corporate authorizations in connection with such note purchase agreement;

(iv) (A) furnishing TruGreen Holdings, TruGreen LP and the Financing Sources as promptly as practicable with (x) audited combined balance sheets of the SLS Business for the two most recent completed fiscal years ended at least 90 days prior to the Closing Date, statements of operations, comprehensive loss, cash flows and changes of equity of the SLS Business for the three most recently completed fiscal years ended at least 90 days prior to the Closing Date, together with all related notes and schedules thereto, accompanied by the reports thereon of the independent auditors of the SLS Business and (y) unaudited combined balance sheets and related combined statements of operations, comprehensive loss, cash flows and changes of equity of the SLS Business prepared in accordance with GAAP for each of the first three fiscal quarters and related year-to-date interim periods ended after the close of its most recent fiscal year and at least 45 days prior to the Closing Date and for the comparable quarter and year-to-date interim period of the prior fiscal year, together with all related notes and schedules thereto, in the case of each of clauses (x) and (y), prepared in accordance with GAAP and in compliance with Regulation S-X promulgated under the Securities Act, (B) furnishing TruGreen Holdings and TruGreen LP (or to the Financing Sources on behalf of TruGreen Holdings and TruGreen LP) as promptly as practicable with all financial data and other information readily available or that can be gathered without undue burden regarding the SLS Business (including financial information of the SLS Business reasonably requested by TruGreen Holdings necessary for TruGreen LP's preparation of any pro forma financial statements taking into account the Transactions (provided that Scotts and its Subsidiaries shall not be responsible for the preparation of such pro forma financial statements or any related pro forma adjustments)), in each case, (x) of the type and form customarily included in an offering memorandum for private placements of "private for life" non-convertible high-yield bonds under Rule 144A promulgated under the Securities Act, or (y) as applicable, otherwise necessary to enable any investment bank that is an initial purchaser in such Rule 144A private placement to receive from the SLS Business'

independent accountants a customary "comfort" letter (including "negative assurance" comfort) (which, in the case of each of clauses (x) and (y), for the avoidance of doubt, shall not include information required by Rule 3-09, Rule 3-10 or Rule 3-16 of Regulation S-X, or Items 402 and 407 of Regulation S-K, or executive compensation and related person disclosure referred to in SEC Release Nos. 33-8732A, 34-54302A and IC-2744A, or other information customarily excluded from a Rule 144A offering memorandum), together with drafts of customary comfort letters which such independent accountants are prepared to deliver (subject to completion of customary procedures) upon "pricing" of any such Rule 144A private placement of high-yield bonds being issued in lieu of all or a portion of the Debt Financing, with respect to the financial information of the SLS Business included in such offering memorandum and, with respect to any interim financial statements of the SLS Business included in the offering memorandum, such interim financial statements shall have been reviewed by independent accountants of the SLS Business as provided in AU 722 to the extent necessary to provide such customary comfort letter, (C) consents from the auditors of the SLS Business for inclusion of their unqualified audit reports in any materials relating to the Debt Financing or any high-yield bonds being issued in lieu of all or a portion of the Debt Financing (in each case to the extent that such consents are required in order to so include such reports) (sub-clauses (A)-(C) of this clause (iv), the "Required Information"); and (D) reasonably assisting TruGreen Holdings and TruGreen LP in the preparation of materials (to the extent relating to the SLS Business and/or the SLS Entities) for customary rating agency presentations, lender presentations and high yield road show presentations or memoranda, customary bank information and offering memoranda, syndication memoranda, private information and offering memoranda, and similar marketing materials or memoranda, including such information and data related to the SLS Business and/or the SLS Entities as is reasonably required by TruGreen Holdings and TruGreen LP to produce the financial statements and information identified in paragraphs 5(a), 6 and 8 of Exhibit D of the Debt Financing Commitment (to the extent relating to the SLS Business or the SLS Entities) or otherwise required in connection with the Debt Financing;

(v) using reasonable best efforts to cooperate with TruGreen Holdings and TruGreen LP in connection with their efforts to obtain customary and reasonable corporate and facilities ratings, consents, landlord waivers and estoppels, non-disturbance agreements, non-invasive environmental assessments, legal opinions, surveys and title insurance (including providing reasonable access to TruGreen Holdings and its representatives to all SLS Owned Real Property and SLS Leased Real Property) as reasonably requested by TruGreen Holdings;

(vi) cooperating with TruGreen Holdings and TruGreen LP to satisfy the conditions precedent to the Debt Financing to the extent reasonably requested by TruGreen Holdings or TruGreen LP and within the control of Scotts and its Subsidiaries, and taking all corporate actions, subject to the occurrence of the Closing, reasonably requested by TruGreen Holdings or TruGreen LP to permit the consummation of the Debt Financing;

(vii) obtaining customary payoff letters, Encumbrance terminations and instruments of discharge to be delivered at Closing to allow for the payoff, discharge and termination in full on the Closing Date of all Indebtedness required to be paid, satisfied and discharged pursuant to <u>Section 1.6</u> (and excluding, for the avoidance of doubt, that comprising the Debt Financing); and

(viii) furnishing TruGreen Holdings and TruGreen LP promptly with all documentation and other information which any lender providing or arranging Debt Financing has reasonably requested and that such lender has determined is required by regulatory authorities in connection with such Debt Financing under applicable "know your customer" and anti-money laundering rules and regulations, including the PATRIOT Act;

<u>provided</u> that (\underline{x}) nothing in this <u>Section 5.11(a)</u> shall require such cooperation to the extent it would (<u>i</u>) require Scotts or any of its Subsidiaries to waive or amend any terms of this Agreement or agree to pay any fees or reimburse any expenses prior to the Closing for which it has not received prior reimbursement by or on behalf of TruGreen Holdings (except to the extent TruGreen Holdings has provided the indemnities set forth in <u>Section 5.11(c)</u>), (<u>ii</u>) require Scotts or any of its Subsidiaries or any of their respective Affiliates to pay (or agree to pay) any fees, or reimburse any expenses prior to the Closing for which it is not promptly reimbursed, or otherwise incur any other obligations or give any indemnities prior to the Closing that are not contingent on the Closing, (<u>iii</u>) cause any representation or warranty or covenant in this Agreement to be breached by Scotts or any of its Subsidiaries, (<u>iv</u>) conflict with the organizational documents of Scotts or any of its Subsidiaries or any Laws, (<u>v</u>) provide access to or disclose information that Scotts or any of its Subsidiaries determines, after consulting with counsel (internal or external) would jeopardize any attorney-client privilege of Scotts or any of its Subsidiaries, or (<u>vi</u>) except as provided in <u>Section 5.11(a)(iv</u>) above, prepare separate financial statements for any Subsidiary of Scotts or change any fiscal period, (<u>y</u>) nothing herein shall require such cooperation from Scotts or its Subsidiaries to the extent it would unreasonably interfere with the ongoing operations of Scotts or its Subsidiaries and (<u>z</u>), except to the extent TruGreen Holdings has provided the indemnities set forth in <u>Section 5.11(c</u>), none of Scotts or any of its Subsidiaries or any of their respective representatives shall have any liability under any certificate, agreement, arrangement, document or instrument relating to the Debt Financing, except, in the case of the SLS Entities, upon the Closing.

(b) TruGreen Holdings shall cause TruGreen LP to promptly, upon request by Scotts, reimburse Scotts for all of its and its Subsidiaries' and their representative's documented reasonable out-of-pocket costs and expenses (including reasonable attorneys' fees) incurred by Scotts and its Subsidiaries in connection with the cooperation of Scotts and its Subsidiaries contemplated by this <u>Section 5.11</u>.

(c) Scotts, its Subsidiaries and their respective officers, advisors and representatives shall be indemnified and held harmless by TruGreen LP for and against any and all liabilities, losses, damages, claims, costs, expenses, interest, awards, judgments and penalties suffered or incurred by them in connection with the arrangement of the Debt Financing and/or

the provision of information utilized in connection therewith (other than information provided in writing specifically for such use by or on behalf of Scotts or any of its Subsidiaries) to the fullest extent permitted by applicable Law, other than to the extent any of the foregoing arises from (<u>i</u>) the willful misconduct or gross negligence of any of Scotts, its Subsidiaries (including SLS Entities) or its or their respective directors, officers, employees, attorneys, accountants or other advisors or representatives or (<u>ii</u>) any information provided by or on behalf of Scotts or any of its Subsidiaries (including the SLS Entities) in connection with the Debt Financing to the extent such information is the subject of any of the representations or warranties set forth in <u>Article III</u> and where such information would constitute a breach of any such representation or warranty.

(d) Scotts shall or shall cause the SLS Entities and Non-SLS Subsidiaries (if applicable) to supplement all information delivered pursuant to <u>Section 5.11(a)(iv)</u> (together with any replacements or restatements thereof, and supplements thereto, if any such information would go stale or otherwise be unusable for its intended purposes) on a reasonably current basis to the extent that any such information, to the Knowledge of Scotts or the SLS Entities, when taken as a whole and in light of the circumstances under which such statements were made, contains any material misstatement of fact or omits to state any material fact necessary to make such information not materially misleading.

(e) The SLS Entities' logos may be used in connection with the Debt Financing; <u>provided</u> that such logos are used solely in a manner that is not intended to or reasonably likely to harm, disparage or otherwise adversely affect the SLS Entities or the reputation or goodwill of any SLS Entity.

5.12 Intellectual Property.

(a) Subject to compliance with the terms and conditions hereof, and effective as of the Closing, Scotts, on behalf of itself and its Affiliates, hereby grants to Newco and its Subsidiaries a non-exclusive, worldwide, perpetual, irrevocable, transferable (solely in connection with a transfer of the SLS Business), sublicensable, paid-up and royalty-free right and license to use and practice the Scotts Licensed IP Rights solely in connection with the SLS Business, as conducted as of the Closing Date, and any natural or reasonably foreseeable expansions thereof. Newco shall, and shall cause its Affiliates and sublicensees to, use reasonable care to maintain and protect the Trade Secrets included in the Scotts Licensed IP Rights.

(b) Between the date hereof and the Closing, TruGreen Holdings and Scotts shall negotiate in good faith a trademark license agreement (the "Trademark License Agreement"), to be effective as of the Closing, pursuant to which Newco and its Subsidiaries shall receive a transferable (solely in connection with a transfer of existing SLS Franchise Agreements), sublicensable (solely in connection with the existing SLS Franchise Agreements), royalty-free license for a term of two (2) years, extendable by Newco for an additional year solely to permit sublicensing to any SLS Franchisee whose SLS Franchise may not be repurchased at such time by Newco under its SLS Franchise Agreement without the consent of

such SLS Franchisee, to use the Scotts Marks to the extent required under existing SLS Franchise Agreements or any renewals thereof.

5.13 SLS Internal IT Systems and Data Separation; Replication of Shared Contracts.

(a) <u>Section 5.19</u> and the Transition Services Agreement entered into pursuant thereto shall govern separation of SLS Internal IT Systems and data access.

(b) Until the expiration of the Transition Services Agreement, Scotts shall, and shall cause its Subsidiaries to, use commercially reasonable efforts to cause the counterparty to any Shared Contract (excluding Shared Contracts for commercially available off-the-shelf Software that is not the subject of a negotiated agreement and for which the aggregate amounts paid or payable to or by Scotts or any of its Subsidiaries related to such Contract are less than \$50,000) and which is listed on TruGreen Schedule 5.13(b) (the "Designated Shared Contracts"), provided that (x) any Shared Contract may be removed from TruGreen Schedule 5.13(b) between the date hereof and the Closing Date at the election of TruGreen Holdings (such that it is not considered a Designated Shared Contract hereunder) and (v) TruGreen Holdings may between the date hereof and the Closing Date add to TruGreen Schedule 5.13(b) any Shared Contract that Scotts failed to include on Scotts Schedule 3.14(n) (such that it is considered a Designated Shared Contract hereunder), to enter into a new agreement with TruGreen LP or its designee effective as of the Closing, or as soon as practicable thereafter, with respect to matters addressed by such Designated Shared Contract. For the avoidance of doubt, Scotts's and its Affiliates' out-of-pocket costs and expenses performing and causing the foregoing shall be a Scotts Transaction Expense. Until the earlier of (i) such time as each Designated Shared Contract is replicated and (ii) the expiration of the Transition Services Agreement, Scotts shall, and shall cause its Subsidiaries to, upon the request of TruGreen Holdings or TruGreen LP and subject to the receipt of any necessary Consents, which Scotts shall use reasonable best efforts to secure, pass through services to TruGreen LP and the SLS Entities provided by the counterparty to the SLS Business to the applicable Designated Shared Contracts, and TruGreen LP and the SLS Entities shall have access to the benefits of such applicable Designated Shared Contracts, on a pass-through cost basis under the Transition Services Agreement (provided that TruGreen LP and the SLS Entities shall bear only their allocable costs of such Shared Contract and provision of benefit thereunder). Nothing herein shall require Scotts to accept greater liability than it would otherwise have in order to comply with this Section 5.13.

5.14 Certain Payments.

At the Closing or, to the extent any amounts below have not been invoiced as of the Closing, promptly following the time they become payable, TruGreen Holdings shall cause TruGreen LP to, without duplication, pay or, as appropriate, reimburse TruGreen Holdings and Scotts, as applicable, for (a) the TruGreen Holdings Transaction Expenses and (b) the Scotts Transaction Expenses, respectively, in each case by wire transfer of immediately available funds in U.S. dollars to the account (or accounts) designated in writing by TruGreen Holdings or Scotts, as the case may be, at least two Business Days prior to the Closing.

5.15 <u>Further Assurances</u>.

Following the Closing, each of TruGreen Holdings, on the one hand, and Scotts, on the other hand, shall (and shall cause their Subsidiaries and their representatives to) from time to time, at the other's reasonable request, execute and deliver, or cause to be executed and delivered, such further instruments, documents, conveyances or assurances and perform such further acts, as such other party may reasonably require in order to fully effect the Transactions and to otherwise consummate the transactions contemplated by this Agreement.

5.16 Transfers of Assets or Liabilities Following the Closing.

Other than as contemplated by any Ancillary Agreement:

(a) If at any time during or after the Closing, Scotts or any of the Non-SLS Subsidiaries shall receive or otherwise possess any SLS Asset or receive any payment with respect to, or in respect of, any SLS Asset or in respect of the SLS Business (other than in its capacity as a holder of interests in Newco after the Contribution), such Person shall promptly transfer, or cause to be transferred, such SLS Asset and/or payment to TruGreen LP or its designee (for no further consideration).

(b) If at any time during or after the Closing, TruGreen Holdings, any TruGreen Entity or any SLS Entity shall receive or otherwise possess any asset owned or otherwise possessed by Scotts or any of its Subsidiaries prior to the Closing that is not an SLS Asset or receives payment with respect to, or in respect of, any such asset or in respect of business of Scotts or any of its Subsidiaries prior to the Closing that is not part of the SLS Business, such Person shall promptly transfer, or cause to be transferred, such asset and/or payment to Scotts or its designees (for no further consideration).

5.17 Audited SLS Financial Statements.

Scotts shall deliver to TruGreen Holdings, as promptly as reasonably practicable after the date of the Agreement, and shall use reasonable best efforts to deliver no later than January 11, 2016, if practicable and taking into account Scotts's interest in the quality of audit and review process in connection therewith, to TruGreen Holdings audited combined balance sheets of the SLS Business as at September 30, 2014 and September 30, 2015, and the related audited combined statements of operations, comprehensive loss, cash flows and changes of equity of the SLS Business for the years ended September 30, 2013, September 30, 2014 and September 30, 2015, together with all related notes and schedules thereto, accompanied by the reports thereon of the independent auditors of the SLS Business (the "<u>Audited SLS Financial Statements</u>"). The Audited SLS Financial Statements shall be accompanied by a certificate in the form attached hereto as Exhibit C, signed by a duly authorized officer of Scotts (the "<u>Financial Statements Certificate</u>").

5.18 <u>Franchises</u>. Scotts shall use commercially reasonable efforts to cause each SLS Franchisee operating under an expired SLS Franchise Agreement to renew its SLS Franchise on

Scotts's current form of franchise agreement prior to the Closing with the clarifications set forth on <u>Scotts Schedule 5.18</u>.

5.19 <u>Transition Services Agreement</u>. Between the date hereof and the Closing, TruGreen Holdings and Scotts shall cooperate reasonably and in good faith to negotiate and agree to a Transition Services Agreement, to be effective as of the Closing, pursuant to which Scotts and its Affiliates will provide or cause to be provided to Newco and its Subsidiaries solely for use in the SLS Business certain mutually agreed services (other than the services set forth on Scotts Schedule 5.19 (the "Excluded Services") and facilities that Scotts and its Affiliates provide or cause to be provided to the SLS Business as of the date hereof (it being understood and agreed that Scotts shall offer to provide to Newco and its Subsidiaries all such services (other than the Excluded Services) as Scotts and its Affiliates provide to the SLS Business as of the date hereof and at the Closing, that are reasonably necessary to allow for the operation of the SLS Business in all material respects as currently conducted as of the date hereof and as of immediately prior to the Closing without any additional material costs (excluding for the avoidance of doubt any payments under the Transition Services Agreement); provided, however, that Shared Contracts shall be governed exclusively by Section 5.13. Such services will be provided (a) at Scotts's and its Affiliates' fully loaded cost (excluding any margin, mark-up or similar charge) and (b) for a term ending no later than one (1) year after the Closing Date. The Transition Services Agreement will include such provisions as are customary in such agreements in connection with divestitures of commercial operating businesses. Notwithstanding anything to the contrary, (i) none of Scotts or its Affiliates shall be required to provide any services to the SLS Business that, if so provided, would violate Law or result in a material violation or material breach to a contract to which Scotts or any of the Non-SLS Subsidiaries are party that results in a material liability to Scotts or the Non-SLS Subsidiaries, provided that Scotts or the applicable Non-SLS Subsidiary has used commercially reasonable efforts to obtain consent under such contract to provide such services) and (ii) none of Scotts or its Affiliates shall be obligated to provide any services that are not provided to the SLS Business as of the date hereof or as of the Closing Date.

ARTICLE VI

TAX MATTERS

6.1 <u>Tax Indemnification</u>.

(a) <u>Indemnification by Scotts</u>.

(1) From and after the Closing, Scotts agrees to indemnify the TruGreen Indemnified Parties against, and to hold the TruGreen Indemnified Parties harmless from and pay, all Losses arising out of (i) any Taxes imposed on or measured by a net or gross income, receipts, capital or worth of any SLS Entity (or in respect of the SLS Assets) with respect to any Pre-Closing Tax Period, together with any interest, penalty or additions to Tax accruing after the Closing on Taxes described in this clause (i); (ii) any Liability of any SLS Entity for the Taxes of any Person by reason of (A)

being a transferee or successor to such Person prior to the Closing Date, (B) pursuant to Section 1.1502-6 of the Treasury Regulations (or comparable provision under any other applicable Law) by reason of being affiliated with such Person prior to the Closing or (C) by reason of any Contract entered into by such SLS Entity prior to the Closing; (iii) any Taxes attributable to any inaccuracy of or breach by Scotts of any representation or warranty made by Scotts in Section 3.9; (iv) any Taxes attributable to any failure by Scotts to perform or comply with any covenant or agreement in this Agreement relating to Taxes; (v) any Taxes that arise as the result of any inclusion under Section 951 of the Code at the end of the taxable year of any SLS Entity that is a controlled foreign corporation (as defined under Section 957 of the Code) that includes the Closing Date to the extent such inclusion results from any transactions or activities occurring between the beginning of the taxable year of such controlled foreign corporation that includes the Closing Date and through the Closing; (vi) any Taxes that arise from the Scotts Internal Reorganization and that would not otherwise arise absent the consummation of the Scotts Internal Reorganization, or any other transactions or activities contemplated by Section 5.8(a); (vii) Liability of any SLS Entity for any Taxes of Scotts or any Non-SLS Subsidiary with respect to any Post-Closing Tax Period; and (viii) any withholding Taxes required to be withheld on any payments or distributions to Scotts (including the issuance of Scotts Newco Interests) required to be made under this Agreement or any Ancillary Agreement; provided, however, that Scotts in all events shall have no obligation to indemnify the TruGreen Indemnified Parties under this Section 6.1(a) from and against any Taxes to the extent such Taxes are taken into account on the Post-Closing Statement (and assuming full completion of the Transactions) as finally determined pursuant to Section 2.1. The amount of any payments required to be made pursuant to this Section 6.1(a) shall be computed without regard to any net operating loss, net capital loss or other Tax deduction, credit or benefit that is attributable to, arises from or relates to any Post-Closing Tax Period. Notwithstanding any other provision of this Agreement and for the avoidance of doubt, the limitations in Section 9.4 shall not apply to this <u>Section 6.1(a)</u>.

(2) From and after the Closing Date, Scotts agrees to indemnify the TruGreen Indemnified Parties against, and to hold the TruGreen Indemnified Parties harmless from and to pay, all Scotts Transaction Taxes to the extent not paid by Scotts pursuant to <u>Section 6.4(a)</u>.

(b) Indemnification by TruGreen Holdings.

(1) From and after the Closing, TruGreen Holdings agrees to cause Newco (out of the assets of Newco and its Subsidiaries) to indemnify the Scotts Indemnified Parties against, and to hold the Scotts Indemnified Parties harmless from and to pay, all Losses (taking into account the Indemnity Gross-Up Principle) arising out of (i) any Taxes imposed on or measured by a net or gross income, receipts, capital or worth of any TruGreen Entity with respect to any Pre-Closing Tax Period, together with any interest, penalty or additions to Tax accruing after the Closing on Taxes described in this clause (i); (ii) any Liability of any TruGreen Entity for the Taxes of any Person by reason of (A) being a transferee or successor to such Person prior to the Closing Date, (B)

pursuant to Section 1.1502-6 of the Treasury Regulations (or comparable provision under any other applicable Law) by reason of being affiliated with such Person prior to the Closing or (C) by reason of any Contract entered into by such TruGreen Entity prior to the Closing; (iii) any Taxes attributable to any inaccuracy of or breach by TruGreen Holdings of any representation or warranty made by TruGreen Holdings in Section 4.9; (iv) any Taxes attributable to any failure by TruGreen Holdings to perform or comply with any covenant or agreement in this Agreement relating to Taxes; (v) any Taxes that arise as the result of any inclusion under Section 951 of the Code at the end of the taxable year of any TruGreen Entity that is a controlled foreign corporation (as defined under Section 957 of the Code) that includes the Closing Date to the extent such inclusion results from any transactions or activities occurring between the beginning of the taxable year of such controlled foreign corporation that includes the Closing Date and through the Closing; (vi) any Taxes that arise from the Newco Restructuring, the Merger or any other transactions or activities contemplated by Section 5.8(b); (vii) Liability of any TruGreen Entity for any Taxes of TruGreen Holdings with respect to any Post-Closing Tax Period; and (viii) any withholding Taxes required to be withheld on any payments or distributions to TruGreen Holdings required to be made under this Agreement or any Ancillary Agreement; provided, however, that TruGreen Holdings in all events shall have no obligation to indemnify the Scotts Indemnified Parties under this <u>Section 6.1(b)</u> from and against any Taxes to the extent such Taxes are taken into account on the Post-Closing Statement (and assuming full completion of the Transactions) as finally determined pursuant to Section 2.1. The amount of any payments required to be made pursuant to this Section 6.1(b). shall be computed without regard to any net operating loss, net capital loss or other Tax deduction, credit or benefit that is attributable to, arises from or relates to any Post-Closing Tax Period. Notwithstanding any other provision of this Agreement and for the avoidance of doubt, the limitations in <u>Section 9.4</u> shall not apply to this <u>Section 6.1(b)</u>.

(2) From and after the Closing Date, TruGreen Holdings agrees to cause Newco (out of the assets of Newco and its Subsidiaries) to indemnify the Scotts Indemnified Parties against, and to hold the Scotts Indemnified Parties harmless from and to pay, all TruGreen Transaction Taxes (taking into account the Indemnity Gross-Up Principle, mutatis mutandis), to the extent not paid by TruGreen Holdings pursuant to <u>Section 6.4(b)</u>.

6.2 Procedures Relating to Certain Tax Indemnification.

(a) If a claim for Taxes, including notice of a pending audit, shall be made by any Tax Authority in writing, which, if successful, might result in an indemnity payment pursuant to <u>Section 6.1</u> (any such claim, a "Tax Claim"), (i) if the notice is received by TruGreen Holdings or its Subsidiaries (including Newco and Subsidiaries of Newco), TruGreen Holdings shall notify Scotts or (ii) if the notice is received by Scotts or its Subsidiaries (for the avoidance of doubt, not including Newco and the Subsidiaries of Newco), Scotts shall notify Newco, in each case, in writing of the Tax Claim within 15 Business Days after receipt of such Tax Claim. If the party seeking indemnification (the "Tax Indemnified Party") does not give notice of a Tax

Claim within such period or in detail sufficient to apprise Newco (in the case of indemnification under <u>Section 6.1(a)</u>) or Scotts (in the case of indemnification under <u>Section 6.1(b)</u>) (as applicable, the "Tax Indemnifying Party"), of the nature of the Tax Claim, the Tax Indemnifying Party shall not be liable to the Tax Indemnified Party to the extent that the Tax Indemnifying Party is prejudiced as a result thereof.

(b) The Tax Indemnifying Party shall control all proceedings in connection with any Tax Claim (including selection of counsel) and, without limiting the foregoing, may pursue or forego (and shall be entitled to cause Newco to pursue or forego) any and all administrative appeals, proceedings, hearings and conferences with any Tax Authority with respect thereto, and may either pay the Tax Claim and sue for a refund where applicable Law permits such refund suits or contest such Tax Claim in any permissible manner (and shall also be entitled to cause Newco to do so); provided, however, that (i) the Tax Indemnifying Party shall keep Newco and the Tax Indemnified Party regularly informed as to the status of such proceedings (including by providing copies of all notices received from the relevant Tax Authority) and the Tax Indemnified Party shall have the right to review and comment on any correspondence from the Tax Indemnifying Party to the relevant Tax Authority prior to submission of such correspondence to the Tax Authority and (ii) the Tax Indemnifying Party shall not settle or otherwise compromise such Tax Claim without the Tax Indemnified Party's prior written consent, which consent shall not be unreasonably withheld, conditioned or delayed, unless the Tax Indemnifying Party indemnifies and holds harmless the Tax Indemnified Party from and against any such Taxes actually incurred as a result of such settlement or compromise. Any indemnification payment resulting from application of the foregoing sentence shall be paid at the time such liability for Taxes is actually incurred by the Tax Indemnified Party. The Tax Indemnifying Party shall not have the right to control any proceedings related to such Taxes arising in a Post-Closing Tax Period. If any Tax Indemnifying Party elects not to control or otherwise does not have the right to control any proceedings relating to a Tax Claim, the Tax Indemnified Party shall control such proceedings; provided, however, (i) the Tax Indemnified Party shall keep the Tax Indemnifying Party reasonably informed as to the status of such proceedings (including by providing copies of all notices received from the relevant Tax Authority) and the Tax Indemnifying Party shall have the right to review and comment on any correspondence from Newco or the Tax Indemnified Party to the relevant Tax Authority prior to submission of such correspondence to the Tax Authority and (ii) the Tax Indemnified Party shall not settle or compromise such Tax Claim without the Tax Indemnifying Party's prior written consent (which consent shall not be unreasonably withheld, conditioned or delayed). Each of Scotts, TruGreen Holdings and each Subsidiary of TruGreen Holdings (including Newco and its Subsidiaries) shall execute any power-of-attorney and take any other actions necessary to effect this Section 6.2.

6.3 Tax Returns.

(a) Scotts shall timely prepare and file or cause to be timely prepared and filed with the appropriate Tax Authorities all U.S. federal and state income Tax Returns required to be filed on or prior to the Closing Date by or with respect to the SLS Entities or the SLS

Assets. All such Tax Returns shall be prepared in a manner consistent with most recent past practice, except as otherwise required by applicable Law.

(b) TruGreen Holdings shall timely prepare and file or cause to be timely prepared and filed with the appropriate Tax Authorities all U.S. federal and state income Tax Returns required to be filed on or prior to the Closing Date by or with respect to the TruGreen Entities. All such Tax Returns shall be prepared in a manner consistent with most recent past practice, except as otherwise required by applicable Law.

(c) After the Closing Date, Newco shall timely prepare and file or cause to be timely prepared and filed with the appropriate Tax Authorities all Tax Returns required to be filed with respect to Newco or any of its Subsidiaries. To the extent any amount payable with respect to such Tax Returns would cause Scotts to be required to indemnify Newco pursuant to <u>Section 6.1(a)</u>, Newco shall provide drafts of such Tax Returns to Scotts for its review and comment at least 14 days prior to filing and to consider in good faith all reasonable comments by Scotts on such Tax Returns.

6.4 Transaction Taxes.

(a) Notwithstanding any other provision of this Agreement to the contrary, all transfer, documentary, sales, use, excise, stamp, registration and other such Taxes imposed by or payable to any jurisdiction or any Governmental Body arising from the Scotts Internal Reorganization (collectively, the "<u>Scotts Transaction Taxes</u>") shall be borne by Scotts. Scotts will, at its own expense, file all necessary Tax Returns and other documentation with respect to all such Scotts Transaction Taxes, and, if required by applicable Law, TruGreen Holdings will, and will cause its Affiliates to, join in the execution of any such Tax Returns and other documentation.

(b) Notwithstanding any other provision of this Agreement to the contrary, all transfer, documentary, sales, use, excise, stamp, registration and other such Taxes imposed by or payable to any jurisdiction or any Governmental Body arising from the Newco Restructuring and the Merger (collectively, the "<u>TruGreen Transaction Taxes</u>") shall be borne by TruGreen Holdings. TruGreen Holdings will, at its own expense, file all necessary Tax Returns and other documentation with respect to all such TruGreen Transaction Taxes, and, if required by applicable Law, Scotts will, and will cause its Affiliates to, join in the execution of any such Tax Returns and other documentation.

6.5 <u>Cooperation</u>.

Scotts, TruGreen Holdings and TruGreen Holdings' Subsidiaries (including Newco) shall reasonably cooperate, and shall cause their respective Affiliates, officers, employees, agents, auditors and representatives to reasonably cooperate, in preparing and filing all Tax Returns relating to Newco or any of its Subsidiaries, including by provision of any required power-of-attorney (or other form of authorization), and in maintaining and making available to each other all records necessary in connection with Taxes and in resolving all disputes and audits, and in

connection with any other legitimate matters (including, for the avoidance of doubt, reasonable requests for information by Scotts or by TruGreen Holdings, any of their respective Affiliates or any of their respective employees or representatives, relating to the tax planning by Newco and its Subsidiaries) with respect to all taxable periods relating to Taxes; <u>provided</u> that the foregoing shall be done in a manner so as not to interfere unreasonably with the conduct of the business of the parties.

6.6 <u>Tax Sharing Agreements</u>.

(a) Scotts shall release or cause the release of each SLS Entity from any obligation under any agreement relating to the allocation, indemnification or sharing of Taxes other than (i) this Agreement and (ii) commercial contracts, the principal subject matter of which is not Taxes, entered into in the ordinary course of business and containing customary Tax indemnification provisions, ("<u>Tax Sharing Agreements</u>") with any Person (other than Scotts Systems or any of its Subsidiaries) prior to the Closing Date. For the avoidance of doubt, Scotts shall cause Scotts Systems and each of its Subsidiaries to have no obligation under any Tax Sharing Agreement to any Person (other than Scotts Systems or any of its Subsidiaries) for the payment of Taxes.

(b) TruGreen Holdings shall release or cause the release of the TruGreen Entities from any obligation under any Tax Sharing Agreements with any Person (other than the TruGreen Entities) prior to the Closing Date. For the avoidance of doubt, TruGreen Holdings shall cause the TruGreen Entities to have no obligations under any Tax Sharing Agreement to any Person (other than the TruGreen Entities) for the payment of Taxes.

6.7 <u>Straddle Period</u>.

For purposes of this <u>Article VI</u>, in the case of any Straddle Period for any SLS Entity or TruGreen Entity, as the case may be, the portion of any Tax attributable to the Pre-Closing Tax Period portion of such Straddle Period shall (<u>a</u>) in the case of any Taxes other than Taxes described in paragraph (b) of this <u>Section 6.7</u>, be deemed to be the amount of such Tax for the entire Straddle Period multiplied by a fraction, the numerator of which is the number of days in the Straddle Period ending on the Closing Date and the denominator of which is the number of days in the entire Straddle Period, and (<u>b</u>) in the case of any Tax based upon or related to income or receipts (including income Taxes and sales and use Taxes), be deemed equal to the amount which would be payable in such Pre-Closing Tax Period if the relevant taxable period ended on the Closing Date.

6.8 <u>Computational Matters</u>.

The indemnification obligations of Scotts and TruGreen Holdings pursuant to this <u>Article VI</u> shall, in respect of computational matters, be subject to <u>Section 9.5</u>.

6.9 Exclusivity.

To the extent of any inconsistency between this <u>Article VI</u> and <u>Article IX</u>, this <u>Article VI</u> shall control as to Tax matters.

ARTICLE VII

CONDITIONS TO THE CLOSING

7.1 <u>Conditions to Obligations of Scotts and the Non-SLS Subsidiaries</u>.

The obligations of Scotts to effect the Closing under this Agreement will be subject to the fulfillment (or, to the extent permitted by Law, written waiver by Scotts), at or prior to the Closing, of each of the following conditions:

(a) <u>Accuracy of Representations and Warranties</u>. (i) The representations and warranties of TruGreen Holdings contained in this Agreement (other than the representations and warranties identified in clause (ii) hereof) shall be true and correct in all respects (without giving effect to any qualifications or limitations as to "materiality" or "<u>TruGreen Material Adverse Effect</u>" or similar qualifications included therein), except where the failure of such representations and warranties to be true and correct would not, individually or in the aggregate, reasonably be expected to have a TruGreen Material Adverse Effect and (<u>ii</u>) the representations and warranties in <u>Sections 4.1</u>, <u>4.2(a)</u>, <u>4.2(b)</u>, <u>4.10(a)</u> and <u>4.19</u> shall be true and correct in all respects, in the case of the representations and warranties referenced in each of the foregoing clauses (i) and (ii), on and as of the date hereof and the Closing Date as if made on and as of such date (other than representations and warranties which address matters only as of a certain date which shall be accurate as of such certain date).

(b) <u>Compliance with Agreements and Covenants</u>. All the covenants and agreements contained in this Agreement to be complied with by TruGreen Holdings at or before the Closing shall have been complied with in all material respects.

(c) <u>Certificate of Compliance</u>. TruGreen Holdings shall have delivered to Scotts a certificate dated as of the Closing Date, signed by a duly authorized officer of TruGreen Holdings, certifying as to compliance with <u>Section 7.1(a)</u> and <u>Section 7.1(b)</u>.

(d) <u>Antitrust Approvals</u>. All applicable waiting periods and clearances pursuant to the HSR Act and Applicable Non-U.S. Antitrust Approvals shall have unconditionally expired, been terminated or been obtained, as applicable.

(e) <u>No Adverse Order</u>. No Governmental Body shall have enacted, issued, promulgated, enforced or entered any Law or Governmental Order, which is in effect and has the effect of making the transactions contemplated by this Agreement or the Ancillary Agreements illegal or otherwise restraining or prohibiting consummation of such transactions.

(f) <u>Closing Deliverables</u>. TruGreen Holdings shall have delivered, or shall deliver at the Closing, to Scotts copies of each of the items described in <u>Section 1.5(b)</u>.

(g) <u>No TruGreen Material Adverse Effect</u>. No change, effect, event, development or occurrence shall exist as of the Closing Date or shall have occurred since the date hereof that, individually or in the aggregate, has had or would reasonably be expected to have a TruGreen Material Adverse Effect.

(h) <u>Debt Financing</u>. The Debt Financing shall have occurred and the Financing Proceeds shall be available to TruGreen LP at or prior to the Closing.

(i) <u>Newco Restructuring</u>. The Newco Restructuring shall have been effected.

(j) <u>Merger Agreement</u>. (i) The Merger Agreement shall be in full force and effect, (ii) all of the closing conditions to the Merger Agreement (except for the closing of the transactions contemplated by this Agreement) shall have been satisfied or waived by the applicable party or parties thereto and (iii) all of the parties thereto shall have confirmed that they are ready, willing and able to close the transactions contemplated thereby.

(k) <u>Solvency Opinion</u>. Newco shall have received a solvency and surplus opinion from Duff & Phelps LLC, in customary form and delivered no more than ten Business Days prior to the Closing Date, to the effect that the Distributions are permitted under and in compliance with applicable Law and after giving effect thereto, Newco shall be Solvent (the "<u>Solvency</u> <u>Opinion</u>").

(l) <u>Audited SLS Financial Statements</u>. The Audited SLS Financial Statements shall satisfy the first requirement set forth on <u>TruGreen Schedule 7.2(l)</u>.

7.2 <u>Conditions to Obligations of TruGreen Holdings</u>.

The obligations of TruGreen Holdings to effect the Closing under this Agreement will be subject to the fulfillment (or, to the extent permitted by Law, written waiver by TruGreen Holdings), at or prior to the Closing, of each of the following conditions:

(a) <u>Accuracy of Representations and Warranties</u>. (i) The representations and warranties of Scotts and the Non-SLS Subsidiaries contained in this Agreement (other than the representations and warranties identified in clause (ii) hereof) shall be true and correct in all respects (without giving effect to any qualifications or limitations as to "materiality" or "<u>SLS Material Adverse Effect</u>" or similar qualifications included therein), except where the failure of such representations and warranties to be true and correct would not, individually or in the aggregate, reasonably be expected to have a SLS Material Adverse Effect, and (<u>ii</u>) the representations and warranties in <u>Sections 3.1</u>, <u>3.2(a)</u>, <u>3.2(b)</u>, <u>3.10(a)</u> and <u>3.19</u> shall be true and correct in all respects, in the case of the representations and warranties referenced in each of the foregoing clauses (i) and (ii), on and as of the date hereof and the Closing Date as if made on and

as of such date (other than representations and warranties which address matters only as of a certain date, which shall be accurate as of such certain date).

(b) <u>Compliance with Agreements and Covenants</u>. The covenants and agreements contained in this Agreement to be complied with by Scotts at or before Closing shall have been complied with in all material respects.

(c) <u>Certificate of Compliance</u>. Scotts shall have delivered to TruGreen Holdings a certificate dated as of the Closing Date, signed by a duly authorized officer of Scotts, certifying as to compliance with <u>Section 7.2(a)</u> and <u>Section 7.2(b)</u>.

(d) <u>Antitrust Approvals</u>. All applicable waiting periods and clearances pursuant to the HSR Act and Applicable Non-U.S. Antitrust Approvals shall have unconditionally expired, been terminated or been obtained, as applicable.

(e) <u>No Adverse Order</u>. No Governmental Body shall have enacted, issued, promulgated, enforced or entered any Law or Governmental Order, which is in effect and has the effect of making the transactions contemplated by this Agreement or the Ancillary Agreements illegal or otherwise restraining or prohibiting consummation of such transactions.

(f) <u>Closing Deliverables</u>. Scotts shall have delivered, or shall deliver at the Closing, to TruGreen Holdings copies of each of the items described in <u>Section 1.5(a)</u>.

(g) <u>No SLS Material Adverse Effect</u>. No change, effect, event, development or occurrence shall exist as of the Closing Date or shall have occurred since the date hereof that, individually or in the aggregate, has had or would reasonably be expected to have a SLS Material Adverse Effect.

(h) <u>Debt Financing</u>. The Debt Financing shall have occurred and the Financing Proceeds shall be available to TruGreen LP at or prior to the Closing.

(i) <u>Scotts Internal Reorganization</u>. The Scotts Internal Reorganization shall have been effected.

(j) <u>Merger Agreement</u>. All of the closing conditions to the Merger Agreement (except for the closing of the transactions contemplated by this Agreement) shall have been satisfied or waived by the applicable party or parties thereto and all of the parties thereto shall have confirmed that they are ready, willing and able to close the transactions contemplated thereby.

(k) <u>Solvency Opinion</u>. Newco shall have received the Solvency Opinion.

(l) <u>Audited SLS Financial Statements</u>. TruGreen Holdings shall have received the Audited SLS Financial Statements satisfying the requirements set forth on <u>TruGreen Schedule 7.1(l)</u>.

7.3 Frustration of Closing Conditions.

Neither Scotts nor TruGreen Holdings may rely, either as a basis for not consummating the transactions contemplated under this Agreement or terminating this Agreement and abandoning the Transactions, on the failure of any condition set forth in <u>Section 7.1</u> or <u>7.2</u>, as the case may be, to be satisfied if such failure was caused by such Party's breach in any material respect of any provision of this Agreement or failure to use the requisite efforts required to consummate the transactions contemplated hereby, as required by and subject to <u>Section 5.3</u> and any other applicable provisions of this Agreement.

ARTICLE VIII

CERTAIN RESTRICTIONS

8.1 Non-Competition.

(a) Scotts.

(i) Scotts agrees that for the period from the Closing Date until the second anniversary of the date that Scotts no longer has the right pursuant to the terms of the LLC Agreement (or as a result of irrevocably and in writing relinquishing such rights under the LLC Agreement) to designate any directors of Newco (the "Non-Competition Period"), it shall not, and shall cause the Non-SLS Subsidiaries (for so long as they remain Non-SLS Subsidiaries) not to, engage in the Specified Activities listed in clause (y) of the definition thereof or the commercial activities constituting the SLS Business as of the date hereof and as of immediately prior to the Closing, anywhere in the United States or Canada (each, a "<u>Scotts Prohibited Activity</u>"); provided that the foregoing shall not prohibit Scotts or any of its Subsidiaries from collectively owning (<u>i</u>) the Scotts Newco Interests or other equity interests in Newco (or any successor entity) or participating in the management of Newco, the TruGreen Entities and the SLS Entities pursuant to this Agreement and the Ancillary Agreements or (<u>ii</u>) up to an aggregate of five percent of the outstanding shares of any class of capital stock of any publicly traded Person that engages in any Scotts Prohibited Activity (a "<u>Competing Person</u>") so long as neither Scotts nor any of its Subsidiaries has any participation in the management of such Competing Person.

(ii) Notwithstanding anything to the contrary in the foregoing, nothing in this <u>Section 8.1</u> shall prohibit Scotts or any of its Subsidiaries from acquiring the whole or any part of a Person or business which engages in any Scotts Prohibited Activity or the whole or any part of a business which includes any Scotts Prohibited Activity; <u>provided</u> that where such Scotts Prohibited Activities of such Person or business acquired during the Non-Competition Period represent greater than 10% of the revenues of such Person or business acquired as set out in the latest available annual financial statements of that Person or business (other than with respect to any acquisition of a business which includes the Specified Activities listed in clause (y) of the definition thereof, any such acquisition (x) that would not cause the annual revenues of Scotts and its Subsidiaries attributable to such Scotts Prohibited Activity to exceed \$30 million and (y) for which the other Scotts Prohibited Activities represent less than 10% of the revenues of such Person or business acquired) (for the avoidance of doubt, without inclusion of revenues attributable to hydroponic systems, which is not a Scotts Prohibited Activity), in each case determined based on the latest available annual financial statements of that Person or business and the latest available annual financial statements of Scotts and its Subsidiaries, Scotts and/or its Subsidiaries shall be required to use commercially reasonable efforts to, within 18 months after the consummation of such acquisition, either (x) divest such Person, business or portion thereof to the extent engaging in such Scotts Prohibited Activity (irrespective of whether or not the end of the Non-Competition Period would occur within 18 months after the consummation of such acquisition) or (y) otherwise cause such Person, business or portion thereof to comply with this <u>Section 8.1</u>.

(b) Newco.

(i) TruGreen Holdings agrees that for Non-Competition Period, it shall not, and shall cause Newco and its Subsidiaries (for so long as they remain Affiliates of TruGreen Holdings and/or Clayton, Dubilier & Rice, LLC (subject to clause (ii)(B) of this <u>Section 8.1(b)</u>), not to, engage in the commercial activities constituting the business of Scotts as of the date hereof and as of immediately prior to the Closing other than (x) Specified Activities, (y) the SLS Business and (z) the business of the TruGreen Entities, in the case of each of clauses (y) and (z), as of the date hereof and as of immediately prior to the Closing and its Subsidiaries from collectively owning up to an aggregate of five percent of the outstanding shares of any class of capital stock of any other publicly traded Person that engages in any Newco Prohibited Activity (a "<u>Scotts Competing Person</u>") so long as neither TruGreen Holdings nor any of its Subsidiaries has any participation in the management of such Scotts Competing Person.

(ii) Notwithstanding anything to the contrary in the foregoing, (A) nothing in this <u>Section 8.1</u> shall prohibit TruGreen Holdings or any of its Subsidiaries from acquiring the whole or any part of a Person or business which engages in any Newco Prohibited Activity or the whole or any part of a business which includes any Newco Prohibited Activity; <u>provided</u> that where such Newco Prohibited Activities of such Person or business represent greater than 10% of the revenues of such Person or business acquired as set out in the latest available annual financial statements of that Person or business, TruGreen Holdings and/or its Subsidiaries shall be required to use commercially reasonable efforts to, within 18 months after the consummation of such acquisition, either (x) divest such Person, business or portion thereof to the extent engaging in such Newco Prohibited Activity (irrespective of whether or not the end of the Non-Competition Period would occur within 18 months after the consummation of such acquisition) or (y) otherwise cause such Person, business or portion thereof to comply

with this <u>Section 8.1</u> and (<u>B</u>) this <u>Section 8.1(b)</u> shall terminate and be of no further force or effect upon the sale of Newco, directly or indirectly, by way of merger, sale of stock or assets, or other form of business combination.

8.2 Non-Solicitation.

(a) Scotts agrees that from and after the date of this Agreement until the second anniversary of the Closing Date (the "<u>Non-Solicitation Period</u>"), it shall not, and shall cause its current and future Subsidiaries (other than the SLS Entities) not to, hire or request or induce any Person who is a SLS Transferred Employee or is at any time from the date of this Agreement to the Closing Date employed by the TruGreen Entities, other than those employees whose responsibilities are solely ministerial in nature, to terminate his or her employment with the TruGreen Entities or the SLS Entities or any of their respective current or future Affiliates; <u>provided</u>, <u>however</u>, that the foregoing shall not apply (<u>i</u>) to solicitations made by job opportunity advertisements, headhunter searches directed to the general public rather than targeting any employees of the TruGreen Entities or the SLS Entities (but shall apply to any hiring resulting therefrom), or (<u>ii</u>) with respect to any employee who has been terminated (except to the extent set forth in clause (iii) below) by the TruGreen Entities or the SLS Entities, or any of their respective current or future Affiliates (or has voluntarily left his or her employment) more than six months prior to such solicitation or hiring, or (<u>iii</u>) with respect to any employee who has been terminance reasons (or who has voluntarily left his or her employment for "good reason" or a similar concept under any such employee's employment agreement) after the Closing.

(b) TruGreen Holdings agrees that during the Non-Solicitation Period, it shall not, and it shall cause its Subsidiaries (including, after the Closing, the SLS Entities) not to, directly or indirectly, hire or request or induce any Person who is at any time from the date of this Agreement to the Closing Date employed by Scotts or any of its Subsidiaries (excluding, after the Closing, for the avoidance of doubt, SLS Employees), in each case, other than those employees whose responsibilities are solely ministerial in nature, with whom it had contact in the course of evaluating and negotiating a possible transaction with Scotts, to terminate his or her employment with Scotts or any of its Subsidiaries (including, prior to the Closing, the SLS Entities); provided, however, that the foregoing shall not apply (i) to solicitations made by job opportunity advertisements and headhunter searches directed to the general public rather than targeting any employees of Scotts or any of its Subsidiaries, as applicable (or has voluntarily left his or her employment), more than six months prior to such solicitation or hiring, or (ii) with respect to any employee who has been terminated by Scotts or any of its Subsidiaries for non-performance reasons (or who has voluntarily left his or her employment) agreement) after the Closing.

8.3 Specific Performance.

Scotts and TruGreen Holdings recognize and affirm that in the event of breach by any Party or its Subsidiaries of any of the provisions of this <u>Article VIII</u>, money damages would be inadequate and the other Party would have no adequate remedy at law. Accordingly, each Party agrees that the other Party shall have the right, in addition to any other rights and remedies existing in its favor, to enforce its rights and the other party's obligations under this <u>Article VIII</u> not only by an action or actions for damages, but also by an action or actions for specific performance, injunction and/or other equitable relief in order to enforce or prevent any violations (whether anticipatory, continuing or future) of the provisions of this <u>Article VIII</u>. Each Party agrees that the other Party is not required to post a bond in order for the other Party to secure an injunction.

8.4 <u>Severability</u>.

If at any time any of the provisions of this <u>Article VIII</u> shall be determined to be invalid or unenforceable by reason of being vague or unreasonable as to duration, area or scope of activity, or otherwise, then this <u>Article VIII</u> shall be considered divisible (with the other provisions to remain in full force and effect) and the invalid or unenforceable provisions shall become and be deemed to be immediately amended to include only such time, area, scope of activity and other restrictions as shall be determined to be reasonable and enforceable by the court or other body having jurisdiction over the matter, and the Parties hereto expressly agree that this Agreement, as so amended, shall be valid and binding as though any invalid or unenforceable provision had not been included herein. Without limiting the foregoing, if the length of the Non-Competition Period is determined to be unacceptable under applicable Law, the Non-Competition Period shall be modified as determined by a court or other agency of competent jurisdiction or statute, as applicable, to be of the maximum length permitted under applicable Law.

ARTICLE IX

INDEMNIFICATION

9.1 Indemnification by TruGreen Holdings.

Subject to the other terms, conditions and limitations of this Agreement (including the provisions of <u>Article VI</u> and <u>Sections 9.3</u>, <u>9.5</u> and <u>9.6</u>), from and after the Closing, TruGreen Holdings agrees to cause Newco (out of the assets of Newco and its Subsidiaries) to indemnify Scotts and its Subsidiaries and its and their officers, directors, employees, agents, successors and assigns (the "<u>Scotts Indemnified Parties</u>") against, and to pay and to hold Scotts Indemnified Parties harmless from, all Losses (it being agreed and understood that for the purposes of measuring Losses suffered by any Scotts Indemnified Party, such Scotts Indemnified Party shall be deemed to have suffered (x) the Loss suffered by such Scotts Indemnified Party (without giving effect to any Loss attributable to the diminution in value of Newco) (the "<u>Direct Loss</u>") <u>plus</u> (y) (<u>i</u>) the Loss (which may include losses arising from the diminution in value of any SLS

Entity or any TruGreen Entity) suffered by Newco (other than as a result of indemnification payments in respect of any Direct Loss) multiplied by (<u>ii</u>) such Scotts Indemnified Party's share of the issued and outstanding equity of Newco at the time the applicable indemnification claim is first made (the "<u>Indirect Loss</u>"), in the case of each of clause (x) and (y), <u>divided by (z)</u> (A) one (1) <u>minus</u> (B) such Scotts Indemnified Party's (together with its Permitted Affiliate Transferees' (as defined in the LLC Agreement), without duplication) share of the issued and outstanding equity of Newco (excluding any equity issued as a result of the exercise or conversion of any option or similar equity security issued pursuant to the Equity Incentive Plan) at the time the applicable indemnification claim is made (the amount so calculated <u>minus</u> the Direct Loss and Indirect Loss, the "<u>Loss Gross Up</u>") (the preceding mechanic, the "<u>Indemnity Gross-Up Principle</u>")) suffered or incurred by any of the Scotts Indemnified Parties to the extent arising out of or relating to:

(a) any inaccuracy of or breach by TruGreen Holdings of any representation or warranty made by TruGreen Holdings in <u>Sections 4.1, 4.2, 4.5, 4.6, 4.11</u> and <u>4.19</u>;

(b) any failure by TruGreen Holdings to perform or comply with (i) any covenant, obligation or other agreement in this Agreement or (ii) any provision set forth herein that requires TruGreen Holdings and/or any of its Subsidiaries to indemnify or reimburse Scotts or any other Person upon the occurrence of certain events as described therein;

(c) any SLS Liabilities and the operations and Liabilities of the SLS Entities and SLS Business prior to the Closing (excluding any Losses to the extent (i) attributable to any direct or indirect ownership by the Scotts Indemnified Parties of Newco or any of its Subsidiaries or (ii) based on or in connection with any Ancillary Agreement or other agreement between any Scotts Indemnified Party, on the one hand, and TruGreen Holdings or any of its Subsidiaries, on the other hand); and

(d) (i) claims by stockholders of TruGreen Holdings related to the Merger Agreement and the transactions contemplated thereby, including the Merger, and (ii) Losses arising out of the matters described in TruGreen <u>Schedule 9.1(d)</u>.

For the purposes of this <u>Article IX</u>, any inaccuracy in or breach of any representation or warranty (excluding <u>Section 4.5</u>) shall be determined without regard to any qualification as to "materiality" or "TruGreen Material Adverse Effect" and phrases of similar import.

9.2 Indemnification by Scotts.

Subject to the other terms, conditions and limitations of this Agreement (including the provisions of <u>Article VI</u> and <u>Sections 9.3</u>, <u>9.5</u> and <u>9.6</u>), from and after the Closing, Scotts agrees to indemnify Newco and its Subsidiaries (including the SLS Entities) and its and their officers, directors, employees, agents, successors and assigns (the "<u>TruGreen Indemnified Parties</u>") against, and to pay and to hold the TruGreen Indemnified Parties harmless from, all Losses (including losses arising from the diminution in value of any SLS Entity or TruGreen Entity)

suffered or incurred by any of the TruGreen Indemnified Parties, to the extent arising out of or relating to:

(a) any inaccuracy of or breach by Scotts of any representation or warranty made by Scotts in <u>Sections 3.1, 3.2</u>, <u>3.5, 3.6, 3.11</u> and <u>3.19</u>;

(b) any failure by Scotts, any Non-SLS Subsidiary or, with respect to any pre-Closing period, any SLS Entity to perform or comply with (i) any covenant, obligation or other agreement in this Agreement or (ii) any provision set forth herein that requires Scotts to indemnify or reimburse TruGreen Holdings or any other Person upon the occurrence of certain events as described therein;

(c) any Retained Liabilities and the operations and Liabilities of Scotts and the Non-SLS Subsidiaries prior to the Closing (excluding any Losses to the extent based on or in connection with any Ancillary Agreement or other agreement between any TruGreen Indemnified Party, on the one hand, and Scotts or any of its Subsidiaries, on the other hand); and

(d) any Losses arising out of the matters described on Scotts <u>Schedule 9.2(d)</u>.

For the purposes of this <u>Article IX</u>, any inaccuracy in or breach of any representation or warranty (excluding <u>Section 3.5</u>) shall be determined without regard to any qualification as to "materiality" or "SLS Material Adverse Effect" and phrases of similar import.

9.3 Limitations on Indemnification.

(a) The representations and warranties set forth in this Agreement, and the right to assert a claim for indemnification with respect thereto pursuant to Section 9.1(a) or 9.2(a), shall not survive the Closing and shall thereafter be of no further force or effect; provided that (i) the representations and warranties in Sections 3.1, 3.2, 3.19, 4.1, 4.2 and 4.19 (the "Fundamental Representations"), and the right to assert a claim for indemnification with respect thereto pursuant to Section 9.1(a) or 9.2(a), shall survive the Closing through the period ending 30 days following the expiration of the applicable statute of limitations; (ii) the representations and warranties in Section 3.9 and Section 4.9, and the right to assert a claim for indemnification with respect thereto pursuant to <u>Section 6.1(a)</u> and <u>Section 6.1(b)</u>, shall survive the Closing until the expiration of the applicable statute of limitations for assessments plus thirty (30) days, giving effect to any waiver, mitigation or extension of such period, and in each case shall thereafter be of no further force or effect; and (iii) the representations and warranties in Sections 3.5, 3.6, 3.11, 4.5, 4.6 and 4.11 and the Financial Statements Certificate, and the right to assert a claim for indemnification with respect thereto pursuant to Section 9.1(a) or 9.2(a), shall survive the Closing until the 15-month anniversary of the Closing Date and shall thereafter be of no further force or effect. The covenants and agreements set forth in this Agreement, and the right to assert a claim for indemnification with respect thereto, shall survive the Closing for a period of 15 months and shall thereafter be of no further force or effect, provided that those covenants and agreements and other provisions that (i) by their terms contemplate performance in whole or in part after the Closing or (ii) require TruGreen Holdings or Scotts, or any of their respective Subsidiaries to indemnify or reimburse any other Person upon the occurrence of certain events as

described herein, shall remain in full force and effect until ninety (90) days after the date by which such covenant or agreement is required to be, and is, fully performed or satisfied. If a party delivers an indemnification notice to the other party before the expiration of a representation or warranty or covenant or agreement, then the right to assert a claim for indemnification with respect to the applicable representation or warranty or covenant or agreement shall survive until, but only for purposes of, the resolution of the matter covered by such notice. No claim for indemnification under <u>Sections 9.1(a)</u>, <u>9.1(b)</u>, <u>9.2(a)</u> or <u>9.2(b)</u> may be made after the expiration of the applicable survival period.

(b) Notwithstanding anything to the contrary contained herein, except with respect to the Fundamental Representations, the Scotts Indemnified Parties and the TruGreen Indemnified Parties shall not be entitled to indemnification under <u>Section 9.1(a)</u>) (with respect to the Scotts Indemnified Parties) or <u>9.2(a)</u> (with respect to the TruGreen Indemnified Parties), until such group shall have incurred as to all claims, indemnifiable Losses in excess of an amount equal to \$8,000,000 (with respect to the Scotts Indemnified Parties) or \$4,000,000 (with respect to the TruGreen Indemnified Parties), in the aggregate; <u>provided</u>, <u>however</u>, that for the purposes of this <u>Section 9.3(b)</u>, the Losses of the Scotts Indemnified Parties attributable to any Loss Gross Up shall be excluded from the Losses of the Scotts Indemnified Parties.

(c) (i) No individual claim or series of related claims based on the same or related facts shall be so asserted unless and until the aggregate amount that would be payable pursuant to each such claim or series of related claims based on the same or related facts exceeds an amount equal to the Minimum Per Claim Amount; provided, however, that for the purposes of this Section 9.3(c)(i), the Losses of the Scotts Indemnified Parties attributable to any Loss Gross Up shall be excluded from the Losses of the Scotts Indemnified Parties. Under no circumstance shall either party's aggregate obligation to provide indemnification for matters covered by Section 9.1(a) or Section 9.2(a), as applicable, exceed an amount equal to \$120,000,000, with respect to indemnification by Scotts; provided, however, that for the purposes of this Section 9.3(c)(ii), any amount paid by Newco to the Scotts Indemnified Parties in respect of any Loss Gross Up shall be excluded from any and all calculations as to the amount of obligations satisfied by Newco as a result of any indemnification payments made to the Scotts Indemnified Parties.

(d) Notwithstanding any other provision in this Agreement to the contrary, there shall be no right to indemnification hereunder for any Losses (excluding Taxes, which shall be governed by <u>Article VI</u>) that were expressly taken into account in the final determination of Closing Working Capital, Closing Cash or Closing Indebtedness pursuant to <u>Section 2.1</u>.

(e) Each party shall take, and shall cause its Subsidiaries to take, actions to mitigate Losses, to the extent required by applicable Law, upon and after becoming aware of any event, which could reasonably be expected to give rise to a claim for indemnification hereunder, including using reasonable efforts to seek full recovery under any applicable insurance policies (but not self-insured programs) covering a Loss to the same extent as they would if such Loss were not subject to indemnification hereunder.

(f) Notwithstanding anything herein to the contrary, in no event shall any Indemnifying Party be liable for any punitive damages, except (<u>i</u>) in the event of fraud or willful misconduct or (<u>ii</u>) to the extent awarded against an Indemnitee pursuant to a claim by a third party.

(g) Notwithstanding anything in this Article IX to the contrary, <u>Sections 9.1</u> and <u>9.2</u> shall not apply to any claim for indemnification for any Losses arising from Taxes, which shall be governed exclusively by <u>Article VI</u>.

9.4 Sole Remedy.

Each of Scotts and TruGreen Holdings acknowledges and agrees that, following the Closing, its sole and exclusive remedy with respect to any and all claims relating to this Agreement and the transactions contemplated hereby (other than (<u>x</u>) claims arising from fraud, (<u>y</u>) any disputes relating to Closing Working Capital, Closing Cash and/or Closing Indebtedness, which shall be governed by <u>Section 2.1</u>, and (<u>z</u>) claims arising under any Ancillary Agreement) shall be pursuant to the indemnification provisions set forth in <u>Article VI</u> and this <u>Article IX</u>. In furtherance of the foregoing, each Party hereby waives on its own behalf and on behalf of its Subsidiaries (including in the case of TruGreen Holdings, TruGreen LP and the SLS Entities following the Closing), from and after the Closing, to the fullest extent permitted under Law, any and all claims (other than claims arising from fraud and other than any disputes relating to Closing Working Capital, Closing Cash and/or Closing Indebtedness, which shall be governed by <u>Section 2.1</u>) it may have against the other Party or any of its Affiliates arising under or based upon this Agreement or any other document or certificate delivered in connection herewith (other than any Ancillary Agreement or any other document executed in connection with the Transactions), except pursuant to the indemnification provisions set forth in <u>Article VI</u> and this <u>Article IX</u>. The foregoing notwithstanding, nothing in this <u>Section 9.4</u> shall limit or restrict the ability or right of TruGreen Holdings or Scotts to seek injunctive or other equitable relief for any breach or alleged breach of any provision of this Agreement (subject to any applicable limitations set forth in <u>Sections 10.2</u> and 10.3); provided that any procedures in respect of, and limitations on, Losses or Liabilities in <u>Article VI</u> and this <u>Article IX</u> shall in no event be diminished or circumvented by such relief.

9.5 <u>Computation of Indemnity Payments</u>.

(a) The amount payable under this <u>Article IX</u> in respect of any Loss shall be calculated net of (<u>A</u>) any express and specific reserves set forth in the Post-Closing Statement (including the balance sheet delivered in connection therewith) that are taken into account in the determination of the Post-Closing Adjustment Amounts and (<u>B</u>) any insurance proceeds or other amounts under indemnification agreements or other contractual arrangements with third parties actually received by the Indemnitee on account thereof; <u>provided</u> in each case that the Indemnitee shall (and shall cause its Subsidiaries to) use commercially reasonable efforts to obtain such insurance (and other) recoveries. In the event that an insurance or other recovery is made at any time after an indemnification payment by the Indemnitee has been made with respect to any indemnified Loss, then, to the extent of the amount of such indemnification

payment, a refund equal to the aggregate net amount of the recovery (less the costs of recovery to the Indemnitee) shall be made promptly to the Person or Persons that provided such indemnity payment to such Indemnitee. The party making any indemnification payment hereunder shall be subrogated to all rights of the Indemnitee in respect of any Losses indemnified by such party.

(b) Any payment by Scotts to Newco pursuant to <u>Article VI</u> or this <u>Article IX</u> shall be treated for tax purposes as a contribution. Any payment by Newco to Scotts pursuant to <u>Article VI</u> and this <u>Article IX</u> shall be treated as a distribution.

9.6 Procedures for Indemnification.

(a) <u>Procedures</u>.

(i) Any Person making a claim for indemnification under <u>Section 9.1</u> or <u>Section 9.2</u> (an "<u>Indemnified</u> <u>Party</u>") shall notify the party against whom indemnification is sought (an "<u>Indemnifying Party</u>") of the claim in writing promptly after receiving notice of any action, lawsuit, proceeding, investigation, demand or other claim against the Indemnified Party by a third party (a "<u>Third-Party Claim</u>"), describing the Third-Party Claim, the amount thereof (if known and quantifiable) and the basis thereof in reasonable detail; <u>provided</u> that the failure to so notify an Indemnifying Party shall not relieve the Indemnifying Party of its obligations hereunder, except to the extent that (and only to the extent that) such failure shall have caused the indemnifiable Losses to be greater than such Losses would have been had the Indemnified Party given the Indemnifying Party prompt notice hereunder.

(ii) Any Indemnifying Party shall be entitled to participate in the defense of such Third-Party Claim at such Indemnifying Party's expense, and at its option shall be entitled to assume the defense thereof by appointing a reputable counsel reasonably acceptable to the Indemnified Party to be the lead counsel in connection with such defense; provided that (x) the Indemnifying Party acknowledges in writing its obligation to indemnify the Indemnified Party for all Losses related to such Third-Party Claim and (y) the Indemnified Party shall be entitled to participate in the defense of such Third-Party Claim and (y) the Indemnified Party shall be entitled to participate in the defense of such Third-Party Claim and to employ counsel of its choice for such purpose (provided that the fees and expenses of such separate counsel shall be borne by the Indemnified Party and shall not be recoverable from such Indemnifying Party under this Article IX). Notwithstanding the foregoing, if the Indemnified Party shall have determined in good faith and upon advice of counsel that (x) an actual or likely conflict of interest makes representation of the Indemnifying Party and the Indemnified Party by the same counsel inappropriate or (y) the defendants in, or targets of, any such action or proceeding include both the Indemnified Party and an Indemnifying Party, and the Indemnified Party shall have reasonably concluded that there may be legal defenses available to it or to other Indemnified Parties which are different from or additional to those available to the Indemnifying Party, then, in each case, the Indemnified Party may, upon notice to the Indemnifying Party, engage

separate counsel, and the reasonable fees and expenses of such separate counsel shall be borne by the Indemnifying Party to the extent the Third-Party Claim is indemnifiable hereunder.

(iii) Upon assumption of the defense of any such Third-Party Claim by the Indemnifying Party, the Indemnified Party will not pay, or permit to be paid, any part of the Third-Party Claim, unless the Indemnifying Party consents in writing (such consent not to be unreasonably withheld or delayed) to such payment or unless a final judgment from which no appeal may be taken by or on behalf of the Indemnified Party is entered against the Indemnified Party for such Liability. Notwithstanding anything to the contrary herein, the Indemnifying Party shall not compromise or settle, or admit any Liability with respect to, any Third-Party Claim without the prior written consent of the Indemnified Party (which consent shall not be unreasonably withheld or delayed), unless (x) the relief consists solely of money damages (all of which the Indemnifying Party shall pay), (y) such settlement or compromise includes a provision whereby the plaintiff or claimant in the matter releases the Indemnified Party from all Liability with respect thereto and (z) such settlement or compromise does not include any admission of fault or wrongdoing on the part of the Indemnified Party.

(iv) In all cases with respect to Third-Party Claims, the Parties shall provide reasonable cooperation to each other in defense of such Third-Party Claims, including by making employees, information and documentation reasonably available (including for purposes of fact finding, consultation, interviews, depositions and, if required, as witnesses) and providing such information, testimony and access to their books and records, during normal business hours and upon reasonable notice, in each case as shall be reasonably necessary in connection with the contest or defense.

(v) If the Indemnifying Party shall not reasonably assume the defense of any such Third-Party Claim, or fails to competently prosecute or withdraws from the defense of any such Third-Party Claim, the Indemnified Party may defend against such matter, at the Indemnifying Party's expense, in a manner consistent with the above provisions regarding conduct of the defense by the Indemnified Party.

(b) In the event that any party or any of its Affiliates alleges that it is entitled to indemnification hereunder, and that its claim is covered under more than one provision of this <u>Article IX</u>, such party or Affiliates shall be entitled to elect the provision or provisions under which it may bring a claim for indemnification.

(c) A claim for indemnification for any matter not involving a Third-Party Claim may be asserted by notice to the party from whom indemnification is sought.

ARTICLE X

TERMINATION AND WAIVER

10.1 <u>Termination</u>.

This Agreement may be terminated:

(a) at any time prior to the Closing by the mutual written consent of Scotts and TruGreen Holdings;

(b) at any time prior to the Closing by either Party if (<u>A</u>) there shall be any Law that makes consummation of the Closing illegal or otherwise prohibited or (<u>B</u>) any Governmental Order enjoining either Party from consummating the Closing (or imposing a condition or requiring a remedy that either Party is not required to accept or give effect to after giving effect to the proviso in Section 5.3(b)) is entered and such Governmental Order shall have become final and nonappealable;

(c) by TruGreen Holdings, if Scotts breaches or fails to perform in any respect any of its representations, warranties or covenants contained in this Agreement and such breach or failure to perform (<u>A</u>) would give rise to the failure of a condition set forth in <u>Section 7.2</u>, (<u>B</u>) is not capable of being cured prior to the Outside Date and (<u>C</u>) has not been waived in writing by TruGreen Holdings;

(d) by Scotts, if TruGreen Holdings breaches or fails to perform in any respect any of its representations, warranties or covenants contained in this Agreement and such breach or failure to perform (i) would give rise to the failure of a condition set forth in <u>Section 7.1</u>, (ii) is not capable of being cured prior to the Outside Date and (iii) has not been waived in writing by Scotts;

(e) by TruGreen Holdings or by Scotts, upon written notice to the other party, if the Closing has not occurred on or prior to May 31, 2016 (the "<u>Outside Date</u>"); and

(f) by TruGreen Holdings or by Scotts, if, based on the Audited SLS Financial Statements, SLS Adjusted EBITDA (as defined in <u>TruGreen Schedule 7.2(l)</u>) exceeds or falls below the SLS EBITDA Reference Amount (as defined in <u>TruGreen Schedule 7.2(l)</u>) by more than 15%.

Notwithstanding anything else contained in this Agreement, the right to terminate this Agreement under <u>Sections 10.1(b)</u> and/or <u>10.1(e)</u> shall not be available to any Party whose (<u>a</u>) material breach of its representations, warranties, covenants or other agreements set forth herein or (<u>b</u>) failure to fulfill its obligations or to comply with its covenants under this Agreement has been the primary cause of, or primarily resulted in, the failure to satisfy any condition to the obligations of the other Party. The Party desiring to terminate this Agreement pursuant to any of <u>Sections 10.1</u>

(b) through (e) shall give written notice of such termination to the other Party in accordance with Section 11.2.

10.2 Effect of Termination.

(a) In the event of any termination in accordance with <u>Section 10.1</u>, this Agreement, except for <u>Section 5.5(c)</u> relating to confidentiality, <u>Sections 5.11(b)</u> and <u>5.11(c)</u>, this <u>Section 10.2</u>, <u>Article XI</u> and any corresponding definitions set forth in <u>Article XII</u> which shall survive such termination, will forthwith become void and have no effect, and there will be no Liability on the part of any party hereto or any of their respective directors, officers, stockholders, employees, agents, consultants or representatives; <u>provided</u>, <u>however</u>, that nothing herein shall relieve TruGreen Holdings or Scotts from Liability for fraud or any Willful Breach of its representations, warranties, covenants or agreements set forth in this Agreement prior to such termination. For purposes of this Agreement, "Willful Breach" means an intentional and willful material breach, or an intentional and willful material failure to perform, in each case that is the consequence of an act or omission by a party with the actual knowledge that the taking of such act or failure to take such act would constitute a breach of this Agreement.

(b) In the event (i) this Agreement is terminated by Scotts or TruGreen Holdings pursuant to <u>Section 10.1(e)</u> and, at such time, all of the conditions to closing set forth in Section 7.2 have been satisfied (other than (\underline{x}) the conditions set forth in Sections 7.2(c), 7.2(f) and 7.2(i), provided that such conditions would have been satisfied had the parties proceeded with Closing in accordance with the terms of this Agreement, and (<u>y</u>) provided that Scotts has not breached any of its obligations under Sections 5.3, 5.11 and 5.17, in each case in such a manner that would have caused the condition to Closing set forth in Section 7.2(b) not to be satisfied, any or all of the conditions set forth in Sections 7.2(h), 7.2(j) and 7.2(k), unless, in the case of Section 7.2(k), such condition has not been satisfied primarily as the result of any negative deviation between the SLS Financial Statements and the Audited SLS Financial Statements (taking into account (i) items disclosed on Scotts Schedules 3.5 and 3.6 and (ii) any unfunded liabilities that Scotts has committed to fund at or prior to Closing), (ii) Scotts has delivered to TruGreen Holdings the Audited SLS Financial Statements, the reviewed financial statements of the SLS Business for the fourth quarter of calendar year 2015 and all other Required Information by March 1, 2016 and (iii) the conditions set forth in Sections 7.2(d) and 7.2(l) shall have been satisfied no later than April 10, 2016 and shall remain satisfied through the Outside Date and the conditions set forth in Sections 7.2(a), 7.2(b), 7.2(c) and 7.2(g) would be satisfied if the Closing occurred on April 10, 2016 and each date subsequent through the Outside Date, then within 2 Business Days after such termination, TruGreen Holdings shall pay, or cause to be paid, to Scotts the Termination Fee in immediately available funds by wire transfer on the second Business Day after the date of such termination to an account designated by Scotts; provided, that the Termination Fee shall not be payable if (i) the SLS Business' independent accountants withdraw their audit opinion with respect to any of the Audited SLS Financial Statements and do not issue a new unqualified audit opinion with respect to such Audited SLS Financial Statements on or prior to March 1, 2016 or (ii) Scotts determines to restate any historical financial statements comprising the SLS Audited Financial Statements, and such

restatement has not been completed with restated Audited SLS Financial Statements on or prior to March 1, 2016.

(c) The Parties acknowledge that the agreements contained in <u>Section 10.2(b)</u> are an integral part of the transactions contemplated by this Agreement and constitute liquidated damages and not a penalty, and that, without these agreements, the Parties would not have entered into this Agreement. For the avoidance of doubt, the Termination Fee shall be payable only once and not in duplication even though the Termination Fee may be payable under one or more provisions hereof. Subject to Section 10.2(a), (i) except for the right of Scotts to seek an injunction, specific performance or other equitable relief to the extent permitted by Section 10.3, Scotts's and its Affiliates' sole and exclusive remedy (whether at Law, in equity, in contract, in tort or otherwise) against TruGreen Holdings or any of its Affiliates in circumstances in which the Termination Fee is pavable pursuant to Section 10.2(b), for any pre-Closing breach, loss or damage with respect to this Agreement and the transactions contemplated hereby shall be receipt of the Termination Fee, if, in the circumstances payable, paid hereunder and (ii) in the event Scotts shall receive the Termination Fee, the receipt thereof shall be deemed to be liquidated damages for any and all Losses or damages suffered or incurred by Scotts or any of its Affiliates in connection with this Agreement and the transactions contemplated hereby (and the termination thereof or any matter forming the basis for such termination), and neither Scotts nor any of its Affiliates shall be entitled to bring or maintain any other legal proceeding against TruGreen Holdings or any of its Affiliates arising out of this Agreement and the transactions contemplated hereby or any matter forming the basis for such termination. Notwithstanding anything to the contrary in this Agreement (in circumstances in which the Termination Fee is payable pursuant to Section 10.2(b)), except for the right of Scotts to seek and recover the Termination Fee pursuant to the terms and conditions of this Section 10.2 and the ability of Scotts to seek an injunction, specific performance or other equitable relief to the extent set forth in Section 10.3 (and without limiting Scotts's rights in law, equity and contract in circumstances in which the Termination Fee is not payable), none of TruGreen Holdings or its Affiliates shall have any liability to TruGreen Holdings or any of its Affiliates or any other Person, whether at Law or equity, in contract in tort or otherwise, arising from or in connection with any pre-Closing breach by TruGreen Holdings of any of its representations, warranties, covenants or agreements contained in this Agreement or arising from any claim or cause of action that TruGreen Holdings or any of its Affiliates may have relating to matters contemplated by this Agreement, and no Person will have any rights or claims against TruGreen Holdings or any of its Affiliates relating to any such matters. If TruGreen Holdings fails to promptly pay the Termination Fee when due and payable pursuant to this Section 10.2, and, in order to obtain such payment, Scotts commences an action or other proceeding that results in an award against TruGreen Holdings for such Termination Fee, TruGreen Holdings shall pay Scotts's costs and expenses (including reasonable attorneys' fees and expenses) in connection with such action or proceeding, together with interest on the amount of the Termination Fee from the date such payment was required to be made until the date of payment at the Prime Rate during such period; provided that if Scotts commences such an action or other proceeding that does not result in an award against TruGreen Holdings for such Termination Fee, Scotts shall pay TruGreen Holdings's costs and expenses (including reasonable attorneys' fees and expenses) in connection with such action or proceeding.

10.3 Enforcement.

(a) The Parties agree that irreparable damage would occur in the event that either Party does not perform, or otherwise breaches any provisions of this Agreement in accordance with their specific terms and that, subject to the limitations provided in <u>Section 10.2</u> and this <u>Section 10.3</u>, any such breach of this Agreement could not be adequately compensated in all cases by monetary damages alone. The Parties acknowledge and agree that each Party shall be entitled to an injunction, specific performance and other equitable relief to prevent breaches of this Agreement by the other Party, and to enforce specifically the terms and provisions hereof, in addition to any other remedy to which it is entitled at Law or in equity. Except as otherwise provided herein, including pursuant to <u>Section 10.2</u>, any and all remedies herein expressly conferred on a Party will be deemed cumulative with, and not exclusive of any other remedy conferred hereby, or by Law or equity upon such Party, and the exercise by a Party of any one remedy will not preclude the exercise of any other remedy.

(b) Each Party hereby agrees not to raise any objections to the availability of the equitable remedy of specific performance (other than on the basis that such remedy is not available pursuant to the terms of this Agreement in respect of the particular instance in question) to prevent or restrain breaches of this Agreement by such Party, and to specifically enforce the terms and provisions of this Agreement, to prevent breaches or threatened breaches of, or to enforce compliance with, the covenants and obligations of such Party under this Agreement all in accordance with the terms of this <u>Section 10.3</u>. Any Party seeking an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement, in each case when expressly available pursuant to the terms of this Agreement, shall not be required to provide any bond or other security in connection with such order or injunction.

ARTICLE XI

GENERAL PROVISIONS

11.1 Expenses.

Except as otherwise set forth herein (including <u>Section 5.11(b)</u>, <u>Section 5.11(c)</u> and <u>Section 5.14</u>) or in the Ancillary Agreements, all costs and expenses, including fees and disbursements of counsel, financial advisors and accountants, incurred in connection with this Agreement and the transactions contemplated hereby will be paid by the Party incurring such costs and expenses, whether or not the Closing shall have occurred.

11.2 <u>Notices</u>.

(a) All notices, requests, claims, demands and other communications hereunder will be in writing and will be given or made (and will be deemed to have been duly given or made upon receipt) (a) on the date of delivery if delivered personally or by facsimile with a response confirming receipt, or by email with a response confirming receipt, or by telecopy with a response confirming receipt, or (b) on the first Business Day following the date

of dispatch if delivered utilizing a next-day service by a recognized next-day courier, charges prepaid, or (c) on the earlier of confirmed receipt or the third Business Day following the date of mailing if delivered by registered or certified mail, postage prepaid, return receipt requested, to the Parties at the following addresses (or at such other address for a party as will be specified by like notice):

(b) if to Scotts:

14111 Scottslawn Road Marysville, Ohio 43041 Attn: General Counsel Fax: 937-644-7568 Email: contracts@scotts.com

with a copy (which shall not constitute notice) to:

Skadden, Arps, Slate, Meagher & Flom LLP Four Times Square New York, NY 10036 Attn: Kenneth M. Wolff Fax: (917) 777-2681 Email: kenneth.wolff@skadden.com

if to TruGreen Holdings:

TruGreen Holding Corporation c/o TruGreen Limited Partnership 860 Ridge Lake Blvd. Memphis, TN 38120 Attn.: Chief Legal Officer Facsimile: 901-597-9578 Email: KevinMann@trugreenmail.com

with copies (which shall not constitute notice) to:

Debevoise & Plimpton LLP 919 Third Avenue New York, NY 10022 Attn: Paul S. Bird Fax: (212) 521-7569 Email: psbird@debevoise.com

and

Richards, Layton & Finger, PA One Rodney Square 920 North King Street Wilmington, DE 19801 Attn: Mark. J. Gentile Fax: (302) 651-7701 Email: gentile@rlf.com

11.3 <u>Severability</u>.

If any term or other provision of this Agreement is held by a court of competent jurisdiction or other authority to be invalid, void, illegal or incapable of being enforced by any rule, Law or public policy of any jurisdiction (a) such invalidity, illegality or unenforceability shall be fully separable and shall not affect any other provision or portion of any provision of this Agreement in such jurisdiction and (b) all other terms and provisions of this Agreement shall nevertheless remain in full force and effect and shall be deemed substituted for the provision at issue, a valid, legal and enforceable provision. Upon a determination that any term or other provision is invalid, illegal or incapable of being enforced, the Parties hereto shall negotiate in good faith to modify this Agreement so as to effect the original intent of the Parties as closely as possible to the fullest extent permitted by Law in order that the transactions contemplated hereby be consummated as originally contemplated to the greatest extent possible, and so long as the economic or legal substance of the transactions contemplated hereby is not affected in a manner adverse to any Party.

11.4 Entire Agreement.

This Agreement (including the Exhibits, Scotts Schedules and TruGreen Schedules hereto) and the Ancillary Agreements, together with the Confidentiality Agreement, constitute the entire agreement of the Parties hereto with respect to the subject matter hereof and thereof and supersede all prior and contemporaneous agreements, arrangements, communications, understandings and undertakings with respect to the subject matter hereof and thereof, both written and oral. Neither this Agreement nor any Ancillary Agreement shall be deemed to contain or imply any restriction, covenant, representation, warranty, agreement or undertaking of any party with respect to the transactions contemplated hereby or thereby other than those expressly set forth herein or therein, and none shall be deemed to exist or be inferred with respect to the subject matter hereof.

11.5 Assignment.

Neither this Agreement nor any rights, interests or obligations hereunder shall be assigned or delegated, in whole or in part, by any Party, whether by operation of Law or otherwise, without the prior written consent of the non-assigning Party. Any such attempted assignment or delegation shall without such prior written consent be null and void *ab initio*. Notwithstanding the foregoing, the rights hereunder may be assigned to the Financing Sources for collateral purposes.

11.6 No Third-Party Beneficiaries.

Except as otherwise expressly provided in <u>Article IX</u>, this Agreement is for the sole benefit of the Parties hereto and their permitted successors and assigns and nothing herein, express or implied, is intended to or will confer upon any other Person any legal or equitable right, benefit or remedy of any nature whatsoever under or by reason of this Agreement; <u>provided</u> that the Financing Sources shall be express third-party beneficiaries of, and have the right to enforce, <u>Sections 11.6</u>, <u>11.7</u>, <u>11.8</u>, <u>11.10</u> and <u>11.16</u>.

11.7 Amendment; Waiver.

This Agreement may not be amended, restated, amended and restated, supplemented or modified, except by an instrument in writing signed by each of the Parties hereto. To the extent that any amendment or modification to <u>Sections 11.6</u>, <u>11.7</u>, <u>11.8</u>, <u>11.10</u> or <u>11.16</u> is sought, which is adverse to the rights of the Financing Sources, the prior written consent of the Financing Sources party to the Debt Financing Commitment shall be required before such amendment or modification is rendered effective.

Any Party hereto may: (a) extend the time for the performance of any of the obligations or other acts of the other party hereto; (b) waive any inaccuracies in or breaches of the representations and warranties of the other Party contained herein or in any document delivered pursuant hereto or (c) waive compliance by the other Party with any of the agreements or conditions contained herein. Any such extension or waiver will only be valid if set forth in an instrument in writing signed by the Party to be bound thereby. The failure of any Party to assert any of its rights under this Agreement or otherwise shall not constitute a waiver of such rights.

11.8 Governing Law; Jurisdiction.

This Agreement will be governed by, and construed in accordance with, the Laws of the State of New York, without regard to any choice or conflict of laws provision or rule (whether of the State of New York or any other jurisdiction) that are not mandatorily applicable by Law and would permit or require the application of the Laws of another jurisdiction.

Each of the Parties irrevocably agrees that any and all legal actions, proceedings or counterclaims (whether based on contract, tort or otherwise) arising out of or relating to this Agreement brought by the other party or its successors or assigns shall be brought and determined in any New York State or federal court sitting in the borough of Manhattan in The City of New York (or, solely if such court lacks subject matter jurisdiction, in any appropriate New York State or federal court), and each of the Parties hereby irrevocably submits to the exclusive jurisdiction of the aforesaid courts for itself and with respect to its property, generally and unconditionally, with regard to any such action or proceeding arising out of or relating to this Agreement and the transactions contemplated hereby. Each of the Parties agrees not to commence any action, suit, and proceeding or counterclaim relating thereto, except in the courts described above in New York, other than actions in any court of competent jurisdiction to

enforce any judgment, decree or award rendered by any such court in New York as described herein. Each of the Parties hereby irrevocably and unconditionally waives, and agrees not to assert, by way of motion or as a defense, counterclaim or otherwise, in any legal action or proceeding arising out of or relating to this Agreement or the transactions contemplated hereby, (i) any claim that it is not personally subject to the jurisdiction of the courts in New York as described herein for any reason, (ii) that it or its property is exempt or immune from jurisdiction of any such court or from any legal process commenced in such courts (whether through service of notice, attachment prior to judgment, attachment in aid of execution of judgment, execution of judgment or otherwise) and (iii) that (A) the action or proceeding in any such court is brought in an inconvenient forum, (B) the venue of such suit, action or proceeding is improper or (C) this Agreement, or the subject matter hereof, may not be enforced in or by such courts. Nothing in this Section 11.8 shall be deemed to prevent any party from seeking to remove any action to a federal court in the State of New York. Notwithstanding the foregoing, each of the Parties hereto hereby agrees that it will not bring or support any action, cause of action, claim, cross-claim or third-party claim of any kind or description, whether in law or in equity, whether in contract or in tort or otherwise, against the Financing Sources in any way relating to this Agreement or any of the transactions contemplated by this Agreement, including any dispute arising out of or relating in any way to the Debt Financing Commitment or the performance thereof, in any forum other than a court of competent jurisdiction sitting in the borough of Manhattan in the City of New York, New York, whether a state or federal court, and that the provisions of Section 11.10 relating to the waiver of jury trial shall apply to any such action, cause of action, claim, cross-claim or third-party claim.

11.9 Public Announcements.

From and after the date hereof through (and including) the Closing Date, the timing and content of all press releases or other written public announcements regarding any aspect of this Agreement to the financial community, employees or the general public by TruGreen Holdings or Scotts shall be subject to prior consultation with and the consent of the other (which shall not be unreasonably withheld or delayed); <u>provided</u>, <u>however</u>, that a party may, without the prior consent of any other party, issue such a press release or other similar public statement as may be required by applicable Law or any listing agreement with a securities exchange to which the disclosing party is a party, if the disclosing party has used all commercially reasonable efforts to consult with the other and to obtain the other's consent but has been unable to do so in a timely manner.

11.10 <u>Waiver of Jury Trial</u>.

EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT OR THE DEBT FINANCING COMMITMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES (AND SHALL CAUSE THEIR RESPECTIVE SUBSIDIARIES TO IRREVOCABLY AND UNCONDITIONALLY WAIVE) ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY ACTION, LEGAL PROCEEDING (WHETHER BASED ON CONTRACT, TORT OR

OTHERWISE) ARISING OUT OF OR RELATED TO THIS AGREEMENT OR THE DEBT FINANCING COMMITMENT, THE ANCILLARY AGREEMENTS OR THE CONFIDENTIALITY AGREEMENT OR BY THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY. EACH OF THE PARTIES CERTIFIES AND ACKNOWLEDGES THAT (i) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (ii) EACH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (iii) EACH PARTY MAKES THIS WAIVER VOLUNTARILY AND (iv) EACH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS <u>SECTION 11.10</u>.

11.11 Disclosure Generally.

Information set forth in any section or subsection of the Scotts Schedules or TruGreen Schedules shall be deemed to apply to and qualify the section or sub-section set forth in this Agreement to which it corresponds in number and, whether or not an explicit reference or cross-reference is made, each other section and subsection of this Agreement for which it is reasonably apparent on its face that such information is relevant to such other section or subsection. The information reflected in any Scotts Schedule or TruGreen Schedule is not necessarily limited to matters required by this Agreement to be reflected in such Scotts Schedule or TruGreen Schedule. Such additional information is set forth for informational purposes and does not necessarily include other matters of a similar nature. Disclosure of such additional information shall not be deemed to constitute an acknowledgment that such information is required to be disclosed and disclosure of such information shall not be deemed to enlarge or enhance any of the representations or warranties in this Agreement or otherwise alter in any way the terms of this Agreement. Inclusion of information in any Scotts Schedule or TruGreen Schedule shall not be construed as an admission that such information is material to the business, assets, liabilities, financial position, operations or results of operations of the SLS Entities, the SLS Business or the TruGreen Entities, as applicable.

11.12 No Presumption Against Drafting Party.

TruGreen Holdings and Scotts each acknowledge that they have been represented by counsel in connection with this Agreement and the transactions contemplated by this Agreement. Accordingly, any rule of law or any legal decision that would require interpretation of any claimed ambiguities in this Agreement against the drafting party has no application and is expressly waived.

11.13 <u>Time Periods</u>.

Unless specified otherwise, any action required hereunder to be taken within a certain number of days shall be taken within that number of calendar days (and not Business Days); <u>provided</u>, <u>however</u>, that if the last day for taking such action falls on a weekend or a holiday in

the United States, the period during which such action may be taken shall be automatically extended to the next Business Day.

11.14 Execution of Agreement.

This Agreement may be executed in one or more counterparts, and by the different Parties hereto in separate counterparts, each of which will be deemed to be an original copy of this Agreement and all of which, when taken together, will be deemed to constitute one and the same agreement. The exchange of copies of this Agreement and of signature pages by facsimile, .pdf signature, or electronic mail transmission shall constitute effective execution and delivery of this Agreement as to the Parties and may be used in lieu of the original Agreement for all purposes.

11.15 Scotts and TruGreen Holdings.

Scotts acknowledges and agrees with TruGreen Holdings that Scotts shall cause the SLS Entities (prior to the Closing) and the Non-SLS Subsidiaries, and TruGreen Holdings acknowledges and agrees with Scotts that TruGreen Holdings shall cause the TruGreen Entities (including TruGreen LP) and the SLS Entities (following the Closing), to perform or cause performance of any obligations under this Agreement to be performed by such entities or by such Party acting through such entity (and an enforcement right of a Party against another Party shall include such Party causing the applicable entity to act or perform).

11.16 No Recourse.

Notwithstanding anything to the contrary contained in this Agreement, except for claims by TruGreen LP or CDRTG Parent, Inc. against any Financing Source under the Debt Financing Commitment (A) none of the parties hereto nor any of their respective subsidiaries, Affiliates, directors, officers, employees, agents, partners, managers, members or stockholders shall have any rights or claims against any Financing Source, in any way relating to this Agreement or any of the transactions contemplated by this Agreement, or in respect of any, or in respect of any oral representations made or alleged to have been made in connection herewith or therewith, including any dispute arising out of or relating in any way to any Debt Financing Commitment or the performance thereof or the financings contemplated thereby, whether at law or equity, in contract, in tort or otherwise and (B) no Financing Source shall have any liability (whether in contract, in tort or otherwise) to any party hereto or any of their respective subsidiaries, Affiliates, directors, officers, employees, agents, partners, managers, members or stockholders for any obligations or liabilities of any party hereto under this Agreement or for any claim based on, in respect of, or by reason of, the transactions contemplated hereby and thereby or in respect of any oral representations made or alleged to have been made in connection herewith or therewith, including any dispute arising out of or relating in any way to any Debt Financing Commitment or the performance thereof or the financings contemplated thereby, whether at law or equity, in contract, in tort or otherwise; provided the performance thereof or the financings contemplated thereby, whether at law or equity, in contract, in tort or otherwise; provided the foregoing shall not limit Scotts's rights to enforce the terms of this Agreement against any other Party hereto.

ARTICLE XII

CERTAIN DEFINITIONS.

12.1 <u>Certain Defined Terms</u>.

As used in this Agreement, the following terms shall have the following meanings:

"Accounting Firm" has the meaning given in Section 2.1(f).

"<u>Affiliate</u>" means, with respect to a specified Person, any other Person that directly or indirectly controls, is controlled by or is under common control with such specified Person; <u>provided</u> that no portfolio company of any fund managed by Clayton, Dubilier & Rice (other than TruGreen Holdings and its Subsidiaries) shall be deemed an Affiliate of TruGreen Holdings or any of its Subsidiaries for purposes of this Agreement, other than for the purposes of clause (iii) of <u>Section 5.8(b)</u> and <u>Articles VI</u>, <u>VIII</u>, and <u>IX</u>, solely as they relate to <u>Section 5.8</u>.

"<u>Agreement</u>" has the meaning given in the Preamble.

"<u>Ancillary Agreements</u>" means the Assignment Agreements, the LLC Agreement, the Transition Services Agreement, the Trademark License Agreement, the Employee Leasing Agreement, the CD&R Consulting Agreement, the Scotts Consulting Agreement, the CD&R Indemnification Agreement and the Scotts Indemnification Agreement.

"Anti-Corruption Laws" has the meaning given in Section 3.16(a).

"<u>Antitrust Approvals</u>" has the meaning set forth in <u>Section 7.1(d)</u>.

"<u>Applicable Non-U.S. Antitrust Approvals</u>" means the necessary approvals, orders, permits or other Consents under any Competition Law set forth on Scotts <u>Schedule 5.3</u>.

"Assignment Agreements" has the meaning given in Section 1.5(a)(i).

"<u>Audited SLS Financial Statements</u>" shall have the meaning given in <u>Section 5.17</u>.

"Audited TruGreen Financial Statements" has the meaning given in Section 4.5(a).

"<u>Available Insurance Policies</u>" has the meaning given in <u>Section 5.9(a)</u>.

"Balance Sheet Date" means September 30, 2015.

"<u>Business Day</u>" means any day other than (i) any Saturday or Sunday or (ii) any other day on which banks located in New York, New York are required or authorized by Law to be closed for business.

"<u>Calculation Principles</u>" means such accounting principles, policies, methods, practices, categories, estimates, judgments and assumptions as are in accordance with <u>Exhibit D</u>.

"<u>CD&R Consulting Agreement</u>" means the letter agreement to be entered into by Newco and the other parties identified therein as of the Closing substantially in the form of Exhibit E hereto.

"<u>CD&R Indemnification Agreement</u>" means the indemnification agreement to be entered into by Newco and the other parties identified therein as of the Closing substantially in the form of Exhibit F hereto.

"<u>Closing</u>" has the meaning given in <u>Section 1.4</u>.

"Closing Adjustment" has the meaning given in Section 2.1(b).

"<u>Closing Cash</u>" means, with respect to the SLS Entities or the TruGreen Entities, as applicable, as of 11:59 p.m., New York time, on the day immediately preceding the Closing Date (assuming full completion of the Scotts Internal Reorganization and after giving effect to any repayment of Indebtedness prior to the Closing pursuant to <u>Section 1.6</u>), all cash, cash equivalents and marketable securities held by the SLS Entities or the TruGreen Entities, as applicable; <u>provided</u> that Closing Cash shall include amounts due from credit card transactions that are in the process of being settled, but shall be reduced by the amount of any checks, deposits or wire transfers written by the SLS Entities or the TruGreen Entities, as applicable, but not yet cleared, and increased by the amount of deposits received but not yet cleared by the SLS Entities or the TruGreen Entities, as applicable, but not yet cleared, and increased by the amount of deposits received but not yet cleared by the SLS Entities or the TruGreen Entities, as applicable, as of 11:59 p.m., New York time, on the day immediately preceding the Closing Date; <u>provided</u>, <u>further</u>, that all such amounts of cash, cash equivalents and marketable securities denominated in a currency other than United States dollars shall be converted into United States dollars at the prevailing exchange rate between the respective currencies set forth in the Eastern Edition of the Wall Street Journal on the last Business Day immediately preceding the Closing Date; <u>provided</u>, <u>further</u>, that Closing Cash shall not include any Trapped Cash. For the avoidance of doubt, "Closing Cash" shall not include any proceeds of the Debt Financing.

"<u>Closing Date</u>" has the meaning given in <u>Section 1.4</u>.

"<u>Closing Indebtedness</u>" means, with respect to the SLS Entities or the TruGreen Entities, as applicable, as of 11:59 p.m., New York time, on the day immediately preceding the Closing Date (assuming full completion of the Scotts Internal Reorganization and after giving effect to any repayment of Indebtedness prior to the Closing pursuant to <u>Section 1.6</u>) all Indebtedness of the SLS Entities or the TruGreen Entities, as applicable. For the avoidance of doubt, "Closing Indebtedness" shall not include any Indebtedness incurred under the Debt Financing.

"<u>Closing Working Capital</u>" means, with respect to the SLS Entities or the TruGreen Entities, as applicable, the working capital of the SLS Entities or the TruGreen Entities, as applicable, as of 11:59 p.m., New York time, on the day immediately preceding the Closing Date

(assuming full completion of the Scotts Internal Reorganization and after giving effect to any repayment of Indebtedness prior to the Closing pursuant to <u>Section 1.6</u>), and which shall be (i) the sum of all current assets of the SLS Entities or the TruGreen Entities, as applicable, minus (ii) the sum of all current liabilities of the SLS Entities or the TruGreen Entities, as applicable, in each case, as determined in accordance with the Calculation Principles; <u>provided</u> that any Taxes imposed on or measured by net or gross income, receipts, capital or worth of the SLS Entities or the TruGreen Entities, as applicable, that are current assets or current liabilities of the SLS Entities or the TruGreen Entities, as applicable, on or prior to the Closing Date shall be included in Closing Working Capital.

"Code" means the Internal Revenue Code of 1986 and any successor statute, as amended from time to time.

"Competing Person" has the meaning given in Section 8.1(a).

"<u>Competition Laws</u>" mean any Laws, including the HSR Act, relating to the regulation of monopolies or competition in any jurisdiction.

"<u>Confidentiality Agreement</u>" means the Confidentiality Agreement, dated as of August 20, 2014 between Clayton, Dubilier & Rice, LLC and Scotts.

"<u>Consent</u>" means a consent, authorization, permit or approval of a Person or Government Body, or a waiver by a Person or Government Body.

"<u>Contract</u>" means any written or enforceable oral contract, agreement, license, lease, sales order, purchase order, indenture, mortgage, note, bond or warrant (including all amendments, supplements and modifications thereto), except any common law contracts with employees of the SLS Entities or the TruGreen Entities, as applicable.

"<u>control</u>" (including the terms "<u>controlled by</u>" and "<u>under common control with</u>") shall mean the possession, directly or indirectly, or as trustee or executor, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, as trustee or executor, by Contract or otherwise.

"<u>Contribution</u>" has the meaning given in <u>Section 1.3(a)</u>.

"<u>Conversion</u>" has the meaning given in <u>Section 1.1(a)</u>.

"Copyrights" has the meaning given in the definition of Intellectual Property.

"Debt Financing" has the meaning given in Section 4.22.

"Debt Financing Commitment" has the meaning given in Section 4.22.

"<u>Declined Services</u>" has the meaning given in <u>Section 3.11(a)</u>.

"Designated Shared Contracts" has the meaning given in Section 5.13(b).

"Direct Loss" has the meaning given in Section 9.1.

"Distributions" has the meaning given in Section 1.3(c).

"<u>Dollars</u>" or "<u>\$</u>" means U.S. dollars.

"Domain Names" has the meaning given in the definition of Intellectual Property.

"ELA Expiration Date" has the meaning given in Section 5.7(c).

"Employee Leasing Agreement" has the meaning given in Section 5.7(c).

"<u>Employee Plan</u>" means any (i) employee benefit plan (within the meaning of Section 3(3) of ERISA, whether or not subject to ERISA) including any pension, profit-sharing, retirement, supplemental retirement, welfare benefit, retiree health, and life insurance plans, agreements or arrangements and (ii) bonus, stock option, stock purchase, restricted stock, phantom stock or other equity-based arrangement, incentive, deferred compensation, termination, severance, retention, change of control, health, group insurance, vacation or other employee benefit plan, program, policy or arrangement or employment agreement, whether written or unwritten.

"<u>Encumbrance</u>" means any charge, claim, encumbrance, pledge, lien (including a Tax lien), collateral assignment, security interest, mortgage, easement, encroachment, right-of-way, covenant, condition, restriction, option, right of first offer or refusal, third party right, limitation on voting rights, deed of trust, title retention, purchase agreement, conditional sale or other security arrangement, or any license, order or charge of any nature, or any adverse claim of title, title defect, ownership or use, or any agreement of any kind restricting transfer of ownership or voting of any nature.

"<u>Environmental Laws</u>" means any Laws regulating or relating to pollution, or protection of human health, safety, natural resources or the environment, including any and all Laws relating to (a) the manufacture, use, generation, management, handling, transport, treatment, disposal, storage, Release or threatened Release of, or exposure to, Hazardous Substances or materials, and (b) recordkeeping, notification, disclosure and reporting requirements for Hazardous Substances.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"<u>ERISA Affiliate</u>" means, with respect to any entity, trade or business, any other entity, trade or business that is, or was at the relevant time, a member of a group described in Section 414(b), (c), (m) or (o) of the Code or Section 4001(b)(1) of ERISA that includes or included the first entity, trade or business, or that is, or was at the relevant time, a member of the same

"controlled group" as the first entity, trade or business pursuant to Section 4001(a)(14) of ERISA.

"Estimated Closing Cash" has the meaning given in Section 2.1(a).

"Estimated Closing Indebtedness" has the meaning given in Section 2.1(a).

"Estimated Closing Working Capital" has the meaning given in Section 2.1(a).

"<u>Exchange Act</u>" means the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.

"Excluded Services" has the meaning given in Section 5.19.

"Exhibit" or "Exhibits" means respectively the document or documents attached to this Agreement as exhibits.

"FDD" means the franchise disclosure documents (including documents prepared as "Franchise Disclosure Documents," "FDDs," "Uniform Franchise Offering Circulars" or "UFOCs") prepared in accordance with the FTC Rule (or its predecessor), NASAA Commentary and any applicable Franchise Law, and all variations of such forms which have been approved for use or used in any state or jurisdiction requiring the filing and/or approval of the SLS Franchise Agreement or TruGreen Franchise Agreement, as applicable, and/or FDDs.

"Fee Letter" has the meaning given in Section 4.22.

"Final Determination Date" has the meaning given in Section 2.1(g).

"Financial Statements Certificate" shall have the meaning given in Section 5.17.

"<u>Financing Proceeds</u>" means the proceeds of the Debt Financing *less* the Transaction Expenses.

"<u>Financing Sources</u>" means the Persons that have committed to provide, or have otherwise entered into agreements in connection with, the Debt Financing Commitment or alternative debt financings in connection with the transactions contemplated hereby, including the parties named in <u>Section 4.22</u> and any joinder agreements, indentures or credit agreements entered into pursuant thereto or relating thereto, together with their Affiliates, officers, directors, employees, agents and representatives involved in the Debt Financing and their successors and assigns (it being understood that none of Scotts, TruGreen Holdings or their respective Subsidiaries shall be "<u>Financing Sources</u>" for any purpose hereunder).

"<u>Franchise</u>" means any grant by a Person to any Person of the right to engage in or carry on a business, or to sell or offer to sell any product or service, under or in association with any trademark, which constitutes a "franchise," as that term is defined under (a) the FTC Rule,

regardless of the jurisdiction in which the franchised business is located or operates; or (b) the Franchise Law applicable in the jurisdiction in which the franchised business is located or operates, if any.

"<u>Franchise Law</u>" means the FTC Rule and any other domestic or foreign Law regulating the offer or sale of franchises, business opportunities, seller-assisted marketing plans or similar relationships, or governing the relationships between franchisors and franchisees, manufacturers and dealers, or grantor and distributors, including without limitation those Laws that address unfair practices related to, or the default, termination, non-renewal, transfer of, franchises, dealerships and distributorships.

"<u>FTC Rule</u>" means the Federal Trade Commission trade regulation rule entitled "Disclosure Requirements and Prohibitions Concerning Franchising," 16 C.F.R Section 436.1 et seq.

"Fundamental Representations" has the meaning given in <u>Section 9.3(a)</u>.

"GAAP" means United States generally accepted accounting principles.

"<u>Government Official</u>" means any (i) officer or employee of a Governmental Body or instrumentality thereof (including any state-owned or state-controlled enterprise), or of a public international organization, (ii) holder of public office or (iii) any Person acting for or on behalf of any such Governmental Body or instrumentality thereof (including any state-owned or state-controlled enterprise).

"<u>Governmental Body</u>" means any domestic or foreign government, including any foreign, federal, state, provincial, local, territorial or municipal government or any political subdivision thereof, or any governmental, regulatory, judicial or administrative division, agency, authority, board, bureau, court, tribunal, or commission.

"<u>Governmental Order</u>" means any order, writ, judgment, award, ruling, injunction, decree or consent decree entered by or with any Governmental Body.

"<u>Group</u>" has the meaning given in <u>Section 2.1(a)</u>.

"<u>Hazardous Substance</u>" means any materials defined, listed, classified or regulated as a hazardous substance, hazardous waste, hazardous material, extremely hazardous waste, restricted hazardous waste, toxic substance, toxic pollutant, contaminant and any other similarly described substance as defined in any applicable Environmental Law, including any radioactive materials, petroleum, petroleum products or by-products, asbestos and asbestos-containing materials.

"<u>HSR Act</u>" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations promulgated thereunder.

"<u>Inactive Employee</u>" shall mean any individual who is absent from active employment on the Closing Date on account of short-term or long-term disability leave.

"Indebtedness" means, with respect to any Person, without duplication, (i) all obligations of such Person for borrowed money, including accrued and unpaid interest and any breakage costs, prepayment penalties, premium payments or fees or other similar amounts payable with respect thereto, (ii) all obligations of such Person evidenced by bonds, debentures, notes or similar instruments or debt securities and any breakage costs, prepayment penalties, premium payments or fees or other similar amounts payable with respect thereto, (iii) all obligations of such Person under conditional sale or other title retention agreements relating to any property purchased by such Person, (iv) all earn-outs, purchase price holdbacks, deferred purchase price obligations, contingent payment obligations and other obligations of such Person issued or assumed as the deferred purchase price of property or services (excluding obligations of such Person for raw materials, inventory, services and supplies incurred in the ordinary course of business consistent with past practices), (v) any obligations of such Person under any lease of property, real or personal, which obligations are required to be classified as capital leases in accordance with GAAP but excluding any breakage costs, prepayment penalties, premium payments or fees or other similar amounts payable in connection with any capitalized leases unless such costs, penalties or fees are due at the Closing or at the "Effective Time" (as defined in the Merger Agreement), (vi) all net obligations of such Person under interest rate, currency or commodity swaps, collars, caps, hedges, futures contract, forward contract, option or other derivative instruments or arrangements or hedging transactions (valued at the termination value thereof), (vii) all letters of credit, to the extent drawn, or performance bonds issued for the account of such Person (excluding surety bonds and customs bonds) and (viii) all guarantees and keep well arrangements of such Person of any Indebtedness of any other Person other than a Subsidiary of such Person. Notwithstanding anything in this Agreement to the contrary. Indebtedness shall expressly exclude Transaction Expenses.

"Indemnified Party" has the meaning given in Section 9.6(a).

"Indemnifying Party" has the meaning given in Section 9.6(a).

"Indemnitee" means any Indemnified Party or Tax Indemnified Party.

"Indemnity Gross Up Principle" has the meaning given in Section 9.1.

"Indirect Loss" has the meaning given in Section 9.1.

"Insurance Arrangements" has the meaning given in Section 5.9(a).

"<u>Intellectual Property</u>" means any of the following: United States or foreign (i) patents and patent applications (collectively, "<u>Patents</u>"); (ii) registered and unregistered trademarks, service marks, trade dress and other indicia of origin, pending trademark and service mark registration applications and intent-to-use registrations or similar reservations of marks (collectively, "Marks"); (iii) registered and unregistered copyrights (including in Software) and

applications for registration (collectively, "Copyrights"); (iv) internet domain names and URLs ("Domain Names"); (v) trade secrets and proprietary information, including unpatented inventions, invention disclosures, know-how, methods, processes, customer lists, data and databases, in each case to the extent any of the foregoing derives economic value (actual or potential) from not being generally known to other Persons who can obtain economic value from its disclosure or use, excluding any Copyrights or Patents that may cover or protect any of the foregoing (collectively, "Trade Secrets"); (vi) rights of paternity, integrity, disclosure and withdrawal and any other rights that may be known as "moral rights"; and (vii) any other intellectual property rights.

"Intellectual Property Contracts" means any Contract to which any SLS Entity or any TruGreen Entity, as applicable, is a party and pursuant to which (A) any SLS Entity or any TruGreen Entity, as applicable, permits any Person to use any SLS Owned Intellectual Property or TruGreen Owned Intellectual Property, as applicable, (B) any Person permits any SLS Entity or any TruGreen Entity, as applicable, to use Intellectual Property not owned by any SLS Entity or any TruGreen Entity, as applicable, to use SLS Entity or any TruGreen Entity, as applicable, to use Intellectual Property not owned by any SLS Entity or any TruGreen Entity, as applicable, to use or register Intellectual Property is restricted.

"IRS" means the Internal Revenue Service of the United States.

"Issuance" has the meaning given in <u>Section 1.3(b)</u>.

"<u>Knowledge</u>" means, when used (i) with respect to Scotts, the actual knowledge, after reasonable inquiry of the other executives and managers having primary responsibility for such matters with respect to the Transactions, of any person set forth on <u>Scotts Schedule 12.1(a)</u> (as of the date of this Agreement) and (ii) with respect to TruGreen Holdings, the actual knowledge, after reasonable inquiry of the other executives and managers having primary responsibility for such matters with respect to the Transactions or the Merger, of any person set forth on <u>TruGreen Schedule 12.1(a)</u> (as of the date of this Agreement).

"<u>Law</u>" means any domestic or foreign, federal, state or local law, statute, regulation, ordinance, rule, order, decree, judgment, consent decree or other binding directive issued, enacted, promulgated, entered into, agreed or imposed by any Governmental Body.

"Lenders" has the meaning given in Section 4.22.

"<u>Liability</u>" or "<u>Liabilities</u>" means, with respect to any Person (a) debts, commissions, duties, fees, salaries, performance or delivery penalties, warranty liabilities or (b) fines or penalties and (c) any other liabilities and obligations (whether pecuniary or not, including obligations to perform or forebear from performing acts or services) of that Person, whether known or unknown, absolute or contingent, accrued or unaccrued, asserted or unasserted, disputed or undisputed, liquidated or unliquidated, secured or unsecured, joint or several, due or to become due, vested or unvested, executory, determined, determinable or otherwise, and whether or not the same is required to be accrued on the financial statements of that Person in accordance with GAAP.

"LLC Agreement" has the meaning given in the Recitals.

"Loss" means any and all Liabilities, losses, damages, or claims of any nature, interest, awards, judgments, penalties, expenses and costs (including reasonable, documented and out of pocket costs and expenses of investigation, enforcement and collection and reasonable, documented and out of pocket attorneys' and accountants' fees and expenses, in each case, in connection with any Proceeding) whether or not involving a Third-Party Claim.

"Loss Gross Up" has the meaning given in Section 9.1.

"Major SLS Customers" has the meaning given in Section 3.21.

"<u>Major SLS Suppliers</u>" has the meaning given in <u>Section 3.21</u>.

"<u>Major TruGreen Customers</u>" has the meaning given in <u>Section 4.21</u>.

"<u>Major TruGreen Suppliers</u>" has the meaning given in <u>Section 4.21</u>.

"<u>Marks</u>" has the meaning given in the definition of Intellectual Property.

"Merger" has the meaning given in the Recitals.

"Merger Agreement" has the meaning given in the Recitals.

"Merger Sub" has the meaning given in the Recitals.

"Minimum Per Claim Amount" means \$50,000.

"<u>NASAA Commentary</u>" means the North American Securities Administrators Association, Inc.'s "2008 Franchise Registration and Disclosure Guidelines" and "Commentary on 2008 Franchise Registration and Disclosure Guidelines."

"<u>New Benefit Plans</u>" has the meaning given in <u>Section 5.7(e)</u>.

"<u>Newco</u>" has the meaning given in the Recitals.

"Newco 401(k) Plan" has the meaning given in Section 5.7(e).

"Newco Prohibited Activity" has the meaning given in Section 8.1(b)(i)

"<u>Newco Restructuring</u>" has the meaning given in <u>Section 1.2</u>.

"<u>Non-Competition Period</u>" has the meaning given in <u>Section 8.1(a)</u>.

"<u>Non-SLS Subsidiary</u>" means any Subsidiary of Scotts other than the SLS Entities (after giving effect to the Scotts Internal Reorganization and the Transactions).

"Non-Solicitation Period" has the meaning given in Section 8.2(a).

"Notice of Disagreement" has the meaning given in Section 2.1(e).

"Outside Date" has the meaning given in Section 10.1(e).

"<u>Parties</u>" has the meaning given in the Preamble.

"Patents" has the meaning given in the definition of Intellectual Property.

"<u>Permit</u>" means any permit, license, approval, authorization, consent, registration, variance, waiver, certification, concession, exemption or order, notice, endorsement or qualification, granted by or obtained from any Governmental Body pursuant to Law.

"Permitted Encumbrances" means (i) any restriction on transfer arising under applicable securities Law; (ii) liens for current utilities, Taxes or other governmental charges not yet due and payable or the amount or validity of which is being contested in good faith by appropriate proceedings by Scotts or TruGreen Holdings, as applicable (or any of their Subsidiaries), or the SLS Entities or the TruGreen Entities, as applicable, and for which adequate accruals or reserves are maintained on the financial statements of Scotts or TruGreen Holdings, as applicable (or any of their Subsidiaries), or the SLS Entities or the TruGreen Entities, as applicable, if such accruals or reserves are required in accordance with GAAP consistently applied; (iii) mechanics', construction, material men's, carriers', workers', repairers' and similar statutory liens imposed by Law or any encumbrance arising in connection with any surety bonds, in each case arising or incurred in the ordinary course of business for amounts which are not due and payable or are being contested in good faith by appropriate proceeding; (iv) zoning, entitlement, building and other land use regulations imposed by any Governmental Body having jurisdiction over the SLS Leased Real Property, TruGreen Leased Real Property, SLS Owned Real Property or TruGreen Owned Real Property, as applicable, which are not materially violated by the current condition, use and operation of the SLS Leased Real Property, TruGreen Leased Real Property, SLS Owned Real Property or TruGreen Owned Real Property, as applicable; (v) covenants, conditions, restrictions, reservations, limitations, exemptions, rights-of-way, easements and other similar matters of record affecting title to the SLS Leased Real Property, TruGreen Leased Real Property, SLS Owned Real Property or TruGreen Owned Real Property, as applicable, which do not individually or in the aggregate (A) materially impair the current occupancy or use of the SLS Leased Real Property, TruGreen Leased Real Property, SLS Owned Real Property or TruGreen Owned Real Property, as applicable, for the purposes for which it is currently used or proposed to be used or (B) materially affect the value of the SLS Leased Real Property, TruGreen Leased Real Property, SLS Owned Real Property or TruGreen Owned Real Property, as applicable; (vi) liens arising under worker's compensation, unemployment insurance, social security, retirement and similar Laws; (vii) liens securing rental payments under capital lease arrangements; (viii) rights of expropriation, access or use or any similar rights conferred on or

reserved by or in any statute of a Governmental Body and (ix) Encumbrances that will be removed prior to or at the Closing.

"<u>Person</u>" means any natural person, general or limited partnership, limited liability company, unlimited liability company, corporation (including non-for-profit corporation), joint stock company, trust, estate, joint venture, group, association or unincorporated organization or any other form of business or professional entity, but does not include a Governmental Body.

"Post-Closing Adjustment Amount" has the meaning given in Section 2.1(g).

"<u>Post-Closing Statement</u>" has meaning given in <u>Section 2.1(d)</u>.

"<u>Post-Closing Tax Period</u>" means (i) any Tax period ending after the Closing Date and (ii) in the case of any Straddle Period, the portion of such period following, but not including, the Closing Date.

"<u>Pre-Closing Tax Period</u>" means (i) any Tax period ending on or before the Closing Date and (ii) in the case of any Straddle Period, the portion of such period up to and including the Closing Date.

"<u>Prime Rate</u>" means the rate per annum published in *The Wall Street Journal* from time to time as the prime lending rate prevailing during any relevant period.

"<u>Proceeding</u>" has the meaning given in <u>Section 3.7</u>.

"<u>Reference Amount</u>" means, with respect to the SLS Entities, \$974,000 and, with respect to the TruGreen Entities, negative (-) \$55,200,000.

"<u>Release</u>" means any actual or threatened release, spill, emission, leaking, dumping, injection, pouring, deposit, disposal, discharge, dispersal, leaching or migration of Hazardous Substances into or through any land, soil, subsurface strata, surface water, groundwater or air, or otherwise entering into any indoor or outdoor property or environment.

"<u>Required Information</u>" has the meaning given in <u>Section 5.11(a)</u>.

"<u>Retained Liabilities</u>" means the Liabilities of Scotts and its Subsidiaries (including the SLS Entities) relating to or arising out of the businesses or operations of Scotts at or prior to Closing to the extent not related to or arising out of the SLS Business or SLS Assets.

"<u>Retained Scotts Employees</u>" means the employees employed by an SLS Entity who are identified as Retained Scotts Employees on Scotts <u>Schedule 5.7(a)</u>.

"Sanctioned Country" has the meaning given in Section 3.16(f).

"<u>Sanctions</u>" has the meaning given in <u>Section 3.16(f)</u>.

"Scotts" has the meaning given in the Preamble.

"Scotts Benefit Plans" has the meaning given in Section 3.8(b).

"Scotts Competing Person" has the meaning given in Section 8.1(b)(i).

"<u>Scotts Consulting Agreement</u>" means the letter agreement to be entered into by Newco and the other parties identified therein as of the Closing substantially in the form of Exhibit G hereto.

"<u>Scotts Indemnification Agreement</u>" means the indemnification agreement to be entered into by Newco and the other parties identified therein as of the Closing substantially in the form of Exhibit H hereto.

"Scotts Indemnified Parties" has the meaning given in Section 9.1.

"Scotts Information" has the meaning given in Section 5.6(b).

"Scotts Internal Reorganization" has the meaning given in Section 1.1.

"<u>Scotts Licensed IP Rights</u>" means, other than the Scotts Marks, any and all Intellectual Property owned (and which will, as of and following the Closing, continue to be owned) by Scotts or any of its Affiliates that was materially used or practiced in the SLS Business in the ordinary course of business in the twelve (12) months prior to the Closing.

"<u>Scotts Marks</u>" means any and all Trademarks consisting of or incorporating the Trademark "Scotts" or any variation thereof, whether or not used primarily in connection with the SLS Business.

"Scotts Newco Interests" has the meaning given in Section 1.3(b).

"Scotts Ownership Percentage" means 30.6%; provided that, (\mathbf{x}) if the SLS Adjusted EBITDA (as defined in TruGreen Schedule 7.2(1)) is less than 92.5% of the SLS EBITDA Reference Amount (as defined in TruGreen Schedule 7.2(1)), the Scotts Ownership Percentage shall be equal to (\mathbf{a}) (\mathbf{i}) 30.6% *multiplied by* (\mathbf{ii}) one *minus* the percentage equal to the excess of (\mathbf{A}) 92.5% over (\mathbf{B}) (\mathbf{I}) the SLS Adjusted EBITDA *divided by* (\mathbf{II}) the SLS EBITDA Reference Amount *divided by* (\mathbf{b}) the sum of the percentage determined by clause (a) and 69.4% and (y) if the SLS Adjusted EBITDA is greater than 107.5% of the SLS EBITDA Reference Amount, the Scotts Ownership Percentage shall be equal to (\mathbf{a}) (\mathbf{i}) 30.6% *multiplied by* (\mathbf{ii}) one *plus* the percentage equal to the excess of (\mathbf{A}) (\mathbf{I}) the SLS Adjusted EBITDA *divided by* (\mathbf{II}) the SLS EBITDA Reference Amount over (\mathbf{B}) 107.5% *divided by* (\mathbf{b}) the sum of the percentage determined by clause (a) and 69.4%.

"Scotts Prohibited Activity" has the meaning given in Section 8.1(a).

"<u>Scotts Schedule</u>" or "<u>Scotts Schedules</u>" means the Schedules referenced in the introductory paragraph to <u>Article III</u> (for the avoidance of doubt, whether or not relating to the provisions of <u>Article III</u>).

"Scotts Systems" has the meaning given in the Recitals.

"<u>Scotts Transaction Expenses</u>" means the costs, fees and expenses incurred by Scotts and its Subsidiaries in connection with the transactions contemplated hereby, but excluding (i) the costs, fees and expenses incurred by Scotts and its Subsidiaries in connection with the preparation of the SLS Financial Statements and (ii) any Taxes to the extent addressed under <u>Article VI</u>.

"Scotts Transaction Taxes" has the meaning given in Section 6.4(a).

"SEC" means the U.S. Securities and Exchange Commission.

"Securities Act" means the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder.

"<u>Shared Contract</u>" means any Contract (i) to which Scotts or any Non-SLS Subsidiary is a party and (ii) which materially benefits the SLS Business.

"SLS 401(k) Plan" has the meaning given in Section 5.7(e).

"SLS Assets" has the meaning given in Section 3.11(a).

"<u>SLS Business</u>" means the LawnService segment of the Scotts business conducted by Scotts and its Subsidiaries as reflected in the public securities filings of Scotts, as of the date hereof and as of the Closing.

"SLS Business Benefit Plans" has the meaning given in Section 3.8(b).

"<u>SLS Employee</u>" means (i) any employee of a SLS Entity who is not a Retained Scotts Employee as of the date of this Agreement, (ii) employees hired by an SLS Entity after the date of this Agreement as permitted under <u>Section 5.1</u> and (iii) any other employee of Scotts or any of its Subsidiaries who is listed as an SLS Employee on Scotts Schedule 5.7(a).

"SLS Entities" means, collectively, Scotts Systems and each of its wholly owned Subsidiaries as of the Closing.

"SLS Entity Benefit Plans" has the meaning given in Section 3.8(a).

"SLS Financial Statements" has the meaning given in Section 3.5(a).

"SLS Franchise" means a Franchise granted by any SLS Entity.

"SLS Franchisee" means a Person who is a party to an SLS Franchise Agreement.

"<u>SLS Franchise Agreement</u>" means any Contracts, commitments, arrangements or understandings pursuant to which any SLS Entity grants or has granted any Franchise or the right or option (whether or not subject to certain qualifications) to acquire any Franchise, including any addendum, amendment, extension or renewal thereof, and together with any guarantee in favor of either SLS Entity, and any other instrument or agreement relating thereto to which either SLS Entity is a party (including any software license and support agreement, distribution services agreement). Without limiting the foregoing, SLS Franchise Agreement includes area development agreements, multi-unit license or franchise agreements, master license or franchise agreements, and similar agreements that cover the development or franchising of Franchises within any area or country or the delegation of duties by any SLS Entity with respect to its obligations as a franchisor or otherwise under any such agreements.

"SLS Holdings" has the meaning given in Section 1.1(a).

"<u>SLS Intercompany Balances</u>" means, as of any date, all balances as of such date between Scotts and/or any Non-SLS Subsidiary, on the one hand, and any SLS Entity, on the other hand, including intercompany accounts receivable and intercompany accounts payable.

"SLS Intercompany Contract" has the meaning given in Section 5.8(a).

"SLS Interests" has the meaning given in the Recitals.

"<u>SLS Internal IT Systems</u>" means the hardware, Software, network and telecommunications equipment and internet-related information technology infrastructure owned or leased by Scotts or any of its Subsidiaries (including the SLS Entities).

"<u>SLS Intracompany Obligation</u>" means any obligation between any two or more of the SLS Entities, including intercompany accounts receivable and intercompany accounts payable between such entities.

"SLS Leased Real Property" has the meaning given in Section 3.12(b).

"<u>SLS Liabilities</u>" means the Liabilities of Scotts and the Non-SLS Subsidiaries to the extent related to the SLS Business (excluding, for the avoidance of doubt, the Retained Liabilities and the Liabilities reflected on Scotts <u>Schedule 9.2(d)</u>).

"<u>SLS Material Adverse Effect</u>" means any change, effect, development, occurrence or state of facts that, individually or in the aggregate, (a) has, or would reasonably be expected to have, a materially adverse effect on the business, properties, rights, assets, condition (financial or otherwise), or results of operations of the SLS Business or the SLS Entities, taken as a whole (after giving effect to the Scotts Internal Reorganization and the Transactions), other than any

change, effect, development, occurrence or state of facts to the extent relating to (i) changes in business, economic or regulatory conditions as a whole or in the industries in which the SLS Business operates, (ii) any acts of God, natural disasters, earthquakes, hurricanes, outbreak or escalation in hostilities, whether or not pursuant to the declaration of a national emergency or war, or the occurrence of any military or terrorist attack in any countries, or territories within which the SLS Business or SLS Entities operate, (iii) changes in financial, banking or securities markets (including changes in interest or exchange rates, stock, bond and/or debt prices, and any disruption thereof), (iv) changes in GAAP or the enforcement or interpretation thereof, (v) changes in Law or the enforcement or interpretation thereof, (vi) changes in commodity prices, (vii) the announcement of, or the taking of any action contemplated or expressly permitted by, this Agreement and the other agreements contemplated hereby, including compliance with the covenants set forth herein (other than for purposes of the representations and warranties contained in Sections 3.3 and 3.4, and the conditions in Section 7.2(a) to the extent they relate to the representations and warranties contained in Sections 3.3 and 3.4) or (viii) any change in credit ratings or failure by the SLS Entities or the SLS Business to meet any projections, forecasts or estimates of revenue or earnings (provided that the underlying cause of such change or failure may be considered in determining whether there is a SLS Material Adverse Effect to the extent not otherwise excluded pursuant to this definition), except, in the cases of clauses (i), (ii), (iii), (iv), (v) and (vi), to the extent that such adverse effects materially and disproportionately have a greater adverse impact on the SLS Entities, taken as a whole, as compared to the adverse impact such changes have on companies in the industry in which the SLS Entities operate; or (b) would, or would reasonably be expected to, prevent, materially delay or materially impede the performance by Scotts of its obligations under this Agreement or the consummation of the transactions contemplated hereby.

"SLS Material Contracts" has the meaning given in Section 3.14.

"SLS Owned Intellectual Property" has the meaning given in Section 3.13(a).

"SLS Owned Real Property" has the meaning given in Section 3.12(a).

"SLS Policies" has the meaning given in Section 3.17.

"SLS Real Property Leases" has the meaning given in Section 3.12(b).

"SLS Registered Intellectual Property" has the meaning given in Section 3.13(a).

"SLS Relevant Persons" has the meaning given in Section 3.16(a).

"SLS Unaudited Financial Statements Representation" has the meaning given in Section 3.5(a).

"<u>Software</u>" means all computer software, including application software, system software firmware, source code and object code versions thereof, in any and all forms and media, and all related documentation.

"Solvency Opinion" has the meaning given in Section 7.1(k).

"<u>Solvent</u>" means that the statements with respect to Newco in clauses (1) through (6) of the "Determinations" section of the engagement letter related to the Solvency Opinion, dated as of November 3, 2015, among Duff & Phelps, LLC, the Company and certain of the Company's Subsidiaries, are true and correct.

"<u>Specified Activities</u>" means (x) core lawn service business and operations, indoor and outdoor pest control services and mosquito control services and (y) the service, inspection and general maintenance of outdoor lawn irrigation systems, excluding hydroponic systems.

"Specified Matters" has the meaning given in Section 5.5(b).

"<u>Straddle Period</u>" means, as the context requires, any Tax period with respect to any SLS Entity or any TruGreen Entity that begins before but ends after the Closing Date.

"<u>Subsidiary</u>" or "<u>Subsidiaries</u>" means, with respect to any Person, any other Person of which (i) if a corporation, a majority of the total voting power of shares of capital stock or other equity interests entitled (without regard to the occurrence of any contingency) to vote in the election of directors, managers or trustees thereof is at the time owned or controlled, directly or indirectly, by that Person or one or more of the other Subsidiaries of that Person or a combination thereof or (ii) if a limited liability company, partnership, association or other business entity (other than a corporation), a majority of member, partnership or other similar ownership interest thereof is at the time owned or controlled, directly or indirectly, by that Person or a combination thereof and for this purpose, a Person or Persons owns a majority ownership interest in such a business entity (other than a corporation) if such Person or Persons shall be allocated a majority of such business entity's gains or losses or shall be or control any managing director or general partner of such business entity (other than a corporation) or shall be entitled to elect a majority of the members of such Person's board of managers or similar governing body. For the purposes hereof, the term Subsidiary shall include all Subsidiaries of such Subsidiary.

"<u>Tax</u>" or "<u>Taxes</u>" means any and all taxes, fees, levies, duties, tariffs, imposts, and similar charges of any kind whatsoever (whether payable directly or by withholding), including franchise, income, gross receipts, windfall or other profits, personal property, real property, intangible personal property, capital stock, ad valorem, value added, sales, use, documentary, stamp, intangible personal property, social security, workers' compensation, unemployment compensation, wages, pension, net worth, withholding or other taxes, together with any interest and penalties, additions to tax or additional amounts with respect thereto imposed by any Tax Authority.

"<u>Tax Authority</u>" means any Governmental Body, including social security administrators, or any agent thereof (third-party or otherwise), legally authorized to assess, lien, levy or otherwise collect, litigate or administer Taxes.

"Tax Claim" has the meaning given in <u>Section 6.2(a)</u>.

"<u>Tax Indemnified Party</u>" has the meaning given in <u>Section 6.2(a)</u>.

"<u>Tax Indemnifying Party</u>" has the meaning given in <u>Section 6.2(a)</u>.

"<u>Tax Return</u>" means any report, return, document, declaration, payee statement or other information or filing (and any schedule or attachment thereto) required to be supplied to any Tax Authority with respect to Taxes.

"Tax Sharing Agreements" has the meaning given in Section 6.6(a).

"Termination Fee" means \$15,000,000.

"Third-Party Claim" has the meaning given in Section 9.6(a).

"Trade Secrets" has the meaning given in the definition of Intellectual Property.

"Trademark License Agreement" has the meaning given in Section 5.12(b).

"<u>Transaction Expenses</u>" means, collectively, the TruGreen Holdings Transaction Expenses, the Scotts Transaction Expenses, the "Initial Consulting Fee" (as defined in the CD&R Consulting Agreement) and the "Initial Consulting Fee" (as defined in the Scotts Consulting Agreement).

"Transactions" has the meaning given in <u>Section 1.3(c)</u>.

"Transferred SLS Employees" has the meaning given in Section 5.7(c).

"Transition Services Agreement" has the meaning given in the Recitals.

"<u>Trapped Cash</u>" means the amount of cash, cash equivalents and marketable securities held by the SLS Entities or TruGreen Entities, as applicable, to the extent held by any such SLS Entity or TruGreen Entity organized outside of the U.S. as of the Closing Date, multiplied by 0.4.

"<u>Treasury Regulations</u>" means the federal income tax regulations, including any temporary or proposed regulations, promulgated under the Code, in effect as of the date hereof.

"TruGreen Benefit Plan" has meaning given in Section 4.8(a).

"<u>TruGreen Employee</u>" means (i) any employee of a TruGreen Entity as of the date of this Agreement and (ii) any employee hired by a TruGreen Entity after the date of this Agreement as permitted under <u>Section 5.2</u>.

"<u>TruGreen Entities</u>" means, collectively, TruGreen LP, GreenLawn International Holdings, LLC, a Delaware limited liability company, and GreenLawn, Ltd., a Canadian corporation, and Newco and any of its other Subsidiaries immediately prior to the Closing; provided that, for purposes of determining the accuracy or inaccuracy of any representation or warranty made by TruGreen Holdings in Article IV, such representations and warranties as they relate to "TruGreen Entities" that are formed after the date hereof shall be deemed to be made as of the applicable dates of formation of such entities and only with respect to periods following such formation dates.

"TruGreen Financial Statements" has meaning given in Section 4.5(a).

"TruGreen Franchise" means a Franchise granted by any TruGreen Entity.

"TruGreen Franchisee" means a Person who is a party to a TruGreen Franchise Agreement.

"<u>TruGreen Franchise Agreement</u>" means any Contracts, commitments, arrangements or understandings pursuant to which any TruGreen Entity grants or has granted any Franchise or the right or option (whether or not subject to certain qualifications) to acquire any Franchise, including any addendum, amendment, extension or renewal thereof, and together with any guarantee in favor of any TruGreen Entity, and any other instrument or agreement relating thereto to which any TruGreen Entity is a party (including any software license and support agreement, distribution services agreement). Without limiting the foregoing, TruGreen Franchise Agreement includes area development agreements, multi-unit license or franchise agreements, master license or franchise agreements, and similar agreements that cover the development or franchising of Franchises within any area or country or the delegation of duties by any TruGreen Entity with respect to its obligations as a franchisor or otherwise under any such agreements.

"TruGreen Holdings" has the meaning given in the Preamble.

"<u>TruGreen Holdings Transaction Expenses</u>" means the costs, fees and expenses incurred by TruGreen Holdings and its Affiliates (including fees and expenses of legal, accounting and financial advisors and including all fees and other amounts required to be paid to the Financing Sources pursuant to the Debt Financing Commitment or otherwise in connection with the Debt Financing) in connection with the transactions contemplated hereby and by the Merger Agreement, but excluding (<u>i</u>) the expenses listed on <u>TruGreen Schedule 12.1(b</u>), which shall be for the account of the Persons listed on such <u>TruGreen Schedule 12.1(b</u>), and (<u>ii</u>) any Taxes to the extent addressed under <u>Article VI</u>.

"TruGreen Indemnified Parties" has the meaning given in Section 9.2.

"<u>TruGreen Intercompany Balances</u>" means, as of any date, all balances as of such date between a TruGreen Entity, on the one hand, and TruGreen Holdings or any of its Subsidiaries, other than Newco and its Subsidiaries as of the Closing, on the other hand, including intercompany accounts receivable and intercompany accounts payable.

"TruGreen Intercompany Contract" has the meaning given in Section 5.8(b).

"<u>TruGreen Internal IT Systems</u>" means the hardware, Software, network and telecommunications equipment and internetrelated information technology infrastructure owned or leased by the TruGreen Entities.

"<u>TruGreen Intracompany Obligation</u>" means any obligation between any two or more of the TruGreen Entities, including intercompany accounts receivable and intercompany accounts payable between such entities.

"TruGreen Leased Real Property" has the meaning given in Section 4.12(b).

"TruGreen LP" has the meaning given in the Recitals.

"TruGreen Material Adverse Effect" means any change, effect, development, occurrence or state of facts that, individually or in the aggregate, (a) has, or would reasonably be expected to have, a materially adverse effect on the business, properties, rights, assets, condition (financial or otherwise), or results of operations of the TruGreen Entities, taken as a whole (after giving effect to the transactions contemplated by this Agreement), other than any change, effect, development, occurrence or state of facts to the extent relating to (i) changes in business, economic or regulatory conditions as a whole or in the industries in which the TruGreen Entities operate, (ii) any acts of God, natural disasters, earthquakes, hurricanes, outbreak or escalation in hostilities whether or not pursuant to the declaration of a national emergency or war, or the occurrence of any military or terrorist attack in any countries or territories within which the TruGreen Entities operate, (iii) changes in financial, banking or securities markets (including changes in interest or exchange rates, stock, bond and/or debt prices, and any disruption thereof), (iv) changes in GAAP or the enforcement or interpretation thereof, (v) changes in Law or the enforcement or interpretation thereof, (vi) changes in commodity prices, (vii) the announcement of, or the taking of any action contemplated or expressly permitted by, this Agreement and the other agreements contemplated hereby, including compliance with the covenants set forth herein (other than for purposes of the representations and warranties contained in Sections 4.3 and 4.4, and the conditions in Section 7.1(a) to the extent they relate to the representations and warranties contained in Sections 4.3 and 4.4) or (viii) any change in credit ratings or failure by the TruGreen Entities to meet any projections, forecasts or estimates of revenue or earnings (provided that the underlying cause of such failure may be considered in determining whether there is a TruGreen Material Adverse Effect to the extent not otherwise excluded pursuant to this definition), except, in the cases of clauses (i), (ii), (iii), (iv), (v) and (vi), to the extent that such adverse effects materially and disproportionately have a greater adverse impact on the TruGreen Entities, taken as a whole, as compared to the adverse impact such changes have on companies in the industry in which the TruGreen Entities operate or (b) would, or would reasonably be expected to, prevent, materially delay or materially impede the performance by TruGreen Holdings of its obligations under this Agreement or the consummation of the transactions contemplated hereby.

"TruGreen Material Contracts" has the meaning given in Section 4.14.

"TruGreen Newco Holders" has the meaning given in the Recitals.

"TruGreen Newco Interests" has the meaning given in the Recitals.

"TruGreen Owned Intellectual Property" has the meaning given in Section 4.13(a).

"TruGreen Owned Real Property" has the meaning given in <u>Section 4.12(a)</u>.

"TruGreen Policies" has meaning given in Section 4.17.

"TruGreen Real Property Leases" has the meaning given in Section 4.12(b).

"TruGreen Registered Intellectual Property" has the meaning given in Section 4.13(a).

"TruGreen Relevant Persons" has the meaning given in Section 4.16(a).

"<u>TruGreen Schedule</u>" or "<u>TruGreen Schedules</u>" means the Schedules referenced in the introductory paragraph to <u>Article IV</u> (for the avoidance of doubt, whether or not relating to the provisions of <u>Article IV</u>).

"TruGreen Transaction Taxes" has the meaning given in <u>Section 6.4(b)</u>.

"<u>WARN Act</u>" means the Worker Adjustment and Retraining Notification Act of 1988, as amended, and any other similar Law.

12.2 Interpretation.

The headings contained in this Agreement are for reference purposes only and will not affect in any way the meaning or interpretation of this Agreement. In the event of a conflict between language or amounts contained in the body of the Agreement and language or amounts contained in the Exhibits, Scotts Schedules or TruGreen Schedules attached hereto, the language or amounts in the body of the Agreement shall control. The use of the masculine, feminine or neuter gender or the singular or plural form of words herein shall not limit any provision of this Agreement. The use of the terms "including" or "include" shall in all cases herein mean "including, without limitation" or "include, without limitation," respectively. The word "extent" in the phrase "to the extent" shall mean the degree to which a subject or thing extends, and such phrase shall not simply mean "if". Reference to any applicable agreement, and reference to a Person in a particular capacity excludes such Person in any other capacity or individually. Reference to any agreement (including this Agreement), document or instrument means such agreement, document or instrument as amended or modified and in effect from time to time in accordance with the terms thereof and, if applicable, the terms hereof. Reference to any Law means such Law as amended, modified, codified, replaced or re-enacted, in whole or in part, including rules, regulations, enforcement procedures and any

interpretations promulgated thereunder, all as in effect on the date hereof. Underscored references to Articles, Sections, Scotts Schedules or TruGreen Schedules shall refer to those portions of this Agreement. References to "\$" or "dollars" in this Agreement shall mean United States dollars. Any reference in this Agreement to a time or date shall be deemed to be such date or time in New York City, unless otherwise specified. "Writing", "written" and comparable terms refer to printing, typing and other means of reproducing words (including electronic format) in a visible form. The use of the terms "hereunder," "hereof," "hereto" and words of similar import shall, unless otherwise stated, refer to this Agreement as a whole and not to any particular Article, Section or clause of or Exhibit, Scotts Schedule or TruGreen Schedule to this Agreement. The use of the word "or" is not intended to be exclusive unless otherwise stated. The use of the phrases "the date of this Agreement", "the date hereof", "of even date herewith" and terms of similar import shall be deemed to refer to the date set forth in the preamble to this Agreement. The phrase "ordinary course of business" shall be deemed to be followed by the words "consistent with past practice" whether or not such words actually follow such phrase. The word "will" shall be construed to have the same meaning and effect as the word "shall" and vice versa.

[Remainder of page intentionally left blank]

IN WITNESS WHEREOF, Scotts and TruGreen Holdings have caused this Agreement to be executed as of the date first written above by their respective duly authorized representatives.

THE SCOTTS MIRACLE-GRO COMPANY

By: <u>/s/ RANDY COLEMAN</u> Name: Randy Coleman Title: Executive Vice President and Chief Financial Officer

TRUGREEN HOLDING CORPORATION

By: <u>/s/ R. DAVID ALEXANDER JR.</u> Name: R. David Alexander Jr. Title: President and CEO

Rule 13a-14(a)/15d-14(a) Certifications (Principal Executive Officer) <u>CERTIFICATIONS</u>

I, James Hagedorn, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of The Scotts Miracle-Gro Company for the fiscal quarter ended January 2, 2016;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: February 11, 2016

By: /s/ JAMES HAGEDORN

Printed Name: James Hagedorn Title: President, Chief Executive Officer and Chairman of the Board

Rule 13a-14(a)/15d-14(a) Certifications (Principal Financial Officer) <u>CERTIFICATIONS</u>

I, Thomas Randal Coleman, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of The Scotts Miracle-Gro Company for the fiscal quarter ended January 2, 2016;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: February 11, 2016

By: /s/ THOMAS RANDAL COLEMAN

Printed Name: Thomas Randal Coleman Title: Executive Vice President and Chief Financial Officer

SECTION 1350 CERTIFICATIONS*

In connection with the Quarterly Report on Form 10-Q of The Scotts Miracle-Gro Company (the "Company") for the fiscal quarter ended January 2, 2016 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), the undersigned James Hagedorn, President, Chief Executive Officer and Chairman of the Board of the Company, and Thomas Randal Coleman, Executive Vice President and Chief Financial Officer of the Company, certify, pursuant to Section 1350 of Chapter 63 of Title 18 of the United States Code, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that, to the best of their knowledge:

- 1) The Report fully complies with the requirements of Section 13(a) of the Securities Exchange Act of 1934; and
- 2) The information contained in the Report fairly presents, in all material respects, the consolidated financial condition and results of operations of the Company and its subsidiaries.

/s/ JAMES HAGEDORN	/s/ THOMAS RANDAL COLEMAN
Printed Name: James Hagedorn	Printed Name: Thomas Randal Coleman
Title: President, Chief Executive Officer and Chairman of the Board	Title: Executive Vice President and Chief Financial Officer

February 11, 2016

February 11, 2016

 THESE CERTIFICATIONS ARE BEING FURNISHED AS REQUIRED BY RULE 13a-14(b) UNDER THE SECURITIES EXCHANGE ACT OF 1934 (THE "EXCHANGE ACT") AND SECTION 1350 OF CHAPTER 63 OF TITLE 18 OF THE UNITED STATES CODE, AND SHALL NOT BE DEEMED "FILED" FOR PURPOSES OF SECTION 18 OF THE EXCHANGE ACT OR OTHERWISE SUBJECT TO THE LIABILITY OF THAT SECTION. THESE CERTIFICATIONS SHALL NOT BE DEEMED TO BE INCORPORATED BY REFERENCE INTO ANY FILING UNDER THE SECURITIES ACT OF 1933 OR THE EXCHANGE ACT, EXCEPT TO THE EXTENT THAT THE COMPANY SPECIFICALLY INCORPORATES THESE CERTIFICATIONS BY REFERENCE.